Edited by Christopher J. Fielding

WILEY-VCH

Lipid Rafts and Caveolae

From Membrane Biophysics to Cell Biology

Lipid Rafts and Caveolae

Edited by Christopher J. Fielding

Related Titles

Tamm, L.K. (Ed.)

Protein-Lipid Interactions

From Membrane Domains to Cellular Networks

2005 ISBN 3-527-31151-3

Wedlich, D. (Ed.)

Cell Migration in Development and Disease

2005 ISBN 3-527-30587-4

Benz, R. (Ed.)

Bacterial and Eukaryotic Porins

Structure, Function, Mechanism

2004 isbn 3-527-30775-3

Krauss, G.

Biochemistry of Signal Transduction and Regulation

2003 isbn 3-527-30591-2

Yawata, Y.

Cell Membrane

The Red Blood Cell as a Model

2003 ISBN 3-527-30463-0

Lipid Rafts and Caveolae

From Membrane Biophysics to Cell Biology

Edited by Christopher J. Fielding



WILEY-VCH Verlag GmbH & Co. KGaA

The Editor

Prof. Dr. Christopher J. Fielding Cardiovascular Research Institute and Department of Medicine University of California San Francisco San Francisco, CA 94143 USA All books published by Wiley-VCH are carefully produced. Nevertheless, authors, editor, and publisher do not warrant the information contained in these books, including this book, to mind that statements, data, illustrations, procedural details or other items may inadvertently be inaccurate.

Library of Congress Card No.: applied for

British Library Cataloguing-in-Publication Data:

A catalogue record for this book is available from the British Library.

Bibliographic information published by Die Deutsche Bibliothek

Die Deutsche Bibliothek lists this publication in the Deutsche Nationalbibliografie; detailed bibliographic data is available in the Internet at <http://dnb.ddb.de>

© 2006 WILEY-VCH Verlag GmbH & Co. KGaA, Weinheim

All rights reserved (including those of translation in other languages). No part of this book may be reproduced in any form – by photoprinting, microfilm, or any other means – nor transmitted or translated into machine language without written permission from the publishers. Registered names, trademarks, etc. used in this book, even when not specifically marked as such, are not to be considered unprotected by law.

Typesetting: Typomedia GmbH, Ostfildern Printing: betz-Druck GmbH, Darmstadt Binding: J. Schäffer GmbH, Grünstadt Cover Design: Grafik-Design Schulz, Fußgönheim

Printed in the Federal Republic of Germany Printed on acid-free paper

ISBN-13:978-3-527-31261-0ISBN-10:3-527-31261-7

Table of Contents

1	Lipid Rafts, Caveolae, and Membrane Traffic 1 Doris Meder and Kai Simons
1.1	Introduction 1
1.2	Basic Organization Principles of a Cell Membrane 1
1.3	Evidence for Phase Separation in Model Membrane Systems: Liquid-Ordered and Liquid-Disordered Phases 3
1.4	Evidence for Phase Separation in Cell Membranes: The "Raft Concept" 5
1.5	Raft Domains are Clustered to Exert their Function 8
1.6	The Apical Membrane of Epithelial Cells:
	A Percolating Raft Membrane at 25 °C 9
1.7	Caveolae: Scaffolded Membrane Domains Rich in Raft Lipids 10
1.8	Caveolae and Lipid Rafts in Membrane Traffic 12
	Abbreviations 17
	References 17
2	The Forces that Shape Caveolae 25
2	The Forces that Shape Caveolae25Pierre Sens and Matthew S. Turner
2 2.1	Pierre Sens and Matthew S. Turner Introduction 25
-	Pierre Sens and Matthew S. Turner
2.1	Pierre Sens and Matthew S. Turner Introduction 25
2.1 2.2	Pierre Sens and Matthew S. Turner Introduction 25 Physical Modeling of Lipid Membranes 26
2.1 2.2 2.3	Pierre Sens and Matthew S. Turner Introduction 25 Physical Modeling of Lipid Membranes 26 Caveolae as Invaginated Lipid Rafts 29
2.1 2.2 2.3 2.4	Pierre Sens and Matthew S. Turner Introduction 25 Physical Modeling of Lipid Membranes 26 Caveolae as Invaginated Lipid Rafts 29 Membrane Inclusions 31
2.1 2.2 2.3 2.4	Pierre Sens and Matthew S. Turner Introduction 25 Physical Modeling of Lipid Membranes 26 Caveolae as Invaginated Lipid Rafts 29 Membrane Inclusions 31 Caveolae as a Thermodynamic Phase Separation of Membrane
2.1 2.2 2.3 2.4 2.5	Pierre Sens and Matthew S. Turner Introduction 25 Physical Modeling of Lipid Membranes 26 Caveolae as Invaginated Lipid Rafts 29 Membrane Inclusions 31 Caveolae as a Thermodynamic Phase Separation of Membrane Proteins 33
2.1 2.2 2.3 2.4 2.5	Pierre Sens and Matthew S. TurnerIntroduction 25Physical Modeling of Lipid Membranes 26Caveolae as Invaginated Lipid Rafts 29Membrane Inclusions 31Caveolae as a Thermodynamic Phase Separation of MembraneProteins 33Caveolae and Membrane Tension: Mechano-Sensitivity and
2.1 2.2 2.3 2.4 2.5 2.6	Pierre Sens and Matthew S. TurnerIntroduction 25Physical Modeling of Lipid Membranes 26Caveolae as Invaginated Lipid Rafts 29Membrane Inclusions 31Caveolae as a Thermodynamic Phase Separation of MembraneProteins 33Caveolae and Membrane Tension: Mechano-Sensitivity andMechano-Regulation 38
2.1 2.2 2.3 2.4 2.5 2.6	Pierre Sens and Matthew S. TurnerIntroduction 25Physical Modeling of Lipid Membranes 26Caveolae as Invaginated Lipid Rafts 29Membrane Inclusions 31Caveolae as a Thermodynamic Phase Separation of MembraneProteins 33Caveolae and Membrane Tension: Mechano-Sensitivity andMechano-Regulation 38Conclusions 42

٧

VI Table of Contents

3	The Biophysical Characterization of Lipid Rafts 45
	Pranav Sharma, Rajat Varma, and Satyajit Mayor
3.1	Introduction: The Fluid Mosaic Model and Membrane Domains 45
3.2	The Origin of the Raft Hypothesis 45
3.3	The Role of Lipid-Anchored Proteins in the Development of the
	Membrane Raft Hypothesis 48
3.4	The Case For and Against DRMs as Evidence for "Rafts" in Cell
	Membranes 49
3.5	Why Are Biophysical Studies Useful for Understanding Lipid
	Rafts? 51
3.6	Diffusion-Based Measurements 52
3.6.1	Single-Molecule Studies 52
3.6.2	Fluorescence Recovery After Photobleaching 55
3.6.3	Fluorescence Correlation Spectroscopy 57
3.7	Proximity Measurements 59
3.7.1	Proximity Measurement Using Homo-FRET 60
3.7.2	Proximity Measurement Using Hetero-FRET 62
3.8	Conclusions 63
	Abbreviations 64
	References 64
4	The Role of Caveolae and Noncaveolar Rafts in Endocytosis 69 Bo van Deurs, Frederik Vilhardt, Maria Torgersen, Kirstine Roepstorff,
41	Anette M. Hommelgaard, and Kirsten Sandvig
4.1	Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69
4.1 4.2	Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69 Caveolae are Largely Immobile, Nonendocytic Membrane
4.2	Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69 Caveolae are Largely Immobile, Nonendocytic Membrane Domains 71
	Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69 Caveolae are Largely Immobile, Nonendocytic Membrane Domains 71 Caveolae May Show Local, Short-Range Motility: A Role in
4.2 4.3	Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69 Caveolae are Largely Immobile, Nonendocytic Membrane Domains 71 Caveolae May Show Local, Short-Range Motility: A Role in Transendothelial Transport? 73
4.2 4.3 4.4	 Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69 Caveolae are Largely Immobile, Nonendocytic Membrane Domains 71 Caveolae May Show Local, Short-Range Motility: A Role in Transendothelial Transport? 73 An Internalization Wave of Caveolae can be Stimulated by Virus 74
4.24.34.44.5	 Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69 Caveolae are Largely Immobile, Nonendocytic Membrane Domains 71 Caveolae May Show Local, Short-Range Motility: A Role in Transendothelial Transport? 73 An Internalization Wave of Caveolae can be Stimulated by Virus 74 Role of Caveolae in Endocytosis of Cholera Toxin 75
4.2 4.3 4.4	 Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69 Caveolae are Largely Immobile, Nonendocytic Membrane Domains 71 Caveolae May Show Local, Short-Range Motility: A Role in Transendothelial Transport? 73 An Internalization Wave of Caveolae can be Stimulated by Virus 74 Role of Caveolae in Endocytosis of Cholera Toxin 75 A Small Fraction of Caveolae may become Constitutively
4.24.34.44.54.6	 Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69 Caveolae are Largely Immobile, Nonendocytic Membrane Domains 71 Caveolae May Show Local, Short-Range Motility: A Role in Transendothelial Transport? 73 An Internalization Wave of Caveolae can be Stimulated by Virus 74 Role of Caveolae in Endocytosis of Cholera Toxin 75 A Small Fraction of Caveolae may become Constitutively Internalized 81
 4.2 4.3 4.4 4.5 4.6 4.7 	 Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69 Caveolae are Largely Immobile, Nonendocytic Membrane Domains 71 Caveolae May Show Local, Short-Range Motility: A Role in Transendothelial Transport? 73 An Internalization Wave of Caveolae can be Stimulated by Virus 74 Role of Caveolae in Endocytosis of Cholera Toxin 75 A Small Fraction of Caveolae may become Constitutively Internalized 81 Caveosomes: Intracellular Caveolin-Associated Structures 82
 4.2 4.3 4.4 4.5 4.6 4.7 4.8 	 Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69 Caveolae are Largely Immobile, Nonendocytic Membrane Domains 71 Caveolae May Show Local, Short-Range Motility: A Role in Transendothelial Transport? 73 An Internalization Wave of Caveolae can be Stimulated by Virus 74 Role of Caveolae in Endocytosis of Cholera Toxin 75 A Small Fraction of Caveolae may become Constitutively Internalized 81 Caveosomes: Intracellular Caveolin-Associated Structures 82 The Role of Dynamin in Caveolar Function 83
 4.2 4.3 4.4 4.5 4.6 4.7 4.8 4.9 	Anette M. Hommelgaard, and Kirsten SandvigIntroduction 69Caveolae are Largely Immobile, Nonendocytic MembraneDomains 71Caveolae May Show Local, Short-Range Motility: A Role inTransendothelial Transport? 73An Internalization Wave of Caveolae can be Stimulated by Virus 74Role of Caveolae in Endocytosis of Cholera Toxin 75A Small Fraction of Caveolae may become ConstitutivelyInternalized 81Caveosomes: Intracellular Caveolin-Associated Structures 82The Role of Dynamin in Caveolar Function 83Caveolin Immobilizes Rafts/Caveolar Invaginations 83
 4.2 4.3 4.4 4.5 4.6 4.7 4.8 	 Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69 Caveolae are Largely Immobile, Nonendocytic Membrane Domains 71 Caveolae May Show Local, Short-Range Motility: A Role in Transendothelial Transport? 73 An Internalization Wave of Caveolae can be Stimulated by Virus 74 Role of Caveolae in Endocytosis of Cholera Toxin 75 A Small Fraction of Caveolae may become Constitutively Internalized 81 Caveosomes: Intracellular Caveolin-Associated Structures 82 The Role of Dynamin in Caveolar Function 83 Caveolin Immobilizes Rafts/Caveolar Invaginations 83 A 2005 Consensus Model for Caveolar Endocytosis 85
 4.2 4.3 4.4 4.5 4.6 4.7 4.8 4.9 	 Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69 Caveolae are Largely Immobile, Nonendocytic Membrane Domains 71 Caveolae May Show Local, Short-Range Motility: A Role in Transendothelial Transport? 73 An Internalization Wave of Caveolae can be Stimulated by Virus 74 Role of Caveolae in Endocytosis of Cholera Toxin 75 A Small Fraction of Caveolae may become Constitutively Internalized 81 Caveosomes: Intracellular Caveolin-Associated Structures 82 The Role of Dynamin in Caveolar Function 83 Caveolin Immobilizes Rafts/Caveolar Invaginations 83 A 2005 Consensus Model for Caveolar Endocytosis 85 Acknowledgments 85
 4.2 4.3 4.4 4.5 4.6 4.7 4.8 4.9 	 Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69 Caveolae are Largely Immobile, Nonendocytic Membrane Domains 71 Caveolae May Show Local, Short-Range Motility: A Role in Transendothelial Transport? 73 An Internalization Wave of Caveolae can be Stimulated by Virus 74 Role of Caveolae in Endocytosis of Cholera Toxin 75 A Small Fraction of Caveolae may become Constitutively Internalized 81 Caveosomes: Intracellular Caveolin-Associated Structures 82 The Role of Dynamin in Caveolar Function 83 Caveolin Immobilizes Rafts/Caveolar Invaginations 83 A 2005 Consensus Model for Caveolar Endocytosis 85
 4.2 4.3 4.4 4.5 4.6 4.7 4.8 4.9 	 Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69 Caveolae are Largely Immobile, Nonendocytic Membrane Domains 71 Caveolae May Show Local, Short-Range Motility: A Role in Transendothelial Transport? 73 An Internalization Wave of Caveolae can be Stimulated by Virus 74 Role of Caveolae in Endocytosis of Cholera Toxin 75 A Small Fraction of Caveolae may become Constitutively Internalized 81 Caveosomes: Intracellular Caveolin-Associated Structures 82 The Role of Dynamin in Caveolar Function 83 Caveolin Immobilizes Rafts/Caveolar Invaginations 83 A 2005 Consensus Model for Caveolar Endocytosis 85 Acknowledgments 85 Abbreviations 86
 4.2 4.3 4.4 4.5 4.6 4.7 4.8 4.9 	Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69 Caveolae are Largely Immobile, Nonendocytic Membrane Domains 71 Caveolae May Show Local, Short-Range Motility: A Role in Transendothelial Transport? 73 An Internalization Wave of Caveolae can be Stimulated by Virus 74 Role of Caveolae in Endocytosis of Cholera Toxin 75 A Small Fraction of Caveolae may become Constitutively Internalized 81 Caveosomes: Intracellular Caveolin-Associated Structures 82 The Role of Dynamin in Caveolar Function 83 Caveolin Immobilizes Rafts/Caveolar Invaginations 83 A 2005 Consensus Model for Caveolar Endocytosis 85 Acknowledgments 85 Abbreviations 86 References 86 Role of Cholesterol in Signal Transduction from Caveolae 91
 4.2 4.3 4.4 4.5 4.6 4.7 4.8 4.9 4.10 	Anette M. Hommelgaard, and Kirsten SandvigIntroduction 69Caveolae are Largely Immobile, Nonendocytic MembraneDomains 71Caveolae May Show Local, Short-Range Motility: A Role inTransendothelial Transport? 73An Internalization Wave of Caveolae can be Stimulated by Virus 74Role of Caveolae in Endocytosis of Cholera Toxin 75A Small Fraction of Caveolae may become ConstitutivelyInternalized 81Caveosomes: Intracellular Caveolin-Associated Structures 82The Role of Dynamin in Caveolar Function 83Caveolin Immobilizes Rafts/Caveolar Invaginations 83A 2005 Consensus Model for Caveolar Endocytosis 85Acknowledgments 85Abbreviations 86References 86
 4.2 4.3 4.4 4.5 4.6 4.7 4.8 4.9 4.10 	Anette M. Hommelgaard, and Kirsten Sandvig Introduction 69 Caveolae are Largely Immobile, Nonendocytic Membrane Domains 71 Caveolae May Show Local, Short-Range Motility: A Role in Transendothelial Transport? 73 An Internalization Wave of Caveolae can be Stimulated by Virus 74 Role of Caveolae in Endocytosis of Cholera Toxin 75 A Small Fraction of Caveolae may become Constitutively Internalized 81 Caveosomes: Intracellular Caveolin-Associated Structures 82 The Role of Dynamin in Caveolar Function 83 Caveolin Immobilizes Rafts/Caveolar Invaginations 83 A 2005 Consensus Model for Caveolar Endocytosis 85 Acknowledgments 85 Abbreviations 86 References 86 Role of Cholesterol in Signal Transduction from Caveolae 91

5.3	Proteins in Caveolae 95
5.4	The Caveolin Scaffold Hypothesis 98
5.4.1	Does the Scaffold Motif in Signaling Proteins that are Present in
	Caveolae Represent the Contact Site of these Proteins with
	Caveolin? 99
5.5	FC Binding by Proteins Including Caveolin 101
5.6	FC in Caveolae: Effects of Depletion and Loading 102
5.7	FC Changes in Caveolae: Effects of Signal Transduction 104
5.8	Summary 107
5.0	Abbreviations 107
	References 107
6	Phosphorylation of Caveolin and Signaling from Caveolae 116
	Cynthia Corley Mastick, Amy Sanguinetti, Haiming Cao, and Suhani
	Thakker
6.1	Introduction 115
6.2	Signaling Pathways Leading to Caveolin Tyrosine Phosphoryla-
	tion 116
6.2.1	Caveolins-1 and -2 are Phosphorylated in Response to Insulin in
	Adipocytes 116
6.2.2	The Caveolins are not Direct Substrates of the Insulin Receptor 117
6.2.3	Src-Family Kinases and Stress-Induced Caveolin Phosphorylation 118
6.2.4	Non-Receptor Tyrosine Kinases and Insulin-Induced Caveolin
	Phosphorylation 119
6.2.5	Abl is a Caveolin Kinase 120
6.2.6	Abl and Fyn Cooperate in the Caveolin Phosphorylation Signaling
	Pathway 121
6.2.7	Model of the Interaction of Fyn and Abl in Caveolin
	Phosphorylation 122
6.3	Signaling Pathways Downstream of Caveolin Tyrosine
	Phosphorylation 123
6.3.1	Csk Binds to Phosphocaveolin 124
6.3.2	Regulation of Src-Family Kinases by Csk 125
6.3.3	Feedback Inhibition of Fyn Through Activation of Csk 125
6.3.4	Phosphocaveolin in the Loop 126
6.3.5	Src-Family Kinases, Csk and Actin Remodeling 127
6.3.6	Phosphocaveolin is Enriched at Sites of Attachment of the Actin
	Cytoskeleton to the Plasma Membrane 127
6.3.7	Abl in the Loop 129
6.3.8	Abl and Actin Remodeling 130
6.3.9	Insulin-Induced Actin Remodeling, GluT4 Translocation, and
	Caveolae 131
6.3.10	The Role of Caveolin Phosphorylation in Cells 131
6.4	Summary 133
	Abbreviations 133
	References 134

VIII Table of Contents

7	Role of Lipid Microdomains in the Formation of Supramolecular Protein Complexes and Transmembrane Signaling 141
	György Vámosi, Andrea Bodnár, György Vereb, János Szöllösi, and Sándor
	Damjanovich
7.1	Introduction 141
7.1.1	Lateral organization of membrane proteins 142
7.1.2	Factors controlling the organization of membrane proteins 143
7.2	Biophysical Strategies for Studying the Lateral Organization of Membrane Proteins 144
7.2.1	Determination of Domain Size and Overlap between Fluorescence
	Distributions using Fluorescence Microscopy 144
7.2.2	Fluorescence Resonance Energy Transfer (FRET) 145
7.2.3	Fluorescence Cross-Correlation Spectroscopy: Analysis of Protein Co-Mobility 147
7.2.4	Atomic Force Microscopy (AFM) 149
7.2.5	Scanning Near-Field Optical Microscopy (SNOM) 149
7.3	The Immunological Synapse 150
7.4	Voltage-Gated K ⁺ Channels in Lipid Rafts: Possible Involvement in Local Regulatory Processes 154
7.5	Cell Fusion as a Tool for Studying Dynamic Behavior of Protein
	Clusters 155
7.6	Lipid Rafts as Platforms for Cytokine Receptor Assembly and Signaling 156
7.7	Organization and Function of Receptor Tyrosine Kinases is Linked to Lipid Microdomains 162
	Acknowledgments 166
	Abbreviations 166
	References 167
8	Caveolin and its Role in Intracellular Chaperone Complexes 175 William V. Everson and Eric J. Smart
8.1	Caveolae and Caveolin-1 175
8.2	Caveolin Protein Structure, Domains, and Membrane Interac- tions 177
8.3	Caveolin Expression and Localization in the Cell 178
8.4	Caveolin Expression and Localization Varies Depending on the
0.4	Physiological State of Cells in Culture 180
8.5	Caveolin-1 Expression Confers a New Level of Regulation 182
8.6	Caveolae Cholesterol and Caveolin Localization to Caveolae 182
8.7	Caveolin and Cholesterol Cross Membranes During Trafficking 183
8.8	Two Chaperone Complexes Regulate a Caveola-Cholesterol Trafficking
	Cycle 184
8.9	Caveolae Linked to Nongenomic Actions and Uptake of
	Estrogen 185
8.10	Protein Acylation and Caveolae 186

- 8.11 Scavenger Receptors Localize to Caveolae 187
- 8.12 Cholesterol Homeostasis Regulates Caveolin Localization and Organization of other Proteins in Caveolae 187
- 8.13 Chaperone Complexes Involved in Cholesterol Transport in Specialized Tissues 188
- 8.14 Caveolin is Linked to Additional Sterol and Lipid Uptake and Trafficking Pathways 188
- 8.15 Conclusions 189 Abbreviations 189 References 189
- 9 The Roles of Caveolae and Caveolin in Cell Shape, Locomotion, and Stress Fiber Formation 195

Sang Chul Park and Kyung A. Cho

- 9.1 Introduction 195
- 9.2 Caveolin and Polarity 195
- 9.3 Caveolin and Rho-family GTPases 197
- 9.4 Caveolin and Focal Adhesion Complex 198
- 9.5 The Dynamics of Caveolin and Actin 199
- 9.6 Caveolae-Dependent Endocytosis via Actin Stress Fiber 200
- 9.7 Summary 201 Abbreviations 202 References 202
- 10 Lipid Rafts in Trafficking and Processing of Prion Protein and Amyloid Precursor Protein 205

Daniela Sarnataro, Vincenza Campana, and Chiara Zurzolo

- 10.1 Introduction 205
- 10.2 Lipid Rafts and Caveolae 206
- 10.2.1 Biochemical Properties and Functions 206
- 10.3 PrP^c and Prion Diseases 207
- 10.3.1 The Site of PrP^c Conversion: The Role of Rafts in the Different Intracellular Compartments 208
- 10.3.2 Role of Lipid Rafts in PrP^{Sc} Formation 210
- 10.3.3 Mechanism of Raft Action in Prion Conversion 211
- 10.3.4 Role of Rafts in Proteolytic Attack on PrP^c 214
- 10.4 Alzheimer's Disease: The Role of Rafts in APP Trafficking and Processing 215
- 10.4.1 The "History" of APP Cleavage 215
- 10.4.2 Intracellular Compartments and Aβ Generation: Involvement of Lipid Rafts 215
- 10.4.3 The Role of Rafts in β -Secretase Activity 217
- The Role of Caveolae/Lipid Rafts in α-Secretase Activity 220
- 10.4.4.1 The Role of Lipid Rafts in this Event 221

X Table of Contents

10.4.5	The Role of Caveolae/Rafts in γ -Secretase Activity 221
10.4.6	The Contribution of Cholesterol and Sphingolipids in APP
	Processing 222
10.5	Conclusions 222
	Acknowledgments 223
	Abbreviations 223
	References 224
11	Caveolae and the Endothelial Nitric Oxide Synthase 233
	Olivier Feron
11.1	Introduction 233
11.2	Caveolin: A Scaffold for eNOS 235
11.3	The Caveolin-eNOS Regulatory Cycle 236
11.4	Lipoproteins and Caveolin-eNOS Interaction 239
11.5	Angiogenesis and Caveolin-eNOS Interaction 241
11.6	Vasodilation, Endothelial Permeability and Caveolin-eNOS
	Interaction 242
11.7	Caveolin-3-eNOS Interaction in Cardiac Myocytes 244
11.8	Conclusions 246
	Abbreviations 246
	References 247
12	The Role of Caveolin-1 in Tumor Cell Survival and Cancer
	Progression 249
	Dana Ravid and Mordechai Liscovitch
12.1	Introduction 249
12.2	The Caveolin-1 Gene and its Regulation During Differentiation and
	Transformation 250
12.3	Divergent Expression of Caveolin-1 in Human Cancer:
	The Case of Lung Cancer 251
12.4	Actions of Caveolin-1 in Cancer Cells: Effects of Heterologous
	Expression and Genetic or Functional Suppression 252
12.4.1	Anti-Proliferative Activity of Caveolin-1 252
12.4.2	Pro-Apoptotic Activity of Caveolin-1 253
12.4.3	Survival-Promoting Activity of Caveolin-1 253
12.5	Molecular Mechanisms Implicated in the Pro-Survival Action of
	Caveolin-1 254
12.6	=
12.6	Caveolin-1 254
12.6 12.7	Caveolin-1 254 The Role of Tyr14 Phosphorylation in Caveolin-1-Mediated
	Caveolin-1 254 The Role of Tyr14 Phosphorylation in Caveolin-1-Mediated Signaling 256
12.7	Caveolin-1 254 The Role of Tyr14 Phosphorylation in Caveolin-1-Mediated Signaling 256 Stress-Induced Changes in Caveolin-1 Expression 257
12.7	Caveolin-1 254 The Role of Tyr14 Phosphorylation in Caveolin-1-Mediated Signaling 256 Stress-Induced Changes in Caveolin-1 Expression 257 Concluding Remarks 258
12.7	Caveolin-1 254 The Role of Tyr14 Phosphorylation in Caveolin-1-Mediated Signaling 256 Stress-Induced Changes in Caveolin-1 Expression 257 Concluding Remarks 258 Acknowledgments 259

Index 265

Preface

This volume is the first book-length survey of caveolae and lipid rafts. Interest has developed rapidly in the role of these surface microdomains in such diverse fields as transmembrane signaling, cell locomotion, vascular relaxation, senescence, and the uptake and exit from cells of viruses and bacteria. Individual chapters in this volume cover areas as diverse as the forces that induce and maintain membrane invaginations, and the clinical relevance of multiprotein complexes at the cell surface, defects in which are associated with cancer, and Alzheimer's and prion-dependent diseases.

The book includes contributors from twelve countries. This reflects the growth and spread of these studies over the last twenty-five years, since the recognition of free cholesterol/sphingolipid (FC/SPH) rich cell microdomains as a distinct class of cell-surface structures. The historical origin of this concept is presented in the Overview by Drs Meder and Simons. Their chapter places caveolar and lipid rafts in context as biological membranes with special roles in information transfer across the plasma membrane, and between cell compartments. Other contributions described research, in a number of cases for the first time, in key areas of the molecular physiology of lipid-protein complexes in rafts and caveolae.

One area addressed in several chapters is the physical nature of FC/SPH-rich microdomains, their origin, and their lifetimes as independent structures within the cell membrane. Are planar lipid rafts, and invaginated caveolae, variations of a common theme? Can they be interconverted? Contributions by Dr Mayor and colleagues and by Drs Sens and Turner on the biophysics of lipid rafts and caveolae respectively, break new ground in describing the techniques now available to analyze these structures. They also summarize the physical forces sustaining them in the cell membrane.

A second group of chapters describes how these physical forces at the surface regulate cell behaviour. The chapter by Dr van Deurs and colleagues addresses one of the most important of these, the significance of caveolae in endocytosis, a pathway alternate to that mediated by clathrin-coated pits. The balance of evidence indicates that though caveolae are usually relatively stable, they can be induced to bud off by a variety of physiological and pathological stimuli. The chapter by Drs Park and Cho reviews the role of caveolae in maintaining cell shape, and promoting locomotion, though their interaction with the actin skeleton.

A third group of contributions deals with recent research on the factors controlling the level of lipid (especially FC) and protein (especially the structural protein caveolin) in cell surface caveolae. Drs Everson and Smart summarizes research indicating that caveolins play key roles in FC homeostasis and intracellular transport, in addition to promoting caveola formation at the cell surface. The chapter by Drs C. and P. Fielding describes mechanisms by which the level of FC at the cell surface affects the structure and function of caveolae, and re-examines the concept of caveolin as scaffold. Data from these and other laboratories have shown that the composition of caveolae is dynamically regulated, and can respond to the extracellular environment as well as the current internal needs of the cell.

One of the best characterized functions of FC/SPH-rich domains is to serve as an assembly point for the multiprotein complexes that promote signal transduction. Recent research has identified structural features that contribute to the assembly of these complexes. These include reversible covalent modifications, such as changes in protein phosphorylation and acylation, that promote assembly and disassembly, leading to signal propagation to the nucleus. The chapter by Dr Damjanovitch and colleagues gives a detailed review of the most recent data on these events in lipid rafts. Dr Mastick and colleagues dissect the regulatory role of caveolin within caveolae during signaling via transmembrane growth factor receptors.

Perhaps the most detailed of all investigation in the field of caveolae has concerned the mechanism by which the activity of endothelial nitric oxide synthase (eNOS) a major determinant of vascular relaxation, is regulated by its association with caveolae. The chapter by Dr Feron, based on the work of his own and other laboratories, describes in detail the complex relationship between e-nos, caveolin and FC.

A final group of chapters discusses the roles of FC/SPH rich domains in human disease. Drs David and Liscovitch discuss the roles of caveolin and caveolin in cancer cells, and show that under different conditions both positive and negative pathways can be identified. The chapter by Dr Zurzolo and colleagues describes the recently identified roles of FC/SPH domains in the pathophysiology of prionbased and Alzheimer's diseases.

The present volume has two main purposes. It brings together current hypotheses about the structure and functions of lipid rafts and caveolae by leading experts. These ideas will be of interest to biophysicists, biochemists, cell biologists and clinicians who study biological membranes. The book also provides a convenient reference work summarizing published work in this rapidly growing area. We hope that this will assist the research of a new generation of investigators drawn to this subtle but fascinating field, with its ramifications in many areas of current biology.

San Francisco, December 2005

Christopher J. Fielding

Author List

Andrea Bodnár Cell Biophysics Research Group of the Hungarian Academy of Sciences Medical and Health Science Center Research Center for Molecular Medicine University of Debrecen Nagyerdei krt. 98 4012 Debrecen Hungary

Vincenza Campana Dipartimento di Biologia e Patologia Cellulare e Molecolare Centro di Endocrinologia ed Oncologia Sperimentale del Consiglio Nazionale delle Ricerche Università degli Studi di Napoli Federico II 80131 Napoli Italy and Unité de Trafic Membranaire et Pathogénèse Institut Pasteur 25 rue du Dr. Roux 75724 Paris Cedex 15 France

Haiming Cao Department of Biochemistry and Molecular Biology School of Medicine and College of Agriculture, Biotechnology, and Natural Resources University of Nevada Reno NV 89557 USA

Kyung A. Cho Department of Biochemistry and Molecular Biology Seoul National University Medical School 28 Yon Gon Dong Chong-No Ku Seoul 110–799 Korea

Sándor Damjanovich Cell Biophysics Research Group of the Hungarian Academy of Sciences and Department of Biophysics and Cell Biology Medical and Health Science Center Research Center for Molecular Medicine University of Debrecen Nagyerdei krt. 98 4012 Debrecen Hungary

XIV Author List

William V. Everson
Department of Pediatrics
University of Kentucky
423 Sanders-Brown Center on Aging
800 South Limestone Street
Lexington
KY 40536–0230
USA

Olivier Feron University of Louvain Medical School Unit of Pharmacology and Therapeutics UCL-FATH 5349 53 Avenue E. Mounier 1200 Brussels Belgium

Christopher J. Fielding Cardiovascular Research Institute and Department of Physiology University of California San Francisco San Francisco CA 94143 USA

Phoebe E. Fielding Cardiovascular Research Institute and Department of Medicine University of California San Francisco San Francisco CA 94143 USA

Anette M. Hommelgaard Structural Cell Biology Unit Department of Medical Anatomy The Panum Institute University of Copenhagen 2200 Copenhagen N Denmark Mordechai Liscovitch Department of Biological Regulation Weizmann Institute of Science Rehovot 76100 Israel

Cynthia Corley Mastick Department of Biochemistry and Molecular Biology School of Medicine and College of Agriculture, Biotechnology, and Natural Resources University of Nevada Reno NV 89557 USA

Satyajit Mayor National Centre for Biological Sciences UAS-GKVK Campus Bangalore 560065 India

Doris Meder Max Planck Institute of Molecular Cell Biology and Genetics Pfotenhauerstr. 108 01307 Dresden Germany

Sang Chul Park Department of Biochemistry and Molecular Biology Seoul National University Medical School 28 Yon Gon Dong Chong-No Ku Seoul 110–799 Korea

Dana Ravid Department of Biological Regulation Weizmann Institute of Science Rehovot 76100 Israel Kirstine Roepstorff Structural Cell Biology Unit Department of Medical Anatomy The Panum Institute University of Copenhagen 2200 Copenhagen N Denmark

Kirsten Sandvig Institute for Cancer Research Department of Biochemistry The Norwegian Radium Hospital Montebello 0310 Oslo Norway

Amy Sanguinetti Department of Biochemistry and Molecular Biology School of Medicine and College of Agriculture, Biotechnology, and Natural Resources University of Nevada Reno NV 89557 USA

Daniela Sarnataro Dipartimento di Biologia e Patologia Cellulare e Molecolare Centro di Endocrinologia ed Oncologia Sperimentale del Consiglio Nazionale delle Ricerche Università degli Studi di Napoli Federico II 80131 Napoli Italy

Pierre Sens Phisico-Chimie Theorique ESPCI 10 rue Vauquelin 75231 Paris cedex 05 France Pranav Sharma Cold Spring Harbor Laboratories Cold Spring Harbor New York 11724 USA

Kai Simons Max Planck Institute of Molecular Cell Biology and Genetics Pfotenhauerstr. 108 01307 Dresden Germany

Eric J. Smart Barnstable-Brown Endowed Chair of Diabetes University of Kentucky 429 Sanders Brown Center on Aging 800 South Limestone Street Lexington KY 40536–0230 USA

János Szöllősi Cell Biophysics Research Group of the Hungarian Academy of Sciences and Department of Biophysics and Cell Biology Medical and Health Science Center Research Center for Molecular Medicine University of Debrecen Nagyerdei krt. 98 4012 Debrecen Hungary

Suhani Thakker Department of Biochemistry and Molecular Biology School of Medicine and College of Agriculture, Biotechnology, and Natural Resources University of Nevada Reno NV 89557 USA

XVI Author List

Maria Torgersen Institute for Cancer Research Department of Biochemistry The Norwegian Radium Hospital Montebello 0310 Oslo Norway

Matthew S. Turner Department of Physics Warwick University Coventry CV4 7AL UK

György Vámosi Cell Biophysics Research Group of the Hungarian Academy of Sciences Medical and Health Science Center Research Center for Molecular Medicine University of Debrecen Nagyerdei krt. 98 4012 Debrecen Hungary

Bo van Deurs Structural Cell Biology Unit Department of Medical Anatomy The Panum Institute University of Copenhagen 2200 Copenhagen N Denmark

Rajat Varma Program in Molecular Pathogenesis Skirball Institute for Biomolecular Medicine New York University School of Medicine 540 First Avenue New York 10016 USA György Vereb Department of Biophysics and Cell Biology Medical and Health Science Center Research Center for Molecular Medicine University of Debrecen Nagyerdei krt. 98 4012 Debrecen Hungary

Frederik Vilhardt Structural Cell Biology Unit Department of Medical Anatomy The Panum Institute University of Copenhagen 2200 Copenhagen N Denmark

Chiara Zurzolo Dipartimento di Biologia e Patologia Cellulare e Molecolare Centro di Endocrinologia ed Oncologia Sperimentale del Consiglio Nazionale delle Ricerche Università degli Studi di Napoli Federico II 80131 Napoli Italy and Unité de Trafic Membranaire et Pathogénèse Institut Pasteur 25 rue du Dr. Roux 75724 Paris Cedex 15 France

1 Lipid Rafts, Caveolae, and Membrane Traffic

Doris Meder and Kai Simons

1.1 Introduction

Cell membranes are dynamic assemblies of a variety of lipids and proteins. They form a protective layer around the cell, but also mediate the communication with the outside world - that is, neighboring cells in a tissue, hormones and growth factors arriving with the blood supply, or pathogens trying to enter the system. The unique feature of cell membranes is that their lipid and protein constituents can self-assemble into 5 nm-thin, two-dimensional fluids composed of two apposing lipid monolayers that form a hydrophobic interior and two polar interfacial regions oriented towards the aqueous medium. This organizing principle - the lipid bilayer - is the oldest, still valid molecular model of biological structures. The first model that incorporated proteins was proposed by Danielli and Davson, and assumed that the bilayer was made up entirely of lipids and that proteins covered the two polar surfaces [1]. Some 40 years later, the fluid mosaic model of the cell membrane proposed by Singer and Nicolson [2] was a conceptual breakthrough. Amphipathic membrane proteins were recognized to reside within, and even span, the whole bilayer that was depicted as a dynamic structure, the components of which are laterally mobile. However, the view that the lipids in the bilayer mainly serve as a homogeneous solvent for proteins [2] has been proven to be too simplistic. Lipids are not only distributed asymmetrically between the two leaflets of the bilayer, but also within the leaflet they are heterogeneously arranged [3]. This chapter will recapitulate the history and recent advances in membrane biology including the lipid raft concept, and then summarize current views on the functions of rafts and caveolae in membrane traffic.

י|

1.2 Basic Organization Principles of a Cell Membrane

The lipid bilayer is a two-dimensional fluid, where lipid molecules exchange slowly between leaflets but are mobile within the leaflet. This mobility consists of two parts:

- 2 1 Lipid Rafts, Caveolae, and Membrane Traffic
 - the "translational freedom" of a molecule that is, its lateral mobility; and
 - the "configurational freedom" that is, the ability to flex parts of the molecule and to rotate bonds in its carbon backbone.

Synthetic bilayers change from a liquid state with high translational and configurational freedom into a rigid gel state at a characteristic freezing point. Cell membranes at physiological temperatures are almost always in the liquid state, but can contain regions with high configurational order, as will be described later. Importantly, the lipid bilayer of cell membranes is asymmetric, with a different lipid composition in the two leaflets. The main lipid components of cellular membranes are glycerophospholipids, with the most abundant species being phosphatidylcholine (PC) in the exoplasmic leaflet and phosphatidylethanolamine (PE) and phosphatidylserine (PS) in the inner leaflet, as well as sphingolipids with glycosphingolipids and sphingomyelin (SM) mostly localized to the exoplasmic leaflet. Sterols make up the third lipid class, and are present in both leaflets. Mammalian cell membranes contain only one sterol, namely cholesterol, but probably more than thousand different glyco- and sphingolipid species, emerging from the combinatorial propensity to assemble lipids from different backbones linked in different ways with two varying hydrocarbon chains and a vast number of headgroups. A large number of flippases and translocators tightly control the asymmetric distribution of all these lipids across the bilayer [4].

Lipids are differentially distributed between cellular organelles. The endoplasmic reticulum and the Golgi-complex contain mainly glycerophospholipids and only small amounts of sphingolipids, whereas the plasma membrane is relatively enriched in SM and glycosphingolipids [5]. Also within the membrane plane of one organelle, lipids are believed to be heterogeneously arranged. Caveolae - small invaginations of the plasma membrane – are enriched in glycosphingolipids [6], and phosphatidylinositol-3'-phosphate (PI(3)P) is concentrated in subdomains of early endosome membranes [7]. Recently, vacuole-fusion in yeast has been shown to be controlled by microdomains of ergosterol, diacylglycerol and phosphoinositide-3-and-4-phosphate [8]. Furthermore, membranes are differentially susceptible to extraction by detergents such as Triton X-100 or CHAPS at 4 °C, with some proteins and lipids being completely solubilized and others forming so-called "detergent-resistant membranes" (DRM; for a review, see [9]). These findings suggested that cell membranes contained microdomains in which lipids were more tightly packed and thus not accessible to the detergent, although it is widely accepted that DRMs do not have an exact *in-vivo* correlate but are defined by being formed during the detergent treatment [10]. These microdomains were later termed "rafts" and were described as sphingolipid-cholesterol assemblies containing a subset of membrane proteins [11]. Currently, the raft hypothesis is heavily debated [12-14], with the main discussion points being the methodologies to study rafts and the size of the domains (see below). The core of the raft concept is that cell membranes phase-separate into different domains and that this is a lipiddriven process. In light of the ongoing discussion in the field, the following sections will provide an overview about what is known about phase separation, first

discussing the studies conducted in model membrane systems and later in cell membranes.

1.3 Evidence for Phase Separation in Model Membrane Systems: Liquid-Ordered and Liquid-Disordered Phases

Various model membrane systems have been used by physicists and chemists to study phase separation in lipid mixtures. They are either monolayers or bilayers. Monolayers are either assembled at an air-water interface with the packing density of the lipids being adjusted by applying lateral pressure, or on a supporting lipid monolayer that is fixed to a solid support. Bilayers are used in the supported version as described above, or in the form of vesicles. The most commonly used vesicles are large or giant unilamellar vesicles (LUV or GUV, respectively) composed of only a single bilayer, but also multilamellar vesicles (MLV) are used. The basic principles were first established in simple binary lipid mixtures, but recently ternary mixtures which more closely mimic the composition of the cell plasma membrane have been used. The mixtures usually contain one lipid with a high melting temperature (T_m), one with a low T_m , and cholesterol. GUVs are probably the system closest to a cell membrane, because artifacts from a support are excluded. Still, cell membranes are asymmetric with different lipid compositions of the outer versus the inner leaflet, while the GUVs used so far were all symmetric. Since maintaining an asymmetric lipid distribution is energy-consuming, perhaps by reconstituting lipid translocators into liposomes this drawback can be overcome in the future. Although model membrane systems produce very simplified pictures of cell membranes, there are many examples of a close correlation with experimental data obtained in living cells [14].

Ipsen et al. were the first to describe the formation of a liquid-ordered phase by cholesterol and saturated phospholipids [15,16]. This phase can coexist with other lipid phases, and its characteristics are described as follows: the translational order of lipid molecules within the liquid-ordered phase is similar to that in a fluid bilayer state, whereas the configurational order of the hydrocarbon chains compares more to that in a gel state. The formation of the liquid-ordered phase was attributed to the unique chemical nature of cholesterol (for a review, see [17]), but later it was shown that all natural sterols promote domain formation and that also small amounts of ceramide (3%) can stabilize domains formed in vesicles [18]. Leventis and Silvius showed that the interaction of cholesterol with different lipid species is dependent on the nature of their hydrocarbon chains and, to a lesser extent, also on their headgroup. The interaction preference decreases with SM > PS > PC > PE and with increasing unsaturation of the acyl chains [19]. Whereas the kink in unsaturated hydrocarbon chains is likely to hinder tight packing with the flat sterol ring of cholesterol, the reason for the preferential interaction of cholesterol with SM is still a debated issue.

The first visualization of "raft-like domains" in model membranes was achieved

4 1 Lipid Rafts, Caveolae, and Membrane Traffic

by Dietrich et al. [20]. They visualized liquid ordered domains in supported bilayers and GUVs composed not only of synthetic lipid mixtures but also of lipid extracts from brush border membrane, the apical membrane of intestinal cells. Domain formation was cholesterol-dependent, since domains disappeared after treatment with the cholesterol-extracting drug methyl-ß-cyclodextrin. Another big step forward was the establishment of a ternary phase diagram of SM/PC/cholesterol at the physiological temperature of 37 °C [21]. This predicts the coexistence of liquid-ordered and liquid-disordered phases for a wide range of compositions mimicking those occurring in the plasma membrane of cells. Most domains observed in model membranes are rather large (i.e., several micrometer in diameter) or they start small when they are being formed and then grow continuously by collision and fusion as the system reaches equilibrium [22]. Contrary to this, raft domains in cells are believed to be small, most likely because the cell membrane is not at equilibrium (see below). Interestingly, fluorescence resonance energy transfer (FRET) measurements on vesicles composed of a ternary lipid mixture mimicking the outer leaflet of the plasma membrane revealed heterogeneities (i.e., domains) of sizes in the tens of nanometer range at 37 °C [23]. Large domains were observed with the same lipid mixture only below 20 °C.

A slightly different interpretation of liquid-liquid immiscibility observed in model membranes was proposed by McConnell and colleagues. These authors argue for the formation of "condensed complexes" between cholesterol and SM rather than a liquid-ordered phase or domain. The name originates from the observation that cholesterol and SM occupy less surface area when mixed together compared to the sum of the areas occupied by each component alone before mixing. Such a complex is supposed to contain 15–30 molecules with a fixed stoichiometry of 2:1 (SM:cholesterol). These complexes could exist in quite high concentration without necessarily leading to a phase separation (for a review, see [24]). However, the condensed complex theory was developed on monolayer membranes and has not yet been validated for bilayers.

Taken together, there is clear evidence for lipid-driven domain formation in model membrane systems mimicking the outer leaflet of the plasma membrane. On the contrary, domain formation could not be observed in lipid mixtures mimicking the inner leaflet of the plasma membrane [25]. The intermolecular forces leading to phase separation are van der Waals interactions between saturated acyl chains and cholesterol, as well as forces such as hydrophobic shielding or the "umbrella effect", described for cholesterol filling the holes left between the acyl chains of glycosphingolipids with large headgroups [26]. However, none of the systems described so far has included proteins in their analysis, and the question remains whether proteins choose the domain they partition into, or whether they organize a domain around them.

Partitioning experiments have been performed, in which proteins were reconstituted into model membranes, and their phase distribution was analyzed. In this way, glycosyl-phosphatidyl-inositol (GPI)-anchored placental alkaline phosphatase (PLAP; [27,28] and Thy-1 [29] were shown to partition into the liquid-ordered phase, and the chain length of the GPI-anchor was shown to be important for partitioning of the protein [30]. Similarly, peptides modified with prenyl groups were excluded from liquid-ordered domains, while peptides modified with cholesterol or palmityl chains partitioned significantly into the ordered phase [31]. Partitioning studies with synthetic transmembrane peptides revealed that longer transmembrane domains are incorporated better into liquid-ordered domains than shorter versions [32]. Another important determinant for the partitioning of a molecule is the size and orientation of its dipole moment [33]. The membrane dipole moment is stronger in ordered phases where the dipoles are better aligned. Only molecules displaying a dipole moment with the same orientation as the dipolar potential of the membrane, are predicted to be able to enter the ordered phase. Nevertheless, our knowledge about lipid-transmembrane protein interactions is still scarce and this area of research is a major challenge.

1.4 Evidence for Phase Separation in Cell Membranes: The "Raft Concept"

There are several indications for cell membranes being inhomogeneous fluids and for the existence of lipid-driven phase separation. One key finding was the selective co-clustering of certain membrane components and segregation from others upon application of antibodies to living cells. Co-clustering of lipids was first observed in lymphocytes, where one ganglioside species was capped with antibodies and another species was found to redistribute into the cap [34]. It was then shown that simultaneous addition of two antibodies against apparently homogeneously distributed surface antigens could, in selected cases, lead to their co-clustering and in other cases to their segregation [35]. These findings were explained by certain proteins residing in small raft domains that are below the light microscopic resolution in size, and others residing outside the raft domains. Upon cross-linking by antibodies the small raft domains coalesce into visible, stable clusters that contain several different raft proteins. The antigens that were previously in the non-raft environment are excluded from the coalescing domains and thus form separate clusters upon cross-linking. How these large-scale domains containing multiple raft components could be formed in a homogeneous membrane without the occurrence of phase separation is not obvious, and an alternative explanation for this phenomenon has not been put forward. Since then, two techniques have been used to directly assess liquid order in living cells. Gidwani et al. measured the steady-state anisotropy of the lipid-probe DPH-PC, which is sensitive to cholesterol-induced liquid order. With this approach, they found that approximately 40% of the plasma membrane of mast cells is in a liquid-ordered state [36]. More recently, Gaus et al. were able to directly visualize liquid-ordered domains in living macrophages on the light microscopic level. They applied two-photon imaging of the amphiphilic dye LAURDAN, which changes its emission peak depending on the state of its lipid environment [37].

Other techniques have also been employed for assessing raft domains in living cells, most of them analyzing the distribution and dynamics of membrane pro-

6 1 Lipid Rafts, Caveolae, and Membrane Traffic

teins rather than lipids. Pralle et al. measured the local diffusion of a bead attached to a single protein molecule in the plasma membrane of fibroblasts within an area smaller than 100 nm in diameter [38]. In this way, diffusion was not hindered by cytoskeletal constraints but was supposed to be free. Proteins previously shown to be resistant to detergent extraction diffused three times slower than detergent-soluble proteins. After cholesterol depletion, the former diffused as fast as the latter. The first group of proteins was thus assumed to reside in a raft environment and to diffuse together with the whole raft entity. After destruction of this entity by cholesterol extraction the proteins behaved as if they were diffusing in a non-raft environment. From the viscous drag and from the diffusion coefficient, the size of the raft entities was calculated to be approximately 50 nm in diameter. Extrapolated from average protein and lipid densities in cell membranes, one raft entity was calculated to contain roughly 3000 lipid molecules and 10–20 proteins.

Remarkably, Prior et al. come to a very similar size for raft domains formed in the cytoplasmic leaflet of the plasma membrane using a completely different technique [39]. They ripped plasma membrane sheets off adherent cells and labeled them with gold-coupled antibodies against H-Ras and K-Ras, supposed to reside inside and outside of raft domains, respectively. Statistical analysis of the distribution of the gold particles revealed that 35% of H-Ras labels were clustered in domains of roughly 44 nm diameter. These domains were cholesterol-dependent. Furthermore, cross-linking of GPI-anchored green fluorescent protein (GFP-GPI) in the exoplasmic leaflet resulted in co-localization of the H-Ras clusters with the formed GPI-patches, but did not change their size. However, 20% of the non-raft protein K-Ras was also found to be clustered in domains of 32 nm diameter, although these domains were cholesterol-independent. H- and K-Ras had been reported to occupy distinct domains in the plasma membrane before [40]. Recently, single molecule imaging of H-Ras revealed cholesterol and actin dependent domains as large as 250 nm [41].

An often-applied technique trying to visualize raft domains *in vivo* is that of FRET. Hetero-FRET, which detects energy transfer between two different fluorophores, has not proven successful [42–44], most likely because the probability that donor and acceptor are in the same microdomain is very low. Even crosslinking one raft marker by antibodies does not lead to appreciable recruitment of others [45]. Recently, Mayor and coworkers refined their previous analysis [46] using homo-FRET (i.e., energy transfer between two fluorophores of the same kind) to study clustering of GPI-anchored proteins in the plasma membrane [47]. By measuring the anisotropy decay over time, these authors found that 20–40% of the GPI-anchored proteins are present in small complexes of two to four molecules, while the remainder is randomly distributed as monomers. The limitation of FRET measurements becomes obvious in these studies. The technique provides information about "closeness" on a very small scale (5 nm), but is not suited for visualizing bigger entities.

The fact that raft domains are difficult to visualize *in vivo* has led to a number of alternative explanations, mostly describing smaller entities and, most importantly, describing the formation of these entities as a protein-driven, induced event. The

smallest entity was proposed by Kusumi and colleagues, who have pioneered single-particle tracking with ultra-high sampling frequencies of 40 000 Hz. The spatial resolution achieved with this frequency is 20 nm, meaning that if the domains were significantly larger and the probe resided either inside or outside the domain for several consecutive steps, then different diffusion behaviors could be observed. Since however raft and non-raft markers displayed the same diffusion characteristics, it was postulated that rafts are extremely small, namely molecular complexes of at least three membrane components, one of which comprises a saturated acyl chain or cholesterol. Stabilized raft domains accessible to diffusion measurements would only form by clustering following stimulation (for a review, see [48]). Anderson and Jacobson have put forward the lipid shell hypothesis, in which roughly 80 lipid molecules are supposed to surround a raft protein and form a shell of 7 nm diameter [49]. The shells would be thermodynamically stable structures resulting from specific binding interactions between proteins and lipids, and could target the protein into larger raft-domains. How the larger raft domains form and why the raft-protein must assemble a shell of raft-lipids around it before it can enter a raft-domain remain open questions.

The size of raft domains is heavily debated and, as a consequence of the different measurements, their existence is questioned. Consensus is reached in that the proposed domain sizes of 200 nm or larger based on single-particle tracking experiments [50,51] were most likely clustered rafts, formed and stabilized by the multi-valent beads used for the tracking. Also, the 50-nm raft calculated from the viscous drag experiments by Pralle et al. [38] could have been a stabilized raft in which the altered dynamics due to the optical trap led to enlargement of a previously smaller structure. This leaves us with a domain size between the 5 nm derived from the FRET measurements [47] and the <20 nm derived from the high-speed single particle tracking studies [52]. Better estimates will have to await the development of new methods which can finally assess the size of isolated raft domains *in vivo*.

In light of the co-clustering data [35], the visualization of distinct liquid-ordered domains in living cells [37], and the evidence that isolated cell membranes phase separate *in vitro* [20], it seems reasonable to assume that native cell membranes can display phase separation. One explanation for the formation of small and transient domains in the plasma membrane lies in its composition. In contrast to ternary lipid mixtures in model systems, the plasma membrane is composed of hundreds of different lipid species and, in addition to that, a variety of proteins. Viewed over a large scale, the complexity of the plasma membrane should counteract phase separation, buffer fluctuations, and in fact protect the cell against rapid phase transitions in response to small changes in the environment. If every fusion or budding event led to a phase transition, it would be difficult to prevent leakages through the bilayer and keep the membrane tight. Viewed on a smaller scale however, the picture can appear very different. Local impurities or changes in membrane composition can allow coalescence and separation of domains containing reaction partners and thus provide a regulatory principle.

1.5

Raft Domains are Clustered to Exert their Function

While the steady-state existence, size and shape of liquid-ordered domains in cells remains the subject of debate, agreement has been reached on the fact that raft domains coalesce upon cross-linking to form signaling and possibly also sorting platforms [53–55]. Cross-linking is achieved by multivalent ligands binding to surface receptors or by cytoplasmic scaffolding proteins. The initial cross-linking event is thought to increase the number of contact sites between raft proteins and lipids, which leads to a potentiation of the formerly weak interactions. The previously small raft domains coalesce and form large, more stable entities. It is the clustered state in which rafts are accessible to microscopy.

Cross-linking of raft antigens not only leads to co-clustering of raft components within one leaflet, but also influences the organization of the opposing monolayer. Cross-linking of the exoplasmic GPI-anchored PLAP led to partial co-clustering of the src-kinase fyn in the cytoplasmic leaflet of Jurkat cells [35, 56]. Cross-correlation analysis revealed co-distribution of an inner leaflet raft protein with FceRI transmembrane receptors that were cross-linked by binding of their multivalent ligand IgE, as well as with antibody cross-linked raft markers of the exoplasmic leaflet, such as the GPI-anchored protein Thy-1 or the ganglioside GD_{1b} [57]. The finding that clustering not only leads to lateral coalescence of small raft domains in the exoplasmic leaflet, but also in the cytoplasmic leaflet, strengthens the hypothesis that clustered raft domains provide a platform for bringing together signaling complexes and propagating signals into the cell (reviewed in [58]). Interestingly, also in symmetric model bilayers, liquid-ordered domains have always been observed to coincide in both leaflets [20, 59]. How the connection of the inner leaflet and the outer leaflet is achieved, remains an open question. Interdigitation of the often long fatty acid chains of glycosphingolipids has been proposed to enforce a higher order also in the cytoplasmic leaflet. Alternatively, or additionally, transmembrane proteins could mediate transbilayer coupling.

Many signaling processes have been proposed to depend on the clustering of raft domains [60,61] (see also Chapter 7), the T-cell synapse being the prime example [62,63]. According to a recent study by Douglass et al., the initial stage of signaling complex assembly does not require rafts but is rather dependent on proteinprotein interactions [64]. Studies by Magee et al., on the other hand, have shown that raft clustering independent of protein-protein interactions can activate signaling pathways downstream of the T-cell receptor [65]. These authors observed that incubating T cells at 0 °C leads to coalescence of raft components into visible domains on the plasma membrane. At the same time, chilling activates the signaling cascade, leading to increased tyrosine phosphorylation and ERK activation. The cold-induced, protein-independent coalescence of raft domains is a clear indicator for a phase separation phenomenon, since it is well established that the phaseseparated domains are larger at lower temperature and fragment at higher temperature due to the increase in Brownian motion [66]. However, it is not yet clear which role this raft coalescence would play in T-cell signaling under physiological conditions.

The formation of large, clustered raft domains is easiest imagined to occur by coalescence of pre-existing, small rafts. However, a recent study on model membranes of different compositions argued that phase separation can be induced by cross-linking one component in a previously homogeneous membrane [67]. GUVs composed of PC, SM and cholesterol exhibit phase separation into a liquid-ordered and a liquid-disordered phase, depending on the ratio of the components. When a small amount of the ganglioside GM1 is included in the vesicles, its cross-linking with the pentavalent cholera toxin B subunit leads to coalescence of the GM1-containing phase into larger, visible domains. Hammond et al. showed that domains can not only be formed at GUV compositions that displayed phase separation prior to clustering, but also at compositions very close to the phase transition boundary in which no previous phase separation was detected [67]. The local increase in GM1 concentration following the cross-linking might have been enough to cross the boundary and cause the membrane to phase separate.

1.6 The Apical Membrane of Epithelial Cells: A Percolating Raft Membrane at 25 °C

Columnar epithelia lining the kidney, intestine or pancreas are composed of a single layer of polarized cells. They have evolved to create stable apical and basolateral membrane domains, which are sealed off from each other by a tight junction barrier. While the basolateral domain of columnar epithelia faces the underlying extracellular matrix and the blood supply, the apical membrane is the one facing the lumen of the renal tubules, of the intestine, or of the pancreas. It has long been known that apical and basolateral membrane domains have a distinct protein composition [68,69]. However, lipids are also distributed differently between the apical and the basolateral membrane. The lipids found in the basolateral membrane resemble those found in the plasma membrane of an unpolarized cell, whereas the apical membrane contains much more glycosphingolipids [70]. In the brush border membrane of the intestine, glycosphingolipids account for more than 30% of the total lipid amount [71]. Considering that they reside exclusively in the exoplasmic leaflet, more than 50% of the lipids in the exoplasmic leaflet should be glycosphingolipids, and together with cholesterol they should leave very little space for glycerophospholipids. Glycosphingolipids mainly contain two long, saturated hydrocarbon chains, as opposed to glycerophospholipids which usually contain unsaturated acyl chains [72], and have been proposed to form a liquidordered phase together with cholesterol. It was this segregation of raft lipids in the outer leaflet of the apical membrane from the more phosphatidylcholine-enriched basolateral membrane that prompted Simons and van Meer to postulate the existence of lipid platforms involved in the biogenesis of the apical membrane [70] and has led to the formulation of the raft hypothesis [11].

Recently, we have experimentally explored the domain organization of the apical membrane of epithelial cells in comparison to that of a fibroblast plasma membrane by measuring long-range diffusion of several fluorescent membrane pro-

10 1 Lipid Rafts, Caveolae, and Membrane Traffic

teins using fluorescence recovery after photobleaching (FRAP) [73]. By using this technique, the diffusion of millions of proteins can be examined at the same time in a noninvasive manner. As previously reported [74], all proteins display free diffusion with 100% recovery in the fibroblast plasma membrane. In the apical membrane of epithelial cells, however, we could distinguish two populations of proteins on the basis of their distinct diffusion characteristics at 25 °C. One group displayed free diffusion with recoveries close to 100%, whereas the other group displayed anomalous diffusion [75, 76] with limited recovery. This is indicative of a phaseseparated system, in which there are (at least) two coexisting phases - one which has a mass fraction just high enough to be continuous (percolating) over the entire membrane surface, and the other being present in isolated domains [77]. Within the percolating phase, long-range diffusion is unconstrained, results in complete recovery, and can be described with a single apparent diffusion coefficient [78, 79] - as observed for the first group of proteins. In the non-percolating phase, proteins will be obstructed in their long-range diffusion, resulting in either incomplete or extremely slow recovery [78, 79] - as observed for the second group of proteins. Strikingly, all proteins falling into the first group have been proposed to reside in rafts, while all members of the second group have been proposed to reside outside of rafts. This may suggest that at 25 °C the apical membrane of epithelial cells is a percolating raft phase with isolated non-raft domains.

Phase separation likely exists also in fibroblasts, with the domain organization of the two membranes being inverted. The fact that in the fibroblast plasma membrane the raft and non-raft proteins diffuse with the same kinetics does, however, not contradict the existence of phase separation. Rather, the results can be explained on the basis of partition coefficients. From all we know, a limited set of proteins has the features required to be accommodated in the ordered lipid environment of a raft domain. While non-raft proteins that lack these features are largely excluded from rafts – that is, non-raft proteins have a low propensity to partition into the surrounding raft phase in the apical membrane of epithelial cells - raft proteins might have a preference for raft domains, but can easily partition into a less-ordered, non-raft environment - that is, raft proteins are not limited to raft domains in the plasma membrane of fibroblasts [28, 80, 81]. With the additional notion that raft domains in fibroblasts are believed to be small and highly dynamic, the differences between the long-range diffusion paths of raft and nonraft proteins in the fibroblasts plasma membrane become too small to be accessible to FRAP measurements.

Caveolae: Scaffolded Membrane Domains Rich in Raft Lipids

Caveolae were first defined morphologically by Palade, who observed plasma membrane invaginations in endothelial cells under the electron microscope [82]. He later named them "plasmalemmal vesicles" [83], implying that they would shuttle molecules across the cell. The name "caveolae" (little caves) was however

coined two years later by Yamada, who described invaginations on the surface of gallbladder epithelial cells [84]. Although he did not distinguish between coated and uncoated invaginations, the name "caveolae" was later specifically attributed to flask-shaped invaginations of 50 to 100 nm diameter that were devoid of the cla-thrin-coat, but instead displayed a characteristic striated coat [85]. While research on clathrin-coated pits and vesicles was rapidly progressing, caveolae long remained elusive.

This was changed when, almost 40 years after the morphological description, caveolin was identified as the major protein constituent of caveolae [86, 87]. Subsequently, two additional caveolin genes were cloned, so that the original caveolin was from then on referred to as caveolin-1. Caveolin-2 was co-purified with caveolin-1 from adipocytes [88], and its expression pattern overlaps with that of caveolin-1. The two proteins are most abundant in endothelial cells, fibroblasts and adipocytes, and they form stable hetero-oligomeric complexes *in vivo* [89]. Caveolin-3 shows a high degree of sequence similarity with caveolin-1, but its expression is restricted to muscle cells in which there is low caveolin-1 expression [90]. Both caveolin-1 and -2 have a smaller β -isoforms in addition to the full-length α -isoform. Caveolin-1 assumes an unusual topology in that it is an integral membrane protein [91] but does not span the bilayer. Instead the central hydrophobic domain is thought to form a hairpin structure which inserts into the cytoplasmic leaflet, leaving both the N- and C-terminus in the cytoplasm [87].

A characteristic feature of caveolins is their propensity to form high molecularweight homo- and hetero-oligomers. Highly stable caveolin-1 oligomers of 14 to 16 monomers, dissociating only upon harsh detergent treatment at elevated temperatures, were found to be assembled relatively rapidly after synthesis of caveolin-1 in the endoplasmic reticulum and prior to Golgi exit [92]. The domain responsible for the oligomerization was mapped to the N-terminus [93]. The N-terminus has also been shown to target caveolin-1 to caveolar invaginations at the plasma membrane, since its absence results in Golgi retention [94, 95]. This ensures that only caveolin oligomers, not monomers, are transported to the plasma membrane. In addition to homo-oligomerization, caveolin-1 can form similarly stable hetero-oligomers with caveolin-2, which are localized mainly to plasma membrane caveolae [89]. In the absence of caveolin-1, caveolin-2 is not able to oligomerize and is retained in the Golgi in the form of monomers and dimers [96–98], again indicating that only the oligomeric form is transported to the plasma membrane.

The fact that caveolin-1 immunostaining decorated the striated coat around plasma membrane caveolae [86], together with the observation that it self-assembled into filaments *in vitro* [92] indicated that it indeed was an integral coat component. Since then, the function of caveolae became very closely linked to the function of caveolin, and it was shown that formation of the stable plasma membrane invaginations depended on caveolin expression. Cells not expressing caveolin-1 (e.g., lymphocytes) lacked cell-surface caveolae, and the expression of caveolin-1 in these cells was sufficient to induce their formation [99]. Quantification of the number of caveolin-1 molecules per caveolae by fluorescence intensity distribution measurements revealed that the uniform size of caveolae as seen by electron mi-

12 1 Lipid Rafts, Caveolae, and Membrane Traffic

croscopy results from a quantal assembly mechanism in which 144 ± 39 caveolin-1 molecules are incorporated into a single caveola [100]; caveolin-2 was not assessed in this study. Caveolin-1 filaments had previously been proposed to assemble from heptamers, measuring 10 nm in diameter [101]. If this model were true, then 144 caveolin-1 molecules would form a filament of roughly 200 nm length, enough to surround an invagination of 50–100 nm diameter with a circumference of 150–300 nm once. The structure and composition of the caveolar coat are far from being understood (see also Chapter 2) but, most likely, caveolin-1 is not the only coat component. Other open questions are, where is the coat assembled and what is the assembly mechanism?

Caveolin-1 has been shown to bind cholesterol and the ganglioside GM1, both in vitro and in vivo [102, 103]. Cholesterol-binding occurs with high affinity, resisting even harsh detergent treatments [103]. The lipid composition of caveolae is thus similar to that of lipid rafts, and it can be extrapolated that the caveolar membrane should also display properties of a liquid-ordered phase. However, a detailed lipid composition of isolated caveolae is still lacking. The strong interaction with two bona fide lipid raft components predisposes caveolin-1 for the role as a raft-clustering agent. Similar to clustered rafts, caveolae have been proposed to function as signaling platforms [104] (see also Chapters 5, 6, and 11). The clear parallels in lipid composition and the partial co-purification of lipid raft and caveolar components in DRMs [105, 106], or in membranes of low buoyant density [107], has often led to an equation of the two membrane systems. However, we will continue to refer to caveolae as plasma membrane invaginations scaffolded by the caveolincoat. The stable membrane curvature of caveolae could be a result of two contributions. Curvature could be induced by: (1) the high cholesterol concentration [108]; and (2) the insertion of caveolin-1 into the cytoplasmic leaflet of the bilayer, which would increase the surface area of the cytoplasmic leaflet relative to that of the exoplasmic leaflet and thus promote inward bending of the membrane. This stabilization of a curved membrane structure and the presence of caveolins would distinguish caveolae functionally from lipid rafts.

1.8

Caveolae and Lipid Rafts in Membrane Traffic

Membrane traffic mediates the exchange of components between the different cellular organelles. Membrane proteins and lipids are synthesized in the endoplasmic reticulum and from there are transported to their subcellular sites of action [109, 110]. While peripheral membrane proteins as well as single lipids bound to lipid transfer proteins can shuttle between different membranes via the cytoplasm or through contacts between membranes [4], most membrane turnover is mediated by vesicular traffic. Directed vesicular transport involves several regulated steps:

 lateral sorting of membrane components according to their destination (i.e., the concentration of cargo following the same pathway and its segregation from cargo following different pathways);

- stabilization of a membrane domain destined for trafficking;
- bending of the membrane domain into the shape of a vesicle or tubule;
- pinching off from the donor compartment;
- traffic through the cytoplasm by passive diffusion or motor-protein-mediated transport along microtubules or actin filaments;
- fusion with the acceptor compartment; and
- release of the cargo.

The best-understood sorting mechanism for transmembrane proteins employs recyclable protein coats, such as clathrin-, COPI- or COPII-coats [111, 112] (Fig. 1.1, left panel). In this case the cargo proteins contain specific sorting signals in their cytoplasmic domains, which are bound by adaptor molecules, to which the coat proteins are recruited. Oligomerization of the coats leads to bending of the membrane domain into a vesicle, which is pinched off by the action of the GTPase dynamin and released into the cytosol. Here the coat disassembles, enabling the vesicle to fuse with its target membrane. This protein-driven mechanism operates by active inclusion of certain components and is not very efficient at excluding.

For other sorting events in membrane traffic, the lipid bilayer itself has been proposed to play the decisive role, and proteins only regulate what lipids can do on their own [113]. From theoretical considerations and model membrane studies it is known that if phases with different properties coexist in the same membrane, then the mismatch of interactions at the phase boundary leads to the so-called "line tension" - the two-dimensional equivalent of surface tension. Multiplied with the length of the phase boundary it gives rise to the "line energy". One way to minimize line energy is therefore to minimize the contact between phases. In the case of domains in cell membranes, this can be achieved by fusion of many small domains into one large domain, and bending the domain out of the surrounding bulk membrane [114]. The bending energy needed to curve the membrane as the domain buds out counteracts the line energy. As the bending energy increases and the line energy decreases, the domain reaches a stable curvature when the sum of the two energies is minimal. For small domains this can be when the domain is still connected to the bulk membrane, but above a critical domain size budding becomes energetically favorable. This mechanism is termed "domain-induced budding" (Fig. 1.1, right panel) and is initially achieved purely by lipid-driven phase separation [114]. However, in order to attain directionality in the budding process (i.e., budding towards the cytoplasm in most cases in cells) and also kinetics that are compatible with the cell's needs, proteins will have to control this process.

The fact that lipids are unevenly distributed between the two surfaces, the apical and the basolateral membrane domains, of epithelial cells [70] together with the finding that newly synthesized glucosyl-ceramide upon leaving the Golgi complex becomes two- to three-fold enriched in the apical versus the basolateral plasma membrane [115], has led to the proposal that lipids are also sorted by vesicular traffic. Interactions between glycosphingolipids and apical proteins were postulated to aid the assembly of sphingolipid microdomains in the Golgi that would

14 1 Lipid Rafts, Caveolae, and Membrane Traffic

concentrate apical cargo as the first step in vesicle formation [70]. This mechanism has two important features which distinguish it from the coat-mediated sorting:

- It also allows for the sorting of lipids.
- It works by actively excluding cargo that does not belong into the pathway and thus prevents the transported membrane from being diluted with inadequate material.

It has been shown previously, that basolateral proteins are excluded from the apical membrane [116], whereas the converse is not true [69, 117]. Physiologically this is sensible, since the apical membrane facing the lumen of an organ must be extremely resistant to external aggression by bile salt detergents, digestive enzymes or low pH, and its composition must therefore be tightly controlled. Whilst it is known that basolateral delivery depends on the interaction with adaptor proteins [118], domain-induced budding seems to be a mechanism ideally suited for delivery to the apical membrane.

Since these microdomains, or rafts, are believed to be small and dynamic, they must be clustered by proteins such as multivalent ligands or caveolin in order to be able to form a bud and later a vesicle or tubule. In apical raft delivery this has been postulated to be mediated by lectins or other multivalent cargo receptors [119, 120]. Raft and caveolar endocytosis is triggered by multivalent cargo, the best described being Simian virus 40 and cholera toxin [121–124], both of which bind several GM1 molecules [125]. Here, the caveolar coat is not necessary for the membrane bending or vesicle formation, since rafts can endocytose upon clustering by a virus or toxin and be delivered to specific destinations in the cell without caveolin [124]. In fact, the internalization has been shown to be faster in the absence of caveolin [126]. Caveolin might thus not be necessary for the endocytic event as such, but rather add another level of regulation to this pathway, which is required for the efficient sorting of some ligands [127].

Indeed, caveolae membrane traffic does display special features that set it apart from other membrane traffic mechanisms. Caveolae were previously believed to be static structures [128], simply increasing the cell-surface area and keeping raft

> Fig. 1.1 Two paradigms of cargo sorting and vesicle formation ► in membrane traffic: inclusion due to sorting signals followed by coat-driven budding (left), or exclusion due to phase separation and domain-induced budding (right). In the left column, proteins containing the appropriate cytoplasmic sorting signals (regardless of if they are residing in a raft or non-raft domain) are bound by adaptor proteins, on which the coat proteins assemble. For the clathrin-coat, membrane bending and subsequent budding is believed to be driven by a conformational change in the coat protein. In the right column, raft proteins are clustered by oligomerizing ligands or cytoplasmic scaffolding proteins, thereby excluding the group of non-raft proteins. Membrane bending and budding is driven by the need to minimize the line energy acting at the domain boundary.



non-raft domain raft domain in exoplasmic leaflet raft domain in cytoplasmic leaflet

GPI-anchored protein

ļ

ŝ

doubly acylated protein

transmembrane raft protein, bound to glycosphingolipid

doubly acylated transmembrane raft protein

oligomerizing ligand



6

transmembrane raft protein, phosphorylated on cytoplasmic side



two different non-raft transmembrane proteins

cytoplasmic crosslinking scaffold

adaptor protein



16 1 Lipid Rafts, Caveolae, and Membrane Traffic

membrane available on the cell surface. Recently, it became evident that even in unstimulated fibroblasts and epithelial cells, 30% of the caveolae undergo local kiss-and-run cycles with the plasma membrane in which they pinch off and fuse again close to the original site [100]. Upon receiving a trigger for endocytosis, caveolae switch from this short-range cycling to long-range cycling, resulting in an intermixing of cell-surface and intracellular caveolar vesicle pools and transport to caveosomes or endosomes [100]. During the trafficking event, the caveolar coat seems to stabilize the clustered raft domain within the bilayer, so that it stays intact even after fusion with the acceptor compartment and can be re-used for multiple rounds of membrane trafficking [127]. Cargo release at the target compartment must therefore also follow different principles than in the clathrin-coated vesicle traffic where the coat disassembles before fusion. Caveolae apparently keep their cargo sequestered, until its release is triggered by a compartment-specific cue. Cholera toxin is released upon encounter of a low pH environment in early endosomes, but stays sequestered in caveolae in the neutral environment at the plasma membrane or in caveosomes. This type of membrane traffic seems especially suited for the sorting of non-membrane spanning cargo, in particular glycosphingolipid-binding ligands [127].

The vesicle fusion machinery on the target compartment also has been proposed to be organized into domains of different lipid composition. The apical t-SNARE syntaxin 3 was proposed to reside in raft domains [129]. More recent investigations have claimed that indeed different SNAREs are compartmentalized in the plasma membrane with the help of lipid domains, with syntaxin 3 residing in raft domains, syntaxin 2 being excluded from raft domains, and syntaxin 4 being equally distributed between the two [130]. In polarized epithelial cells, syntaxin 4 resides on the basolateral surface, whereas syntaxin 2 and 3 are localized to the apical surface [131]. The data would thus imply, that there could be two pathways trafficking to the apical side of epithelial cells – one raft- and one non-raft pathway. Indeed, it was previously observed that two different apical proteins, sucrase-isomaltase and lactase-phlorizin-hydrolase, use separate containers for transport to the apical membrane of Madin-Darby canine kidney (MDCK) cells, and the existence of two different pathways was proposed [132].

Research on rafts and caveolae is entering a new phase. The technologies that have been used to study these membrane domains are being revised, and new technologies must be developed. If rafts are small and dynamic, many of the standard techniques that have been employed to visualize them (e.g., FRET, single particle tracking, FRAP in most cases) can not provide anything else but negative results because they are not suited for the size and time resolution needed. Another critical point is the purification methods used to isolate rafts or caveolae. The two were often confused with each other since they were supposed to co-fractionate when isolated based on detergent insolubility or light buoyant density. It is now accepted that these fractions are useful to obtain information about the proteins found in them, but since they form during the purification process, they can not be assumed to represent an equivalent of any pre-existing cellular domain, neither rafts nor caveolae [10]. Instead, new approaches have been taken – for example, to isolate plasma membrane fragments with small antibody-coated beads [133]. Techniques such as this must be developed in order to obtain pure raft and caveolae fractions that can be used to analyze their lipid and protein composition. With the new mass spectroscopic techniques it should then be possible to compare the lipidome of rafts and caveolae with each other to determine how similar they actually are, and also to compare them with the lipidome of the plasma membrane. Only then will we have a chance to assess properly the involvement of lipids in processes such as raft dynamics, raft clustering, and to address the special functions of caveolae.

Abbreviations

DRM	detergent-resistant membrane
FRAP	fluorescence recovery after photobleaching
FRET	Förster's resonance energy transfer
GFP	green fluorescent protein
GPI	glycosyl-phosphatidyl-inositol
GUV	giant unilamellar vesicles
LUV	large unilamellar vesicles
MDCK	Madin-Darby canine kidney
MLV	multilamellar vesicles
PC	phosphatidylcholine
PE	phosphatidylethanolamine
PI(3)P	phosphatidylinositol-3'-phosphate
PLAP	placental alkaline phosphatase
PS	phosphatidylserine
SM	sphingomyelin
T _m	melting temperature

References

- Danielli, J. F. and H. Davson, A contribution to the theory of permeability of thin films. *J. Cell. Comp. Physiol.* 1935; 5: 495–508.
- 2 Singer, S. J. and G. L. Nicolson, The fluid mosaic model of the structure of cell membranes. *Science* 1972; 175(23): 720–731.
- 3 Devaux, P. F. and R. Morris, Transmembrane asymmetry and lateral domains in biological membranes. *Traffic* 2004; 5(4): 241–246.
- **4** Holthuis, J.C. and T.P. Levine, Lipid traffic: floppy drives and a superhighway.

Nat. Rev. Mol. Cell. Biol. 2005; 6(3): 209–220.

- 5 Allan, D., Mapping the lipid distribution in the membranes of BHK cells (mini-review). *Mol. Membr. Biol.* 1996; 13(2): 81–84.
- 6 Tran, D., et al., Ligands internalized through coated or noncoated invaginations follow a common intracellular pathway. *Proc. Natl. Acad. Sci. USA* 1987; 84(22): 7957–7961.
- 7 Gillooly, D. J., C. Raiborg, and H. Stenmark, Phosphatidylinositol 3-phosphate is found in microdomains of early endo-

18 References

somes. Histochem. Cell. Biol. 2003; 120(6): 445–453.

- 8 Fratti, R. A., et al., Interdependent assembly of specific regulatory lipids and membrane fusion proteins into the vertex ring domain of docked vacuoles. *J. Cell Biol.* 2004; 167(6): 1087–1098.
- 9 Brown, D. A. and E. London, Functions of lipid rafts in biological membranes. *Annu. Rev. Cell. Dev. Biol.* 1998; 14: 111–136.
- 10 Lichtenberg, D., F.M. Goni, and H. Heerklotz, Detergent-resistant membranes should not be identified with membrane rafts. *Trends Biochem. Sci.* 2005; 30(8): 430–436.
- Simons, K. and E. Ikonen, Functional rafts in cell membranes. *Nature* 1997; 387(6633): 569–572.
- 12 Munro, S., Lipid rafts: elusive or illusive? Cell 2003; 115(4): 377–388.
- 13 Edidin, M., The state of lipid rafts: from model membranes to cells. Annu. Rev. Biophys. Biomol. Struct. 2003; 32: 257–283.
- 14 Simons, K. and W. L. Vaz, Model systems, lipid rafts, and cell membranes. Annu. Rev. Biophys. Biomol. Struct. 2004; 33: 269–295.
- 15 Ipsen, J. H., et al., Phase equilibria in the phosphatidylcholine-cholesterol system. *Biochim. Biophys. Acta* 1987; 905(1): 162–172.
- 16 Ipsen, J. H., O.G. Mouritsen, and M.J. Zuckermann, Theory of thermal anomalies in the specific heat of lipid bilayers containing cholesterol. *Biophys. J.* 1989; 56(4): 661–667.
- Miao, L., et al., From lanosterol to cholesterol: structural evolution and differential effects on lipid bilayers. *Biophys. J.* 2002; 82(3): 1429–1444.
- 18 Xu, X., et al., Effect of the structure of natural sterols and sphingolipids on the formation of ordered sphingolipid/sterol domains (rafts). Comparison of cholesterol to plant, fungal, and disease-associated sterols and comparison of sphingomyelin, cerebrosides, and ceramide. *J. Biol. Chem.* 2001; 276(36): 33 540–33546.
- 19 Leventis, R. and J. R. Silvius, Use of cyclodextrins to monitor transbilayer movement and differential lipid affinities of cholesterol. *Biophys. J.* 2001; 81(4): 2257–2267.

- 20 Dietrich, C., et al., Lipid rafts reconstituted in model membranes. *Biophys. J.* 2001; 80(3): 1417–1428.
- 21 de Almeida, R. F., A. Fedorov, and M. Prieto, Sphingomyelin/phosphatidylcholine/cholesterol phase diagram: boundaries and composition of lipid rafts. *Biophys. J.* 2003; 85(4): 2406–2416.
- 22 Veatch, S. L. and S. L. Keller, Separation of liquid phases in giant vesicles of ternary mixtures of phospholipids and cholesterol. *Biophys. J.* 2003; 85(5): 3074–3083.
- 23 Silvius, J. R., Fluorescence energy transfer reveals microdomain formation at physiological temperatures in lipid mixtures modeling the outer leaflet of the plasma membrane. *Biophys. J.* 2003; 85(2): 1034–1045.
- 24 McConnell, H. M. and M. Vrljic, Liquidliquid immiscibility in membranes. *Annu. Rev. Biophys. Biomol. Struct.* 2003; 32: 469–492.
- 25 Wang, T. Y. and J. R. Silvius, Cholesterol does not induce segregation of liquid-ordered domains in bilayers modeling the inner leaflet of the plasma membrane. *Biophys. J.* 2001; 81(5): 2762–2773.
- 26 Huang, J. and G. W. Feigenson, A microscopic interaction model of maximum solubility of cholesterol in lipid bilayers. *Biophys. J.* 1999; 76(4): 2142–2157.
- 27 Schroeder, R., E. London, and D. Brown, Interactions between saturated acyl chains confer detergent resistance on lipids and glycosylphosphatidylinositol (GPI)-anchored proteins: GPI-anchored proteins in liposomes and cells show similar behavior. *Proc. Natl. Acad. Sci. USA* 1994; 91(25): 12130–12134.
- 28 Kahya, N., D.A. Brown, and P. Schwille, Raft partitioning and dynamic behavior of human placental alkaline phosphatase in giant unilamellar vesicles. *Biochemistry* 2005; 44(20): 7479–7489.
- 29 Dietrich, C., et al., Partitioning of Thy-1, GM1, and cross-linked phospholipid analogs into lipid rafts reconstituted in supported model membrane monolayers. *Proc. Natl. Acad. Sci. USA* 2001; 98(19): 10642–10647.
- 30 Benting, J., et al., Acyl and alkyl chain length of GPI-anchors is critical for raft association in vitro. *FEBS Lett.* 1999; 462(1–2): 47–50.

- **31** Wang, T. Y., R. Leventis, and J. R. Silvius, Partitioning of lipidated peptide sequences into liquid-ordered lipid domains in model and biological membranes. *Biochemistry* 2001; 40(43): 13 031–13040.
- 32 McIntosh, T.J., A. Vidal, and S.A. Simon, Sorting of lipids and transmembrane peptides between detergent-soluble bilayers and detergent-resistant rafts. *Biophys. J.* 2003; 85(3): 1656–1666.
- 33 Estronca, L. M., et al., Solubility of amphiphiles in membranes: influence of phase properties and amphiphile head group. *Biochem. Biophys. Res. Commun.* 2002; 296(3): 596–603.
- 34 Spiegel, S., et al., Direct visualization of redistribution and capping of fluorescent gangliosides on lymphocytes. *J. Cell Biol.* 1984; 99(5): 1575–1581.
- **35** Harder, T., et al., Lipid domain structure of the plasma membrane revealed by patching of membrane components. *J. Cell Biol.* 1998; 141(4): 929–942.
- 36 Gidwani, A., D. Holowka, and B. Baird, Fluorescence anisotropy measurements of lipid order in plasma membranes and lipid rafts from RBL-2H3 mast cells. *Biochemistry* 2001; 40(41): 12422–12429.
- 37 Gaus, K., et al., Visualizing lipid structure and raft domains in living cells with twophoton microscopy. *Proc. Natl. Acad. Sci.* USA 2003; 100(26): 15554–15559.
- 38 Pralle, A., et al., Sphingolipid-cholesterol rafts diffuse as small entities in the plasma membrane of mammalian cells. J. Cell Biol. 2000; 148(5): 997–1008.
- 39 Prior, I. A., et al., Direct visualization of Ras proteins in spatially distinct cell surface microdomains. *J. Cell Biol.* 2003; 160(2): 165–170.
- **40** Zacharias, D.A., et al., Partitioning of lipid-modified monomeric GFPs into membrane microdomains of live cells. *Science* 2002; 296(5569): 913–916.
- 41 Lommerse, P. H., et al., Single-molecule imaging of the H-ras membrane-anchor reveals domains in the cytoplasmic leaflet of the cell membrane. *Biophys. J.* 2004; 86(1 Pt 1): 609–616.
- 42 Kenworthy, A. K. and M. Edidin, Distribution of a glycosylphosphatidylinositol-anchored protein at the apical surface of MDCK cells examined at a resolution of <100 A using imaging fluorescence reso-</p>

nance energy transfer. J. Cell Biol. 1998; 142(1): 69-84.

- **43** Kenworthy, A. K., N. Petranova, and M. Edidin, High-resolution FRET microscopy of cholera toxin B-subunit and GPI-anchored proteins in cell plasma membranes. *Mol. Biol. Cell* 2000; 11(5): 1645–1655.
- 44 Glebov, O.O. and B.J. Nichols, Lipid raft proteins have a random distribution during localized activation of the T-cell receptor. *Nat. Cell Biol.* 2004; 6(3): 238–243.
- 45 Fra, A. M., et al., Detergent-insoluble glycolipid microdomains in lymphocytes in the absence of caveolae. *J. Biol. Chem.* 1994; 269(49): 30745–30748.
- 46 Varma, R. and S. Mayor, GPI-anchored proteins are organized in submicron domains at the cell surface. *Nature* 1998; 394(6695): 798–801.
- 47 Sharma, P., et al., Nanoscale organization of multiple GPI-anchored proteins in living cell membranes. *Cell* 2004; 116(4): 577–589.
- 48 Kusumi, A., I. Koyama-Honda, and K. Suzuki, Molecular dynamics and interactions for creation of stimulation-induced stabilized rafts from small unstable steady-state rafts. *Traffic* 2004; 5(4): 213–230.
- **49** Anderson, R.G. and K. Jacobson, A role for lipid shells in targeting proteins to caveolae, rafts, and other lipid domains. *Science* 2002; 296(5574): 1821–1825.
- 50 Dietrich, C., et al., Relationship of lipid rafts to transient confinement zones detected by single particle tracking. *Biophys. J.* 2002; 82(1 Pt 1): 274–284.
- 51 Schutz, G. J., et al., Properties of lipid microdomains in a muscle cell membrane visualized by single molecule microscopy. *EMBO J.* 2000; 19(5): 892–901.
- 52 Fujiwara, T., et al., Phospholipids undergo hop diffusion in compartmentalized cell membrane. *J. Cell Biol.* 2002; 157(6): 1071–1081.
- 53 Lafont, F., et al., Annexin XIIIb associates with lipid microdomains to function in apical delivery. J. Cell Biol. 1998; 142(6): 1413–1427.
- 54 Cheong, K. H., et al., VIP17/MAL, a lipid raft-associated protein, is involved in apical transport in MDCK cells. *Proc. Natl. Acad. Sci. USA* 1999; 96(11): 6241–6248.
- 20 References
 - 55 Simons, K. and D. Toomre, Lipid rafts and signal transduction. Nat. Rev. Mol. Cell. Biol. 2000; 1(1): 31–39.
 - 56 Gri, G., et al., The inner side of T cell lipid rafts. *Immunol. Lett.* 2004; 94(3): 247–252.
 - 57 Pyenta, P.S., D. Holowka, and B. Baird, Cross-correlation analysis of inner-leafletanchored green fluorescent protein co-redistributed with IgE receptors and outer leaflet lipid raft components. *Biophys. J.* 2001; 80(5): 2120–2132.
 - 58 Harder, T. and K. R. Engelhardt, Membrane domains in lymphocytes from lipid rafts to protein scaffolds. *Traffic* 2004; 5(4): 265–275.
 - 59 Kahya, N., et al., Probing lipid mobility of raft-exhibiting model membranes by fluorescence correlation spectroscopy. *J. Biol. Chem.* 2003; 278(30): 28109–28115.
 - 60 Pierce, S. K., Lipid rafts and B-cell activation. *Nat. Rev. Immunol.* 2002; 2(2): 96–105.
 - **61** Holowka, D., et al., Lipid segregation and IgE receptor signaling: A decade of progress. *Biochim. Biophys. Acta* **2005**; in press.
 - **62** Horejsi, V., Lipid rafts and their roles in T-cell activation. *Microbes Infect.* 2005; 7(2): 310–316.
 - 63 He, H. T., A. Lellouch, and D. Marguet, Lipid rafts and the initiation of T cell receptor signaling. *Semin. Immunol.* 2005; 17(1): 23–33.
 - **64** Douglass, A. D. and R. D. Vale, Singlemolecule microscopy reveals plasma membrane microdomains created by protein-protein networks that exclude or trap signaling molecules in T cells. *Cell* 2005; 121(6): 937–950.
 - 65 Magee, A.I., J. Adler, and I. Parmryd, Cold-induced coalescence of T-cell plasma membrane microdomains activates signalling pathways. *J. Cell Sci.* 2005; 118(Pt 14): 3141–3151.
 - **66** Veatch, S. L. and S. L. Keller, Seeing spots: Complex phase behavior in simple membranes. *Biochim. Biophys. Acta* 2005; in press.
 - 67 Hammond, A. T., et al., Crosslinking a lipid raft component triggers liquid ordered-liquid disordered phase separation in model plasma membranes. *Proc. Natl. Acad. Sci. USA* 2005; 102(18): 6320–6325.

- 68 Herzlinger, D. A. and G. K. Ojakian, Studies on the development and maintenance of epithelial cell surface polarity with monoclonal antibodies. *J. Cell Biol.* 1984; 98(5): 1777–1787.
- **69** Balcarova-Stander, J., et al., Development of cell surface polarity in the epithelial Madin-Darby canine kidney (MDCK) cell line. *EMBO J.* 1984; 3(11): 2687–2694.
- 70 Simons, K. and G. van Meer, Lipid sorting in epithelial cells. *Biochemistry* 1988; 27(17): 6197–6202.
- 71 Danielsen, E. M. and G. H. Hansen, Lipid rafts in epithelial brush borders: atypical membrane microdomains with specialized functions. *Biochim. Biophys. Acta* 2003; 1617(1–2): 1–9.
- 72 Barenholz, Y. and T. E. Thompson, Sphingomyelins in bilayers and biological membranes. *Biochim. Biophys. Acta* 1980; 604(2): 129–158.
- 73 Meder, D., et al., Phase coexistence and connectivity in the apical membrane of polarized epithelial cells. Proc. Natl. Acad. Sci USA, in press.
- 74 Kenworthy, A. K., et al., Dynamics of putative raft-associated proteins at the cell surface. J. Cell Biol. 2004; 165(5): 735–746.
- 75 Feder, T. J., et al., Constrained diffusion or immobile fraction on cell surfaces: a new interpretation. *Biophys. J.* 1996; 70(6): 2767–2773.
- 76 Bouchaud, J. P. and A. Georges, Comment on "Stochastic pathway to anomalous diffusion". *Phys. Rev. A* 1990; 41(2): 1156–1157.
- 77 Vaz, W. L. and P. F. Almeida, Phase topology and percolation in multi-phase lipid bilayers: is the biological membrane a domain mosaic? *Curr. Opin. Struct. Biol.* 1993; 3: 482–488.
- 78 Almeida, P. F., W. L. Vaz, and T. E. Thompson, Lateral diffusion and percolation in two-phase, two-component lipid bilayers. Topology of the solid-phase domains in-plane and across the lipid bilayer. *Biochemistry* 1992; 31(31): 7198–7210.
- **79** Coelho, F. P., W. L. Vaz, and E. Melo, Phase topology and percolation in twocomponent lipid bilayers: a Monte Carlo approach. *Biophys. J.* 1997; 72(4): 1501–1511.

- 80 Abreu, M. S., M. J. Moreno, and W. L. Vaz, Kinetics and thermodynamics of association of a phospholipid derivative with lipid bilayers in liquid-disordered and liquid-ordered phases. Biophys. J. 2004; 87(1): 353-365.
- 81 Shogomori, H., et al., Palmitoylation and intracellular domain interactions both contribute to raft targeting of linker for activation of T cells. J. Biol. Chem. 2005; 280(19): 18931-18942.
- 82 Palade, G.E., Fine structure of blood capillaries. J. Appl. Physiol. 1953; 24: 1424.
- 83 Bruns, R.R. and G.E. Palade, Studies on blood capillaries. I. General organization of blood capillaries in muscle. J. Cell Biol. 1968; 37(2): 244-276.
- 84 Yamada, E., The fine structure of the gall bladder epithelium of the mouse. I. Biophys. Biochem. Cytol. 1955; 1(5): 445-458.
- 85 Peters, K. R., W. W. Carley, and G. E. Palade, Endothelial plasmalemmal vesicles have a characteristic striped bipolar surface structure. J. Cell Biol. 1985l 101(6): 2233-2238.
- 86 Rothberg, K.G., et al., Caveolin, a protein component of caveolae membrane coats. Cell 1992; 68(4): 673-682.
- 87 Dupree, P., et al., Caveolae and sorting in the trans-Golgi network of epithelial cells. EMBO J. 1993; 12(4): 1597-1605.
- 88 Scherer, P.E., et al., Caveolin isoforms differ in their N-terminal protein sequence and subcellular distribution. Identification and epitope mapping of an isoform-specific monoclonal antibody probe. J. Biol. Chem. 1995; 270(27): 16395-16401.
- 89 Scherer, P.E., et al., Cell-type and tissuespecific expression of caveolin-2. Caveolins 1 and 2 co-localize and form a stable hetero-oligomeric complex in vivo. J. Biol. Chem. 1997; 272(46): 29337-29346.
- 90 Tang, Z., et al., Molecular cloning of caveolin-3, a novel member of the caveolin gene family expressed predominantly in muscle. J. Biol. Chem. 1996; 271(4): 2255-2261.
- 91 Kurzchalia, T.V., et al., VIP21, a 21-kD membrane protein is an integral component of trans-Golgi-network-derived transport vesicles. J. Cell Biol. 1992; 118(5): 1003-1014.

- 92 Monier, S., et al., VIP21-caveolin, a membrane protein constituent of the caveolar coat, oligomerizes in vivo and in vitro. Mol. Biol. Cell 1995; 6(7): 911-927.
- 93 Sargiacomo, M., et al., Oligomeric structure of caveolin: implications for caveolae membrane organization. Proc. Natl. Acad. Sci. USA 1995; 92(20): 9407-9411.
- 94 Luetterforst, R., et al., Molecular characterization of caveolin association with the Golgi complex: identification of a cis-Golgi targeting domain in the caveolin molecule. J. Cell Biol. 1999; 145(7): 1443-1459.
- 95 Machleidt, T., et al., Multiple domains in caveolin-1 control its intracellular traffic. J. Cell Biol. 2000; 148(1): 17-28.
- 96 Scherer, P.E., et al., Identification, sequence, and expression of caveolin-2 defines a caveolin gene family. Proc. Natl. Acad. Sci. USA 1996; 93(1): 131-135.
- 97 Mora, R., et al., Caveolin-2 localizes to the Golgi complex but redistributes to plasma membrane, caveolae, and rafts when coexpressed with caveolin-1. J. Biol. Chem. 1999; 274(36): 25708-25717.
- 98 Parolini, I., et al., Expression of caveolin-1 is required for the transport of caveolin-2 to the plasma membrane. Retention of caveolin-2 at the level of the Golgi complex. J. Biol. Chem. 1999; 274(36): 25718-25725.
- 99 Fra, A. M., et al., De novo formation of caveolae in lymphocytes by expression of VIP21-caveolin. Proc. Natl. Acad. Sci. USA 1995; 92(19): 8655-8659.
- 100 Pelkmans, L. and M. Zerial, Kinase-regulated quantal assemblies and kiss-and-run recycling of caveolae. Nature 2005; 436(7047): 128-133.
- 101 Fernandez, I., et al., Mechanism of caveolin filament assembly. Proc. Natl. Acad. Sci. USA 2002; 99(17): 11193-11198.
- 102 Fra, A. M., et al., A photo-reactive derivative of ganglioside GM1 specifically crosslinks VIP21-caveolin on the cell surface. FEBS Lett. 1995; 375(1-2): 11-14.
- 103 Murata, M., et al., VIP21/caveolin is a cholesterol-binding protein. Proc. Natl. Acad. Sci. USA 1995; 92(22): 10339-10343.
- 104 Lisanti, M.P., et al., Caveolae, caveolin and caveolin-rich membrane domains: a signalling hypothesis. Trends Cell Biol. 1994; 4(7): 231-235.

- 22 References
 - 105 Brown, D.A. and J.K. Rose, Sorting of GPI-anchored proteins to glycolipid-enriched membrane subdomains during transport to the apical cell surface. *Cell* 1992; 68(3): 533–544.
 - 106 Sargiacomo, M., et al., Signal transducing molecules and glycosyl-phosphatidylinositol-linked proteins form a caveolin-rich insoluble complex in MDCK cells. J. Cell Biol. 1993; 122(4): 789–807.
 - 107 Smart, E. J., et al., A detergent-free method for purifying caveolae membrane from tissue culture cells. *Proc. Natl. Acad. Sci. USA* 1995; 92(22): 10104–10108.
 - 108 Bacia, K., P. Schwille, and T. Kurzchalia, Sterol structure determines the separation of phases and the curvature of the liquid-ordered phase in model membranes. *Proc. Natl. Acad. Sci. USA* 2005; 102(9): 3272–3277.
 - 109 Rodriguez-Boulan, E., G. Kreitzer, and A. Musch, Organization of vesicular trafficking in epithelia. *Nat. Rev. Mol. Cell. Biol.* 2005; 6(3): 233–247.
 - **110** Mellman, I., Membranes and sorting. *Curr. Opin. Cell Biol.* 1996; 8(4): 497–498.
 - 111 Salama, N. R. and R. W. Schekman, The role of coat proteins in the biosynthesis of secretory proteins. *Curr. Opin. Cell Biol.* 1995; 7(4): 536–543.
 - 112 Kreis, T.E., M. Lowe, and R. Pepperkok, COPs regulating membrane traffic. *Annu. Rev. Cell. Dev. Biol.* 1995; 11: 677–706.
 - 113 Schuck, S. and K. Simons, Polarized sorting in epithelial cells: raft clustering and the biogenesis of the apical membrane. *J. Cell Sci.* 2004; 117(Pt 25): 5955–5964.
 - 114 Lipowsky, R., Domain-induced budding of fluid membranes. *Biophys. J.* 1993; 64: 1133–1138.
 - 115 van Meer, G., et al., Sorting of sphingolipids in epithelial (Madin-Darby canine kidney) cells. J. Cell Biol. 1987; 105(4): 1623–1635.
 - 116 Fuller, S. D. and K. Simons, Transferrin receptor polarity and recycling accuracy in "tight" and "leaky" strains of Madin-Darby canine kidney cells. *J. Cell Biol.* 1986; 103(5): 1767–1779.
 - 117 Pfeiffer, S., S. D. Fuller, and K. Simons, Intracellular sorting and basolateral appearance of the G protein of vesicular stomatitis virus in Madin-Darby canine kid-

ney cells. J. Cell Biol. 1985; 101(2): 470–476.

- 118 Folsch, H., et al., A novel clathrin adaptor complex mediates basolateral targeting in polarized epithelial cells. *Cell* 1999; 99(2): 189–198.
- **119** Paladino, S., et al., Protein oligomerization modulates raft partitioning and apical sorting of GPI-anchored proteins. *J. Cell Biol.* 2004; 167(4): 699–709.
- 120 Delacour, D., et al., Galectin-4 and sulfatides in apical membrane trafficking in enterocyte-like cells. *J. Cell Biol.* 2005; 169(3): 491–501.
- 121 Parton, R.G., B. Joggerst, and K. Simons, Regulated internalization of caveolae. J. Cell Biol. 1994; 127(5): 1199–1215.
- 122 Pelkmans, L., J. Kartenbeck, and A. Helenius, Caveolar endocytosis of simian virus 40 reveals a new two-step vesiculartransport pathway to the ER. *Nat. Cell. Biol.* 2001; 3(5): 473–483.
- 123 Kirkham, M., et al., Ultrastructural identification of uncoated caveolin-independent early endocytic vehicles. J. Cell Biol. 2005; 168(3): 465–476.
- 124 Damm, E. M., et al., Clathrin- and caveolin-1-independent endocytosis: entry of simian virus 40 into cells devoid of caveolae. J. Cell Biol. 2005; 168(3): 477–488.
- 125 Tsai, B., et al., Gangliosides are receptors for murine polyoma virus and SV40. *EMBO J.* 2003; 22(17): 4346–4355.
- 126 Le, P. U., et al., Caveolin-1 is a negative regulator of caveolae-mediated endocytosis to the endoplasmic reticulum. *J. Biol. Chem.* 2002; 277(5): 3371–3379.
- 127 Pelkmans, L., et al., Caveolin-stabilized membrane domains as multifunctional transport and sorting devices in endocytic membrane traffic. *Cell* 2004; 118(6): 767–780.
- 128 van Deurs, B., et al., Caveolae: anchored, multifunctional platforms in the lipid ocean. *Trends Cell Biol.* 2003; 13(2): 92–100.
- 129 Lafont, F., et al., Raft association of SNAP receptors acting in apical trafficking in Madin-Darby canine kidney cells. *Proc. Natl. Acad. Sci. USA* 1999; 96(7): 3734–3738.
- 130 Pombo, I., J. Rivera, and U. Blank, Munc18–2/syntaxin3 complexes are spatially separated from syntaxin3-containing

SNARE complexes. *FEBS Lett.* 2003; 550(1–3): 144–148.

- 131 Low, S. H., et al., Differential localization of syntaxin isoforms in polarized Madin-Darby canine kidney cells. *Mol. Biol. Cell* 1996; 7(12): 2007–2018.
- **132** Jacob, R. and H. Y. Naim, Apical membrane proteins are transported in distinct

vesicular carriers. *Curr. Biol.* 2001; 11(18): 1444–1450.

133 Harder, T. and M. Kuhn, Selective accumulation of raft-associated membrane protein LAT in T cell receptor signaling assemblies. *J. Cell Biol.* 2000; 151(2): 199–208.

Pierre Sens and Matthew S. Turner

2.1 Introduction

Caveolae are Ω -shaped invaginations of the plasma membrane, found in many types of cells [1]. Caveolae are enriched in cholesterol, and have a membrane composition similar to that of lipid rafts [2]. In addition, caveolae show a high concentration of the protein caveolin, a hairpin-structured membrane protein possessing a hydrophobic domain [32 amino acids (AA), flanked by two hydrophilic termini (N-terminal: 101 AA, C-terminal: 43 AA)] [3]. Both termini of caveolin extend from the cytosolic side of the membrane, conferring a strong asymmetrical structure to the caveolar domain. The goal of this chapter is to provide an overview of the possible physical effects that can stem from both of these characteristics of caveolae, namely the composition difference with the rest of the plasma membrane ("raft-aspect") and their patent asymmetry ("hairy-aspect").

Although Caveolae were first observed more than 50 years ago, many of their properties and functions remain unknown. Caveolae formation seems to require the presence of the protein caveolin [4] and is very dependent on the cholesterol level in the cell [5]. There is also evidence that caveolin is coupled to other types of membrane deformation (e.g., tubular structures in endothelial cells [6]). These are strong indications that at least some biological functions of caveolae rely heavily upon their biophysical properties. Plasma membranes typically resist bending, and the formation of membrane invaginations requires the action of mechanical forces on the membrane. Even though caveolae are very complex biochemical objects, they are bound to obey the laws of physics. We must therefore understand the origin of the forces at play in the formation of the invaginations if we are to understand how, and why, caveolae form. As we make progress towards this we may gain important insights into the biological functions of caveolae. Indeed, since caveolae are inherently coupled to the mechanical state of the plasma membrane, one may envision that the cell has taken advantage of this coupling, and may use caveolae as mechano-sensors or mechano-regulators for the plasma membrane. Before discussing these possibilities in Section 2.6, we will first review some of the physical concepts behind the formation and structure of mem-

brane domains and how this relates to the physical properties of membrane proteins.

The description present ed relies on coarse-grained physical models where the molecular structure of the membrane and proteins is only taken into account in an approximate way. This is justified by the fact that caveolae (of size ~ 100 nm) are much larger than the size of the individual caveolin proteins and of the thickness of the plasma membrane (~5 nm). This physical description is based on the wellknown properties of fluid bilayer membranes, described in Section 2.2. Following this, two different points of view are taken to describe the formation and invagination of caveolae. In Section 2.3, caveolae are regarded as membrane domains that are chemically immiscible in the plasma membrane. This neglects effects associated directly with the details of domain composition. It is assumed there that membrane phase separation into domains does n't depend on the mechanical properties of the membrane, although the domain shape might. This description somewhat overlooks the importance of the protein caveolin in the formation of caveolae. In an attempt to approximate the complexity of the biological membrane, theoretical physicists have studied the behavior of membrane inclusions, and in particular how protein aggregation is coupled to membrane deformation, and viceversa. These models are briefly overviewed in Section 2.4, and applied to the particular case of caveolin aggregation in caveolar membranes in Section 2.5, by taking some account of the protein structure. The final section includes a discussion of how such a description relates to the "life" of caveolae at the plasma membrane of cells. Finally, we speculate on further possible biological functions of caveolae.

2.2

Physical Modeling of Lipid Membranes

Mathematical models of deformable, fluid membranes have been available for many years [7,8], and have been successfully compared with experimental results, both on artificial [9] and biological [10] membranes. At the most fundamental level these theories rely on the single basic principle underlying statistical mechanics: that the probability of observing a given membrane deformation depends on the energy change involved in making this deformation [11]. The higher the energy, the less likely the deformation. Statistical mechanics tells us that the probability p_i of an event i is related to its energy F_i according to:

$$p_i \sim \exp\left[-\frac{F_i}{k_{\rm B}T_m}\right] \tag{1}$$

This probability compares the deformation energy F_i to some energy source in the system. This energy is written F to remind us that it is a *free energy* and therefore includes changes in entropy, as well as internal and chemical energies [12]. Reactions that reduce the entropy of the system are disfavored in the same way as are those that involve a spontaneous increase in the energy by, for example, disrupting

chemical bonds. Strictly speaking, Eq. (1) only holds for (sub)systems that are at equilibrium but this can often be a reasonable approximation, for example for small patches of membrane that can move and relax quickly, even though it may be inappropriate for the cell as a whole. In passive systems, the only energy source comes from the thermal fluctuations of energy k_BT , where k_B is the Boltzmann constant and *T* is the temperature (in Kelvin). Biological systems are called "active", because chemical energy, coming from, for example, ATP hydrolysis, can be harnessed by specific enzymes (molecular motors) to perform mechanical work. The cell membrane is generally the site of many active processes, including cytoskeleton polymerization and ion pumping. One may adopt the approach that these active processes provide an effective "membrane" temperature $T_m > T$ [13], and it is this that appears in Eq. (1).

In practice, much information can be obtained by the study of membrane deformations that minimize the membrane energy (those having the higher probability to occur). One contribution to membrane energy can arise from any change in the area of the cell, which must act against tension in the membrane. This is reminiscent of the work required to deform a child's balloon, for example by pinching a small patch of its surface between the fingers. As with any interface, a lipid membrane bears a "surface tension" (denoted γ throughout), which is the energy cost per unit area associated with decreasing the membrane area. However, while the surface tension observed at a water air interface is of the order 10⁻¹ J m⁻², and typically dominates any other type of deformation energy, the surface tension of lipid bilayers can be extremely low (10⁻⁸ J m⁻² in very floppy artificial systems, and $\sim 10^{-5}$ J m⁻² for the plasma membrane). As the surface tension is low, other modes of deformation can also play an important role. One such mode is the energy associated with bending the membrane. A symmetrical bilayer membrane prefers to be flat, so that both monolayers have the same structure. Bending the membrane one way or the other breaks this symmetry, and costs an energy which varies quadratically with the membrane curvature (deformation) C. This is fundamentally analogous to the fact that the energy of an ideal spring varies with the square of its extension (known as Hooke's law), and is ultimately the reason why any flexible material that is bent will spring back into its original shape. If the membrane is asymmetrical, and cell membranes are indeed rather asymmetrical, it may prefer a non-zero curvature. This means that the membrane energy might be minimized by, and the membrane therefore most happy with, a non-zero curvature. This curvature is called the "spontaneous curvature" C₀. In this case, the deformation energy is again quadratic, but now is the difference between the membrane's (local) curvature and its spontaneous curvature. This can also be identified with a version of Hooke's law. While the membrane tension tells us how much the energy increases when the membrane area is increased, the energy increase caused by a deviation from the preferred membrane curvature is controlled by the "bending rigidity", conventionally denoted κ . Adding the curvature energy to the energy of membrane tension, the total energy of a patch of membrane of area S, with a curvature C is

$$F = \gamma S + \frac{1}{2} \kappa S (C - C_0)^2 \tag{2}$$

A Typical value for the bending rigidity of biomembranes [10] is $\kappa = 20 k_B T$. It is convenient to measure energies in units of the thermal energy scale which, at room temperature is, $k_B T = 4 \times 10^{-21} J$. Thus, 1 kJ mol⁻¹ = 0.4 $k_B T$.

In this chapter, we will be mostly concerned by the flask-shaped membrane deformations mimicking the caveolae (Fig. 2.1). For simplicity, we will assimilate the invagination to a spherical cap of constant curvature. In practice, there exists a membrane neck connecting the concave central cap to the flat surrounding membrane. Specialized proteins are likely to be present near the caveolae neck [14], and this is not included in the present models. From Eq. (2), the energy of a spherical membrane (with no spontaneous curvature), is $F_{sphere} = (\gamma S + 8\pi\kappa)$. The energy of large patches is dominated by membrane tension, and the energy of small patches by membrane rigidity. Clearly, this has strong consequences for the stability of membrane invaginations in general, and of caveolae in particular. Indeed, small invaginations all have the same energy (~ $8\pi\kappa$), which is dominated by the bending energy of the membrane. Large invaginations on the other hand, have an energy which increases with their size ($\sim \gamma S$), and are much less likely to be observed. The cross-over size between small and large invaginations in the physical sense corresponds to an area $S \sim 8\pi\kappa/\gamma$. Choosing a bending rigidity $\kappa \sim 20k_{\rm B}T$ and a surface tension $\gamma \sim 10^{-5}$ J m⁻², the cross-over size corresponds to a sphere of radius R ~ 120 nm. The fact that this scale is close to the typical size scale of the invaginations is very encouraging for our physical approach. It indicates that even such simple physical arguments can reveal a competition between different physical energies (and hence forces) that could give rise to invaginations with roughly the observed size.

Of course biological membranes have a complexity that is not reflected in the seminal elastic model of Eq. (2). In particular, the complex lipid composition (up to 25 different lipid species), the inclusion of a host of membrane proteins (~30% of the whole genome), and the support of the membrane cytoskeleton. The question of how to incorporate the two former features will form the subject of most of the following text. The cytoskeleton provides a visco elastic scaffold to the cell, and is able to exert direct forces to the membrane. Cytoskeletal anchoring of the plasma membrane is crucial to the membrane's mechanical behavior. The breaking of



Fig. 2.1 Sketch of a spherical membrane cap. The membrane curvature C is roughly constant over the deformed area S which has radius of curvature (the radius of the circle passing through the cap) 1/C, as shown. Inset: the bilayer structure of a lipid membrane.

some anchoring sites upon cell deformation and membrane extension is a major component of the energy cost of such deformations. In particular, cytoskeleton anchoring can account for up to 75% of the measured tension of cell membranes [15]. Here, we adopt the philosophy that the cytoskeleton acts to maintain the plasma membrane under tension, but does not exert direct forces to pull flask-shaped invaginations from the membrane. In fact, we will find that a consistent physical explanation of invagination can be constructed without the cytoskeleton playing any direct role in the formation of caveolae.

2.3 Caveolae as Invaginated Lipid Rafts

Caveolae are one example of lipid domains in the cell membrane [2]. These domains and other lipid "rafts" are characterized by a high concentration of cholesterol and saturated lipids. Although many controversies exist regarding the size and lifetime of lipid rafts, the morphology of caveolae seems much better defined. One likely reason for this is the stabilization of these membrane domains by the protein caveolin, and in particular the fact that they are invaginated. In this section, we present some general physical arguments concerning the behavior of membrane domains in a flexible, fluid lipid bilayer [16], without discussing the origin of their formation. Here, we assume that the phase separation is driven by chemical incompatibility between the raft and the non-raft phase, regardless of the mechanical state of the membrane. In Physical terms, chemical incompatibility can be accounted for by an energy cost of creating an interface between the two phases. Since the membrane is in two dimensions, the interface between the two phases is a line, and the immiscibility parameter is a "line tension", termed σ . For a given domain size, the line energy is the smallest when the interface is smallest, meaning that we can expect domains to be circular with a radius $R = \sqrt{S/\pi}$ (S is the domain area). Unless the line tension is very small, in which case the rafts are very small and their shape can fluctuate significantly. This signifies that they are only weakly phase separating from the non-raft membrane and are close to dissolving back into it.

Figure 2.2 illustrates how chemical incompatibility alone can have a strong influence on the domain shape. Indeed, a flat domain (Fig. 2.2a) has a large interface with its surrounding, costing a line energy of order σR . If the domain is large, this can be quite a large energy, and may cause the domain to bud off the membrane in an attempt to reduce the size of the interface to the membrane neck connecting the bud to the rest of the membrane. On the other hand, an invaginated domain costs the energy of bending the membrane into a spherical shape. The bending energy of a sphere is proportional to the bending rigidity of the membrane κ , and it has the remarkable feature that it does not depend on the size of the sphere for a symmetrical domain (it is equal to $8\pi\kappa$). Since the line energy increases with domain size, and the bending energy does not, there is a critical domain size for which we can expect the domain to invaginate spontaneously [16] (Fig. 2.2c). The

30 2 The Forces that Shape Caveolae



Fig. 2.2 Invagination of a membrane domain (in gray) due to chemical incompatilibity. (a) A flat domain has a large line energy, due to a large interface with the surrounding membrane. (b) An invaginated domain has a small interface, but is accompanied with a

bending energy. (c) Comparison of the energy of a flat domain (sloping line) and of an invaginated domain (horizontal line). The invagination is favored for domains larger than a critical size, estimated to be of order 100 nm.

domain size at which this occurs is of the order $R \sim 4\kappa/\sigma$. The actual value of the physical parameters κ and σ vary from membrane to membrane, but we have a good idea of the order of magnitude of such parameters. The bending rigidity of biological membrane is typically of the order $\kappa \sim 20k_{\rm B}T$. The line tension arises from unfavorable contacts at the molecular scale (the size of a lipid molecule, or the thickness of the bilayer). It is estimated to be of the order $\sigma \sim k_{\rm B}T/nm$, although its actual value is very sensitive to the nature of the two phases in contact (in a raft, it is the contact between liquid-ordered and-liquid disordered lipid phases). Using those numbers, the critical size for domain invagination is R = 80 nm (corresponding to a spherical bud of radius 40 nm). Caveolae are precisely in this range of size, which is a good indication that the physical phenomena of membrane bending energy and raft line energy play a crucial role in caveolae formation and stability.

One possible picture of the formation of caveolae is the following [16] (see Fig. 2.2). Imagine a given amount of caveolin, cholesterol, and raft-forming lipids (in particular sphingolipids), dispersed in the cell membrane. With time, these various components diffuse into the membrane and find each other, forming growing membrane domains. Although the rate of this phase separation might be quite slow [17], domains should eventually grow to a large size if they are not perturbed by other dynamical phenomena at the cell membrane, and even more so if the presence of caveolin promotes the phase separation. Membrane recycling, endocytosis, and exocytosis might perturb domain growth, and may be invoked to explain the small size of the non-caveolae rafts observed in vivo (see [17].) When domains grow beyond the critical size discussed above (Fig. 2.2), they are at their lowest energy *N* and therefore most stable *N* when invaginated rather than flat.

Such a scenario is still qualitatively valid if the membrane supports a tension and the domain has a spontaneous curvature. In this case, however, the critical budding size depends upon the membrane tension. Indeed, work against membrane tension must be performed to invaginate a domain, and this stabilizes the flat shape. As a result, this simple theory applied to caveolae would predict that the size of the invagination increases with membrane tension. However, caveolae have very similar sizes across several cell types that can, presumably, bear different membrane tensions. As will be seen in Section 2.5, this indicates that the structure of the protein caveolin might play a crucial role in controlling the size of the invaginated domains.

2.4 Membrane Inclusions

Membrane proteins have hydrophobic regions inserted within the lipid bilayer, and this insertion may perturb the bilayer structure. For example, a mismatch in thickness between the hydrophobic core of the protein and that of the bilayer has an associated energy cost. It implies either that some hydrophobic residues are left unshielded from contact with water, or that the membrane (or the protein) changes its thickness to obtain a good match [18]. The consequence of hydrophobic mismatch may be protein clustering, as shown in Figure 2.3a,b. Clustering may occur even in the absence of direct (specific) interaction between the proteins, as the result of an effective attraction mediated by the membrane. As shown in Figure 2.3a,b, the membrane order is locally perturbed in the vicinity of the inclusion (the figure shows a local stretching of the lipid tails, to accommodate the hydrophobic thickness of the membrane). Bringing two inclusions together reduces the membrane area that needs to be perturbed and hence reduces the energy of membrane deformation. The result of this is an effective force that brings the proteins together.

An important phenomenon in the context of caveolae and their asymmetric membranes is when protein clustering is coupled to a change of membrane morphology. This can be expected for very asymmetric proteins, or for peripheral proteins that mostly extend on one side of the membrane. Such asymmetric membrane proteins can be thought of as imprinting a local spontaneous curvature (the C_0 term in Eq. (2) to the neighboring membrane (Fig. 2.3c). The membrane is locally curved near the protein, which again leads to a frustration of the bilayer order. As it is the case for hydrophobic mismatch, some of the frustration can be released if the proteins aggregate. In this case however, the concentration of a large number of proteins over a limited membrane area leads to a morphological change of the membrane, which adopts the preferred curvature of the proteins (Fig. 2.3d). As will be discussed in Section 2.5, this phenomenon, called "curvature instability" in the physics literature [19], is likely to play an important part in the formation of caveolae. The caveolin proteins found in caveolae are very good candidates for such large morphological changes for two reasons. On the one hand, both their hydro-



Fig. 2.3 Sketch of the aggregation of membrane proteins induced by an hydrophobic mismatch. (a) The mismatch imposes a perturbation of the bilayer structure around the protein. (b) The aggregation of two such proteins reduces the area of the perturbed membrane, and is energetically favorable.

(c,d) Sketch of the aggregation of membrane proteins induced by an asymmetric coupling with the membrane. The asymmetric membrane perturbation around each inclusion (c), is reduced by protein aggregation (d), which induce a large-scale deformation in the membrane region with high protein density.

phobic termini face the cytoplasm, which makes their interaction with the neighboring membrane very asymmetric. A similar situation can to some extent be reproduced in artificial systems, by mixing bilayer-forming lipids with hydrophilic polymers (typically polyethylene glycol) with a hydrophobic anchor attached. Lipid membranes with a grafted (but mobile) polymer chain can be obtained in such manner, and it has been shown theoretically [20], and observed experimentally [21], that such a membrane can exhibit a phase separation. Another reason to expect membrane deformation near caveolin proteins is the fact that they are known to form homo-oligomers of about 15 proteins. This is due to specific biochemical interactions between caveolin via a short section of the N-terminal cytoplasmic domain, quite close to the transmembrane domain [3]. As shown in Section 2.5, this oligomerization may have much to do with the success of caveolin in aggregating and promoting membrane invagination. By concentrating the caveolin, oligomerization also concentrates the effect of the membrane asymmetry over a small membrane area, creating a large asymmetric pressure.

2.5 Caveolae as a Thermodynamic Phase Separation of Membrane Proteins

Simple mixtures of two or more material components can reside in a variety of states or "phases". The simplest of these is the mixed state, when all components are evenly mixed throughout the system. However, if molecules of one component have a sufficiently large mutual attraction for one another, or equivalently a sufficiently large repulsion from the remainder, they can "phase separate". There can then be large regions that are rich in this component suspended in a background which contains relatively little of it. It is this effect that causes oil to de-mix from water at room temperature but it is a far more generic effect than is often realized. It is now clear that there are components in the plasma membranes of cells that phase separate, for example, into caveolae. This is a slightly unusual phenomenon in that the phase-separated domains are typically only 100 nm across rather than macroscopic in size, but the principle is the same.

There is now good evidence that caveolin proteins form homo-oligomers [3] containing approximately 15 molecules. Whilst this is not a necessary feature for the generation of bending forces, which require only a molecular asymmetry between the two sides of the membrane, it may act to amplify those forces by increasing the density of interacting cytoplasmic domains (see Fig. 2.4).

It has been suggested [22] that the mechanism by which caveolin homo-oligomers form is reminiscent of micellization on a membrane. Usual in spherical surfactant micelles are formed by the aggregation of amphiphilic molecules which experience a mutual attraction between their hydrophobic tails [8]. Such micellar aggregates do not grow indefinitely because of the packing and stretching con-



Fig. 2.4 Sketch indicating the origin of the forces that act to bend the membrane near an asymmetric membrane protein (left), or homo-oligomer thereof (center), with domains extending on one (cytoplasmic) side only. The cytoplasmic domains may be entirely disordered, resembling a random coil (as shown), or may contain some folded structure(s) forced into dense contact within the oligomer. In any case these domains

exert forces on the membrane. Even disordered coils have their configurational entropy restricted by the presence of the membrane. This restriction is large for a planar membrane but is reduced if the interface bends away from the coils, leaving more room in which to fluctuate. This means that the proteins exert forces that give rise to "bending moments" (as shown, right).

straints of their tails. The same is true of caveolin. It has been determined that there is an attractive interaction between N-terminal segments of caveolin [3] and, as the aggregate grows, an increasing number of the caveolins enjoy such contacts. Eventually the repulsive forces between the greatly confined cytoplasmic domains is enough just to balance the force of attraction experienced by the next caveolin molecule that seeks to join the aggregate and, at this point, the optimal size has been reached. For caveolin oligomers this size appears to be ~15 molecules. Given the size of the cytoplasmic domains this indicates that the N-terminal attractive domains probably give rise to a substantial attraction, perhaps of the order of $10k_{\rm B}T$ [22]. This, in turn, results in substantial repulsive forces between the cytoplasmic domains in the oligomers which acts to "amplify" the bending forces indicated in Figure 2.4.

In the same way that there exists a critical micelle concentration (cmc) in surfactant systems there is a similar concentration at which caveolin oligomers will start to form. For simplicity we denote this the cmc. Above this concentration there will be a few single caveolin molecules on the membrane at concentrations equal to that of the cmc, with the remainder forming as many oligomers (micelles) as are required to incorporate all the caveolin. Given that caveolins experience a substantial attraction, their cmc is probably so low that oligomers will always form at physiological concentrations. However, there is another scale of self assembly which also, in its way, involves a concentration that resembles a cmc. This is a critical budding concentration (cbc) which lies above the cmc. Below the cbc all oligomers exist on a roughly flat membrane, whilst above it the flat membrane supports oligomers at, or very close to the cbc, while the remainder of the oligomers form buds that each have an area fraction of oligomers $\varphi^* > \varphi_{cbc}$. As more oligomers are added to the membrane, more N but similar N buds are formed. We are able to establish the cbc by comparing the free energy of a membrane bearing buds to one which does not. In the following section it will be assumed implicitly that the cbc is exceeded and hence buds form.

A free inclusion, as shown in Figure 2.4 (left and center panels), is one which is not anchored to any external structure such as the cytoskeleton. In this case it is further possible to prove that there can be no net overall force acting to move the membrane up or down, nor torques which tilt it left or right. This is a direct consequence of Newton's third law: The membrane cannot experience a force (or torque) without another body or structure experiencing one that is equal and opposite. If there is no such structure there can be no such forces. This leaves the bending moment shown in Figure 2.4 (right panel) as the dominant mechanism for local membrane deformation [23]. The inclusion pushes down on the membrane with its "arms" and pulls up with its "body", but it is not connected to any other structure.

We propose to investigate how an asymmetric inclusion such as the protein caveolin (or an oligomer thereof) can generate a local curvature in the membrane, and how this curved membrane can then provide an environment preferred by other identical curvature-sensitive caveolin molecules. Whilst this approach has the advantage of providing a formal method for calculating the size of a caveolae bud directly from physical arguments, it suffers from the limitation that several parameters are known only to within an order of magnitude. Its utility should therefore be understood in the following terms:

- It represents a check on whether invaginations of 100 nm diameter might possibly be driven by the physical process described. In particular, since we find that this is indeed plausible, it provides a mechanism for the formation of caveolae that explicitly does not involve cytoskeletal forces playing any significant role.
- Such a model is then able to predict how the stability and size of the invaginations will vary with the control parameters, for example, surface tension, membrane rigidity and the tendency of any inclusions to curve the membrane. We are then able to compare these predictions with experiments involving several mutant caveolins.

Whilst an exact calculation of these bending forces is difficult there is one calculation that can at least give us an indication of the magnitude of such forces. This, involves treating the cytoplasmic tails as random coils. In this case there are wellestablished rules from the theory of polymers [24,25] that allow these bending forces to be calculated exactly [20,23,26]. The basic principle is that the chains gain more configurational entropy if they locate next to a convex surface rather than a flat one. This is formally equivalent to saying that each caveolin oligomer imprints a local spontaneous curvature C_0 on the membrane, which is of order $C_0 \sim f_0/\kappa$, where f_0 is a characteristic force exerted on the membrane by the cytoplasmic tails ("arms") of the caveolin oligomer, of the order 10pN [22].

One final physical "ingredient" is required in order to complete our model for the formation of caveolae. It is necessary to include the effect of mixing *n* caveolin oligomers, each of areas *s*, and hence area fraction $\varphi = ns/S$, on the surface of a caveolae. Simplistically, we view the surface of the caveolae as being made up of two components, caveolin oligomers (with fraction φ) surrounded by the rest of the caveolar membrane (with fraction $1-\varphi$). The energy of the membrane deformation is given by Eq. (2), where the spontaneous curvature of the bud increases with the density of caveolin, and is equal to $C_0 \varphi$. Assuming that these components have no interactions of longer range than *a*, then this is a classical two-component ideal mixture. The free energy of mixing of this fluid is well known [12,22], contains similar contributions from the oligomer and non-oligomer membrane patches, and has the natural feature that it is very costly to remove all of either component. Indeed, the energy required to do this actually increases without bound as φ or $1 - \varphi \rightarrow 0$. For inclusions that interact with one another the interaction energy includes a contribution that scales like the density of oligomer oligomer interactions (~ φ^2).

The formation of buds is controlled by a variety of physical processes that have been introduced above. These can be combined into a single equation for the free energy per caveolin oligomer on a caveolae of radius R containing oligomers with area fraction φ . This merely encodes mathematically all of the physical contributions to the energy discussed earlier in this section. These are:

• The existence of a spontaneous curvature, indicating that energy is gained when

2 The Forces that Shape Caveolae



Fig. 2.5 The origin of the contributions to total free energy. (a) Energy is gained by bending the membrane in response to the bending moments exerted by the caveolin oligomers. (b) There is an energy cost when the membrane is bent away from its preferred shape: flat in the absence of caveolin and at precisely the preferred (spontaneous)

curvature when they are present. (c) There is an energy cost when the area of the sphere is removed from the cell membrane against surface tension. (d) There is an energy cost associated with both mixing of and interactions between inclusions on the sphere surface. The latter increases with the area fraction of buds.

the bud, with curvature 1/R bends in response to the bending moments of the oligomers; this is the only effect which drives bending of the membrane.

- The energy cost of bending a membrane away from its preferred shape. If there are no oligomers, the membrane would like to remain flat. When it forms a bud it need not always have a curvature exactly equal to its spontaneous curvature and this, similarly, costs bending energy.
- The energy cost of drawing the area of the bud away from the remainder of the cell membrane into the bud; this involves doing work against the membrane tension γ .
- Finally, the repulsion between caveolin oligomers and the mixing energy must be included. The origin of all of these contributions is sketched in Figure 2.5.

The preferred caveolae state can be obtained by identifying the minimum of this free energy which, in turn, yields predictions for the preferred caveolae bulb radius R* and oligomer density φ^* (see Fig. 2.6). The invagination radius decreases with increasing caveolin density, so that the curvature of the bud approaches the spontaneous curvature of the caveolar membrane: $C^* = C_0 \varphi^*$.

These results follow from the same model for the bud as a nearly-closed sphere attached to the membrane by a small neck, as was introduced in Section 2.4. We find that a physical description of caveolae can yield predictions for such observable quantities as the caveolae radius R that are in good agreement with observations. In this model we find that the primary reason why caveolae form is because of the coupling between membrane curvature and protein density: proteins accu-

36



Fig. 2.6 Variation of the preferred radius of the caveolar bulb *R* with surface tension of the membrane. The size of the caveolae reduces with increasing tension for small tensions, but at larger tensions reaches a preferred value that is rather insensitive to tension. The two sketches of the calveolar bulb indicate this, with darker shading representing higher density of caveolin. Inset: The variation of the critical budding concentration

of inclusions, expressed as the area fraction φ_{cbc} , and the area fraction of oligomers on buds, φ^{\star} . For all but the smallest tensions our simple two-component model predicts that the caveolae should be almost entirely covered with caveolin oligomers, at a concentration $\varphi^{\star} s^{-}$ that is always above that of the surrounding membrane between caveolae $\varphi_{cbc} s^{-}$.

mulate to a curved membrane (bud), stabilizing its shape and thereby attracting more proteins. If the formation of a curved membrane is too energetically costly because of a high surface tension, the domains are destabilized [22].

Also, we can again return to make contact with caveolae as lipid rafts. The consensus seems to be that these are the only rafts for which the identity and stability is not controversial. They are large, comparatively easy to observe, and very stable. We take this as evidence that the particular membrane curvature of caveolae stabilizes the phase separation into buds, a mechanism which is entirely consistent with the results of our physical analysis of their stability. Further evidence for a coupling between caveolae and membrane curvature can be found in the fact that an increase of the number of caveolae can enhance other types of membrane deformation, namely endothelial capillary tubule formation [6].

Finally, our theory provides a framework with which to seek to understand caveolae formation in mutant caveolin systems [27]. Mutants lacking the mutually attractive domain of the N-terminus are still able to drive membrane invagination, but with a much larger size $R \approx 1 \,\mu\text{m}$. This is consistent with the fact that the force

exerted by isolated proteins should be an order of magnitude smaller than the force exerted by oligomers which results in a 10-fold increase in bud radius. Other mutants lacking the mutually attractive C-terminus also form larger buds. We would understand this as being due to a weaker oligomer – oligomer attraction, resulting in a lower density of caveolin in caveolae and therefore larger buds.

The origin of the striated texture observed on the surface of caveolae [1], which superficially resembles tree bark, is still not understood. It may be that such structures are due to the spatial organization of caveolin oligomers on the surface of these "gnarly buds". If this is the case it may not be immediately obvious how circularly symmetric oligomers can organize themselves into asymmetric phases consisting of long, linear structures. However, it is now known that caveolin proteins also interact via the distal regions of their C-termini tails [3]. This attraction, together with a purely physical, membrane-mediated, longer-range repulsion that it is possible o estimate [23] could lead to a phase separation of the caveolin-poor surface. Such phenomena have been studied in physically similar systems [28], and a reasonable conclusion would be that these structures may occur naturally as a result of a balance between a short ranged (C-terminal) attraction and a longer-ranged (membrane deformation-mediated) repulsion.

2.6

Caveolae and Membrane Tension: Mechano-Sensitivity and Mechano-Regulation

So far, we have two possible mechanisms by which invaginated membrane domains may form at the plasma membrane. The raft model of Section 2.3 tells us that a phase separation in the membrane promotes membrane curvature because of chemical immiscibility, the driving force is the domain line tension. The curvature instability model of Section 2.5, takes the point of view that membrane curvature promotes the phase separation because of the aggregation of proteins into patches of preferred curvature, the driving force being the protein spontaneous curvature. In both cases, the tendency to membrane invagination must overcome the membrane tension, which favors a smooth, flat membrane, and in both cases, increasing the membrane tension may lead to the flattening of the invagination. Figure 2.7 shows the effect of membrane tension on the energy of flat and invaginated domains for the raft model. At low tension, chemical incompatibility and the asymmetry of the membrane conspire to promote domain invagination, the budded state being the most stable state. At large tension, budding the domain costs too much energy, and invaginated domains flatten. Flat caveolae have indeed been reported in the literature [29], although they appear much less common than their invaginated counterparts. Figure 2.7 shows that an additional complexity arises from the fact that there is an energy barrier between the flat and budded states. This means that intermediate states (such as the hemispherical state – β = 0.5 of Fig. 2.7) are very unfavorable. This fact has two very important physical consequences. On the one hand, this means that the transition from invaginated



Fig. 2.7 Variation of the energy of a domain as a function of the domain shape for different membrane tension. The shape is characterized by a single parameter β , which vanishes for a flat domain and is equal to unity for a fully budded domain (corresponding to a closed sphere, a state that can be attained only if the domain endocytoses). The energy plot shows the domain energy (in k_BT unit) for a domain of size R = 100 nm, of line energy $\sigma = k_BTnm$, and of bending rigidity \varkappa = 20 k_BT . The energy is shown for three different membrane tensions. Under low tension (red), the domain tends to bud to minimize contact with the surrounding phase, and the

minimum of energy is for $\beta = 1$. As the membrane tension increases (green), the energy of the budded domain increases with respect to the energy of the flat domain. At even higher tension, the flat domain becomes the minimum of energy. Flattening the invagination requires an energy barrier (ΔF) to be exceeded. The transition over an energy barrier (inset) is activated by thermal fluctuations, and requires a time that increases exponentially with the height of the barrier (see text). The higher the membrane tension, the lesser the time needed to operate the flattening of the invagination.

to flat domains does not occur continuously. Instead, the domains will abruptly snap open when the membrane tension is raised to a sufficient value. We argue below that interesting biological functions for caveolae could stem from this physical fact. Furthermore, we have seen in Eq. (1) that states of high energy are exponentially unlikely, which means that the passage of a barrier ΔF requires an appropriate thermal fluctuation. This is a rare event which occurs only after a time proportional to $e^{\Delta F/k_{\rm B}T_m}$. This means that the response of caveolae to a mechanical perturbation is sensitive to the time scale over which this perturbation occurs.

A new putative function of caveolae at the plasma membrane emerges from these physical considerations, in addition to their supposed role in cell signaling

and cholesterol transport (to cite only a few). Specifically, this is a possible role in cell mechano-sensitivity and mechano-regulation. Although direct experimental evidence of such role is lacking at this time, Figure 2.7 provides a clear picture of the effect that membrane tension should have on caveolae. The one remaining question at this stage is the level of membrane tension required to affect caveolae morphology. We are able to relate this tension to relevant physical parameters of the caveolar membrane, such as the line tension, bending rigidity, and spontaneous curvature [22,30]. The expected values for these parameters leads to the identification of a characteristic membrane tension that can be observed in cells. At present, however, these parameters are not known with sufficient precision for the theory to produce quantitatively precise predictions.

We may however investigate some biological consequences of the disruption of caveolae at high membrane tension. In that respect, the scenarios presented in Sections 2.5 and 2.3 are somewhat different. If the very formation of the caveolin raft is coupled with membrane curvature (which is what the thermodynamic model of Section 2.5 predicts), increasing the membrane tension will lead not only to flattening of the invagination but also to dispersion of the caveolin aggregate. One can relate this phenomenon to a putative function of caveolae in cell signaling, which is to hold signaling components inactive until they are released and activated by an appropriate stimulus [31]. The thermodynamical model of Section 2.5 suggests that an increase in the mechanical tension of the cell membrane could provide such a stimulus. If, on the other hand, the phase separation leading to caveolin aggregates is independent of the shape of the membrane, the caveolin rafts will remain regardless of the membrane mechanical tension. Their morphology will however change upon tension increase, as is described Figure 2.7.

The morphological changes of caveolae with the tension of the plasma membrane provides the basis for mechanical regulation at the cell membrane. It is known that the tension of a cell increases if it is mechanically perturbed [32], and that cells have developed regulatory mechanisms to accommodate mechanical perturbations [32,33]. Caveolae might play a role in this mechanism, and many pathologies associated with caveolin seem to involve the mechanical behavior of the cell. One can cite their involvement in various muscular diseases [34], and defects in vascular relaxation and contractility in mice deficient of caveolin-1 [4]. There is an increase of the number of caveolae in Duchènne muscular dystrophy [35], and such an increase has also been observed in cells subjected to long-lasting shear stress [36]. Furthermore, there exists evidence that caveolin can contribute significantly to cell-cycle regulation [37], and cell entry into mitosis can be inhibited by artificially maintaining a high level of caveolin prior to mitosis. One can argue on the basis of physical arguments that the coexistence of flat and invaginated membrane domains regulates cell membrane tension [30]. Figure 2.8 shows that the mechanism of a perturbation of the cell membrane area can be buffered by the flattening and invagination of domains. By this mechanism, caveolin expression could regulate and buffer the membrane tension of cells by controlling the number of caveolae at the cell membrane. Along with many other factors, this could be one reason why caveolin is down-regulated prior to mitosis [37], as a means low-



Fig. 2.8 The possible involvement of caveolae in mechanical regulation at the cell membrane. (Left) A mechanical perturbation is said to be positive if membrane area is taken out, and negative if it is put in. In the absence of invaginations, such perturbation would influence the membrane tension (visualized as a spring of various length), increasing it for positive perturbation, and decreasing it for negative perturbation. The presence of membrane invaginations, for which caveolae are a good candidate, allow buffering of the changes in membrane tension, as positive perturbations flatten the invagination (P1) prior to tension increase (P2), and negative perturbation leads to more invagination (N1), before it starts affecting the tension (N2). (Right) Variation of membrane tension with perturbation in the absence of invagination (dashed line), and with invagination (solid line). The membrane reservoir in the invagination helps keep the tension unaltered.

ering membrane tension by releasing the membrane area stored in caveolae, thereby assisting cell division. This is consistent with the fact that caveolin-1 knockout mice show an increased rate of cellular proliferation [4].

Compensatory pathways exist for cells lacking caveolin that do not exhibit caveolae at their plasma membranes [4]. In order to investigate further the possible role of caveolae in tension regulation, one potentially promising line of research might be to perturb (e.g. mechanically) cells that do have caveolae, and to observe the effect of that perturbation on caveolae before any alternative regulatory mechanisms can have a significant effect.

2.7

Conclusions

Fluid membranes can be described using physical laws. The membrane has some tension which resists the formation of invaginations of finite area. The area of each invagination must be removed from the rest of the cell membrane under tension in order to form a bud. The membrane has a rigidity which parameterizes the stiffness of the membrane. Patches (rafts) on the membrane which have a distinct chemical composition also experience a line tension that acts at the interface of the patch with the rest of the membrane. This line tension acts to minimize the length of the contact line and tends either to make the domain circular or to form a bud, with a much reduced contact region near the neck of the bud. Finally, membrane inclusions such as caveolin that are distributed asymmetrically in the membrane can always be expected to make the membrane curve. It is argued that the predominantly random coil of the N-terminal section of caveolin acts to bend the membrane away from it and hence favors the formation of endo, rather than exo, buds. These physical ingredients can be combined in simple theories to understand when, and if, caveolae-like invaginations should be stable and, if so, at what length scales. The characteristic scales that arise from comparing surface tension and line tension with rigidity both are in the 100 nm range, and thereby provide an early hint that a balance between the various physical effects listed above could be expected to give rise to invaginations similar to caveolae. We describe how this can be analyzed in more detail, leading to a theory for the stability of the buds under variation of surface tension. We find that buds of a single characteristic size stabilized by line tension can act as mechanical tension regulators and how these could very effectively buffer the cell's tension in the physiological range.

Abbreviations

AA	amino acid
cms	critical micelle concentration
cbc	critical budding concentration

References

- 1 Rothberg, K. G., J. E. Heuser, W. C. Donzell, Y.S. Ying, J.R. Glenney, and R.G. Anderson. Caveolin, a protein component of caveolae membrane coats. Cell 68, 673-682 15 Sheetz, M.P. Cell control by membrane-cy-(1992).
- 2 Simons, K., and E. Ikonen. Functional rafts in cell membranes. Nature 387, 569-572 (1997).
- 3 Schlegel, A., D. Volonté, J.A. Engelman, F. Galbiati, P. Mehta, X.L. Zhang, P.E. Scherer, and M.P. Lisanti, Crowded little caves: structure and function of caveolae. Cell Signal 10, 457-463 (1998).
- 4 Drab, M., P. Verkade, M. Elger, M. Kasper, M. Lohn, B. Lauterbach, J. Menne, C. Lindschau, F. Mende, F.C. Luft, A. Schedl, H. Haller, and T.V. Kurzchalia. Loss of caveolae, vascular dysfunction, and pulmonary defects in caveolin-1 gene-disrupted mice. Science 293, 2449 (2001).
- 5 Hailstones, D., L.S. Sleer, R.G. Parton, and K.K. Stanley. Regulation of caveolin and caveolae by cholesterol in MDCK cells. J. Lipid Res. 39, 369-379 (1998).
- 6 Liu, J., X.B. Wang, D.S. Park, and M.P. Lisanti. Caveolin-1 expression enhances endothelial capillary tubule formation. J. Biol. Chem. 277, 10661-10668 (2002).
- 7 Helfrich, W., and Servus R. Nuovo Cimento D 3, 137 (1984).
- 8 Safran, S.A. Statistical Thermodynamics of Surfaces, Interfaces and Membranes. Perseus, Cambridge, MA (1994).
- 9 Evans, E., and Rawicz, W. Entropy-driven tension and bending elasticity in condensed-fluid membranes. Phys. Rev. LEtt. 64, 2094-2097 (1990).
- 10 Brochard, F., and J.F. Lennon. J. Physique (Paris) 36, 1035 (1975).
- 11 Chaikin, P.M., and T.C. Lubensky. Principles of condensed matter physics. Cambridge University Press, Cambridge UK (1995).
- 12 Reichl, L.E. A modern course in statistical physics. Edward Arnold, London (1980).
- 13 Gov, N.S., and S.A. Safran. Red blood cell membrane fluctuations and shape controlled by ATP-induced cytoskeletal. Defects Biophys. J. 88, 1859-1874 (2005).
- 14 Oh, P., D. P. McIntosh, and J. E. Schnitzer. Dynamin at the neck of caveolae mediates their budding to form transport vesicles by

GTP-driven fission from the plasma membrane of endothelium. J. Cell Biol. 141, 101-114 (1998).

- toskeleton adhesion, Nat. Rev. Mol. Cell Biol. 2, 392-396 (2001).
- 16 Lipowsky, R. (1992) Budding of membranes induced by intramembrane domains J. Phys. II France 2, 1825-1840. Lipowsky, R. Domain-induced budding of fluid membranes Biophys. J. 64, 1133-1138 (1993).
- 17 Turner, M.S., P. Sens, and N.D. Socci. Recycling and the control of raft-like membrane domains Phys. Rev. Let. (Submitted).
- 18 Mouritsen, O., and M. Bloom. Biophys. J. 46, 141 (1984); Annu. Rev. Biophys. Biomol. Struct. 22, 145 (1993).
- 19 Leibler, S. Curvature instability in membranes. Journal de Physique I 47, 507-516 (1986).
- 20 Lipowsky, R. Bending of membranes by anchored polymers. Europhys. Lett. 30, 197-202 (1995).
- 21 Szleifer, I., O.V. Gerasimov, and D.H. Thomson. Spontaneous liposome formation induced by grafted poly(ethylene oxide) layers. Proc. Natl. Acad. Sci. USA 95, 1032-1037 (1998).
- 22 Sens, P., and M.S. Turner. Theoretical Model for the Formation of Caveolae and Similar Membrane Invagination. Biophys. *J.* **86**, 1–9 (2004).
- 23 Evans, A., M.S. Turner, and P. Sens. Interactions between proteins bound to biomembranes. Phys. Rev. E. 67 041907 (2003).
- 24 Doi, M., and S.F. Edwards. The Theory of Polymer Dynamics. Oxford University Press, Oxford UK (1984).
- 25 de Gennes, P.G. Scaling Concepts in Polymer Physics. Cornell University Press, Ithaca (1979).
- 26 Bickel, T., C. Jeppesen, and C. M. Marques. Local entropic effects of polymers grafted to soft interfaces. Eur. Phys. J. E 4, 33-43 (2001).
- 27 Li, S., F. Galbiati, D. Volonte, M. Sargiacomo, J.A. Engelman, K. Das, P.E. Scherer, and M.P. Lisanti. Mutational analysis of caveolin induced vesicle formation.

References

to caveolae membranes. FEBS Lett. 434. 127-134 (1998).

- 28 Sear, R. P., S.W. Chung, G. Markovich, W.M. Gelbart, and J.R. Heath. Spontaneous patterning of quantum dots at the airwater interface. Phys. Rev. E. 59, R6255-R6258 (1999).
- 29 Prescott, L., and M.W. Brightman. The sarcolemma of Aplysia smooth muscle in freeze-fracture preparations. Tissue Cell 8, 248-258 (1976).
- 30 Sens, P., and M.S. Turner. Budded membrane microdomains as regulators for cellular tension (2005).
- 31 Parton, R.G. Life without caveolae. Science 293, 2404 (2001).
- 32 Raucher, D., and M. P. Sheetz. Characteristics of a membrane reservoir buffering membrane tension, Biophys. J. 77, 1992-2002 (1999).
- 33 Morris, C.E., and U. Homann. Cell surface area regulation and membrane tension J. Membrane Biol. 179, 79-102 (2000).

- Expression of caveolin-1 recruits caveolin-2 34 Woodman, S. E., F. Sotgia, F. Galbiati, C. Minetti, and M.P. Lisanti, Caveolinopathies: mutations in caveolin-3 cause four distinct autosomal dominant muscle diseases. Neurology 62, 538-543 (2004).
 - 35 Repetto, S., M. Bado, P. Broda, G. Lucania, E. Masetti, F. Sotgia, I. Carbone, A. Pavan, E. Bonilla, G. Cordone, M. P. Lisanti, and C. Minetti. Increased number of caveolae and caveolin-3 overexpression in Duchènne muscular dystrophy. Biochem. Biophys. Res. Commun. 261, 547-550 (1999).
 - 36 Park, H., Y.-M. Go, P. L. St. John, M. C. Maland, M.P. Lisanti, D.R. Abrahamson, and H. Jo. Plasma membrane cholesterol is a key molecule in shear-dependent activation of extracellular signal-regulated kinase. J. Biol. Chem. 273, 32304-32311 (1998).
 - 37 Fielding, C.J., A. Bist, and P.E. Fielding. Intracellular cholesterol transport in synchronized human skin fibroblasts. Biochemistry 38, 2506-2513 (1999).

3 The Biophysical Characterization of Lipid Rafts

Pranav Sharma, Rajat Varma, and Satyajit Mayor

3.1 Introduction: The Fluid Mosaic Model and Membrane Domains

In an influential review in the early 1970s, Singer and Nicolson summarized the available information on the property of the lipid bilayer, and proposed a "fluid mosaic" model of the plasma membrane [1]. These authors suggested that biological membranes are two-dimensional fluids of lipids, in which integral membrane proteins are dissolved; peripheral proteins are attached to the surface of the membrane and protein milieu (Fig. 3.1). Lipids and proteins were proposed to have unrestricted lateral mobility, but restricted transverse mobility. These considerations were construed to result in a homogeneous distribution of lipids and proteins in biological membranes. However, the spatial and temporal inhomogeneity of lipids, while not explicitly advocated, was certainly not ruled out. In fact, domain models of cell membranes, as mosaic rather than fluid, have existed for a long time, based primarily on the properties of lipids in artificial liposome membranes [2,3]. The fluid-mosaic model of Singer and Nicolson also considered the possibility of small membrane domains in the fluid cell membrane bilayer. However, all these models of cell membranes did not focus on relating any specific biological functions that required domain formation.

3.2 The Origin of the Raft Hypothesis

Membrane heterogeneities resurfaced primarily as a necessity to explain a specific biological observation of preferential sorting of glycosphingolipids to apical membranes in polarized epithelial cells [4]. This idea was finally developed into a notion of lipid rafts by Simons and coworkers, and was centered on cholesterol-containing domains as the cornerstone of the raft model of life in a living cell membrane [5] (Fig. 3.2A).

Lipid rafts were hypothesized as specialized regions of cell membrane where sphingolipids and cholesterol come together as a result of chemical affinity and/or



Fig. 3.1 The fluid-mosaic model of the plasma membrane. In this famous model, Singer and Nicholson summarized experimental observations from studying diffusion processes in cell and artificial membranes to

suggest that proteins are dissolved in a twodimensional fluid. They also considered the possibility of small membrane domains (~100 nm at most) in the fluid cell membrane bilayer.

their preferential packing. These regions could include or exclude other lipids and proteins, and this specific segregation was proposed to mediate their biological function [5]. Lipid rafts have been implicated in a variety of functions such as sorting, endocytosis, signaling, and cell migration [6]. Although lipid rafts are implicated in many fundamental biological functions, there is significant confusion in their definition, evidence for their existence, and their precise role in biological function. Numerous models to explain their structure and function have been proposed and are summarized in Figure 3.2.

Lipid rafts have been studied and defined with a variety of techniques, and this has resulted in various terminologies such as 'microdomains', detergent-insoluble glycolipid/glycosphingolipid-enriched complexes/membranes (DIGs), and detergent-resistant membranes (DRMs). In order to avoid confusion, the term "lipid rafts" in this chapter will be used henceforth as a general description of lipid-based membrane heterogeneity, and wherever necessary followed by the method used to define it.

This chapter focuses on the biophysical tools that have been used to examine membrane heterogeneities.





Fig. 3.2 Current raft models. (A) The most commonly cited hypothesis for membrane rafts proposed by K. Simons (Dresden, Germany) [5] depicts rafts that are relatively large structures (~50 nm), enriched in cholesterol and sphingolipid (SL), with which proteins are likely to associate. (B) Anderson and Jacobson visualize rafts as lipid shells which are small, dynamic molecular-scale assemblies in which "raft" proteins preferentially associate with certain types of lipids [83]. The recruitment of these "shells" into functional structures could be a dynamic and regulated process. (C) Another point of view is that a large fraction of the cell membrane is raft-like and exists as a "mosaic of domains"; cells

regulate the amount of the different types of domains via a cholesterol-based mechanism [84]. (D) Actively generated spatial and temporal organization of raft components. A different picture that is consistent with data from GPI-anchored protein studies in living cells suggest that pre-existing lipid assemblies are small and dynamic, and coexist with monomers [30]. They are actively induced to form large-scale stable "rafts". Black circles = GPI-anchored proteins; red and pink circles = non-raft-associated lipids; yellow circles = raft-associated lipids; green = cholesterol. Scale bar = ~5 nm. (Reprinted with permission from [30].)



Fig. 3.3 (A) Membrane topology of GPI-anchored proteins in comparison to transmembrane (TM) proteins and fatty-acid linked proteins. (B) Core residues of the GPI-anchor. (Reprinted with permission from [7].)

3.3

The Role of Lipid-Anchored Proteins in the Development of the Membrane Raft Hypothesis

Before reverting to the heart of the matter, it is useful to examine why glycosylphosphatidylinositol (GPI)-anchored proteins have been used as markers for rafts. GPI-APs are a class of extracytoplasmic proteins that undergo a post-translational lipid modification that results in the exchange of an often perfectly competent hydrophobic transmembrane protein anchor with a glycolipid, the GPI-anchor [7] (Fig. 3.3). The GPI-anchor is the sole anchor by which most of these proteins attach to membranes, and often is required for targeting to plasma membrane after their synthesis and post-translational lipid modification in the endoplasmic reticulum (ER) [8]. Diverse proteins are GPI-anchored, and thus the GPI-anchor has been implicated in providing functions such as specific sorting in the secretory and endocytic pathway [7,9], and signaling [6,10-12]. In addition, in most cells almost all of the GPI-anchored proteins are present in DRMs. The requirement for maintenance of specific lipid composition for these special sorting and signaling characteristics, and the fact that the GPI-anchor did not extend beyond the first leaflet of the bilayer, provided the impetus to examine the distribution of GPIanchored proteins in cell membranes as indicators of lipid-based heterogeneity markers. Similar arguments are applicable to other lipid-tethered proteins where the lipid anchor appears to specific sorting and signaling specificity to the lipidtethered proteins in question [13].

3.4 The Case For and Against DRMs as Evidence for "Rafts" in Cell Membranes

For a long time, the existence of DRMs was thought to reflect "lipid rafts" in biological membranes, and this is still one of the most widely used methods to define lipid rafts. This was despite the notion that many membrane-associated proteins and lipids interacting with the cytoskeleton directly or via complexes could be resistant to detergent extraction [14,15]. Studies examining the interaction of detergent with membranes are about as old as models that predicted the fluid nature of biological membranes [16]. Membranes of Semliki Forest virus were treated with different detergents, and various stages of virus membrane dissociation (following detergent treatment) were investigated for understanding virus function [17–19]. Simons and coworkers, following their studies on the preferential apical transport of some proteins and glycosphingolipids [4], suggested that there may be a correlation between apical transport of these proteins and glycosphingolipids. This makes apical sorting of GPI-APs along with glycosphingolipids among the first biological function proposed for lipid rafts. Simons and coworkers proposed that glycosphingolipid-enriched patches might act as a sorting platforms for proteins that are preferentially sorted to apical compartments. In addition, GPI-APs and several sphingolipids were shown to become resistant to cold Triton-X 100 solubilization (also referred to as DRMs) in the trans-Golgi network (TGN) before their delivery to apical membranes in polarized epithelial cells [20,21]. The roots of the widespread use of cold nonionic detergent as an evidence of membrane inhomogeneities or lipid rafts (thus referred to as DRMs or DIGs) and its link to biological function may be partially traced to this seminal study. Brown and Rose showed that the cold detergent-insoluble complexes from lysates of polarized Madin-Darby canine kidney (MDCK) cells contained moderate amounts of apically delivered protein human placental alkaline phosphatase (PLAP) along with other GPI-APs. These detergent-insoluble membranes were concentrated in glycosphingolipids, but proteins destined for delivery to basolateral compartments were missing. This led the authors to speculate that these cold detergent-insoluble complexes could form the sorting platforms, as suggested by Simons and coworkers, for the apical delivery of PLAP.

The idea of DRMs as a biochemical purification of a specific membrane domain in living cells was reinforced from the studies investigating the phase behavior of lipids in membranes. Constituent lipids determine the state of membrane; lipids with saturated long acyl chains exists in the "gel state" with restricted lateral mobility and long range order because they are able to tightly pack the hydrocarbon chains of the lipids, whereas lipids with unsaturated (kinked) acyl chains exist as a "liquid crystalline state" with faster lateral mobility because they are unable to pack in the same fashion. Clearly, thermal effects determine in which state(s) these lipids may exist, and also the phase diagram of coexisting phases [22]. Extensive studies with artificial lipid membranes has suggested that lipids in the liquid crystalline state can exist in a liquid-disordered (l_d) or liquid-ordered (l_o) state [23], differentiated only by local order parameter. A lo state is characterized by a gel-like

50 3 The Biophysical Characterization of Lipid Rafts

short-range local order but liquid crystalline state-like translational mobility. Sphingolipids with saturated long acyl chains and short rigid cholesterol undergo preferential packing into l_o phases [24,25]. The arguments that biological membranes contain patches of lo domains in otherwise ld membranes are supported by the following observations:

- 1. There is a coexistence of $l_{\rm o}$ and $l_{\rm d}$ phases in model membranes.
- Model membranes known to be in l_o state are insoluble in cold, nonionic detergent (e.g., Triton X-100), and the physical state of model membranes treated with cold detergent were similar to that of the l_o state.
- A fraction of the lipids in biological membranes are insoluble in cold nonionic detergent.

This physical picture provided the much-needed rationale to relate the existence of these detergent-insoluble complexes to 'rafts' in cell membranes, leading to a flood of reports investigating the components of cold detergent-insoluble membranes and linking them with biological function; indeed, several studies on the "proteome" of DRMs are also available today [26,27]. Even to date, detergent insolubility remains the most widespread method of identifying association with lipid rafts.

Despite numerous studies involving detergent insolubility on cells and artificial membranes, there remains a lack of understanding of phase behavior, composition, and also physical principles governing the association of lipids in the membranes of living cells. It would be difficult to consider the treatment of membranes with detergent as being nondisruptive. However, the phenomena of detergent insolubility could result from the chemical interaction (or possibly lack of it) of certain lipids with detergents. So, the interpretation of this technique rests heavily on the premise that detergent insolubility reflects some "chemical interaction" between lipids that existed in live cells prior to detergent treatment. This "chemical interaction" needs to be over and above the chemical interaction of various lipids to the detergent itself. To date, we lack a direct correlation between live cell lipid raft/ microdomain organization and the resulting detergent-insoluble complexes. In the absence of any clear understanding of the relationship between "a priori" live cell organization and resulting phenomena of cold detergent insolubility, the suggestion that cold detergent insolubility reflects presence of some organization on living cell is, at best, speculative.

Support for these suspicions came from the studies of Heerklotz [28], who examined the phase transitions and phase structures of model lipid membranes using microcalorimetry and solid state nuclear magnetic resonance (NMR) [28]. When examining vesicles composed of an equimolar mixture of 1-palmitoyl-2-oleoyl-*sn*glycero-3-phosphocholine (PC), egg sphingomyelin (SM) and cholesterol, Heerklotz did not find any significant fraction of membrane in ordered state at 37 °C. However, the addition of increasing amounts of the detergent Triton X-100, or a lowering of the temperature, promoted the formation of ordered domains in these membranes, which were resistant to subsequent detergent extraction procedure. Hence, instead of reflecting upon the ordering of lipids in membrane, Triton X-100 could cause the formation of such ordered domains. In a follow-up study, Heerklotz et al. showed that very small transition energies are required to induce changes in the size and abundance of ordered domains in artificial model membrane systems [29]. Hence, the perturbations caused by incorporating detergents could result in artifacts, especially in studies involving artificial lipid membranes. In biological membranes with complex lipid mixtures and proteins, there is a possibility of some component(s) acting as a stabilizer of the lipid domains, but the use of detergents to probe this situation is fraught with the concerns raised above. At best, DRM-association reveals a biochemical propensity that may be associated with ordered lipid domains [9,30].

3.5 Why Are Biophysical Studies Useful for Understanding Lipid Rafts?

Given the large size (larger than the limit of resolution of light microscopy) of membrane heterogeneities observed in artificial model membranes, one would expect to observe them in biological membranes with reasonable ease. However, even with the spectrum of protein- and lipid-related techniques available, ranging from electron microscopy, optical tweezers and single molecule studies to biochemical detection, the direct visualization of lipid rafts has been elusive. An association of proteins expected to be present in lipid rafts or DRMs is not detectable by using biochemical procedures such as SDS-PAGE, which indicates that an association between raft constituents – if present – is weak, noncovalent, and/or transient.

Within the resolution of light microscopy, DRM constituents such as GPI-anchored proteins appear to be diffusely distributed [31], indicating that if segregated "raft" structures exist they must be smaller than the resolution of light microscopy (at best 200 nm). Studying the distribution of GPI-anchored proteins by electron microscopy (EM), which has the correct scale of detection, also failed to detect any clustering of lipid raft constituents [31–33]. Elaborate fixation procedures and the efficiency of labeling with electron-dense tags might be possible causes of this loss of detection sensitivity [34]. As each technique is scrutinized for its suitability for raft detection, it has become clear that these structures – if they exist – are extremely difficult to visualize and/or detect in direct manner.

In the absence of any other option, indirect methods remain in widespread use to define the constituents and structure of lipid rafts; biophysical studies hold the promise of understanding lipid rafts. In the following sections a variety of different biophysical techniques used to study lipid rafts are discussed, together with an idea of their impact on our understanding of "raft" structures in living cell membranes.

The biophysical techniques used to examine the organization of lipid-based components may be divided into two types:

• those based on studying the diffusion characteristics of membrane components, based on the assumptions that raft-association will be detected as a change in

52 3 The Biophysical Characterization of Lipid Rafts

local membrane viscosity, or will generate large-sized entities that should show deviations from diffusive behavior attributable to monomer diffusion; and

• those based on detecting enhanced proximity between raft-associated molecules.

3.6

Diffusion-Based Measurements

Two main types of diffusion measurement have been analyzed, namely single molecule-based methods such as single-particle tracking (SPT), and ensemble-based methods such as fluorescence recovery after photobleaching (FRAP) or fluorescence correlation spectroscopy (FCS).

3.6.1

Single-Molecule Studies

Single-particle tracking studies, which involves observing the motion of a single molecule tagged with nanometer-sized colloidal gold or fluorescent dye [35], has been used to elucidate the fine structure of biological membranes (Fig. 3.4). Sheets and coworkers used video-enhanced bright field microscopy on 40-nm colloidal gold labeled to antibodies against Thy-1 or cholera toxin B (CtxB) subunit to study the movement of Thy-1 or GM1 respectively, in C3H 10T1/2 fibroblasts [36]. They divided the observed motion of molecules into four categories: fast diffusion; slow diffusion; confined diffusion; and stationary at 6.6 s time scale (200 frames at video rate of 30 frames per second). While significantly higher fraction of Thy-1 and GM1 showed confined diffusion (37% and 35%, respectively), only 16% of fluorescein phosphatidylethanolamine (fl-PE; phospholipid analogue) molecules displayed confined diffusion. Moreover, the treatment of cells with a glycosphingolipid inhibitor (which reduced the glycosphingolipid level by $\sim 40\%$) reduced the fraction of Thy-1 (28%) molecules undergoing confined diffusion. This indicates a role of glycosphingolipids in the reduced mobility of Thy-1. At this timescale of 6.6 s, Thy-1 and GM1 were found in transient confinement zones (TCZs) averaging 325 to 370 nm in diameter. At the 60 s observation window, Thy-1 and GM1 were confined to TCZs for 7-9 s, with an average diameter of between 260 and 330 nm. Interestingly, the extraction with cold Triton X-100 did not affect either fraction of molecules showing confined diffusion or diameter of confined region. The authors concluded that these confined reasons might be in-vivo correlates of detergent-insoluble complexes. The same group conducted a further characterization [37] and found that the lipid analogues 1,2-dio-leoyl-sn-glycero-3-phosphoethanolamine-fluorescein (FL-DOPE) and 1,2-dipalmitoyl-sn-glycero-3-phosphoethanolamine-fluorescein (FL-DPPE) spent significantly less time in TCZs compared to Thy-1 and GM1. For Thy-1, the confinement times were significantly reduced upon cholesterol depletion. The authors investigated the mobility of Thy-1 within the TCZs at a higher time resolution, by capturing data using a high-speed



Fig. 3.4 Single-molecule tracking of membrane molecules suggests a model of a plasma membrane that resembles a partitioned fluid supported by a cytoskeleton mesh. High-speed tracking of single molecules in cell membranes reveals an unexpected behavior of diffusing particles which suggests the plasma membrane is a fluid that is partitioned by a membrane cytoskeleton fence. Using this technique, Kusumi et al. identified three types of lipidic structures that formed in the plasma membrane [85]. Type (a) is prevalent in the absence of extracellular stimulation; these are small (perhaps consisting of several molecules) and unstable (the lifetimes may be less than 1 ms), and may be the type that monomeric GPI-anchored proteins associate with [78]. Type (b) may appear when receptor molecules form oligomers upon liganding or crosslinking. The receptors may be GPI-anchored receptors or transmembrane receptors with some affinity to cholesterol and saturated alkyl chains. Oligomerized receptors may then induce small but stable rafts around them, perhaps due to the slight reduction in the thermal motion around the cluster and the subsequent assembly of cholesterol. Given the rather stable oligomerization of the receptor molecules, the type (b) raft may be stable for minutes. although the associated raft-constituent molecules may be exchanged frequently between the raft and the bulk domains. Type (c) structures may be formed around these receptor rafts. (Image reprinted with permission from [85].)

digital video camera (500 frames per second for up to 10 s). The diffusion coefficient within TCZs was reduced by a factor of ~2 compared to the one outside of the TCZs. In another study, the diffusion of GPI and transmembrane-anchored forms of neural cell adhesion molecule (NCAM) was studied in NIH-3T3 fibroblasts [38]. Surprisingly, a similar confinement was shown for GPI as well as the transmembrane forms of NCAMs.

Schutz et al. used the single-particle tracking of lipid labeled with fluorescent dye, a method referred to as "single dye tracing" [39]. In this approach, dilute quantities of Cy5-labeled, saturated lipid probe 1,2-dimyristol-*sn*-glycero-3-phosphoethanolamine (DMPE-Cy5) and a mono-unsaturated lipid probe DOPE-Cy5 were introduced into human coronary artery smooth muscle (HASM) cells. The single dye tracing data indicated rapid and confined diffusion for DMPE-Cy5 with

a long residence time and a confinement area in the order of \sim 600–700 nm. By contrast, DOPE-Cy5 displayed a relatively unconfined diffusion within the membrane.

Pralle et al. examined the viscous drag of GPI and transmembrane-anchored proteins in regions much smaller than TCZs (as defined by video-based singleparticle tracking experiments [40]). These authors performed high-resolution particle tracking with a bead (labeled with antibody as well as a fluorophore) held by laser trap and bound to membrane protein, and then observed the amplitude of fluctuations of a 2-mm bead in a laser trap of defined spring constant. The amplitude of fluctuations in turn may be related to the local viscosity by the Saffman-Delbruk model of diffusion in two dimensions [41]. The authors reported that raft proteins (by cold detergent-insolubility criterion) such as hemagglutinin (HA), PLAP or chimeric YFPGLGPI (a GPI-anchored protein containing the signal sequence of lactase-phlorizin hydrolase (LPH) fused to yellow fluorescent protein; YFP) experienced an approximately three-fold higher viscous drag than non-raft LYFPGT46 (YFP ectodomain fused to the transmembrane domain of the LDL receptor). Cholesterol depletion resulted in a significant decrease in the viscous drag of HA and GPI-APs, whereas viscous drag of LYFPGT46 was unchanged after cholesterol depletion. By applying the Saffman-Delbruck model for diffusion in biological membranes, Pralle et al. estimated the size of rafts as being \sim 26 ± 13 nm. However, the application of this theory to membrane diffusion in an heterogeneous environment is fraught with complications, and even if applicable the extremely insensitive relationship between particle size and diffusion coefficient [Diffusion coefficient $\infty \ln (1/R)$] makes the estimation of size inaccurate.

Kusumi et al. [42] presented a comprehensive review of the various factors that might influence the diffusion of a protein in the plasma membrane (Fig. 3.4). These authors suggested that there might be two different types of raft in the plasma membrane: (1) steady-state rafts or "reserve rafts" that exist at all times; and (2) "clustered rafts", which are more stable long-lived structures that might be created when processes such as signaling are initiated. Kusumi et al. also defined all possible physical interactions that take place between lipids and proteins that would give rise to segregation and, most importantly, associated lifetimes to each of these processes that are critical when designing any type of experiment to detect such structures. For many years Kusumi's group has focused its efforts on understanding the diffusion of proteins in membranes, by using single-particle tracking [43]. This technology allows single particles to be tracked at a time resolution of 25 ms, whereupon most proteins and lipids are seen not to undergo free unrestricted diffusion but rather to diffuse freely into small compartments between 30 and 250 nm in size that are bounded by cytoskeletal "fences". The proteins and lipids cross these fences at a frequency ranging from 1 to 25 ms, depending on the cell type [43]. Such a phenomenon is called "hop diffusion", and the frequency with which molecules hop from one compartment to the other is called the "rate of hop diffusion". Both theoretical models and experiments appear to coincide in this understanding of the structure of the cell membrane where the plasma membrane is actively partitioned by the cytoskeleton that it rests on into membrane skeleton pickets and fences.

When such experiments were performed on GPI-anchored proteins, the latter were found to have a rate of hop diffusion that was indistinguishable from that of phospholipids [42]. Surface scanning resistance (SSR) measurements are performed by coating a bead with antibodies to a receptor on the cell surface, and this bead is held by a laser trap while the stage is scanned in two dimensions [44]. The resistance felt by the bead is reflective of the barriers to diffusion present on the cell surface, and can be used to determine the linkages of the receptor to the cytoskeleton. SSR measurements performed on Qa-2 (GPI-anchored MHC class I) revealed that, using a low concentration of antibodies on the bead, the barriers encountered by the bead were similar to that observed by a monomeric GPI-AP [45]. When a higher concentration of antibody was used, it resulted in crosslinking of Qa-2 and the barriers observed were indicative of an association with a transmembrane protein that has linkage with the actin cytoskeleton. This result was surprising, and questioned most existing models of rafts. The authors interpreted their results by saying that these were the representatives of reserve rafts that consisted of a very small number of GPI-anchored proteins having a hop diffusion rate of 25 ms on average. If the fraction of molecules that resided in aggregated species was small, then in single-particle measurements the probability of labeling those species is also lowered and it becomes difficult to obtain statistically significant data on those events.

As indicated above, by using single-particle tracking, albeit at a lower time resolution, Jacobson and coworkers reported a phenomenon for raft-like molecules. They found that GPI-anchored proteins – but not other transmembrane proteins – underwent slow diffusion in small zones that were 200 nm in size [36,38, 86], called TCZs. Kusumi and coworkers clarified that these TCZs could not be compared to the membrane compartments that they observed because the time resolution of the two techniques was very different [42]. It is also possible that due to the inability to have single antibody per gold particle, Jacobson and co-workers might have observed diffusion of small clustered molecules. The issue of multi-valency of the tagging particle remains a matter of great concern in the correct interpretation of results from these techniques [46].

3.6.2 Fluorescence Recovery After Photobleaching

Fluorescence recovery after photobleaching (FRAP) is a well-established technique which is used to study the lateral mobility and fluorescence dynamics of proteins in membranes [47,48]. After labeling of the surface molecules with a fluorescent tag, a defined micrometer-sized area on the cell membrane is photobleached (irreversible photo-destruction) within a very short time, using a focused laser beam. This is followed by an observation of the recovery of fluorescence in the bleached area (by diffusion of fluorescent molecules from neighboring regions). Two parameters may be obtained from a rigorous analysis of the data: (1) the rate(s) of recovery, which results in an estimation of a rate of diffusion; and (2) the immobile fraction. The rates of fluorescence recovery are used to estimate the diffusion
56 3 The Biophysical Characterization of Lipid Rafts

constants, whereas the discrepancy between fluorescence intensity before photobleaching and after complete recovery of fluorescence provides the fraction of immobile molecules [48]. Although this procedure does not throw light on the motion of individual molecules, one clear advantage of FRAP over single molecule methods is its superior statistical confidence resulting from the averaging of large number of diffusion events. Since the technique results in the destruction of fluorophores to achieve the measurement, there are issues regarding the damage caused by this photobleaching phenomenon. In addition, the time scales and sensitivity of the technique are limited by instrumentation, and the many different models of diffusion may be used to explain the recovery characteristics. A prior physical picture of the diffusion process is usually essential in deciphering the data [49–51].

In one of the earliest FRAP studies of DRM-associated proteins, Ishihara et al. showed that although Thy-1 exhibited a diffusion constant similar to that of labeled lipids, while up to 50% of the protein was immobile on the surface of various cell types [52]. Hannan et al. [53] also showed the presence of an immobile fraction of GPI-APs on the surface of MDCK cells. When placed in perspective of recent developments in the picket-fence model, this immobile fraction would include the proteins which diffused within pickets but were restricted in mobility at larger timescales. Moreover, the results did not provide the immobile fraction of "nonraft" protein for comparison. Oliferenko et al. [54] performed FRAP measurements on a transmembrane-anchored lipid raft (DRM-associated) marker CD44 expressed in EpH4 cells (polarized mammary epithelial cells). These authors found that CD44 was significantly immobilized compared to a transferrin receptor (TfR), which is not present in lipid rafts (recovery after saturation being ~19% and ~50%, respectively). Cholesterol depletion and treatment with latranaculin A (an actin-disrupting agent) led to comparatively higher mobile fractions of CD44 (28% and 40% recovery respectively compared to ~19% for untreated CD44). Oliferenko et al. took these results as evidence of the presence of CD44 in lipid rafts, but the alternative possibility of CD44 being retained by pickets and fences better than TfR (perhaps through selective interaction) could not be ruled out. Cholesterol depletion and actin disruption could merely perturb pickets and fences by acting on the cytoskeleton [55]. Shvartsman et al. investigated the diffusion of influenza HA tagged with various anchors [56], and found that DRM-associated wild-type and GPI-anchored forms of HA diffused more slowly than an HA mutant that was not associated with the DRMs. However, the diffusion became comparable following the depletion of cholesterol from the cells. In cells that coexpressed the wild-type and GPI-anchored forms of HA, the patching of one form by using antibodies slowed down the diffusion of the other, thus indicating the anchor- (and hence raft-) dependent interaction between the two forms.

Using FRAP, Henis and coworkers [57,58] examined the lateral diffusion constants of wild-type and activated H-Ras and K-Ras (lipid-linked small GTPases which associate with the inner leaflet of the membrane). Whilst wild-type H-Ras was found to a greater extent in DRMs, the other isoforms (wild-type K-Ras, activated H-Ras and activated K-ras) were not at all detergent-insoluble, suggesting a functional potential of the "lipid raft" association [59]. Interestingly, these authors did not report any significant difference in the lateral mobility of any of the proteins. Only wild-type H-Ras showed an increased lateral mobility upon cholesterol depletion. Moreover, the lateral mobility of activated H-Ras and K-Ras increased with expression level in a saturable manner, whereas the lateral mobility of wild-type H-Ras was independent of its expression level. These complicated results again highlighted the lack of a precise understanding of detergent insolubility. A more comprehensive understanding of the nature of the domains that these small-molecule GTPases occupy was reviewed recently [13].

3.6.3

Fluorescence Correlation Spectroscopy

Fluorescence correlation spectroscopy (FCS) provides information on the diffusion of fluorescently tagged molecules by observing the correlation of fluorophore fluctuation through a small optically delimited detection volume [60–65]. The resultant fluorescence fluctuations provide an autocorrelation curve, which could be used to calculate diffusion constant of the molecules. Whilst FCS retains the statistical advantages of FRAP, it can be performed with dilute fluorophores and with relatively less laser power. This makes it more suitable for live cell studies, with lesser photo-damage and lower fluorophore concentrations, closer to that of endogenous molecules [60,63,66].

Korlach et al. used giant unilamellar vesicles (GUVs) containing various mixtures of dilauroyl phosphatidylcholine (DLPC), dipalmitoyl phosphatidylcholine (DPPC) and cholesterol. These authors imaged coexisting phases (resulting from phase separation) with confocal fluorescence microscopy using differential probe partitioning of fluorescent probes 1,1'-dieicosanyl-3,3,3',3'-tetramethylindocarbocyanine perchlorate (DiI-C20) and 2-(4,4-difluoro-5,7-dimethyl-4-bora-3a,4a-diazas-indacene-3-pentanoyl)-1-hexadecanoyl-*sn*-glycero-3-phosphocholine (Bodipy-PC). The identified phases were characterized by measuring translational diffusion of DiI-C20 by FCS measurements. The probe displayed fast mobility in fluid membrane phases, and slower mobility in ordered membrane phases. Cholesterol was found to induce changes in coexisting phase domains. In binary mixtures of DLPC/cholesterol, the fluid phase of DLPC that contained a higher cholesterol content displayed slower diffusion coefficients for DiI-C20. However, by confocal fluorescence microscopy these phases appears identical.

In a similar study, Kahya et al. studied phases in GUVs prepared from ternary mixtures of 1,2-dioleoyl-*sn*-glycero-3-phosphocholine (DOPC), SM and cholesterol [67]. DiI-C18 was largely excluded from the SM-rich regions, in which the raft marker ganglioside GM1 was localized when visualized with CtxB subunit. The cholesterol content was found to be critical for the phase separation into a liquid-disordered, DOPC-enriched phase exhibiting high probe mobility and a dense, liquid-ordered, SM-enriched phase. The addition of cholesterol led to increased probe mobility for the liquid-disordered, DOPC/cholesterol mixture and decreased probe mobility for the liquid-ordered, SM/cholesterol mixture. Bacia et al. per-

58 3 The Biophysical Characterization of Lipid Rafts

formed FCS on a raft marker GM1 probed with fluorescently labeled CtxB subunit and compared it with a non-raft marker dialkylcarbocyanine (Dil) [68]. In homogeneous GUVs, both probes displayed slightly different diffusion (attributed to factors other than membrane composition). Both probes also showed significantly different diffusion in GUVs that contained a raft lipid mixture (unsaturated phosphatidylcholine, cholesterol and SM). CtxB-GM1 diffused significantly more slowly than DiI, consistent with its presence in liquid-ordered domains. The depletion of cholesterol by methyl-β-cyclodextrin (mbCD) resulted in an increased mobility of CtxB-GM1, consistent with the disruption of liquid-ordered domains. Similarly, CtxB-GM1 displayed extremely slow diffusion compared to DiI in rat basophilic leukemia (RBL) cells. However, there was no increase in the mobility of CtxB-GM1 by depletion of cholesterol using m β CD. In contrast, disruption of the cytoskeleton by treatment with latrunculin A resulted in a higher mobility of CtxB-GM1. The authors speculated that these results could arise if some skeleton rafts remained associated with the cytoskeleton after cholesterol depletion, but they did not conduct an experiment in which cholesterol was depleted along with cytoskeleton disruption. A further increase in mobility could have suggested a hierarchical organization of CtxB-GM1 into cholesterol- and cytoskeleton-dependent structures. Bacia et al. also showed that the SNARE proteins syntaxin and synaptobrevin, when reconstituted into GUVs, were preferentially present in liquid-ordered phases. Interestingly, Lang et al. showed that syntaxin does not co-patch with typical raft markers such as GPI-linked proteins, and does not co-fractionate with DRMs [69], whereas others showed that SNAREs are highly enriched in DRMs [70,71]. Although these results related to DRM association are ambiguous, they demonstrate the gap between data obtained from model membranes and from biological membranes. In an interesting follow-up study, Bacia et al. examined the role of sterol structure in phase separation in GUVs [72]. Sterol structure was shown not only to influence phase separation but also to cause remarkable differences in the curvatures of GUVs. Whereas both cholesterol and lophenol induced positive curvature and outward budding of liquid-ordered phases, lanosterol and cholesteryl sulfonate treatment resulted in a negative curvature and inward budding of liquid-ordered phases.

One of the major criticisms of diffusion-based studies is lack of understanding of diffusion process in membranes with complex lipid mixtures, though a recent study conducted by Hac et al. showed some progress in this direction [73]. These authors studied diffusion in two-component binary lipid membranes as a function of composition (fraction of two lipids) and temperature. They performed Monte Carlo simulations using the thermodynamic properties of lipid mixtures (measured by calorimetry) to predict FCS autocorrelation profiles resulting from diffusion in the lipid mixtures. The predicted FCS data agreed very well with data obtained experimentally. Although biological membranes are far more complex than a two-component system, these studies at least provided some ground rules for an understanding of diffusion in biological membranes.

3.7 Proximity Measurements

Proximity measurements between two or more fluorophore-tagged molecules using the well-understood characteristics of fluorescence – namely energy transfer (Foerster's resonance energy transfer; FRET) – is another popular method of determining whether molecules are brought into nanometer proximity of each other.



Fig. 3.5 Schematic depiction of the fluorescence resonance energy transfer process and its implementation. (A) Orientation of donor and acceptor transition dipoles. The relative angle between the two transition dipole is responsible for depolarization of fluorescence upon energy transfer. (B) Overlap integral $J(\lambda)$ between the donor emission (E_D) and acceptor absorption spectra (A_A). A_D and E_A are the donor absorption and acceptor emission spectra, respectively. Arrows depict decrease in donor emission and increase in acceptor emission intensities upon energy transfer. Observation windows show excitation and emission wavelength bandwidths for a typical imaging experiment, indicating the potential for cross-talk between the different imaging channels. D = donor; A = acceptor; exc = excitation; em = emission. (Reprinted with permission from [86].)

60 3 The Biophysical Characterization of Lipid Rafts

FRET may be theoretically described as an interaction between a dipole (excited state of a fluorophore) and an induced dipole (nearby ground-state fluorophore). This interaction decays extremely sharply with distance (1/R6), which provides this technique with the characteristics of a spectroscopic ruler in a typical range of 1 to 10 nm for organic fluorophores [74]. Since there is a transfer of energy between an excited-state fluorophore (donor) and a ground-state fluorophore (acceptor), the detection of this phenomenon may be related to changes in donor characteristics (donor lifetime, bleaching rate, or net fluorescence emission intensity or anisotropy) or acceptor characteristics (emission intensity or anisotropy) (for a review, see [75]). In the detection of "rafts" - which are defined by this technique as an anomalous proximity between fluorophore-tagged molecules – two types of FRET experiments have been carried out, with differing results. In hetero-FRET, the donor and acceptor are of different molecular and spectral characteristics, whereas in homo-FRET the donor and acceptor are the same species. In all cases the extent of energy transfer was dependent upon spectral overlap between the donor and acceptor species, the distance between dipoles, and the orientation of dipoles with respect to each other (Fig. 3.5).

3.7.1

Proximity Measurement Using Homo-FRET

Homo-FRET measures the loss of fluorescence anisotropy, resulting from the FRET process. It is a relatively less well-known approach to FRET, and is also referred to as depolarization-FRET [74,75]. In this type of FRET, fluorescence emission excited by polarized light is measured for the extent of loss of fluorescence anisotropy using appropriately placed excitation and emission polarizers (Fig. 3.6). Due to a finite spread of allowed angular dependence between the donor and acceptor fluorophores, there is a significant transfer of energy between those donor and acceptor fluorophores that are *not* aligned to each other, and this results in an instantaneous loss of emission anisotropy. This loss can be measured in the steady state as a change in anisotropy of fluorescence emission in the presence or absence of nearby fluorophores, or alternatively in the time domain (during the lifetime of the fluorophore) as a rapid decay of emission anisotropy.

Using this technique in the steady-state measurement, Varma and Mayor [76] demonstrated the presence of lipid-dependent protein clusters on the surface of living cells. Since the extent of depolarization (or loss of anisotropy) resulting from FRET is dependent on distance (limited by R₀; [74]), increasing depolarization would be seen with increasing concentration. Using this assay, Varma and Mayor showed that fluorophore-tagged Folate Receptor (FR-GPI; a model GPI-AP) is organized in clusters maintained by cholesterol levels in the membrane, whereas the same FR ecto-domain – when linked to two different transmembrane domains – showed random organization. Cholesterol depletion led to a disruption of the clusters formed by FR-GPI. Later, Sharma and coworkers showed that GFP anchored to cell membranes via a GPI-anchor (GFP-GPI) and mYFP-GPI (mYFP is monomeric version of YFP; [77]) were also organized in cholesterol-dependent sub-reso-



Fig. 3.6 Imaging set-up used to measure steady-state anisotropy. Parallel (I_I) and perpendicular (I_{\perp}) fluorescence intensity images are obtained using a set of excitation and emission polarizers. The perpendicular

and parallel intensity images thus obtained are processed mathematically using software to obtain anisotropy and total intensity images.

lution clusters, showing that mere GPI-anchoring could result into this sub-resolution clustering [78].

Although steady-state anisotropy measurement is a valuable tool for the study of sub-resolution nanometer-scale homotypic interactions of molecules in living cells, time-resolved data provide additional information that is not available from steady-state measurements. Steady-state anisotropy reports a time-averaged picture of the fluorescence anisotropy displayed by the molecules in a population. It is difficult to interpret the multiple states/possibilities that could result in this average value for the population. It is worthwhile considering the possibility of the existence of dimers (for simplicity) in a population. Steady-state anisotropy measurements would be capable of reporting homo-FRET between dimers, but they would not allow any distinction to be made between the multiple possibilities that could result in particular homo-FRET efficiency. In a situation where a homo-FRET efficiency of 50% is recorded, this could result either from 50% FRET efficiencies between all molecules, or 100% FRET efficiencies between half of the molecules. Steady-state anisotropy measurements would not be able to distinguish between the two possibilities. Time-resolved anisotropy measurements would provide the distinct decay profile for each of the above distributions, and also provide an estimate of the fraction of molecules in the clustered organization. Moreover,

62 3 The Biophysical Characterization of Lipid Rafts

anisotropy decay rates associated with FRET can also be used to estimate the interfluorophore distances. In addition to FRET distances, time-resolved anisotropy is also capable of resolving multiple sources of depolarization arising from segmental motion or the rotation of a molecule. Hence, time-resolved anisotropy measurements provide additional information related to the fraction of molecules in cluster and intermolecular distances between molecules undergoing FRET.

Sharma and coworkers performed homo-FRET detection with time-resolved anisotropy measurements to obtain further information on the structure of GPI-AP organization [78]. Time-resolved anisotropy measurements on GFP-GPI and mYFP-GPI expressed in living cells showed that they were present in extremely high-density clusters, with $\sim 20-40\%$ of molecules present in the clusters and the remainder as monomers. The estimated structure of GFP-GPI clusters (with ~30% of molecules in clusters) was verified with results obtained with FR-GPI using the method of anisotropy photobleaching (for a description, see [78]). However, neither method could provide an estimate of cluster size. Sharma et al. also developed a novel tool to estimate cluster size at nanometer scale that required the measurement of homo- as well as hetero-FRET efficiencies. GPI-APs display significant homo-FRET, but hetero-FRET was not detected between GPI-APs using multiple methods of detection. In order to explain this discrepancy, Sharma et al. performed theoretical modeling of hetero-FRET efficiencies. Initially, they calculated the efficiency of hetero-FRET for variable donor: acceptor ratios, cluster sizes and fractions of molecules in clusters. The calculations showed that small clusters with a lesser fraction of molecules accounted for the lack of hetero-FRET. As expected from theoretical calculations, the formation of heptamers of GPI-APs by aerolysin toxin [79] resulted in hetero-FRET detection. This methodology provides an upper bound on the cluster size (less than four molecules). The challenge now lies in using other techniques to detect these clusters. Using this information, FCS is a likely candidate if a suitable model for the diffusion of different species can be incorporated into theoretical analyses of the correlation function.

3.7.2

Proximity Measurement Using Hetero-FRET

Another FRET technique which has been used extensively to detect proximities between fluorophores is that of hetero-FRET [80]. Several hetero-FRET methods exist (see Fig. 3.5), and most have been applied to study GPI-anchored proteins (for a review, see [75]). In the early studies, Kenworthy and Edidin [80], when examining FRET between donor and acceptor labeled antibodies against 5'-nucleotidase (5'-NT; a GPI-AP), reported that most proteins on the surface of fixed cells were monomers. These authors overexpressed the large quantities of proteins in MDCK cells and found that FRET, as reported by the acceptor photobleaching method (measuring donor dequenching after photobleaching of acceptor), was dependent on the density of the labeled acceptors. A comparison of data with theoretical predictions for FRET in membranes ruled out any significant clustering between 5'-NT. In order to address the discrepancy between their results and those of

Varma and Mayor, Kenworthy et al. investigated FRET between FR-GPI in unfixed living cells [81] and failed to detect any significant clustering. These authors also examined FRET between either glycosphingolipid GM1 (as probed by the binding of fluorophore-labeled CtxB) or GM1 and GPI-AP. Although FRET was detected in each case (indicating the presence of these molecules in proximity to each other), the data ruled out any significant clustering as predicted by theoretical modeling. It is interesting to note however, that Kenworthy et al. also failed to detect any significant dimerization of CD59 as reported by Hatanaka et al. using cross-linking and biochemical (PAGE) detection [82]. However, all of these results showing a lack of hetero-FRET detection were consistent with the predictions of these studies provide a complex picture of GPI-anchored protein organization, and also generate new tools and concepts for an understanding of the organization of "raft" components in membranes.

3.8 Conclusions

Although it is easy to lose one's way when interpreting the many biophysical approaches of analyzing the existence of lipid rafts, few conclusions could be drawn without significant dissent. Model membrane studies are vital in understanding lipid-lipid and lipid-protein interaction, but there is still a long way to go in understanding multicomponent complex biological membranes. This could be analogous to the "three-body problem" where the gravitational interaction of three masses is surprisingly difficult to solve. In the light of information that small transition energies could bring about remarkable structural changes, there may be serious problems in extending model membrane studies to biological membranes. This includes all model membrane studies which support the concept of "detergent-insoluble" complexes to describe lipid rafts. Although diffusion-based studies provide valuable live cell data, they have several drawbacks. It is not easy to interpret results from diffusion data due to a poor understanding of: (1) the diffusion of molecules in complex biological membranes; (2) the ultrastructure of biological membranes; and (3) interaction of the cytoskeleton with biological membranes. Diffusion data might provide a valuable time-kinetics, though it is difficult to deduce structural information from diffusion data with current theoretical understanding. Although homo-FRET-based studies provide an excellent source of structural information on proximity relationships derived by lipidic interactions, the relatively longer time scales of measurement mask the temporal information. In the absence of any ideal "sub-resolution ultra-fast imaging" technique, it is desirable to study chosen model raft components in a model cellular system with available biophysical techniques, and appropriate functional consequences.

64 References

Abbreviations

AP	anchored protein
CtxB	cholera toxin B
DIG	detergent-insoluble glycolipid
DiI	dialkylcarbocyanine
DLPC	dilauroyl phosphatidylcholine
DPPC	dipalmitoyl phosphatidylcholine
DRM	detergent-resistant membrane
EM	electron microscopy
ER	endoplasmic reticulum
FCS	fluorescence correlation spectroscopy
FL-DOPE	1,2-dio-leoyl-sn-glycero-3-phosphoethanolamine-fluorescein
FL-DPPE	1,2-dipalmitoyl-sn-glycero-3-phosphoethanolamine-fluorescein
fl-PE	fluorescein phosphatidylethanolamine
FRAP	fluorescence recovery after photobleaching
FRET	Foerster's resonance energy transfer
GPI	glycosyl-phosphatidylinositol
GUV	giant unilamellar vesicle
HA	hemagglutinin
HCASM	human coronary artery smooth muscle
LPH	lactase-phlorizin hydrolase
mβCD	methyl-β-cyclodextrin
MDCK	Madin-Darby canine kidney
NCAM	neural cell adhesion molecule
NMR	nuclear magnetic resonance
5'-NT	5'-nucleotidase
PAGE	polyacrylamide gel electrophoresis
PC	phosphocholine
PLAP	placental alkaline phosphatase
RBL	rat basophilic leukemia
SM	sphingomyelin
SPT	single-particle tracking
SSR	surface scanning resistance
TCZ	transient confinement zone
TfR	transferrin receptor
TGN	trans-Golgi network
YFP	yellow fluorescent protein

References

1 Singer, S. J. and G. L. Nicolson. The fluid mosaic model of the structure of cell membranes. *Science* 1972; 175(23): 720–731.
2 Jain, M. K. and H. B. White, III. Long-range order in biomembranes. *Adv. Lipid Res.* 1977; 15: 1–60.

- 3 Klausner, R. D., et al. Lipid domains in membranes. Evidence derived from structural perturbations induced by free fatty acids and lifetime heterogeneity analysis. J. Biol. Chem. 1980; 255(4): 1286-1295.
- 4 Simons, K. and G. van Meer. Lipid sorting in epithelial cells. Biochemistry 1988; 27(17): 6197-6202.
- 5 Simons, K. and E. Ikonen. Functional rafts in cell membranes. Nature 1997; 387(6633): 569-572.
- 6 Simons, K. and D. Toomre. Lipid rafts and signal transduction. Nat. Rev. Mol. Cell. Biol. 2000; 1(1): 31-39.
- 7 Chatterjee, S. and S. Mayor. The GPI-anchor and protein sorting. Cell. Mol. Life Sci. 2001; 58(14): 1969-1987.
- 8 Udenfriend, S. and K. Kodukula. How glycosylphosphatidylinositol-anchored membrane proteins are made. Annu. Rev. Biochem. 1995; 64: 563-591.
- 9 Mayor, S. and H. Riezman. Sorting GPIanchored proteins. Nat. Rev. Mol. Cell. Biol. 2004: 5(2): 110-120.
- 10 Robinson, P.J. Signal transduction via GPI-anchored membrane proteins. Adv. Exp. Med. Biol. 1997; 419: 365-370.
- 11 Tsui-Pierchala, B.A., et al. Lipid rafts in neuronal signaling and function. Trends Neurosci. 2002; 25(8): 412-417.
- 12 Horejsi, V., et al. GPI-microdomains: a role in signalling via immunoreceptors. Immunol. Today 1999; 20(8): 356-361.
- 13 Parton, R.G. and J.F. Hancock. Lipid rafts and plasma membrane microorganization: insights from Ras. Trends Cell Biol. 2004; 14(3): 141-147.
- 14 Gonen, A., P. Weisman-Shomer, and M. Fry. Cell adhesion and acquisition of detergent resistance by the cytoskeleton of cultured chick fibroblasts. Biochim. Biophys. Acta 1979; 552(2): 307-321.
- 15 Streuli, C. H., B. Patel, and D. R. Critchley. The cholera toxin receptor ganglioside GM remains associated with Triton X-100 cytoskeletons of BALB/c-3T3 cells. Exp. Cell Res. 1981; 136(2): 247-254.
- 16 Helenius, A. and K. Simons. Solubilization 30 Mayor, S. and M. Rao. Rafts: scale-dependof membranes by detergents. Biochim. Biophys. Acta 1975; 415(1): 29-79.
- 17 Simons, K., A. Helenius, and H. Garoff. Solubilization of the membrane proteins from Semliki Forest virus with Triton X-100. J. Mol. Biol. 1973; 80(1): 119-133.

- 18 Becker, R., A. Helenius, and K. Simons. Solubilization of the Semliki Forest virus membrane with sodium dodecyl sulfate. Biochemistry 1975; 14(9): 1835-1841.
- 19 Helenius, A., et al. Solubilization of the Semliki Forest virus membrane with sodium deoxycholate. Biochim. Biophys. Acta 1976; 436(2): 319-334.
- 20 Brown, D.A. and J.K. Rose. Sorting of GPI-anchored proteins to glycolipid-enriched membrane subdomains during transport to the apical cell surface. Cell 1992: 68(3): 533-544.
- 21 Brown, D.A. Interactions between GPI-anchored proteins and membrane lipids. Trends Cell Biol. 1992; 2(11): 338-343.
- 22 Yeagle, P. The Membranes of Cells. 2nd edn. San Diego, CA: Academic Press, 1993.
- 23 Lamaze, C., et al. Interleukin 2 receptors and detergent-resistant membrane domains define a clathrin-independent endocytic pathway. Mol. Cell 2001; 7(3): 661-671.
- 24 Brown, D.A. and E. London. Functions of lipid rafts in biological membranes. Annu. Rev. Cell Dev. Biol. 1998; 14: 111-136.
- 25 London, E. and D.A. Brown. Insolubility of lipids in Triton X-100: physical origin and relationship to sphingolipid/cholesterol membrane domains (rafts). Biochim. Biophys. Acta 2000; 1508(1-2): 182-195.
- 26 Shaw, A.R. and L. Li. Exploration of the functional proteome: lessons from lipid rafts. Curr. Opin. Mol. Ther. 2003; 5(3): 294-301.
- 27 Foster, L.J., C.L. De Hoog, and M. Mann. Unbiased quantitative proteomics of lipid rafts reveals high specificity for signaling factors. Proc. Natl. Acad. Sci. USA 2003; 100(10): 5813-5818.
- 28 Heerklotz, H. Triton promotes domain formation in lipid raft mixtures. Biophys. J. 2002; 83(5): 2693-2701.
- 29 Heerklotz, H., et al. The sensitivity of lipid domains to small perturbations demonstrated by the effect of Triton. J. Mol. Biol. 2003; 329(4): 793-799.
- ent, active lipid organization at the cell surface. Traffic 2004; 5(4): 231-240.
- 31 Mayor, S., K.G. Rothberg, and F.R. Maxfield. Sequestration of GPI-anchored proteins in caveolae triggered by cross-linking. Science 1994; 264(5167): 1948-1951.

- 66 References
 - 32 Mayor, S. and F. R. Maxfield. Insolubility and redistribution of GPI-anchored proteins at the cell surface after detergent treatment. *Mol. Biol. Cell* 1995; 6(7): 929–944.
 - 33 Parton, R.G., B. Joggerst, and K. Simons. Regulated internalization of caveolae. J. Cell Biol. 1994; 127(5): 1199–1215.
 - 34 Maunsbach, A. Immunolabeling and staining of ultrathin sections in biological electron microscopy. In: Celis, J. (Ed.). Cell Biology: A Laboratory Handbook. Academic Press: San Diego, CA, 1998.
 - 35 Sheets, E. D., R. Simson, and K. Jacobson. New insights into membrane dynamics from the analysis of cell surface interactions by physical methods. *Curr. Opin. Cell Biol.* 1995; 7(5): 707–714.
 - 36 Sheets, E. D., et al. Transient confinement of a glycosylphosphatidylinositol-anchored protein in the plasma membrane. *Biochemistry* 1997; 36(41): 12449–12458.
 - 37 Dietrich, C., et al. Relationship of lipid rafts to transient confinement zones detected by single particle tracking. *Biophys. J.* 2002; 82(1 Pt 1): 274–284.
 - 38 Simson, R., et al. Structural mosaicism on the submicron scale in the plasma membrane. *Biophys. J.* 1998; 74(1): 297–308.
 - 39 Schutz, G. J., et al. Properties of lipid microdomains in a muscle cell membrane visualized by single molecule microscopy. *EMBO J.* 2000; 19(5): 892–901.
 - **40** Pralle, A., et al. Sphingolipid-cholesterol rafts diffuse as small entities in the plasma membrane of mammalian cells. *J. Cell Biol.* 2000; 148(5): 997–1008.
 - 41 Saffman, P.G. and M. Delbruck. Brownian motion in biological membranes. *Proc. Natl. Acad. Sci. USA* 1975; 72(8): 3111–3113.
 - **42** Kusumi, A., I. Koyama-Honda, and K. Suzuki. Molecular dynamics and interactions for creation of stimulation-induced stabilized rafts from small unstable steady-state rafts. *Traffic* 2004; 5(4): 213–230.
 - 43 Kusumi, A., et al. Paradigm shift of the plasma membrane concept from the twodimensional continuum fluid to the partitioned fluid: high-speed single-molecules tracking of membrane molecules. *Annu. Rev. Biophys. Biomol. Struct.* 2005; 34: 351–378.

- 44 Suzuki, K., R.E. Sterba, and M.P. Sheetz. Outer membrane monolayer domains from two-dimensional surface scanning resistance measurements. *Biophys. J.* 2000; 79(1): 448–459.
- **45** Suzuki, K. and M.P. Sheetz. Binding of cross-linked glycosylphosphatidylinositol-anchored proteins to discrete actin-associated sites and cholesterol-dependent domains. *Biophys. J.* 2001; 81(4): 2181–2189.
- 46 Ritchie, K. and A. Kusumi. Single-particle tracking image microscopy. *Methods Enzymol.* 2003; 360: 618–634.
- 47 Edidin, M. In: Damjanovich, S., Edidin, M., Szollosi, J., Tron, L. (Eds.). Mobility and Proximity in Biological Membranes. CRC: Boca Raton, Florida, 1994: 109–135.
- 48 Reits, E.A. and J.J. Neefjes. From fixed to FRAP: measuring protein mobility and activity in living cells. *Nat. Cell. Biol.* 2001; 3(6): E145-E147.
- 49 Houtsmuller, A. B. and W. Vermeulen. Macromolecular dynamics in living cell nuclei revealed by fluorescence redistribution after photobleaching. *Histochem. Cell. Biol.* 2001; 115(1): 13–21.
- 50 Lippincott-Schwartz, J. and G. H. Patterson. Development and use of fluorescent protein markers in living cells. *Science* 2003; 300(5616): 87–91.
- 51 Lippincott-Schwartz, J., N. Altan-Bonnet, and G. H. Patterson. Photobleaching and photoactivation: following protein dynamics in living cells. *Nat. Cell. Biol.* 2003; Suppl: S7-S14.
- 52 Ishihara, A., Y. Hou, and K. Jacobson. The Thy-1 antigen exhibits rapid lateral diffusion in the plasma membrane of rodent lymphoid cells and fibroblasts. *Proc. Natl. Acad. Sci. USA* 1987; 84(5): 1290–1293.
- 53 Hannan, L.A., et al. Correctly sorted molecules of a GPI-anchored protein are clustered and immobile when they arrive at the apical surface of MDCK cells. *J. Cell Biol.* 1993; 120(2): 353–358.
- 54 Oliferenko, S., et al. Analysis of CD44-containing lipid rafts: recruitment of annexin II and stabilization by the actin cytoskeleton. *J. Cell Biol.* 1999; 146(4): 843–854.
- 55 Edidin, M. Patches, posts and fences: proteins and plasma membrane domains. *Trends Cell Biol.* 1992; 2(12): 376–380.
- 56 Shvartsman, D. E., et al. Differently anchored influenza hemagglutinin mutants

display distinct interaction dynamics with mutual rafts. J. Cell Biol. 2003; 163(4): 879-888.

- 57 Niv, H., et al. Activated K-Ras and H-Ras display different interactions with saturable nonraft sites at the surface of live cells. J. Cell Biol. 2002; 157(5): 865-872.
- 58 Niv, H., et al. Membrane interactions of a constitutively active GFP-Ki-Ras 4B and their role in signaling. Evidence from lateral mobility studies. J. Biol. Chem. 1999; 274(3): 1606-1613.
- 59 Prior, I.A., et al. GTP-dependent segregation of H-ras from lipid rafts is required for biological activity. Nat. Cell. Biol. 2001; 3(4): 368-375.
- 60 Bulseco, D.A. and D.E. Wolf. Fluorescence 72 Bacia, K., P. Schwille, and T. Kurzchalia. correlation spectroscopy: molecular complexing in solution and in living cells. Methods Cell Biol. 2003; 72: 465-498.
- 61 Berland, K.M. Fluorescence correlation spectroscopy: a new tool for quantification of molecular interactions. Methods Mol. Biol. 2004; 261: 383-398.
- 62 Haupts, U., et al. Dynamics of fluorescence fluctuations in green fluorescent protein observed by fluorescence correlation spectroscopy. Proc. Natl. Acad. Sci. USA 1998; 95(23): 13573-13578.
- 63 Koppel, D.E., et al. Dynamics of fluorescence marker concentration as a probe of mobility. Biophys. J. 1976; 16(11): 1315-1329.
- 64 Schwille, P., J. Korlach, and W.W. Webb. Fluorescence correlation spectroscopy with 76 Varma, R. and S. Mayor. GPI-anchored single-molecule sensitivity on cell and model membranes. Cytometry 1999; 36(3): 176-182.
- 65 Rigler, R. Fluorescence correlations, single molecule detection and large number screening. Applications in biotechnology. J. Biotechnol. 1995; 41(2-3): 177-186.
- 66 Korlach, J., et al. Characterization of lipid bilayer phases by confocal microscopy and fluorescence correlation spectroscopy. Proc. Natl. Acad. Sci. USA 1999; 96(15): 8461-8466.
- 67 Kahya, N., et al. Probing lipid mobility of raft-exhibiting model membranes by fluorescence correlation spectroscopy. J. Biol. Chem. 2003; 278(30): 28109-28115.
- 68 Bacia, K., et al. Fluorescence correlation spectroscopy relates rafts in model and na-

tive membranes. Biophys. J. 2004; 87(2): 1034-1043.

- 69 Lang, T., et al. SNAREs are concentrated in cholesterol-dependent clusters that define docking and fusion sites for exocytosis. EMBO I. 2001; 20(9); 2202-2213.
- 70 Chamberlain, L.H., R.D. Burgoyne, and G.W. Gould. SNARE proteins are highly enriched in lipid rafts in PC12 cells: implications for the spatial control of exocytosis. Proc. Natl. Acad. Sci. USA 2001; 98(10): 5619-5624.
- 71 Xia, F., et al. Disruption of pancreatic betacell lipid rafts modifies Kv2.1 channel gating and insulin exocytosis. J. Biol. Chem. 2004; 279(23): 24685-24691.
- Sterol structure determines the separation of phases and the curvature of the liquidordered phase in model membranes. Proc. Natl. Acad. Sci. USA 2005; 102(9): 3272-3277.
- 73 Hac, A.E., et al. Diffusion in two-component lipid membranes – a fluorescence correlation spectroscopy and Monte Carlo simulation study. Biophys. J. 2005; 88(1): 317-333.
- 74 Lakowicz, J. R. Principles of Fluorescence Spectroscopy. 2nd edn. Kluwer Academic/ Plenum Publishers, 1999.
- 75 Krishnan, R.V., R. Varma, and S. Mayor. Fluorescence methods to probe nanometer-scale organization of molecules in living cell membranes. J. Fluorescence 2001; 11(3): 211-226.
- proteins are organized in submicron domains at the cell surface. Nature 1998; 394(6695): 798-801.
- 77 Zacharias, D.A., et al. Partitioning of lipidmodified monomeric GFPs into membrane microdomains of live cells. Science 2002; 296(5569): 913-916.
- 78 Sharma, P., et al. Nanoscale organization of multiple GPI-anchored proteins in living cell membranes. Cell 2004; 116(4): 577-589.
- 79 Fivaz, M., et al. Differential sorting and fate of endocytosed GPI-anchored proteins. EMBO J. 2002; 21(15): 3989-4000.
- 80 Kenworthy, A.K. and M. Edidin. Distribution of a glycosylphosphatidylinositol-anchored protein at the apical surface of

MDCK cells examined at a resolution of <100 Å using imaging fluorescence resonance energy transfer. *J. Cell Biol.* 1998; 142(1): 69–84.

- 81 Kenworthy, A. K., N. Petranova, and M. Edidin. High-resolution FRET microscopy of cholera toxin B-subunit and GPI- anchored proteins in cell plasma membranes. *Mol. Biol. Cell* 2000; 11(5): 1645–1655.
- 82 Hatanaka, M., et al. Cellular distribution of a GPI-anchored complement regulatory protein CD59: homodimerization on the surface of HeLa and CD59-transfected CHO cells. J. Biochem. (Tokyo) 1998; 123(4): 579–586.
- 83 Anderson, R. G. and K. Jacobson. A role for lipid shells in targeting proteins to caveolae, rafts, and other lipid domains. *Science* 2002; 296(5574): 1821–1825.
- 84 Maxfield, F. R. Plasma membrane microdomains. Curr. Opin. Cell Biol. 2002; 14(4): 483–487.
- 85 Subczynski, W. K. and A. Kusumi. Dynamics of raft molecules in the cell and artificial membranes: approaches by pulse EPR spin labeling and single molecule optical microscopy. *Biochim. Biophys. Acta* 2003; 1610(2): 231–243.
- 86 Jacobson, K., and Dietrich. Looking at Lipid rafts? *Trends Cell Biol.* 1999; 9 (3): 87–91.

The Role of Caveolae and Noncaveolar Rafts in Endocytosis

Bo van Deurs, Frederik Vilhardt, Maria Torgersen, Kirstine Roepstorff, Anette M. Hommelgaard, and Kirsten Sandvig

4.1 Introduction

4

Endocytosis is involved in multiple cellular processes, including the uptake of transport proteins, uptake of "opportunistic ligands" such as bacterial and plant toxins and viruses, attenuation of signaling activity, receptor recycling and resensitization, down-regulation of cell contacts during epithelial-to-mesenchymal transition, antigen uptake and processing in MHC class II-expressing cells, transcytosis and delivery of antibodies to the newborn baby, and synaptic function. About 15-20 years ago it was frequently - and persistently - claimed that only one endocytic mechanism existed, namely the one mediated by clathrin-coated pits and vesicles. However, in studies in which clathrin-dependent uptake was inhibited by potassium depletion and acidification of the cytosol, it became clear that clathrinindependent endocytic mechanisms also must be taken into consideration [1-6]. That different forms of clathrin-independent endocytosis exist is now well-established. Thus, although the clathrin-mediated endocytic machinery is by far the most well-studied type of endocytosis [7–9], a plethora of clathrin-independent endocytic mechanisms also appears in today's literature; however, the underlying molecular machinery is far from being clarified [10-16]. One such clathrin-independent endocytic mechanism can involve caveolae.

Caveolae are small, 50- to 70-nm membrane invaginations that are present on the plasma membrane of many different cell types [17]. Caveolae are often clustered within a short stretch of the plasma membrane, and they can even form large, sometimes branched invaginations deeply into the cytoplasm (Fig. 4.1). *In vivo*, caveolae are particularly abundant in adipocytes, endothelial cells, fibroblasts and smooth muscle cells, but they are also found in, for instance, the basal layer of epithelial cells in stratified epithelia [18]. In contrast, they are scarce or absent in hematopoietic cells and neurons, for example. *In vitro*, the presence of caveolae is more unpredictable because cultured cells do not always reflect the caveolin/caveolae status of the tissue of origin. In addition, the caveolin/caveolae status of the cells of origin is often unknown, not least when cells derive from cancers [17].



Fig. 4.1 Electron micrographs of human myoepithelial cells grown in a chemically defined medium in the absence of insulin. Under these conditions the cells stop proliferating and express a differentiated phenotype including distinct bundles of actin filaments (shown as Ac in panel A) and numerous caveolae. In the presence of insulin, the cells do not show these characteristics but proliferate (see [19]). Note the deeply invaginated caveolae clusters in panels B and C (arrows). Scale bars = 200 nm.

Moreover, cells may express cav-1 without having structurally identifiable caveolae (our unpublished results). Also, the culture conditions may influence whether caveolae are frequent or absent in one and the same cell line [19] (Fig. 4.1).

Caveolae are a subset of lipid rafts which, in addition to an enrichment in cholesterol and sphingolipids, are characterized by the protein caveolin. There are three members of the caveolin family, caveolin-1 and –2, which are (co)expressed in many cell types, and caveolin-3 that is specific for muscle cells. Caveolin-1 and caveolin-2 exist in different isoforms. The caveolins form a characteristic hairpin loop into the membrane lipid bilayer, exposing both the N- and C-termini to the cytoplasm. In the N-terminal region the caveolin scaffolding domain is responsible for the interaction with numerous other proteins. The C-terminal region is attached to the membrane by palmitoyl anchors. Expression of caveolin-1 can be sufficient to generate caveolae, although other factors than caveolin may be involved. For instance, in polarized epithelial cells such as intestinal Caco-2 cells, expression of caveolin-1 results in formation of caveolae only at the basolateral surface [20].

Caveolae seem to represent a kind of multifunctional platform as they have been implicated in such different functions as cholesterol transport, calcium transport and homeostasis, signaling, regulation of endothelial nitric oxide synthase (eNOS) activity, and tumor suppression [17,21–27]. In addition, it has been debated over the years whether caveolae are also involved in endocytosis, although the many other functions of caveolae would contradict such a role. Historically, the reason for believing that caveolae could be involved in endocytosis comes from their characteristic, invaginated shape as seen in the electron microscope and, more recently, because dynamin, a GTP-binding protein involved in the formation of clathrin-coated vesicles, has been found in association with caveolae [28,29] (see Section 4.8). Since caveolae also bind a variety of ligands that become internalized (e.g.,

cholera toxin; see Section 4.5), it is tempting to conclude – by analogy to clathrincoated pits – that they are endocytic, although the question of whether they are actually able to pinch off and deliver material to intracellular compartments has often been ignored. Recent studies, including electron microscopy (EM) and live cell digital imaging, strongly indicate that the answer to the question of whether caveolae are endocytic structures or not, is far from a simple "yes" or "no", but more complex [30]. Therefore, in the present chapter we will treat these aspects of caveolar biology and present a working model that includes the various findings and speculations (see Section 4.10).

4.2

Caveolae are Largely Immobile, Nonendocytic Membrane Domains

In the discussion of whether caveolae are endocytic structures, or not, it is important to distinguish between constitutive endocytosis and stimulated endocytosis. Similarly, frequency and time are also important parameters in relation to endocytic processes. Hence, if a certain endocytic mechanism operates rarely and very slowly it does not add much to the cell's total internalization of membrane, receptors and ligands. Moreover, such a mechanism can be very difficult to document.

It was shown recently that caveolae, in contrast to clathrin-coated pits, are not involved in efficient, constitutive endocytosis in nonstimulated cells [31]. This was in part documented by using photobleaching approaches on different cell lines expressing N- and C-terminally green fluorescent protein (GFP)-tagged caveolin-1 (GFP-tagged caveolin). Caveolin which was found to be incorporated into caveolae at the plasma membrane was highly immobile, whilst some intracellular, caveolin-1-associated structures were dynamic.

The basic idea of using GFP-tagged caveolin fusion proteins and photobleaching techniques such as fluorescence recovery after photobleaching (FRAP) to evaluate whether caveolae are involved in efficient, constitutive endocytosis operating as a parallel internalization mechanism to that of clathrin-coated pits is as follows. Fluorescent caveolin becomes incorporated in caveolae at the plasma membrane, as well as in presumptive caveolar endocytic and recycling vesicles. If caveolae are involved in efficient, constitutive endocytosis, then there will be a rapid turnover of caveolae at the plasma membrane. If a group of fluorescent caveolae at the plasma membrane is bleached, then it will be rapidly replaced by new fluorescent caveolae, a process that is reflected in a rapid recovery of fluorescence intensity in the bleached field of interest. Alternatively, if caveolae are immobile structures at the plasma membrane - that is, there is no internalization/recycling or lateral mobility of caveolin/caveolae taking place - then the bleached field of interest will remain bleached over time. A prerequisite indeed is that the expressed fluorescent caveolin behaves as endogenous caveolin, namely that it becomes associated with caveolae and that it does not change the amount of caveolae at the plasma membrane. This was tested by using quantitative immunogold labeling electron micros-

72 4 The Role of Caveolae and Noncaveolar Rafts in Endocytosis

copy [31]. Moreover, no differences were found between results obtained with N-and C-terminally tagged caveolin-1.

FRAP experiments applying bleaching fields over both the cell periphery/plasma membrane and the interior of HeLa cells, A431 cells, and Madin-Darby canine kidney (MDCK) cells revealed that the fluorescence recovery at the plasma membrane was very low compared to that of the interior of the cells [31]. This means that caveolin-associated membrane inside the cell (for instance, associated with the trans-Golgi network (TGN) or caveosomes; see Section 4.7) is mobile, whereas caveolae at the plasma membrane are highly immobile. Thus, the experiments revealed that the mobile fraction of fluorescent caveolin associated with internal structures was about 80%, which is quite comparable to the value obtained for other membrane proteins moving without constraints [32]. However, the mobile fraction of fluorescent caveolae at the plasma membrane with caveolae at the plasma membrane without constraints [32]. However, the mobile fraction of fluorescent caveolae at the plasma membrane without constraints [32]. However, the mobile fraction of fluorescent caveolae at the plasma membrane without constraints [32]. However, the mobile fraction of fluorescent caveolae at the plasma membrane was as low as about 5–20%, a value comparable to that of E-cadherin after this adhesion molecule has become trapped by the actin cytoskeleton to form immobile structures [33]. Also, the low diffusion coefficient obtained for caveolae-associated fluorescent caveolin was in the same order of magnitude as that of E-cadherin.

An alternative approach to FRAP in the study of the mobility of a fluorescent fusion protein is that of fluorescence loss in photobleaching (FLIP). Here, a certain region of interest is exposed to several bleaching cycles, and the fluorescence intensity in the region of interest as well as of the entire cell is measured. When FLIP is applied to a fluorescent-caveolin-expressing cell it is striking that there is basically no fluorescence recovery in the region after the first bleaching cycle, and the fluorescence intensity of the entire cell falls very slowly [31]. Importantly, in particular the fluorescence signal deriving from caveolae at the plasma membrane (seen as strongly fluorescent dots or elongated structures at the rim of the cells; the resolution of the confocal microscope does not allow distinction to be made between single caveolae and clusters of caveolae) was hardly influenced by the bleaching cycles. This confirms the FRAP data, and further stresses that caveolin/ caveolae are highly immobile. When, for comparison, FLIP was applied to GFP-Rab7-expressing cells, a completely different result was obtained. Thus, within each bleaching cycle large amounts of GFP-Rab7 diffused into the bleaching region, and the fluorescence level of the entire cell fell continuously. In fact, the GFP-Rab7-expressing cells could be completely fluorescence-depleted by FLIP [31]. This is in agreement with Rab7's properties as a small, mobile GTPase that rapidly switches between being membrane-associated and, after GTP hydrolysis, cytosolic [34].

Like noncaveolar rafts, caveolae are cholesterol-based structures, and cholesterol depletion with methyl- β -cyclodextrin (m β CD) leads to the disappearance of caveolae [35–37]. When cholesterol is removed from fluorescent caveolin-expressing cells, FLIP revealed a gradual disappearance of the distinct caveolin fluorescence at the plasma membrane. Thus, caveolin exhibits an increased mobility after cholesterol depletion, possibly due to a loss of caveolae. Also, a much more efficient removal of fluorescence from the bleaching region was obtained compared to control cells [31]. These findings support the notion that cholesterol plays an important role in the maintenance of caveolar integrity.

Moreover, the actin cytoskeleton may help to stabilize caveolae at the plasma membrane [38,39]. Thus, we have shown previously by using a yeast two-hybrid system that the actin-binding protein filamin is a binding partner for caveolin-1. Activation of Rho leads to reorganization of the actin cytoskeleton and subsequently to the reorganization of caveolae [39]. This was further substantiated in FLIP studies of fluorescent caveolin-expressing cells treated with the actin-depolymerizing drug cytochalasin D. Here, the normal fluorescent staining of caveolae at the plasma membrane was perturbed and caveolae began to move laterally and to cluster in the plasma membrane [31].

Taken together, the results of these studies show that plasma membrane-associated caveolin-1 is a highly immobile molecule, and that caveolae are not involved in efficient constitutive endocytosis. Furthermore, the actin cytoskeleton plays an important role in keeping caveolae immobilized. However, it should be emphasized that this immobility of caveolae in the normal steady-state situation is not incompatible with a role of caveolin and caveolae in endocytosis under special conditions where profound changes of the actin cytoskeleton takes place (see Section 4.4). This makes the unraveling of caveolar function even more intriguing.

4.3

Caveolae May Show Local, Short-Range Motility: A Role in Transendothelial Transport?

In the above-mentioned bleaching studies a special FRAP protocol was included in which the entire interior of the cells was bleached, leaving only a narrow, fluorescent rim of GFP-tagged caveolin at the cell periphery [31]. These FRAP experiments highlighted two important points:

- The total fluorescence intensity of the peripheral rim including caveolae at the plasma membrane decreased only very slowly, further supporting the notion that caveolae are not becoming efficiently internalized (see also Section 4.6).
- There was a lively activity of the fluorescent caveolae at the cell periphery, although they did not disappear from the periphery (see Ref. [31], Fig. 5, online movie).

Whether this reflects that caveolae (or groups of caveolae) actually pinch off and refuse with the plasma membrane without really moving away from the subplasmalemmal zone (for example, because they are tethered by the actin cytoskeleton), or it reflects movements in the plasma membrane with its population of caveolae as such, was not clear. However, such continuous recycling of caveolae immediately beneath the plasma membrane was recently demonstrated [40].

A short-range motility and fission-fusion processes may be highly relevant in order to understand the function of caveolae in transendothelial transport, or transcytosis. In endothelial transcytosis [41] caveolae are thought to give rise to free vesicles at one (lumenal or ablumenal) pole and thereafter move to and fuse with the opposite pole, thereby being involved in the exchange of macromolecules be-

74 4 The Role of Caveolae and Noncaveolar Rafts in Endocytosis

tween the blood and the connective tissue. Although it is widely accepted, a major problem with the concept of transendothelial transcytosis in its strict meaning is that truly free, caveolar vesicles have been difficult or impossible to demonstrate ultrastructurally, both in conventionally fixed and rapidly frozen tissue [42-44]. However, considering that the distance between the lumenal and ablumenal membranes in the flattened endothelial cell is often around 200 nm, a short-range motility of the 50-70 nm caveolae or chains of caveolae and their apparent ability to fuse and detach again might explain how caveolae could be involved in transendothelial transport by a "kiss-and-run" activity, even without ever pinching off to form free, endocytic vesicles [17]. Such a process would most likely have to be highly regulated. In fact, the caveolar membrane and the endothelial cell cytoplasm contain the required molecular machinery [28,45,46], and transendothelial transport of albumin has been shown to require interaction of the albumin-docking protein pg60 with caveolin and activation of Gi-coupled Src kinase signaling [47]. Unfortunately, recent studies on caveolin-1-deficient mice have not clarified the role of caveolae in transendothelial transport. Thus, although lacking caveolae, these mice do not show any marked reduction in the transport of albumin [48,49]. This is most likely because caveolae also plays a more indirect role in vascular permeability. Vasorelaxation and permeability is regulated by eNOS, which is interacting with, and negatively regulated by, caveolin associated with endothelial caveolae [50]. In caveolin knockout mice, the microvasculature has even been reported to be hyperpermeable because of "opened" intercellular spaces and tight junctions apparently caused by uncontrolled release of and signaling via nitric oxide [51]. This hyperpermeability blurs any possible reduction in the caveolae-mediated transendothelial transport in the knockout mice.

4.4

An Internalization Wave of Caveolae can be Stimulated by Virus

Studies on the entry of simian 40 (SV40) virus by Pelkmans and coworkers [52–54] have documented that caveolae can actually play a role in nonconstitutive endocytosis. Thus, after binding of SV40 virus to the cell surface via the MHC class I molecule, the virus particles move laterally in the plasma membrane to end up in caveolae. Although these caveolae are initially immobile, the virus initiates a complex signaling cascade leading to a profound disorganization of the cortical actin cytoskeleton and a transient recruitment of the GTP-binding protein dynamin known to be involved in membrane fission (see Section 4.8). Importantly, without SV40-stimulation, less than 10% of the caveolae were associated with dynamin. These changes, in turn, resulted in a wave of incoming caveolar vesicles containing virus where reorganized actin filaments formed "tails" necessary for internalization of the SV40-containing caveolae [53]. Subsequently, the virus was delivered to caveosomes (see Section 4.7) and transported further downstream to the endoplasmic reticulum (ER). However, a delay of several hours then occurs before caveolin returns from the caveosomes to the plasma membrane in vesicles now devoid of virus particles [52].

It is interesting to note that antibody-induced crosslinking of MHC class I molecules (the SV40 receptor) results in an accumulation of MHC class I clusters in "small uncoated surface invaginations" identical to caveolae, as reported 25 years ago by Huet and coworkers. No clusters were found in clathrin-coated pits. From the caveolae-like invaginations the clusters were apparently internalized and delivered to lysosomes [55]. It is therefore tempting to speculate that it is the same underlying mechanism that is responsible for caveolae-mediated uptake of SV40 virus particles after their binding to MHC class I and of MHC class I clusters generated by antibody crosslinking.

Echovirus 1 is also taken up by caveolae, and binds to the collagen receptor $\alpha_2\beta_1$ integrin [56]. Interestingly, recent studies revealed that $\alpha_2\beta_1$ is localized to noncaveolar rafts, but that antibody crosslinking clustered $\alpha_2\beta_1$ in caveolae, which subsequently became internalized and delivered the integrin to caveosome-like structures. This internalization is dependent on protein kinase C α activity [57]. In the same study it was revealed that α V integrin clusters are not taken up by caveolae, but via clathrin-coated pits and vesicles. The common human polyomavirus BK also seems to become internalized by caveolae [58]. It should be stressed, however, that several types of virus are internalized in a clathrin-dependent manner, the classical example being Semliki Forest virus [59], although other recent examples include rubella virus [60] and bovine viral diarrhea virus (a pest virus) [61].

In an ultrastructural study, Parton and co-workers [62] found that the treatment of cells with the phosphatase inhibitor okadaic acid leads to aggregation and subsequent internalization of caveolae, and that the process is dependent on an intact actin cytoskeleton since treatment with cytochalasin D could prevent it. These results suggest that okadaic acid somehow induces a reorganization of actin. The okadaic acid-induced internalization of caveolae was later confirmed by FLIP studies on fluorescent caveolin-expressing cells. Thus, after okadaic acid treatment, plasma membrane caveolin/caveolae gradually disappear from the plasma membrane and aggregate in the perinuclear region, and the mobility of caveolin clearly increased so that the repeated bleaching cycles almost completely extracted the entire cellular fluorescence signal [31].

Finally, a rather special example of a caveolin/caveolae-stimulated internalization should be mentioned. Thus, FimH-expressing *Escherichia coli* are, via binding of the bacteria to CD48 on macrophages and mast cells, apparently able to recruit and cluster caveolae, eventually leading to the formation of intracellular, bacteriacontaining compartments [63,64].

4.5 Role of Caveolae in Endocytosis of Cholera Toxin

In addition to the recent studies on internalization of virus mentioned above, studies on cholera toxin (CT) uptake have also been important for the ongoing discussion of the role of caveolae in endocytosis. CT belongs to a large group of protein toxins with enzymatically active A-moieties, and B-moieties that act as

76 4 The Role of Caveolae and Noncaveolar Rafts in Endocytosis

ligands and bind to cell- surface components. These toxins must be endocytosed before they exert their toxic effect(s). In the case of CT, the B-moieties bind to the ganglioside GM1 and, following internalization and transport via the Golgi complex to the ER, the A-moiety is translocated to the cytosol where it affects the activity of adenylyl cyclase [11].

Although clathrin-mediated and clathrin- and caveolin-independent endocytosis can play a role in the uptake of CT [65–69], this does not in any way exclude a role of caveolae as well, and it is widely believed that caveolae or caveolae-like structures are the major vehicle for CT internalization and the subsequent delivery to the TGN [70-74]. The main reason for this is that CT binds to GM1, which is in part present in caveolae [68,75]. However, the consequences of caveolae-mediated CT/ GM1 internalization for the cell's steady-state balance of membrane-associated caveolin and caveolae have generally been ignored. In principle, two possibilities for CT-stimulated internalization of caveolae exist. Binding of CT to GM1 may induce a wave of caveolae internalization as described above for SV40 and okadaic acid, leading to down-regulation of caveolae at the plasma membrane. Alternatively, CT could induce a constitutive internalization mechanism for caveolae including an efficient recycling of caveolae or caveolin back to the plasma membrane. This would not lead to a depletion of the plasma membrane pool of caveolae, but requires a rapid replacement of those CT-containing caveolae that become internalized with new, empty caveolae.

The first report of a major role for caveolae in CT endocytosis was an ultrastructural study published in 1982 by Montesano et al. [72]. Here, the authors used CT-B chain (CT-B)-gold and found that the conjugate preferentially bound to caveolae at 2 °C. Interestingly, they noticed an increase in CT-B-gold in caveolae relative to CT-B-gold bound to noncaveolar plasma membrane over time at 37 °C. These observations were originally interpreted as the result of an up-concentration of CT in endocytic caveolae. However, an alternative explanation could be selective internalization of CT bound to the noncaveolar plasma membrane. In fact, CT/ GM1 in caveolae could represent an immobilized fraction of the ligand-receptor complex. In the study by Montesano et al., CT-B-gold was only found very rarely, if ever, in clathrin-coated pits. However, other studies have documented small, but significant amounts of CT-B-gold [68], as well as biotin-GM1 detected by antibiotin antibodies conjugated to gold [75] in clathrin-coated pits. CT-B-HRP has also been localized to clathrin-coated pits [66]. Parton [68] was in fact very careful not to draw too far-reaching conclusions based on the preferential localization of CT-B to caveolae, and stressed that at least some GM1 must be internalized by clathrin-coated pits. We would like to emphasize that the finding that only minimal CT-B can be observed at any given time point in clathrin-coated pits does not exclude a significant role for the coated pits in CT-B uptake. Thus, clathrin-coated pits are highly dynamic structures with a half-life of minutes at the cell surface [8], and CT-B may continuously move into newly formed clathrin-coated pits to become efficiently internalized. This is in agreement with the findings that clathrin-dependent endocytosis can account for about 50% of CT internalization [66,76].

Several uptake mechanisms have been described for CT and these may - to some extent - be cell type-dependent. Thus, CT can be internalized both by clathrin-dependent and -independent mechanisms, and the effect of cholesterol-binding drugs (nystatin, filipin, cyclodextrin) on CT-internalization varies [65-67.71.77.78]. In studies with dominant negative mutants of epsin and eps15 (molecules required for clathrin-mediated endocytosis), Nichols et al. [67] found that a proportion of CT-B is internalized by clathrin-coated pits. Moreover, Shogomori and Futerman [65] reported that although CT-B is present in detergent-insoluble rafts or microdomains, it becomes internalized by a raft-independent mechanism, presumably via clathrin-coated pits. The same is true for the epidermal growth factor receptor [79]. Recently, Hansen et al. [69] showed that CT bound to GM1 localized to a detergent-insoluble, raft-like fraction of the enterocyte brush border membrane, and was subsequently internalized by a cholesterol-independent, but clathrin-dependent mechanism. It could be argued that uptake mediated by clathrin-coated pits would not lead to delivery of cargo to the TGN. However, Shiga toxin, which binds to globotriasylceremide (Gb3), is internalized via clathrincoated pits and vesicles by an as-yet unknown mechanism [80] and subsequently transported to the TGN and the ER [81]. In accordance with this, expression of antisense clathrin heavy chain inhibits the toxicity of Shiga toxin [82]. Interestingly, Shiga toxin-HRP, like CT-B-HRP, also accumulates in caveolae at 37 °C (our unpublished data). It should be noted that ligands taken up from caveolae and clathrin-coated pits can ultimately be located in the same endosome (see Section 4.7).

In our attempts to characterize the endocytic mechanisms of CT uptake [66], Caco-2 cells were transfected with caveolin, leading to the formation of basolateral caveolae in the otherwise caveolae/caveolin-negative, highly polarized cell line [20]. This did not change the uptake or effect of CT. Furthermore, treatment of the cav-1-expressing Caco-2 cells with the cholesterol-binding drug filipin only reduced internalization of CT slightly (<20%), suggesting that although caveolae may be involved in CT endocytosis, they do not play a major role. In contrast, cholesteroldepletion with m β CD, a treatment that both removes caveolae and inhibits the formation of deeply invaginated clathrin-coated pits and subsequently coated vesicle formation [37,83], resulted in a 30-40% reduction in CT uptake [66], indicating that coated pits/vesicles also play a role, though not exclusive, in CT uptake. Similarly, in baby hamster kidney (BHK) cells expressing antisense clathrin heavy chain [84], endocytosis of CT was reduced by about 50%. Moreover, in HeLa cells expressing the dominant-negative dynamin mutant K44A, a 50-70% reduction in CT uptake was seen [66]. In these cells both endocytosis mediated by caveolae and by clathrin-coated pits/vesicles are inhibited, as well as possible caveolin- and clathrin-independent mechanisms that may depend on dynamin. In BHK and HeLa cells, in which either antisense clathrin or the dominant-negative dynamin mutant was expressed, CT is still internalized after cholesterol depletion with mβCD [85]. In a recent report it was similarly shown that the overexpression of dominant mutants that inhibit clathrin-, caveolin- or Arf6-dependent endocytic pathways does not prevent CT uptake and trafficking to the Golgi and ER, or CT cytotoxicity.

78 4 The Role of Caveolae and Noncaveolar Rafts in Endocytosis

Interestingly, even under conditions where all three endocytic pathways were inhibited simultaneously, CT has a toxic effect [86]. Importantly, it must be borne in mind that the blocking of one or more pathways may lead to up-regulation of other pathways that are normally quiescent [87], or it may induce new pathways [88]. Taken together, these results suggest that CT can be internalized by caveolae and clathrin-coated pits/vesicles as well as by one or more mechanism not involving these structures. However, caveolae do not seem to play a major role in the uptake.

In order to further analyze the importance of caveolae in CT-uptake, and not least to evaluate the internalization efficiency of CT bound to GM1 inside and outside caveolae, we incubated HeLa cells with red fluorescent CT-B on ice and subsequently at 37 °C for various periods of time (between 0 and 120 min), followed by fixation and immunocytochemical detection of caveolae using an antibody against endogenous caveolin-1. A partial co-localization of CT-B and caveolin on the cell surface following labeling at 0 °C was seen. However, upon heating to 37 °C, CT-B was internalized and gradually accumulated in the perinuclear region. After 30 min and more, some - but not all - of the internalized CT-B was colocalized with TGN38, a marker of the TGN (Fig. 4.2A), while a small fraction of CT was still seen at the plasma membrane where it apparently co-localized with caveolin. Interestingly, during this CT-B uptake, no change in the distribution of endogenous caveolin was detected, irrespective of the time of CT-B incubation. At all time points (0-120 min) the peripheral rim of caveolae remained stable. In blind experiments, where the microscope operator only examined caveolin in the confocal microscope and was unaware of the incubation time with CT-B at 37 °C, no difference between the time points could be established. Similarly, no differences in the distribution of caveolae could be observed when cells were incubated with the holotoxin (CT A+B chain) (our unpublished data).

Since the individual cells bound CT-B to varying degrees, and some were unlabeled, such cells could serve as internal controls for a changed pattern in the distribution of endogenous caveolin stimulated by CT-B binding. Importantly, at all incubation times the caveolar localization was the same, irrespective of the GM1 expression level of the cell and consequently the degree of CT-B binding (Fig. 4.2B). Taken together, these data strongly indicate that CT-B uptake does not cause a synchronized wave of caveolae internalization. To further analyze the effect of CT-B uptake on the distribution of caveolae, we expressed GFP-tagged caveolin-1 fusion proteins [31] in HeLa cells and human skin fibroblasts. When these cells were incubated with CT-B on ice and subsequently incubated at 37 °C for various periods (0-120 min), and then fixed and examined in the confocal microscope, results similar to those with endogenous caveolin were obtained: CT-B became internalized and accumulated in the TGN over time, while the pattern of caveolar expression was unchanged - no wave of incoming caveolae or caveolin stimulated by CT-B was observed (Fig. 4.2C and D). When the pattern of CT-B internalization in cells expressing GFP-tagged caveolin was compared to that of nontransfected, neighboring cells in the same culture, no differences were seen (Fig. 4.2C). Thus, the expression of GFP-tagged caveolin does not influence CT-B uptake and transport.



Fig. 4.2 (A) HeLa cells were labeled with Alexa 594 CT-B (red) on ice for 1 h, washed, and incubated at 37 °C for 2 h to allow internalization of CT-B. The cells were subsequently fixed and immunolabeled for endogenous caveolin-1 (white) and TGN-38 (green). After 2 h, the majority of CT-B has been internalized to the TGN. (B) HeLa cells were allowed to internalize CT-B for 2 h, as described above. The cells were subsequently fixed and immunolabeled for endogenous caveolin-1 (green). The panel shows two cells expressing GM1 that have internalized CT-B, and adjacent cells that have not internalized CT-B. The cellular distribution of caveolin-1 the same, irrespective of whether the cells have internalized CT-B. (C) HeLa cells transfected with Cav-GFP were allowed to internalize CT-B for 2 h, as described above. The panel shows two transfected cells and adjacent, nontransfected cells. The pattern of internalized CT-B is the same in transfected and nontransfected cells. (D) HeLa cells transfected with Cav-GFP and incubated with CT-B for 5 to 30 min at 37 °C. While CT-B moves from the cell surface to the perinuclear region (the TGN), the distribution of Cav-GFP-labeled caveolae is not changed. Scale bars = 20 μ m.

80 4 The Role of Caveolae and Noncaveolar Rafts in Endocytosis



Fig. 4.3 Electron micrographs of BHK cells incubated with a CT-B-HRP conjugate. (A) Cells were incubated at 0 °C with the conjugate, which is seen all over the cell surface including some caveolae. (B,C) Cells have been incubated with the conjugate at 0 °C for 1 h, then washed and further incubated for 30 min at 37 °C. Very little or no labeling is seen at the cell surface, except in caveolae, which are heavily labeled. Such images sug-

gest that the CT-B-HRP has become internalized by a caveolae-independent mechanism. However, on very rare occasions unlabeled caveolae can be seen (arrow in C). These caveolae most likely are formed after the surfaceassociated CT-B-HRP conjugate was internalized, and may represent a substitution for a rare caveola that has been involved in conjugate uptake. Scale bar = 200 nm.

Next, the internalization of CT-B in relation to caveolae was followed over time in HeLa cells and human skin fibroblasts by photobleaching techniques (our unpublished results). For this purpose, cells expressing GFP-tagged caveolin were preincubated for 1 h at 4 °C with CT-B, washed and further incubated for 5-10 min in CT-B-free medium at 37 °C on stage to stabilize the culture. The cells were then exposed to a modified FRAP protocol where the entire cytoplasm was bleached in both the red and the green channels, leaving only the most peripheral rim of fluorescence corresponding to the plasma membrane (see also Ref. [31]). Fluorescence recovery in the cytoplasm was then followed over time. The rationale of this type of experiment is that if CT-B binding stimulates caveolae to become constitutively internalized and replaced by a recycling mechanism, then the GFPtagged caveolae at the plasma membrane should disappear and be replaced by caveolae which are not labeled - that is, the peripheral fluorescence should decrease and the intracellular fluorescence increase. It was, however, striking that although CT-B appeared in vesicles in the bleached interior of the cells, GFPtagged caveolin did not, but rather remained at the cell periphery. These findings strongly suggest that (GFP-tagged) caveolae are not efficiently internalized and

recycled in response to CT-binding but are more likely stable, plasma membraneassociated structures.

These findings were further supported by EM observations on HeLa cells showing that even after 30–120 min of CT-B-HRP internalization, caveolae were often distinctly labeled by CT-B-HRP although the remaining plasma membrane was unlabeled (Fig. 4.3). This emphasizes that at time points where the main, mobile GM1/CT-B fraction has been internalized by a caveolae-independent mechanism(s), a nonmobile fraction is still present on the cell surface, trapped in the caveolae. It was also possible by using EM to detect the internalized CT-B-HRP in the TGN [17]. The number of caveolae at the cell surface of cells pulse-chased with CT-B-HRP for 30 min at 37 °C was the same as in control cells not exposed to CT-B-HRP (our unpublished results).

Furthermore, since it has been reported that expression of the K44A dominantnegative dynamin-1 mutant inhibits caveolae-mediated internalization of CT-B in endothelial cells [28], we also quantified the number of caveolae in HeLa K44A cells expressing the mutant – that is, in cells grown without tetracycline. Interestingly, although expression of the dynamin mutant leads to a two- to three-fold increase in the number of clathrin-coated pits at the cell surface as expected [66], the number of caveolae at the cell surface was unchanged, indicating that caveolae are not constitutively internalized by a dynamin-dependent mechanism. In K44A cells grown both in the presence or absence of tetracycline and incubated for 30 min at 37 °C with CT-B-HRP, practically all caveolae were labeled (our unpublished observations).

4.6 A Small Fraction of Caveolae may become Constitutively Internalized

Clearly, the possibility cannot be excluded that a few caveolae (or clusters of caveolae) pinch off occasionally and therefore participate in the delivery of CT and other molecules to intracellular compartments, although this must – in terms of endocytic activity – be relatively insignificant. Recent studies conducted by Kirkham et al. [76] showed that a very small fraction (or a subpopulation) of caveolae is actually internalized constitutively. These authors developed an ultrastructural assay which made it possible to quantitate free, budded caveolae versus surfaceconnected caveolae. For this, they used CT-B-HRP as an endocytic marker, and quenched the HRP reaction of labeled structures connected to the cell surface with ascorbic acid, which is membrane-impermeable. This was combined with an immunolabeling protocol that allowed them to show that truly free caveolae-like vesicles containing the HRP reaction product were in fact caveolae. In this way, it was shown that about 2% of the total population of caveolae bud off per minute.

Interestingly, the internalization of prion proteins that are GPI-anchored also appears to be a rather slow or inefficient process that can be mediated by caveolae [89]. Thus, in an immunogold labeling pulse-chase experiment it was found that approximately 90% of the caveolae at the plasma membrane that were labeled

82 4 The Role of Caveolae and Noncaveolar Rafts in Endocytosis

after a 10-min period were still labeled after a 50-min chase; only about 10% of all caveolae were internalized per hour. This slow internalization may be constitutive, although it cannot be excluded that it is somehow stimulated by the prion proteins because the caveolae-mediated pathway was not observed for several other GPI-anchored proteins [89].

In our FRAP experiments in which the interior of the cells were bleached, only leaving a narrow, fluorescent zone of GFP-tagged caveolin structures, the fluorescence disappeared only very slowly from the peripheral zone (see Fig. 5t in Ref. [31]). Based on extrapolations of such data, the fluorescence would disappear completely from the periphery within about 3 h. Taken together, the described EM and FRAP data are consistent with a very slow, constitutive internalization of caveolae.

4.7

Caveosomes: Intracellular Caveolin-Associated Structures

In addition to reaching the TGN, internalized CT is also found in caveosomes [90], the organelle reached by internalized SV40 virus [52]. The caveosome is a non-acidic, caveolin-associated compartment that does not stain for EEA1, a marker of early endosomes reached by the clathrin-dependent endocytic pathway. Moreover, caveosomes do not accumulate Lysotracker, a lysosomal marker, and they do not stain for TGN46 and mannosidase II, both of which are Golgi markers [52]. Evidence that the clathrin-dependent endosome pathway and the caveolin-dependent caveosome pathway somehow merge are accumulating [91,92]. Moreover, in studies with SV40 virus and CT it was recently reported that these caveolar markers are transported by a Rab5-dependent pathway to early endosomes and that early endosomes and caveosomes may even communicate via this Rab5 pathway [91]. An important difference between the two ligands is that the Rab5-dependent step is required for CT transport to the Golgi complex, but not for SV40 entry into the ER.

These studies further stress the important notion that because a certain ligand initially is seen in caveolae, and later on also in intracellular compartments associated with caveolin, the ligand has not necessarily been internalized and transported to the compartment by caveolae. For instance, transforming growth factor beta (TGF β) was found to localize to both rafts/caveolae and clathrin-coated pits at the cell surface [93]. Moreover, after internalization, TGF β was seen in caveolin-positive vesicles as well as in vesicles staining for EEA1. It was therefore concluded that two endocytic pathways for TGF β were operating in parallel: one mediated by clathrin-coated pits and vesicles and leading to endosomes and subsequently to TGF β signaling; and one mediated by rafts/caveolae leading to caveolin-positive vesicles (caveosomes?) involved in receptor degradation and turnover. However, caveolae could in fact bind TGF β without being involved in endocytosis, and all internalized receptors could have been taken up by endocytic noncaveolar rafts or clathrin-dependent endocytosis, leading to endosomes. These endosomes, in turn,

may exchange material with caveolin-positive vesicles or even with caveosomes via intracellular, caveolin-associated carrier vesicles.

4.8 The Role of Dynamin in Caveolar Function

From the sections above, it is clear that dynamin is central in the discussions on caveolar endocytosis, and this GTP-binding protein therefore warrants further attention in this chapter. It is well established that dynamin plays a role in the final scission of clathrin-coated vesicles to form free endocytic vesicles [94,95], and it has also been reported to be important for the fission of caveolae [28,29]. However, the role of dynamin associated with caveolae can be complex. Dynamin is an actininteracting protein which, for example, binds to cortactin. Indeed, recently dynamin was found to be involved in the regulation of actin polymerization during micropinosome comet formation - that is, the formation of rapidly moving vesicles propelled forward by an actin treadmilling machine [96]. Additionally, caveolin interacts with the actin cytoskeleton via filamin, and reorganization of the actin cytoskeleton leads to redistribution of caveolae [39]. Depolymerization of the actin cytoskeleton with cytochalasin D leads to an increase in caveolar mobility, and it has therefore been suggested that the normal immobility of caveolae is due to an anchoring role of actin filaments [31]. Dynamin could be a partner in such stabilization of caveolae. As mentioned above (see Section 4.4), Pelkmans et al. [53] found a pronounced SV40 virus-induced reorganization of the actin cytoskeleton and showed that dynamin is transiently recruited to SV40 virus-containing caveolae, which are subsequently internalized. Hence, it could be speculated that the internalization of caveolae, induced for instance by SV40 virus or okadaic acid [62], leads to a reorganization of the actin cytoskeleton mediated by dynamin. In fact, cytochalasin D inhibits the okadaic acid-stimulated uptake of caveolae [62], and there is an increasing body of evidence that actin plays a key role in endocytosis [62,97-99]. Therefore, the possible association of dynamin with caveolae does not per se indicate that these caveolae are going to pinch off unless other processes such as actin reorganization have also been initiated in response to a specific stimulation. In agreement with this and, as pointed out above, we found no accumulation of caveolae at the plasma membrane in cells expressing dominant-negative dynamin, as would have been expected if dynamin was constitutively mediating scission of caveolae (our unpublished observations).

4.9 Caveolin Immobilizes Rafts/Caveolar Invaginations

In addition to the actin cytoskeleton, which seems to be involved in the immobility of caveolae (see Section 4.2), caveolin-1 itself apparently immobilizes caveolae. Thus, Le et al. [100] and Nabi and Le [14] suggested that lipid rafts can invaginate

84 4 The Role of Caveolae and Noncaveolar Rafts in Endocytosis

in a cholesterol-dependent but caveolin-independent way to give rise to caveolaelike invaginations of the plasma membrane, which become rapidly internalized in a dynamin-dependent way. Such endocytic raft structures may then be immobilized by caveolin-1 and only become internalized after specific stimulation (e.g., virus binding). Pang et al. [77] recently showed that the content of GM1 varies among cells, and that cells with a low GM1 content also had a relatively low CT uptake which was independent of caveolin-1 expression.

It remains an open question as to whether there exist subpopulations of caveolae, or a subpopulation of caveolae-like invaginations with so little caveolin that it cannot be immobilized, but rather is endocytic [12]. Such a subpopulation could account for the small fraction of caveolae that appears to become constitutively internalized (see Section 4.6).

As mentioned above (see Section 4.5), CT-B can be internalized by mechanisms that are caveolin-, clathrin- and dynamin-independent [66]. Although such an endocytic mechanism (or several mechanisms) is far from well characterized, it is clear that for instance other protein toxins may enter cells in such a manner [11]. It should be noted that caveolin-, clathrin- and dynamin-independent endocytosis is not necessarily dependent on lipid rafts [37], but one such internalization pathway could be via a noncaveolar, raft-based and cholesterol-dependent mechanism [101,102]. For example, in a study on lymphocytes it was found that after ligand stimulation the interleukin-2 receptor became internalized in a clathrinand caveolin-independent, but dynamin-dependent way via detergent-resistant membrane domains [102]. It should be stressed, however, that noncaveolar rafts, which do not form characteristic, regular, 50- to 70-nm invaginations during their pinching-off [14,100] (in contrast to caveolae or caveolae-like invaginated rafts) are poorly defined morphologically. This indeed makes the study of a possible role of noncaveolar rafts in endocytosis difficult. Several years ago we identified small, uncoated endocytic vesicles which differed from coated pits/vesicles and caveolae ultrastructurally [103]. In a recent study, Kirkham et al. [76] found that CT was mainly internalized in a clathrin-, caveolin-, and dynamin-independent way by means of uncoated tubular or ring-shaped structures, as identified ultrastructurally. These structures also contained GPI-anchored proteins. That uptake of this type of protein can occur independently of clathrin, caveolin and dynamin is in agreement with our study of a GPI-anchored diphtheria toxin receptor [104]. The mechanism reported by Kirkham et al. [76] was operating both in primary fibroblasts from wild-type and cav-1^{-/-}(cav-1-null) mice. Moreover, it was recently reported also that SV40 virus can be rapidly internalized in both cav-1-null and wildtype fibroblasts in a caveolin-, clathrin- and dynamin-independent manner [105].

The special caveolar lipid-environment appears to be of functional importance for caveolae-mediated endocytosis. Thus, increasing the cellular cholesterol level increases caveolae-mediated uptake of albumin and lactosylceramide. A similar effect is seen after the addition of exogenous lactosylceramide or GM1. In fact, the balance between the cellular amount of caveolin and glycosphingolipids may be of vital importance for the ability of caveolae to pinch off. In control cells, increasing the amount of caveolin-1 leads to a decrease in the caveolar-mediated endocytosis of albumin. In contrast, increasing both the amount of caveolin-1 and glycosphingolipid leads to an increased endocytosis of albumin, presumably because in the latter case the correct balance between caveolin-1 and glycosphingolipids is maintained [106].

4.10 A 2005 Consensus Model for Caveolar Endocytosis

Based on the various findings presented here, we propose the following model for a role of caveolae and noncaveolar rafts in endocytosis (see also Ref. [30]). Noncaveolar lipid rafts, perhaps smaller than caveolae, fuse laterally and form preendocytic domains that may pinch off by both dynamin-dependent and -independent mechanisms. Such rafts may also become associated with caveolin. This association leads to a stable, defined curvature and size of the raft, and to a reduced ability to pinch off. The newly formed caveolae become associated with the actin cytoskeleton via filamin and are now immobilized. However, a small fraction of these caveolae may actually become constitutively internalized, perhaps as a means of caveolin turnover, either because they have not been sufficiently immobilized by the actin cytoskeleton and/or they contain too little caveolin. Molecules present in such caveolae may therefore become internalized. Moreover, caveolae may be involved in local, short-range motility, possibly including fission and fusion processes. Finally, in some cases – for example, after virus stimulation – a signaling cascade may be initiated, leading to reorganization of the actin cytoskeleton and recruitment of dynamin, and a subsequent wave of internalized caveolae, which are only very slowly replaced. Considering all the other functions ascribed to caveolin/caveolae at the plasma membrane (see Section 4.1), such a massive, longlasting down-regulation of caveolae can, however, hardly be considered a physiologically desirable situation for the cell.

Acknowledgments

The studies referred to in this chapter from the authors' laboratories have been supported by the Danish and Norwegian Cancer Societies, the Danish Medical Research Council, the Norwegian Research Council for Science and Humanities, The Novo Nordisk Foundation, The John and Birthe Meyer Foundation, the Jahre Foundation, and Jeanette and Søren Bothners Legacy. 86 References

Abbreviations

BHK	baby hamster kidney
CT	cholera toxin
EM	electron microscopy
eNOS	endothelial nitric oxide synthase
ER	endoplasmic reticulum
FLIP	fluorescence loss in photobleaching
FRAP	fluorescence recovery after photobleaching
Gb3	globotriasylceremide
GFP	green fluorescent protein
mβCD	methyl-β-cyclodextrin
MDCK	Madin-Darby canine kidney
TGFβ	transforming growth factor beta
TGN	trans-Golgi network

References

- Larkin, J. M., Brown, M. S., Goldstein, J. L., Anderson, R. G. W., *Cell* 1983, 33, 273–285.
- 2 Larkin, J. M., Donzell, W. C., Anderson, R. G. W., J. Cell. Physiol. 1985, 124, 372–378.
- 3 Moya, M., Dautry-Varsat, A., Goud, B., Louvard, D., Boquet, P., *J. Cell Biol.* 1985, 101, 548–559.
- 4 Sandvig, K., Olsnes, S., Petersen, O. W., van Deurs, B., *J. Cell Biol.* 1987, 105, 679–689.
- 5 Madshus, I. H., Sandvig, K., Olsnes, S., van Deurs, B., J. Cell. Physiol. 1987, 131, 14–22.
- 6 van Deurs, B., Petersen, O. W., Olsnes, S., Sandvig, K., Int. Rev. Cytol. 1989, 117, 131–177.
- 7 Merrifield, C. J., Feldman, M. E., Wan, L., Almers, W., *Nature Cell Biol.* 2002, 4, 691–698.
- 8 Marsh, M., McMahon, H.T., *Science* 1999, 285, 215–220.
- 9 Higgins, M.K., McMahon, H.T., Trends Biochem. Sci. 2002, 27, 257–263.
- 10 Johannes, L., Lamaze, C., *Traffic* 2002, 3, 443–451.
- 11 Sandvig, K., van Deurs, B., Annu. Rev. Cell Dev. Biol. 2002, 18, 1–24.
- 12 Nichols, B., J. Cell Sci. 2003, 116, 4707–4714.

- 13 Helms, J. B., Zurzolo, C., Traffic 2004, 5, 247–254.
- 14 Nabi, I.R., Le, P.U., J. Cell Biol. 2003, 161, 673–677.
- 15 Conner, S.D., Schmid, S.L., *Nature* 2003, 422, 37–44.
- 16 Kirkham, M., Parton, R., Biochim. Biophys. Acta 2005, 1745, 273–286.
- 17 van Deurs, B., Roepstorff, K., Hommelgaard, A. M., Sandvig, K., *Trends Cell Biol.* 2003, 13, 92–100.
- 18 Fujimoto, T., Nakade, S., Miyawaki, A., Mikoshiba, K., Ogawa, K. J., *Cell. Biol.* 1992, 119, 1507–1513.
- Petersen, O.W., Hansen, S.H., Laursen, I., van Deurs, B., *Eur. J. Cell Biol.* 1989, 50, 500–509.
- 20 Vogel, U., Sandvig, K., van Deurs, B., J. Cell Sci. 1998, 111, 825–832.
- 21 Fielding, C. J., Fielding, P. E., Adv. Drug Deliv. Rev. 2001, 49, 251–264.
- 22 Fielding, C. J., Fielding, P. E., Biochem. Soc. Trans. 2004, 32, 65–69.
- 23 Ikonen, E., Parton, R.G., *Traffic* 2000, 1, 212–217.
- 24 Isshiki, M., Anderson, R.G., Cell Calcium 1999, 26, 201–208.
- 25 Chini, B., Parenti, M., J. Mol. Endocrinol. 2004, 32, 325–338.
- 26 Cohen, A. W., Hnasko, R., Schubert, W., Lisanti, M. P., *Physiol. Rev.* 2004, 84, 1341–1379.

- 27 Tortelote, G. G., Valverde, R. H., Lemos, T., Guilherme, A., Einicker-Lamas, M., Vieyra, A., *FEBS Lett.* 2004, 576, 31–35.
- 28 Oh, P., McIntosh, D. P., Schnitzer, J. E., J. Cell Biol. 1998, 141, 101–144.
- 29 Henley, J. R., Krueger, E.W.A., Oswald, B. J., McNiven, M. A., J. Cell Biol. 1998, 141, 85–99.
- **30** Hommelgaard, A. M., Roepstorff, K., Vilhardt, F., Torgersen, M. L., Sandvig, K., van Deurs, B., *Traffic* 2005, 6, 720–724.
- 31 Thomsen, P., Roepstorff, K., Stahlhut, M., van Deurs, B., *Mol. Biol. Cell* 2002, 13, 238–250.
- 32 Kenworthy, A. K., Nichols, B. J., Remmert, C. L., Hendrix, G. M., Kumar, M., Zimmerberg, J., Lippincott-Schwartz, J., J. Cell Biol. 2004, 165, 735–746.
- 33 Adams, C. L., Chen, Y. T., Smith, S. J., Nelson, W. J., J. Cell Biol. 1998, 142, 1105–1119.
- 34 Bucci, C., Thomsen, P., Nicoziani, P., McCarthy, J., van Deurs, B., *Mol. Biol. Cell* 2000, 11, 467–480.
- 35 Chang, W.J., Rothberg, K.G., Kamen, B.A., Anderson, R.G.W., J. Cell Biol. 1992, 118, 63–69.
- 36 Hailstone, D., Sleer, L. S., Parton, R. G., Stanley, K. K.. J. Lipid Res. 1998, 39, 369–379.
- 37 Rodal, S. K., Skretting, G., Garred, O., Vilhardt, F., van Deurs, B., Sandvig, K., *Mol. Biol. Cell* 1999, 10, 961–974.
- 38 Mundy, D. I., Machleidt, T., Ying, Y., Anderson, R.G. W., Bloom, G. S., J. Cell Sci. 2002, 115, 4327–4339.
- 39 Stahlhut, M., van Deurs, B., Mol. Biol. Cell 2000, 11, 325–337.
- 40 Pelkmans, L., Zerial, M., Nature 2005, 436, 128–133.
- 41 Simionescu, M., Gafencu, A., Antohe, F., Microsc. Res. Tech. 2002, 57, 269–288.
- 42 Frøkjær-Jensen, J., Prog. Appl. Microcirc. 1983, 1, 17–34.
- 43 Frøkjær-Jensen, J., Wagner, R.C., Andrews, S.B., Hagman, P., Reese, T.S., *Cell Tissue Res.* 1988, 254, 17–24.
- 44 Bundgaard, M., Hagman, P., Crone, C., *Microvasc. Res.* 1983, 25, 358–368.
- 45 Predescu, S.A., Predescu, D.A., Palade, G.E., *Mol. Biol. Cell* 2001, 12, 1019–1033.
- **46** Schnitzer, J. E., Liu, J., Oh, P., *J. Biol. Chem.* 1995, 270, 14 399–14404.

- 47 Minshall, R. D., Tiruppathi, C., Vogel, S. M., Niles, W. D., Gilchrist, A., Hamm, H. E., Malik, A. B., *J. Cell Biol.* 2000, 150, 1057–1069.
- 48 Drab, M., Verkade, P., Elger, M., Kasper, M., Lohn, M., Lauterbach, B., Menne, J., Lindschau, C., Mende, F., Luft, F.C., Schedl, A.S., Haller, H., Kurzchalia, T.V., *Science* 2001, 293, 2449–2452.
- 49 Razani, B., Engelman, J. A., Wang, X. B., Schubert, W., Zhang, X. L., Marks, C. B., Macaluso, F., Russell, R. G., Li, M., Pestell, R. G., Di Vizio, D., Hou, H., Jr., Kneitz, B., Lagaud, G., Christ, G. J., Edelmann, W., Lisanti, M. P., J. Biol. Chem. 2001, 276, 38121–38138.
- 50 Goligorsky, M. S., Li, H., Brodsky, S., Chen, J., Am. J. Physiol. 2002, 283, F1-F10.
- 51 Schubert, W., Frank, P. G., Woodman, S. E., Hyogo, H., Cohen, D. E., Chow, C. W., Lisanti, M. P., *J. Biol. Chem.* 2002, 277, 40091–40098.
- 52 Pelkmans, L., Kartenbeck, J., Helenius, A., Nature Cell Biol. 2001, 3, 473–483.
- 53 Pelkmans, L., Puntener, D., Helenius, A., *Science* 2002, 296, 535–539.
- 54 Pelkmans, L., Helenius, A., *Traffic* 2002, 3, 311–320.
- 55 Huet, C., Ash, J. F., Singer, S. J., Cell 1980, 21, 429–438.
- 56 Marjomäki, V. S., Pietianen, V., Matilainen, H., Upla, P., Ivaska, J., Nissinen, L., Reunanen, H., Huttunen, P., Hyypia, T., Heino, J., J. Virol. 2002, 76, 1856–1865.
- 57 Upla, P., Marjomäki, V. S., Kankaanpaa, P., Ivaska, J., Hyypia, T., van der Goot, F. G., Heino, J., *Mol. Biol. Cell* 2004, 15, 625–636.
- 58 Eash, S., Querbes, W., Atwood, W.J., J. Virol. 2004, 78, 11583–11590.
- 59 Helenius, A., Kartenbeck, J., Simons, K., Fries, E., J. Cell Biol. 1980, 84, 404–420.
- 60 Kee, S. H., Cho, E. J., Song, J. W., Park, K. S., Baek, L. J., Song, K., J. Microbiol. Immunol. 2004, 48, 823–829.
- 61 Grummer, B., Grotha, S., Greiser-Wilke, I. J., Vet. Med. B Infect. Dis. Vet. Public Health 2004, 51, 427–432.
- 62 Parton, R.G., Joggerst, B., Simons, K., J. Cell Biol. 1994, 127, 1199–1215.
- 63 Shin, J.-S., Abraham, S. N., Immunology 2001, 102, 2–7.
- 64 Shin, J.-S., Gao, Z., Abraham, S.N., Science 2000, 289, 785–788.

- 88 References
 - **65** Shogomori, H., Futerman, A.H., J. Biol. Chem. 2001, 276, 9182–9188.
 - 66 Torgersen, M. L., Skretting, G., van Deurs, B., Sandvig, K., J. Cell Sci. 2001, 114, 3737–3747.
 - 67 Nichols, B. J., Kenworthy, A. K., Polishchuk, R. S., Lodge, R., Roberts, T. H., Hirschberg, K., Phair, R. D., Lippincott-Schwartz, J., *J. Cell Biol.* 2001, 153, 529–541.
 - 68 Parton, R.G., J. Histochem. Cytochem. 1994, 42, 155–166.
 - 69 Hansen, G. H., Dalskov, S. M., Rasmussen, C. R., Immerdal, L., Niels-Christiansen, L.-L., Danielsen, E. M., *Biochemistry* 2005, 44, 873–882.
 - 70 Puri, V., Watanabe, R., Singh, R. D., Dominguez, M., Brown, J.C., Wheatley, C. L., Marks, D. L., Pagano, R. E., *J. Cell Biol.* 2001, 154, 535–547.
 - 71 Orlandi, P.A., Fishman, P.H., J. Cell Biol. 1998, 141, 905–915.
 - 72 Montesano, R., Roth, J., Robert, A., Orci, L., *Nature* 1982, 296, 651–653.
 - 73 Matveev, S., Li, X., Everson, W., Smart, E. J., Adv. Drug Deliv. Rev. 2001, 49, 237–250.
 - 74 Norkin, L. C., Adv. Drug Deliv. Rev. 2001, 49, 301–315.
 - 75 Möbius, V., Herzog, V., Sandhoff, K., Schwarzmann, G., J. Histochem. Cytochem. 1999, 47, 1005–1014.
 - 76 Kirkham, M., Fujita, A., Chadda, R., Nixon, S. J., Kurzchalia, T. V., Sharma, D. K., Pagano, R. E., Hancock, J. F., Mayor, S., Parton, R. G., *J. Cell Biol.* 2005, 168, 465–476.
 - 77 Pang, H., Le, P. U., Nabi, I. R., J. Cell Sci. 2004, 117, 1421–1430.
 - 78 Singh, R. D., Puri, V., Valiyaveettil, J. T., Marks, D. L., Bittman, R., Pagano, R. E., *Mol. Biol. Cell* 2003, 14, 3254–3265.
 - 79 Roepstorff, K., Thomsen, P., Sandvig, K., van Deurs, B., J. Biol. Chem. 2002, 277, 18954–18960.
 - 80 Sandvig, K., Olsnes, S., Brown, J. E., Petersen O. W., van Deurs, B., *J. Cell Biol.* 1989, 108, 1331–1343.
 - 81 Sandvig, K., Garred, Ø., Prydz, K., Kozlov, J.V., Hansen, S.H., van Deurs, B., *Nature* 1992, 358, 510–511.
 - 82 Sandvig, K., Grimmer, S., Lauvrak, S.U., Torgersen, M.L., Skretting, G., van

Deurs, B., Iversen, T.G., Histochem. Cell. Biol. 2002, 117, 131–141.

- 83 Subtil, A., Hémar, A., Dautry-varsat, A. J., Cell Sci. 1994, 107, 3461–3468.
- 84 Iversen, T.-G., Skretting, G., van Deurs, B., Sandvig, K., *Proc. Natl. Acad. Sci. USA* 2003, 100, 5175–5180.
- 85 Sandvig, K., Spilsberg, B., Lauvrak, S. U., Torgersen, M. L., Iversen, T.-G., van Deurs, B., Int. J. Med. Microbiol. 2004, 293, 483–490.
- 86 Massol, R. H., Larsen, J. E., Fujinaga, Y., Lencer, W. I., Kirchhausen, T., *Mol. Biol. Cell* 2004, 15, 3631–3641.
- 87 Damke, H., Baba, T., van der Bliek, A. M., Schmid, S. L., J. Cell Biol. 1995, 131, 69–80.
- 88 Llorente, A., Lauvrak, S. U., van Deurs, B., Sandvig, K., J. Biol. Chem. 2003, 278, 35850–35855.
- 89 Peters, P. J., Mironov, A., Peretz, D., van Donselaar, E., Leclerc, E., Erpel, S., DeArmond, S. J., Burton, D. R., Williamson, R. A., Vey, M., Prusiner, S. B., *J. Cell Biol.* 2003, 162, 703–717.
- **90** Nichols, B. J., *Nature Cell Biol.* 2002, 4, 374–378.
- **91** Pelkmans, L., Burli, T., Zerial, M., Helenius, A., *Cell* 2004, 118, 767–780.
- 92 Sharma, D.K., Choudhury, A., Singh, R.D., Wheatley, C.L., Marks, D.L., Pagano, R.E., J. Biol. Chem. 2003, 278, 7564–7572.
- 93 Di Guglielmo, G. M., Le Roy, C., Goodfellow, A. F., Wrana, J. L., *Nature Cell Biol.* 2003, 5, 410–421.
- 94 Hill, E., van Der Kaay, J., Downes, C. P., Smythe, E., J. Cell Biol. 2001, 152, 309–323.
- 95 Sever, S., Damke, H., Schmid, S. L., Traffic 2000, 1, 385–392.
- 96 Orth, J. D., Krueger, E. W., Cao, H., McNiven, M.A., Proc. Natl. Acad. Sci. USA 2002, 99, 167–172.
- 97 Jeng, R. L., Welch, M. D., Curr. Biol. 2001, 11, R691-R694.
- 98 Schafer, D. A., Curr. Opin. Cell Biol. 2002, 14, 76–81.
- 99 Yarar, D., Waterman-Storer, C. M., Schmid, S. L., *Mol. Biol. Cell* 2005, 16, 964–975.
- 100 Le, P. U., Guay, G., Altschuler, Y., Nabi, I. R., J. Biol. Chem. 2002, 277, 3371– 3379.

- 101 Ros-Baro, A., Lopez-Iglesias, C., Peiro, S., Bellido, D., Palacin, M., Zorzano, A., Camps, M., Proc. Natl. Acad. Sci. USA 2001, 98, 12050–12055.
- 102 Lamaze, C., Dujeancourt, A., Baba, T., Lo, C. G., Benmerah, A., Dautry-Varsat, A., *Mol. Cell* 2001, 7, 661–671.
- 103 Hansen, S. H., Sandvig, K., van Deurs, B., J. Cell Biol. 1991, 113, 731–741.
- 104 Skretting, G., Torgersen, M. L., van Deurs, B., Sandvig, K., *J. Cell Sci.* 1999, 112, 3899–3909.
- 105 Damm, E. M., Pelkmans, L., Kartenbeck, J., Mezzacasa, A., Kurzchalia, T., Hele-
- nius, A., *J. Cell Biol.* 2005, 168, 477–488. 106 Sharma, D.K., Brown, J.C., Choudhury, A., Peterson, T.E., Holicky, E., Marks,
 - D.L., Simari, R., Parton, R.G., Pagano,
 - R.E., Mol. Biol. Cell 2004, 15, 3114-3122.

Role of Cholesterol in Signal Transduction from Caveolae

Christopher J. Fielding and Phoebe E. Fielding

5.1 Introduction

5

Caveolae are free cholesterol/sphingolipid (FC/SPH)-rich microdomains of the cell surface that are assembly sites for many transmembrane signaling complexes. Other proteins associated with caveolae include the transporters of some small ligands (glucose, inorganic ions) and catalysts of cell FC homeostasis. The highest levels of caveolae are found in differentiated primary cells, including pulmonary type-1 cells, adipocytes, endothelial cells, smooth muscle cells, and fibroblasts. In blood lymphocytes and many transformed and cancer cell lines, caveolae are sparse or may be completely absent.

While there is no broadly accepted nomenclature, many investigators now recognize two classes of FC/SPH-rich microdomains, caveolae and lipid rafts (Fig. 5.1) [1–3]. Caveolae are invaginated, and contain unique structural proteins (caveolins) that play an essential role in maintaining membrane curvature (see Chapter 2). The mean diameter of caveolar pits is ~60-80 Å. A second class of proteins (dynamins) is present in the necks of caveolae [4], but these are also present in other structures, such as clathrin-coated pits. The central role of caveolins-[particularly caveolin-1, the largest (22 kDa) caveolin-family protein] in maintaining the structure of caveolae is shown by their disappearance from the cell surface if the expression of caveolin is knocked down in wild-type cells [5]. This is confirmed by the absence of caveolae from caveolin-1 -/- cells [6,7]. The structural significance of FC in caveolae is shown by the flattening that first occurs when the plasma membrane is FC-depleted [8,9]. The caveolin skeleton remains at the cell surface. Caveolae reappear when FC is replaced in the plasma membrane. Additional loss of FC may be followed by disassembly of caveolin multimers [10]. Whilst the transfer of caveolin between the cell surface and intracellular pools can be demonstrated using inhibitors of microtubule assembly, it is unclear whether this normally occurs.

Under physiological conditions the caveolin skeleton at the surface of living cells appears to be relatively long-lived. Individual caveolae can be identified at the cell surface over periods of hours or days [11]. In contrast, FC and signaling proteins


Fig. 5.1 The equilibrium between invaginated caveolae and planar lipid rafts. The model suggests that interconversion of lipid rafts and caveolae proceeds via equilibration in free-cholesterol (FC)-poor microdomains of the plasma membrane.

move in and out of caveolae much more rapidly (in some cases over a few minutes) in response to signal transduction, mitosis, and cell migration [12,13]. Caveolae are defined here as invaginated cell-surface microdomains stabilized with caveolin whose association with more labile complexes of lipids and other proteins responds to physiological stimuli. Almost all research investigations into the relationship between caveolin and FC have focused on caveolin-1.

Lipid rafts have a similar diameter to caveolae, but are planar and lack caveolin (see Fig. 5.1). Most studies of lipid rafts have been carried out in cells that lack appreciable levels of caveolin or caveolae; however, even in cells rich in caveolae, some caveolin-free lipid rafts are probably present. Glycosylphosphatidylinositol (GPI)-anchored proteins are enriched in lipid rafts, and reduced or absent in caveolae [1,2,14]. The lifetime of lipid rafts appears to be orders of magnitude less than that of caveolae. Single molecules of GPI-anchored protein move between rafts with a $t_{1/2}$ of seconds or minutes (see Chapter 3), though rafts incorporating multi-protein signaling complexes may be more stable. FC/SPH-rich microdomains with physical properties similar to those of lipid rafts form spontaneously in synthetic lipid bilayers to an extent dependent on the levels of FC and SPH, and reflecting the separation of a FC-rich liquid-ordered (L_0) phase [15–17]. In addition to GPI-anchored proteins, many acylated proteins are raft-associated. It seems likely that acylation is an important promoter of the association of signaling proteins in FC/SPH-rich raft membrane domains.

Much less is known of the organization of lipids in caveolae, but almost certainly there are significant differences from that in lipid rafts. In particular, it is not clear that the lipids of caveolae represent a L_0 phase. This need not imply that there is no exchange of lipids between caveolae and rafts, but it could involve equilibration with non-raft membrane microdomains (see Fig. 5.1). The properties of caveolae and lipid rafts are reviewed in Chapters 2, 3, and 7. Here, the focus is on the

regulatory role of FC, and in particular its interaction with signaling proteins and caveolin.

5.2 Lipids of Caveolae

The lipid composition of caveolae has been determined in only a few cases. The results of early studies in this area are difficult to interpret due to the presence of the detergents that are now recognized to modify the native distribution of lipids and proteins in membrane fractions [18]. A second problem has been the combination of caveolae and lipid rafts in a FC/SPH-rich "lipid raft" plasma membrane fraction, in which the contribution of caveolin-containing vesicles was undetermined. An additional variable has been the extent to which a "rim" of circumferential membrane and "neck" joining the caveolar bulb to the cell surface was included with the membrane "bowl" which made up the rest of the caveolar surface [1]. Recently, techniques for detergent-free isolation of caveolae from the cell surface have been described [19]. One detailed recent report is of the composition of caveolae from primary rat adipocytes [18]. These were purified from a total plasma membrane fraction by gradient ultracentrifugation, and then by immuno-precipitation with caveolin antibody.

Caveolae contain significantly higher levels of FC and sphingomyelin than do unfractionated plasma membranes. That of glycerophospholipids is essentially unchanged, so the total number of lipid molecules per unit area is increased (Fig. 5.2). The level of gangliosides is higher in caveolae than in plasma mem-



Fig. 5.2 Composition of total plasma membrane and purified caveolae from rat adipocytes. Data are expressed per 100 nm^2 plasma membrane (from [18]). FC = free cholesterol; SPH = sphingolipids; GPL = glycerophospholipids.

branes overall, though it represents only a small part (about 1%) of the total sphingolipid present. The major ganglioside of adipocytes is GM3. The same has been found for other differentiated peripheral cells rich in caveolae. GM1, though often identified as a specific marker for caveolae by its reaction with cholera toxin, in primary adipocytes was enriched only 2.6-fold in this fraction.

Additional analyses were reported from a line of human epidermal carcinoma (KB) cells stably transfected with mouse caveolin-1 cDNA [20]. The membrane fraction in this study was not immunopurified, and so probably included some caveolin-free lipid rafts. Nevertheless, as in the adipocyte study, the caveola-enriched fraction of KB cells contained increased levels of FC and sphingomyelin, relative to the rest of the plasma membrane. An unexpected finding was the enrichment there of arachidonyl ethanolamine plasmalogens, though this was also found in the raft fraction from a control, caveolin-free KB cell line.

Another unexpected finding was the rapidity with which FC moved between caveolae and other microdomains or extracellular acceptors in response to signal transduction [21]. FC was transferred readily to either cyclodextrin, a synthetic extracellular FC acceptor, or lipid-poor high-density lipoprotein, prebeta-HDL [22,23]. Dehydroergosterol, a fluorescent sterol marker for FC, was transferred more rapidly from of a caveola-rich fraction of fibroblast-derived l-cells than from other membrane domains [24]. In synthetic membranes, the effect of lowering FC is to decrease membrane stiffness and increase diffusion rates from the bilayer [25,26]. In contrast, FC depletion from caveola-rich membranes, which are linked to the sub-skeleton via the actin-family protein filamin [27] led to increased membrane stiffness [28,29].

Further evidence for a unique organization of FC in caveolae was presented in a recent study of the water permeability of caveolae from rat lung. This was 5- to 10-fold greater than that of noncaveolar membranes of comparable lipid composition [30]. Aquaporin has been reported in lung caveolae [31], but aquaporin did not significantly stimulate water flux in caveolae in this study [30]. If confirmed with caveolae from other cells, this observation could help to explain the rapid changes in FC in caveolae to physiological stimuli [21] compared to the stability of their caveolin skeleton [11]. FC-FC bonds within caveolae appear to be weaker than in other membrane microdomains.

Taken together, these data suggest that the organization of FC in caveolae and lipid rafts is fundamentally different. FC and SPH are enriched in both (relative to other membrane fractions), but within caveolae FC is labile not stabilized and the lipid bilayer is more, rather than less, permeable to water. The key factor may be that caveolin binds FC directly [31–33]. If this is correct, the properties of FC in lipid rafts would be determined mainly by its interaction with other lipids particularly SPH; in caveolae, they would be determined mainly by caveolin [13,33].

In addition to the major lipid classes (FC, sphingolipids, glycerolipids), caveolae also contain relatively high levels (compared to other plasma membrane fractions) of signaling lipids, including phosphatidic acid, diglyceride, ceramide and asialogangliosides. Most of these are probably generated *in situ* since caveolae are enriched in phospholipase D2 [34], lipid phosphate phosphohydrolase [35], sphingomyelinase and ceramidase [36,37] and sialidase [38] While the primary role of these lipids appears to be in signaling, some may also act locally to regulate the composition and properties of caveolae. Ceramide generated by extracellular bacterial sphingomyelinase reduced the interaction of FC and caveolin [39]; however, whether the low levels of bioactive lipids present under physiological conditions would be sufficient for such effects to be significant remains uncertain.

5.3 Proteins in Caveolae

Caveolar membranes contain one or more members of small (18–22 kDa), highly conserved proteins of the caveolin family of which caveolin-1 is the major structural protein of caveolae in cells other than striated muscle. It is present in two isoforms that may represent different gene transcripts [40]. The longer (α) isoform is the major species in most cells. A second family member (caveolin-2) plays a role at the cell surface that is secondary, because caveolae are not detected in caveolin-1 –/– cells, despite the expression of caveolin-2 [41]. Caveolin-3 is the major caveolin of striated muscle cells [42]. A central domain in each caveolin (represented by residues 110–132 in caveolin-1) is highly hydrophobic. Both N- and C-termini are cytoplasmic [43]. This suggests that the protein penetrates (but does not traverse) the plasma membrane bilayer.

In fixed membranes from rat adipocytes or human fibroblasts, myc-tagged caveolin-1 was present in the necks but not the bulbs of caveolae [44]. In contrast, in quick-frozen unfixed membranes from 3T3-L1 mouse fibroblastic cells, caveolin-1 was identified as a belt around the caveolar bulb, but was absent from the neck that joins it to the plasma membrane [10]. In both cell lines, the depletion of membrane FC with an extracellular acceptor (cyclodextrin) led to disassembly of caveolin multimers and flattening.

The caveolin(82–101) domain has been implicated in self-recognition of caveolin monomers, leading to assembly of a supportive structural basketwork. Based on an estimate of the predicted α -helical content of the caveolin-1 peptide including residues 79–96, it was proposed that filaments made up of caveolin heptamers formed a basketwork surrounding the caveola [45]. However, short peptides can readily assume many different conformations, and secondary structure predictions for the 79–96 aa region in the context of the full primary sequence suggest a more complex structural situation. Other evidence discussed below suggests the presence of significant tertiary structure within the N-terminal half of caveolin-1. High-resolution three-dimensional structural analysis of caveolin will almost certainly be needed to distinguish these alternatives.

The composition of multi-protein complexes containing caveolin in living cells remains a contentious issue. Several dozens of proteins covering a wide range of functions at the cell surface have been identified at one time or another in purified membranes containing caveolae. It has been difficult to prove the existence of many of these in caveolae in living cells. Several factors have contributed to this

situation. In earlier research, the range of factors that could modify native complexes in plasma membrane fractions, including detergents and cross-linking antibodies, was not fully appreciated. Major differences between cell lines in the protein composition of caveolae also were not recognized. Proteins in caveolae in primary cells may be found in caveolin-free lipid rafts or nonraft membrane domains in transformed and continuous cell lines, many of which express few, if any, caveolae [46–48]. Growth conditions can have major effects on the structure of caveolae, and on the distribution of caveolin between caveolae and other microdomains [49]. Expression of caveolin in cell lines that are normally caveolin-deficient (e.g., FRT cells) may lead to the appearance of only a few cell-surface caveolae; most of the caveolin expressed accumulates on intracellular membrane vesicles [50]. The sum of these factors means that the distribution of membrane proteins between caveolae and other cell-surface domains, and the influence of FC, is highly dependent on the cell or tissue used, technical details of fractionation, and the conditions of analysis.

This being said, certain classes of signaling proteins in tissues and primary cells have been repeatedly shown, by using several different techniques, to be reliably caveola-associated. The following criteria have been used in this chapter to assess if a multi-protein complex including caveolin represents a *bona fide* caveolar complex: co-localization with caveolin at the cell surface by confocal microscopy; recovery in caveolar vesicles in the absence of detergents; co-precipitation with caveolin antibodies from caveolae; and competitive displacement by caveolin peptides.

In addition to caveolin, caveolae contain other structural proteins which are insufficient by themselves to support membrane invagination, but are possibly important for the biological properties of these domains. Flotillins are a conserved protein family first identified as co-precipitates with caveolin in cells that had been extracted with Triton X-100 and octyl glucoside. Most recent research identifies these proteins mainly in association with GPI-anchored proteins in caveolin-free lipid rafts [51]. Annexins are Ca²⁺-binding proteins. Annexin-2 forms a stable cytoplasmic chaperone complex with caveolin-1 in mammalian cells in culture and in zebrafish *in vivo* (see Chapter 8). However, the inclusion of these proteins in caveolae at the cell surface has not yet been conclusively shown.

It is a characteristic of many differentiated peripheral cells that mitosis is strongly dependent on the activity of protein growth factors. These serve as ligands for transmembrane receptors. The protein kinase activity of these receptors initiates signaling cascades that terminate in the activation of nuclear transcription factors. The anchorage-independent growth of virally-transformed and cancer cells reflects their independence from exogenous growth factors, consistent with the frequent reduction of caveolin in cancer cells, and the role proposed for caveolae in growth control and cell attachment [49,52]. Hyperplasia reported in multiple tissues in caveolin-1 -/- mice [6,7] similarly reflects the impairment of normal growth regulation.

Transmembrane signaling kinases have been localized to caveolae in many different peripheral cell lines, using many different criteria. Receptor proteins for the platelet-derived, epidermal and vascular endothelial growth factors (PDGF, EGF, VEGF) co-purified with the caveolar membrane fraction, and were co-precipitated with caveolin-1 antibodies. PDGFR co-purified with caveolae in 3T3 cells [53,54], normal human fibroblasts [55] and human vascular smooth muscle cells [50]. EGFR co-purified with caveolae in primary human fibroblasts [55] and in a mouse fibroblast (3T3) line [56], but not laryngeal (Hep-2) or epidermal (A431) cancer cell lines [57,58]. VEGFR co-purified with caveolae and caveolin in both bovine aortic and human umbilical endothelial cells [59–61].

Additional evidence of a structural and functional association of caveolin with growth factors comes from studies of its (Y14)-phosphorylation, mediated by non-receptor c-Src family kinases, which are themselves substrates for the tyrosine kinase activities of PDGFR, EGFR, and VEGFR (see Chapter 6). These data provide strong support for the existence of functional complexes between caveolin-1, transmembrane kinases, and nonreceptor signaling kinases such as c-Src and Fyn in living cells [62]. Membrane transporter proteins are also frequently recovered in preparations of caveolae. Glucose (Glut-4) transporters were localized to caveolae in 3T3 cells [63]. Potassium (K+(KATP), Na+, and Ca²⁺ channels were found in this fraction in SMC [64,65].

ATP-binding cassette transporter-A1 (ABCA1) ferries phospholipids and possibly FC across cell membranes, but it was not found with caveolin in detergentextracted membranes [66]. However, the opposite result was more recently obtained, under detergent-free conditions, using the plasma membrane fraction of rat endothelial cells, after purification with caveolin antibody [67]. These contrasting data illustrate the ambiguity in protein composition data of caveolae from detergent-treated membranes.

Another recent study showed that a second ABC transporter (p-glycoprotein) can regulate the distribution of FC across the membrane bilayer [68]. The localization of p-glycoprotein to caveolae in primary tissues and cells has been shown in several studies. [47,69,70]. In contrast, in 2780AD ovarian carcinoma cells (which lack caveolae completely), p-glycoprotein was recovered with lipid rafts [71].

Several other proteins able to modify the level of plasma membrane FC have been identified less consistently in caveolae. Scavenger receptor BI (SR-BI) catalyzes the selective uptake of cholesteryl esters and FC from HDL. It also plays a key physiological role in supplying substrate for steroid hormone production by adrenal and gonadal cells. SR-BI was identified in the caveolae of both overexpressing Chinese hamster ovary (CHO) cells and mouse adrenal (Y1) cells on the basis of immunofluorescence co-localization [72]. A second study using SV-40 transformed human fibroblasts (WI38-VA13 cells) found little SR-BI in caveolae; the majority of this protein was in microvilli of the cell surface [73]. The transformed cells contained relatively few caveolae. In adrenal cells in vivo, SR-BI was mainly associated with a novel class of double-membrane channels associated with microvilli containing few, if any, caveolae [74]. The association of SR-BI with caveolae, when present, remains unsettled at this time. In contrast, the beta-subunit of ATP synthase, identified as a high-density lipoprotein (HDL) binding protein, was localized to caveolae, and immunoprecipitated with caveolin [75]. These data, and that relating to ABCA1 [67], suggest that caveolae may be able to regulate their own FC content.

One particularly well-characterized complex of caveolar lipids and proteins is that of FC with caveolin and endothelial nitric oxide synthase (eNOS) Numerous studies have shown the co-purification of eNOS and caveolin, and its inhibition of this activity by caveolin peptide [76–80] (see also Chapter 11).

Several proteins which control growth and differentiation by other pathways are also present in caveolae. These include receptors for bone morphogenetic proteins [81] and the cell-surface receptor proteins for estrogen and vitamin D, identical to the transcription factors regulated by these hormones in the nucleus [82,83].

To summarize, the distribution of proteins between different cell-surface domains in living cells is not maintained after detergent extraction. Even in the absence of detergents, the distribution of these proteins in continuous cell lines may be modified from that found in intact tissues and primary cells. Nevertheless, a subset of proteins is reproducibly associated with caveolae in tissues and primary cells. These include transmembrane signaling kinases, eNOS, and some membrane transporters. These caveola-associated proteins are characterized both by caveolin-binding, and a marked dependence on FC. The structural and functional relationship between these properties, and its implications, are considered in the following sections.

5.4

The Caveolin Scaffold Hypothesis

The scaffold hypothesis is a widely cited early interpretation of the link between caveolin and caveola-associated proteins [84]. This hypothesis proposes that proteins associated *in vivo* with caveolin can be identified by a "scaffold recognition" sequence that is responsible for their co-assembly, and identifies the protein-protein contact domain. Based on the selectivity of a peptide library, the scaffold sequence motif was identified as $-\Phi xx\Phi xxx\Phi$, $\Phi xxxx\Phi$ are or $-\phi x\phi xxxx\phi$, where Φ is an aromatic amino acid (F, W, or Y) and x is any residue. The orientation of the signature aromatic amino acids in these linear motifs varies, depending on whether it forms part of a loop, beta-sheet or alpha-helical sequence within the tertiary structure of the native protein.

In terms of this definition, caveolin itself contains a scaffold sequence or rather, two overlapping sequences (Fig. 5.3A). One of these overlaps a sequence (-K₉₆YW-FYR-) which is implicated in membrane binding ("membrane attachment sequence") [85]. GFP-caveolin peptide lacking this sequence retains about 50% of membrane binding. One scaffold sequence is mainly predicted as helix, the other as beta-sheet. The cav(82–101) domain also includes a "cholesterol recognition amino acid consensus (CRAC-1)" sequence -L/V(x_{1-4})Y (x_{1-4})K/R- [86] that partially overlaps the "scaffold" sequences.



B. FUNCTIONAL SEQUENCES IN CAVEOLIN - 1

Fig. 5.3 (A) Location of predicted scaffold motifs and FC recognition sites of the cav(82-101) domain of human caveolin-1 (from [84,86]. (B) Location of amino acid residues needed for association G1 α , eNOS, and FC with caveolin (from [80,82,85,96]).

5.4.1

Does the Scaffold Motif in Signaling Proteins that are Present in Caveolae Represent the Contact Site of these Proteins with Caveolin?

In the eNOS complex with caveolin (which is perhaps the best-studied example), the scaffold sequence is $-F_{347}$ SAAPFSGW-. This was recently identified as part of an internal loop stabilizing the heme prosthetic group [87], and so it is unlikely to be accessible to caveolin [80]. $-F_{656}$ SYVNPQF- in protein kinase C is within a closed conformation of the V5/C3/4 domain [88]. The scaffold sequence in endothelin receptor-A (ER-A, -WPFDHNDFGVF-) is part of extracellular loop-1 [89]. Since caveolin lacks an extracellular domain [43], the scaffold motif in ER-A cannot represent a functional protein-protein contact site. In other cases, structural information is inadequate to assign the location of the scaffold motif within the three-dimensional structure of the protein. Overall, however, the experimental data are lacking that, in a native complex of caveolar proteins, the "scaffold sequence" identifies a site that binds caveolin.

 Table 5.1 Peptide contributions to free-cholesterol (FC)-binding sites.

Protein	Sequence	Reference
NPC2 ^a	-V97KEEYPSIK-	91
StARD4 ^a	-L210ANFYSDLRK-	92
RORα ^a	-L285ETCQYLR-	93
Benzodiazepine receptor ^b	-L150NYCVWR-	94
HIV-1 Nef ^b	-L HPEYYK-	95
Caveolin-1 ^b	-V93TKYWFYR-	96

^a From X-ray crystallography.

^b From FC or sterol analogue-binding.

Another test of the scaffold hypothesis for signaling proteins can be made. Does the presence or absence of a scaffold sequence predict whether a cell membrane protein binds to caveolin? In Table 5.1, the frequency of scaffold sites was seen to be one per 541 amino acids within the primary sequence of caveola-associated signaling proteins. However, a quite similar incidence (one in 493 residues) was present in lipid-associated proteins not associated with caveolin or caveolae (NPC-1, HMG-CoA reductase, lecithin:cholesterol acyltransferase, acyl CoA cholesterol acyltransferase, clathrin heavy chain) (sequences data from GenBank). This finding argues strongly against the diagnostic significance of the scaffold motif in signaling proteins.

Two rather different questions relate the F,W,Y-rich "scaffold" sequences in caveolin.

- Is the domain of caveolin that includes these sequences the contact site with signaling proteins in caveolae?
- If so, are the aromatic amino acids that define the scaffold sequence necessary for this contact?

There can be little doubt that the central domain of caveolin (cav82–101) is somehow important in both promoting protein-protein and protein-lipid associations in caveolae. Evidence supporting this includes its ability both *in vitro* [90] and *in vivo* [78] to compete with caveolin to affect signal protein binding and activity (Table 5.1). But is it the pattern of three aromatic residues that is important?

Inconsistent with the scaffold hypothesis, alanine-scanning mutagenesis showed that an intact scaffold sequence was not required for caveolin-derived peptides to bind to G1 α [84]. Within the cav(82–101) domain, only residues -F₉₂TVT-were needed for binding activity (Fig. 5.3B). Nor was a complete scaffold sequence required to compete with caveolin in displacing eNOS [80]. In a third case, that of adenyl cyclase [90], a different pattern of amino acids, including elements of the FC recognition site, was needed for caveolin binding.

Though information of this same kind is needed for additional proteins, the data so far argue against the significance of a meaningful "scaffold sequence." Proteins bound to caveolin are no more likely to include a scaffold sequence than other proteins. The scaffold sequences in caveolin appears to be parts of a broader pattern of amino acids within the cav(82–101) domain responsible both for the association of caveolin with signaling proteins, and with FC.

5.5 FC Binding by Proteins Including Caveolin

In contrast to reviews of protein-protein and protein-DNA binding, very few studies have been made of the structural aspects of FC binding to proteins. The sum of evidence available suggests that FC lies within a hydrophobic cleft or tunnel, the sides of which are composed of several different hydrophobic sequences organized by the tertiary structure of the native protein. A positively charged amino acid (K or R) may contribute to a "charge clamp" stabilizing the 3β-OH group. Whilst the volume of FC has been estimated at 741 Å³ [91], its binding cavity is usually about twice as large. There is evidence in some instances that FC itself induces significant changes in the shape and size of this cavity. In the case of many of the FCbinding proteins for which a tertiary structure has been established, one of the hydrophobic sequences contributing to the FC-binding cavity conforms to the "cholesterol recognition" consensus motif (-V/L_{x1-4} Y_{x1-4} K/R-) [86] (Table 5.1).

The caveolin central domain (cav82–101) also contains a cholesterol recognition sequence (CRAC-1) (Fig. 5.3B) that binds FC [96]. The isolated peptide also binds FC from FC/phosphatidylcholine vesicles to an extent that exceeds FC solubility in the lipid bilayer. This result, based on tryptophan fluorescence and NOESY data, indicates the ability of the peptide to induce macromolecular order in FC, possibly by stabilizing FC-FC links [97]. The sequence required for FC binding (-V94TKYWFYR-) abuts the residues in cav(82–101) needed for interaction with signal proteins Gi2 α [80] and eNOS [76]. A key question considered later is whether the binding of protein and FC to the same domain is complementary or competitive.

Although residues 94–101 within the caveolin(83–102) peptide promote binding with synthetic FC-containing liposomes, it seems likely that additional hydrophobic residues contribute to membrane binding by caveolin *in vivo*. Phosphorylation at S80 has a substantial effect on FC binding [50], suggesting a role for neighboring hydrophobic residues within the sequence -S80FDGLW-. In contrast, phosphorylation at Y14 completely inhibited FC binding, which may indicate that the short hydrophobic sequence -L13YTV- may also contribute. Though the data remain limited, it seems likely that the structural basis of FC binding in caveolin resembles that in other proteins.

Do other proteins present in caveolae bind FC? A number of caveolin-associated proteins crosslink to the photoactivable FC analogue FCBP [13], and many caveolin-associated proteins including eNOS and PDGFR include one or more "cho-

Activity	Effect of FC \downarrow	Addition of cav (82–101)	Reference(s)
Neutral ceramidase	↑		109
Adenyl cyclase	Ļ	Ļ	90
Ca ²⁺ -act K ⁺ channel	↑		64
TGFbeta-1 receptor	↑		110
Ca ²⁺ -dependent PLA2	Ļ	Ļ	111, 112
Ca ²⁺ -ATPase	↓		113
ERK1/2	Ļ		114
p-glycoprotein	↓		115
Agonist stim cAMP accum	↑		116
Insulin receptor	Ļ		117
eNOS	↓	Ļ	80, 118
PKC α translocation	↓	Ļ	119, 120

Table 5.2 Effects of FC depletion (\downarrow) and cav (82–101) on activities of caveolar proteins.

FC was depleted from cultured cells with beta-methyl cyclodextrin. Cav (82–101) was determined either *in vitro* in cultured cell homogenates, or *in vivo* as a complex with a cell-permeable antennopedia peptide.

lesterol recognition" motifs within the primary sequence, though at present experimental data showing if these sequences are involved in FC binding are lacking.

5.6

FC in Caveolae: Effects of Depletion and Loading

It is characteristic of proteins in caveolae that their activities are very sensitive to membrane levels of FC. In some cases these activities are increased, in other cases inhibited, when FC is depleted (Table 5.2). Addition of the cav(82–101) peptide – either *in vitro* or *in vivo* – spliced to an antennopedia (AP) "Trojan peptide" [78] also affects the activity of caveola-associated signaling proteins. In those examples where comparative data are available, the addition of peptide (at 10–100 μ M) in each case mimics the effect of FC depletion. Is this coincidence, or informative about the role of FC in this system?

One way to look at the role of FC in caveolar signaling is to modify the level of FC, and to determine the effects of this on the activity of caveola-associated proteins. Much of the research on the influence of FC on caveolae has been carried out with cyclodextrins, cyclic oligosaccharides containing six to eight α -1,4-linked-glycopyranoside units enclosing a central hydrophobic tunnel [98]. These bind



Fig. 5.4 Alternative mechanisms for the effects of caveolin(82–101) peptide on functional multiprotein complexes in caveolae. Mechanisms shown: A = displacement of caveolin from signaling kinase (PDGFR) by

anteonnopedia (AP)-linked caveolin(82–101) peptide. B = displacement of PDGFR from its association with caveolin by AP peptide binding to caveolin. C = the same AP-peptide acting as a sink for caveolar FC.

small hydrophobic molecules in a reaction depending on the replacement of core water molecules by the organic ligand. β -Methyl and β -hydroxypropyl cyclodextrins have an internal cavity of ~0.8 nm. They interact with FC, but not with other common membrane lipids. Consequently, FC-free cyclodextrins are effective and selective receptor-independent acceptors of cell FC by diffusion. Conversely, preformed cyclodextrin-FC complexes, or low-density lipoprotein (LDL), can donate FC to increase levels of FC-rich microdomains, including caveolae. Decreased FC mediated by cyclodextrin led to the dissociation of signaling proteins from caveolae, disassembly of caveolin multimers [10] and eventually, to a reduction in the rate of caveolin synthesis by transcriptional down-regulation [99]. In human primary fibroblasts and smooth muscle cells the chronic effects of FC loading included an increase in cell-surface caveolae by transfer of caveolin from intracellular pools [49], increased endocytosis of FC-rich vacuoles [100] and, over time, the induction of caveolin synthesis at the transcriptional level by a FC-sensitive promoter site [101,102]. In human primary endothelial cells, an increase in caveolin at

the cell surface in response to LDL was drawn mainly from intracellular pools [103]. These data suggest that the presence of FC is closely linked to both the structure and function of caveolae.

Three different mechanisms could give rise to the displacement by caveolin peptide(82–101) of signaling proteins from their complex with caveolin (Fig. 5.4):

- 1. Caveolin peptide(82–101) competes with the target kinase, binding caveolin and generates free kinase.
- 2. Caveolin peptide(82-101) competes with caveolin to bind the target kinase.
- 3. Caveolin peptide(82–101) binds FC competitively, displacing it from caveolin, secondarily dissociating the caveolin-kinase complex.

Since peptides cav(1–101) and cav(1–79) polymerize similarly *in vitro* [45], FCbinding to cav (82–101) [96] is unlikely to play a direct role in caveolin self-assembly. These data argue against the first hypothesis. Since the cav(82–101) peptide mimics caveolin and, like the full-length protein, binds FC, it is difficult to envisage a mechanism by which cav(82–101) would reverse the effect of native caveolin. Other evidence supports the hypothesis that cav(82–101) may mimic cyclodextrin by binding FC competitively. The relatively high concentrations of this freely internalized peptide required for maximal effect (50–100 μ M) would be adequate to bind significant amounts of plasma membrane FC. F₉₂ of intact caveolin is necessary for the effect of peptide on eNOS [80], and the same residue is required for FC binding [96]. We suggest that the scaffold peptide cav(82–101) may act at least in part to competitively sequester caveolar FC.

5.7

FC Changes in Caveolae: Effects of Signal Transduction

The metabolic effects of chronic changes in the FC content of the caveolae can be explained in terms of competition between binding of protein and binding of FC to overlapping sequences within a central (resides 82–101) domain of caveolin, However the content of FC in caveolae also responds both spatially and temporally to physiological changes at the cell surface, in particular to the binding of protein growth factors. Could these changing FC levels be involved in regulating the magnitude and duration of signal transduction and possibly, in the case of branching pathways, its selectivity?

The mechanism of signal transduction via protein growth factors and transmembrane receptor kinases is well understood in outline. Ligand binding is coupled to dimerization of the receptor protein, receptor autophosphorylation, and induction of its activity with downstream kinases, particularly proteins of the c-Src family (see Chapter 6). This leads to dissociation of signaling proteins from the cell surface and eventually, to transcriptional activation of a specific pattern of genes important in mitosis. A consistent feature of ACTH, PDGF, -EGF, VEGF, insulin and insulin-like growth factor-mediated signaling is that receptor autophosphorylation is followed by the generation of $p(Y_{14})$ caveolin, the synthesis of which is



phosphatase (from [121]).

mediated by c-Src or a Src-family kinase such as Fyn [50,104–108]. This observation provides *prima facie* evidence for the existence of signaling complexes containing bound caveolin.

In the case of PDGFR, signal transduction was recently shown to be associated with a loss of FC binding to caveolin, assayed either by co-purification of ³H-FC with caveolin, or by the formation of stable crosslinks between caveolin and a photoactivable FC analogue, FCBP [50]. A comparable change occurs when VEGF activates its receptor in human aortic endothelial cells, and confirms that the associated with PDGFR, a maximum of 70–80% of caveolin-associated FC was lost in the course of signal transduction. This is consistent with the concept that FC bound to caveolin has higher-order effects on membrane FC in caveolae. Reassembly of the PDGFR signaling complex requires the re-establishment of FC levels in the caveola (Fig. 5.5). Studies of the time course of signaling in the presence and absence of inhibitors of specific kinases and phosphatases, and substitution within caveolae of mutant caveolins with altered polarity and FC binding properties, were carried out.

The yield of caveolin-associated ³H-FC or -FCBP under different conditions was assayed. Recovery of the FC content of caveolae following exposure to PDGF took 30–40 min at 37 °C. The cell became responsive again to PDGF only after the FC

content of its caveolae was completely recovered. If the activity of the caveolaassociated low molecular-weight phosphotyrosine phosphatase was inhibited by hydrogen peroxide [121], p(Y₁₄)caveolin accumulated and the caveola did not recover its reactivity with PDGF. If caveolin(S80A), which binds FC more tightly than the native protein [50] was overexpressed, then although PDGF bound normally to its receptor, FC was not displaced from this caveolin, and the signal was not propagated from the PDGF receptor (PDGFR) to c-Src and to caveolin. These data suggest that, at different steps, FC has positive and negative roles. The presence of FC is required for the assembly of the transmembrane signaling complex; after ligand binding, loss of FC is required for signal transduction. Following signal transduction, FC re-enters the caveola. These observations are incorporated into Fig. 5.5:

- A signaling complex containing at a minimum caveolin, FC, a transmembrane receptor kinase, a Src-family non-receptor kinase is first assembled within caveolae. At least in SMC, caveolin is constitutively phosphorylated at S₈₀, influencing FC binding to the cav(82–101)
- 2. Addition of ligand leading to receptor and downstream receptor kinase activation displaces FC from its binding site. Mutant Caveolin(S80A), which cannot be phosphorylated, does not release FC and signal transduction is inhibited.
- 3. Loss of FC from caveolin precedes p(Y₁₄)-caveolin synthesis mediated by c-Src or Src-family kinase. Since the level of p-caveolin, which is normally low, is dynamically regulated by caveolar LMW-PTP, the rise in p-caveolin which follows the loss of FC must be the result of either the dissociation of this phosphatase, or its inhibition *in situ*. These alternatives have not yet been distinguished. p-Caveolin synthesis in response to PDGF was not associated with any detectable disassembly of caveolin oligomers, or withdrawal of caveolin from the cell surface. This is in contrast to what occurs when p(Y₁₄)-caveolin is chronically increased for example by vanadate, a nonspecific phosphotyrosine phosphatase inhibitor.
- 4. After dissociation of both FC and downstream kinases, LMW-PTP is reactivated, and reduces p-caveolin levels to baseline. Levels of total caveolin continue unchanged.
- 5. FC levels in caveolae (and those of FCBP in cells equilibrated with this photoactivable FC analogue) are restored only after p-caveolin is hydrolyzed. In cells exposed to peroxide, phosphocaveolin levels were maintained at their maximum; FC was not restored to caveolae, and the cells remained insensitive to PDGF.
- 6. Finally (at ca. 30 min following initial exposure to PDGF) the cell regains responsiveness to ligand.

These data indicate that the overall equilibrium between FC-associated and FCfree caveolin in caveolae is influenced by phosphorylation of both receptor protein and caveolin at different points in the signaling reaction.

5.8 Summary

FC binding to caveolin plays important structural and functional roles in caveolae. Structurally, it is needed to maintain the invaginated form of these microdomains. Functionally, FC sensitively regulates the activity of caveolin-associated proteins including signaling kinases, transporters and eNOS. At the molecular level, these effects are mediated by competition between adjacent caveolin-protein and caveolin-FC binding sites.

Abbreviations

ABCA1	ATP-binding cassette transporter-A1
AP	antennopedia
СНО	Chinese hamster ovary
CRAC	cholesterol recognition amino acid consensus
EGF	epidermal growth factor
eNOS	endothelial nitric oxide synthase
ER-A	endothelin receptor-A
FC/SPH	free cholesterol/sphingolipid
GPI	glycosylphosphatidylinositol
HDL	high-density lipoprotein
PDGF	platelet-derived growth factor
PDGFR	PDGF receptor
SR-BI	scavenger receptor BI
VEGF	vascular endothelial growth factor

References

- 1 Schnitzer JE, McIntosh DP, Dvorak AM, Liu J and Oh P (1995) Separation of caveolae from associated microdomains of GPI-anchored proteins. *Science* 269, 1435–1439.
- 2 Abrami L, Fivaz M, Kobayashi T, Kinoshita T, Parton RG and van der Groot FG (2001) Cross-talk between caveolae and glycosylphosphatidylinositol-rich domains. J. Biol. Chem. 276, 30729–30736.
- **3** Sowa G, Pypaert M and Sessa WC (2001) Distinction between signaling mechanisms in lipid rafts vs caveolae. *Proc. Natl. Acad. Sci. USA* 98, 14072–14077.
- 4 Oh P, McIntosh DP and Schnitzer JE (1998) Dynamin at the neck of caveolae mediates their budding to form transport

vesicles by GTP-driven fission from the plasma membrane of endothelium. *J. Cell Biol.* 141, 101–114.

- 5 Griffoni C, Spisni E, Santi S, Riccio M, Guarnieri T and Tomasi V (2000) Knockdown of caveolin-1 by antisense oligonucleotides impairs angiogenesis in vitro and in vivo. *Biochem. Biophys. Res. Commun.* 276, 756–761.
- **6** Drab M, Verkade P, Elger M, Kasper M, Lohn M, Lauterbach B, Menne J, Lindschau C, Mende F, Luft FRC, Schedl A, Haller H, Kurzchalia TV (2001) Loss of caveolae, vascular dysfunction, and pulmonary defects in caveolin-1 gene disrupted mice. *Science* 293, 2449– 2452.

- 108 References
 - 7 Razani B, Engelman JA, Wang XB, Schubert W, Zhang XL, Marks CB, Macaluso F, Russell RG, Li M, Pestell RG, Di Vizio D, Hou H, Kneitz B, Lagaud G, Christ GJ, Edelmann W and Lisanti MP (2001) Caveolin-1 null mice are viable but show evidence of hyperproliferative and vascular abnormalities. *J. Biol. Chem.* 276, 38121–38138.
 - 8 Parpal S, Karlsson M, Thorn H, Stralfors P (2001) Cholesterol depletion disrupts caveolae and insulin receptor signaling for metabolic control via insulin receptor substrate-1, but not for mitogen-activated protein kinase control. *J. Biol. Chem.* 276, 9670–9678.
 - 9 Dreja K, Volstedlind M, Vinten J, Tranum-Jensen J, Hellstrand P and Sward K (2002) Cholesterol depletion disrupts caveolae and differentially impairs agonistinduced arterial contraction. *Arterioscler. Thromb. Vasc. Biol.* 22, 1267–1272.
 - 10 Westermann M, Steiniger F and Richter W (2005) Belt-like localization of caveolin in deep caveolae and its redistribution after cholesterol depletion. *Histochem. Cell Biol.* May 12, e-pub.
 - 11 Thomsen P, Roepstorff K, Shahlhut M and van Deurs B (2002) Caveolae are highly immobile plasma membrane microdomains, which are not involved in constitutive endocytic trafficking. *Mol. Cell Biol.* 13, 238–250.
 - 12 Liu P, Wang P, Michaely P, Zhu M and Anderson RGW (2000) Presence of oxidized cholesterol in caveolae uncouples active platelet-derived growth factor receptors from tyrosine kinase substrates. J. Biol. Chem. 275, 31648–31654.
 - 13 Fielding PE, Russel JS, Spencer TA, Hakamata H, Nagao K and Fielding CJ (2002) Sterol efflux to apolipoprotein A-1 originates from caveolin-rich microdomains and potentiates PDGF-dependent protein kinase activity. *Biochemistry* 41, 4929–4937.
 - 14 Kirkham M, Fujita A, Chadda R, Nixon SJ, Kurzchalia TV, Sharma DK, Pagano RE, Hancock JF, Mayor S and Parton RG (2005) Ultrastructural identification of uncoated caveolin-independent early endocytic vesicles. J. Cell Biol. 168, 465–476).
 - 15 Dietrich C, Volovyk ZN, Levi M, Thompson NL and Jacobson K (2001) Partition-

ing of Thy-1, GM1 and cross-linked phospholipid analogs into lipid rafts reconstituted in supported model membrane monolayers. *Proc. Natl. Acad. Sci. USA* 98, 10642–10647.

- 16 Dietrich C, Bagatolli LA, Vovovyk ZN, Thompson NL, Levi M, Jacobson K and Gratton E (2001) *Biophys. J.* 3, 1417–1428.
- 17 Sheets ED, Lee GM, Simson R and Jacobson K (1997) Transient confinement of a glycophosphatidylinositol-anchored protein in the plasma membrane. *Biochemistry* 41, 12449–12458.
- 18 Ortegren U, Karlsson M, Blazic N, Blomqvist M, Nystrom FH, Gustavsson J, Fredman P and Stralfors P (2004) Lipids and glycosphingolipids in caveolae and surrounding plasma membrane of primary adipocytes. *Eur. J. Biochem.* 271, 2028–2036.
- 19 Smart EJ, Ying YS, Mineo C, Anderson RGW (1995) A detergent-free method for purifying caveolae membrane from tissue culture cells. *Proc. Natl. Acad. Sci. USA* 92, 10104–10108.
- 20 Pike LJ, Han X, Chung KN and Gross RW (2002) Lipid rafts are enriched in arachidonic acid and plasmenylethanolamine and their composition is independent of caveolin-1 expression: a quantitative electrospray ionization/mass spectrometric analysis. *Biochemistry* 41, 2075–2088.
- Fielding CJ and Fielding PE (2004) Membrane cholesterol and the regulation of signal transduction. *Biochem. Soc. Trans.* 32, 65–69.
- 22 Dreja K, Voldstedlund M, Vinten J, Tranum-Jernsen J, Hellstrand P and Sward K (2002) Cholesterol depletion disrupts caveolae and differentially impairs agonistinduced arterial contraction. *Arterioscler. Thromb. Vasc. Biol.* 22, 1267–1272.
- **23** Fielding PE and Fielding CJ (1995) Plasma membrane caveolae mediate the efflux of cellular free cholesterol. *Biochemistry* 34, 14288–14292.
- 24 Gallegos AM, McIntosh AL, Atshaves BP and Schroeder F (2004) Structure and cholesterol domain dynamics of an enriched caveolae/lipid raft isolate. *Biochem.* J. 382, 451–461.
- 25 Song J and Waugh RE (1993) Bending rigidity of SOPC membranes containing cholesterol. *Biophys. J.* 64, 1967–1970.

- 26 Lundback JA, Birn P, Girsshman J, Hansen AJ and Anderson OS (1996) Membrane stiffness and channel function. *Biochemistry* 35, 3825–3830.
- 27 Stahlhut M and van Deurs B (2000) Identification of filamin as a novel ligand for caveolin-1: evidence for the organization of caveolin-associated membrane domains by the actin cytoskeleton. *Mol. Biol. Cell* 11, 325–337.
- 28 Kwik J, Boyle S, Fooksman D, Margolis L, Sheetz MP and Edidin M (2003) Membrane cholesterol, lateral mobility, and the phosphatidylinositol 4,5-bisphosphate-dependent organization of cell actin. Proc. Natl. Acad. Sci. USA 24, 13964–13969.
- 29 Byfield FJ, Aranda-Espinoza H, Romanenko VG, Rothblat GH and Levitan I (2004) Cholesterol depletion increases membrane stiffness of aortic endothelial cells. *Biophys. J.* 87, 3336–3343.
- 30 Hill WG, Almasri E, Ruiz WG, Apodaca G and Zeidel ML (2005) Water and solute permeability of rat lung caveolae: high permeabilities explained by acyl chain unsaturation. Am. J. Physiol. Cell Physiol. 289, C33-C41.
- **31** Schnitzer JE and Oh P (1996) Aquaporin-1 in plasma membrane and caveolae provides mercury-sensitive water channels across lung endothelium. *Am. J. Physiol.* 270, H416-H422.
- 32 Murata M, Peranen J, Schriener R, Wieland F, Kurzchalia TV and Simons K (1995) VIP21/caveolin is a cholesterolbinding protein. *Proc. Natl. Acad. Sci.* USA 92, 10339–10343.
- 33 Li S, Song KS and Lisanti MP (1996) Expression and characterization of recombinant caveolin. Purification by polyhistidine tagging and cholesterol-dependent incorporation into defined lipid membranes. J. Biol. Chem. 271, 568–573.
- **34** Cho CH, Lee CS, Chang M, Jang IH, Kin SJ, Hwang I, Ryu SH, Lee CO and Koh GY (2004) Localization of VEGFR-2 and PLD-2 in endothelial caveolae is involved in VEGF-induced phosphorylation of MEK and ERK. *Am. J. Physiol.* 286, H1881–H1888.
- 35 Nanjundan N and Possmayer F (2001) Pulmonary lipid phosphate phosphohydrolase in plasma membrane signaling platforms. *Biochem. J.* 358, 637–646.

- 36 Veldman RJ, Maestre N, Aduib OM, Medin JA, Salvayre R and Levade T (2001) A neutral sphingomyelinase residues in sphingolipid-enriched microdomains and is inhibited by the caveolinscaffolding domain: potential implications in tumor necrosis factor signaling. *Biochem. J.* 355, 859–868.
- **37** Romiti E, Meacci E, Tanzi G, Becciolini L, Mitsutake S, Farnararo M, Ito M and Bruni P (2001) Localization of neutral ceramidase in caveolin-enriched light membranes of murine endothelial cells. *FEBS Lett.* 506, 163–168.
- 38 Wang Y, Yamaguchi K, Wada T, Hata K, Zhao X, Fujimoto T and Miyagi T (2002) A close association of the ganglioside-specific sialidase Neu3 with caveolin in membrane microdomains. J. Biol. Chem. 277, 26252–26259.
- **39** Yu C, Alterman M and Dobrowsky RT (2005) Ceramide displaces cholesterol from lipid raft membranes and decreases the association of the cholesterol binding protein caveolin-1. *J. Lipid Res.* May, e-pub.
- 40 Kogo H and Fujimoto T (2000) Caveolin-1 isoforms are encoded by distinct mRNAs. Identification of mouse caveolin-1 mRNA variants caused by alternative transcription initiation and splicing. *FEBS Lett.* 465, 119–123.
- **41** Prolini I, Sargiacomo M, Galbiati F, Rizzo G, Grignani F, Engelman JA, Okamoto T, Ikezu T, Scherer PE, Mora R, Rodriguez-Boulan E, Peschle C and Lisanti MP (1999) Expression of caveolin-1 is required for the transport of caveolin-2 to the plasma membrane. Retention of caveolin-2 at the level of the Golgi complex. *J. Biol. Chem.* 274, 25718–25725.
- **42** Way M and Parton RG (1995) M-caveolin, a muscle-specific caveolin-related protein. *FEBS Lett.* 376, 108–112.
- **43** Dupree P, Parton RG, Raposo G, Kurzchalia TV and Simons K (1993) Caveolae and sorting in the trans-Golgi network of epithelial cells. *EMBO J*. 12, 1597–1605.
- 44 Thorn H, Stenkula KG, Karlsson M, Ortegren U, Nystrom FH, Gustavsson J and Stralfors P. Cell surface orifices of caveolae and localization of caveolin to the necks of caveolae in adipocytes. *Mol. Biol. Cell* 14, 3967–3976.

- 110 References
 - 45 Fernandez I, Ying Y, Albanesi J and Anderson RGW (2002) Mechanism of caveolin filament assembly. *Proc. Natl. Acad. Sci. USA* 99, 11193–11198.
 - 46 Koleske AJ, Baltimore D and Lisanti MP (1995) Reduction of caveolin and caveolae in oncogenically transformed cells. *Proc. Natl. Acad. Sci. USA* 92, 1381–1385.
 - 47 Lavie Y, Fiucci G and Liscovitch M (1998) Up-regulation of caveolae and caveolar constituents in multi-drug resistant cancer cells. J. Biol. Chem. 273, 32380–32383.
 - **48** Lee SW, Reimer CL, Oh P, Campbell DB and Schnitzer JE (1998) Tumor cell growth inhibition by caveolin re-expression in human breast tumor cells. *Oncogene* 16, 1391–1397.
 - 49 Thyberg J (2002) Caveolae and cholesterol distribution in vascular smooth muscle cells of different phenotypes. J. Histochem. Cytochem. 50, 185–195.
 - 50 Fielding PE, Chau P, Liu D, Spencer TA and Fielding CJ (2004) Mechanism of platelet-derived growth factor-dependent caveolin-1 phosphorylation: relationship to sterol binding and the role of serine-80. *Biochemistry* 43, 2578–2586.
 - 51 Stuermer CA and Plattner H. (2005) The 'lipid raft' microdomain proteins reggie-1 and reggie-2 (flotillins) are scaffolds for protein integration and signaling. *Biochem. Soc. Symp.* 72, 109–118.
 - 52 Fielding CJ, Bist A and Fielding PE (1999) Intracellular cholesterol transport in synchronized human skin fibroblasts. *Biochemistry* 38, 2506–2513.
 - 53 Yamamoto M, Toya Y, Jensen RA and Ishikawa Y (1999) Caveolin is an inhibitor of platelet-derived growth factor receptor signaling. *Exp. Cell Res.* 247, 380–388.
 - 54 Matveev SV and Smart EJ (2002) Heterologous desensitization of EGF receptors and PDGF receptors by sequestration in caveolae. Am. J. Physiol. 282, C935–C946.
 - 55 Liu P, Ying Y, Ko YG and Anderson RGW (1996) Localization of platelet-derived growth factor-stimulated phosphorylation cascade to caveolae. J. Biol. Chem. 271, 10299–10303.
 - 56 Park WY, Cho KA, Park JS, Kim DI and Park SC (2001) Attenuation of EGF signaling in senescent cells by caveolin. *Ann. N. Y. Acad. Sci.* 928, 79–84.
 - 57 Ringerike T, Blystad FD, Levy FO, Madshus IH and Stang E (2002) Cholesterol

is important in control of EGF receptor kinase activity but EGF receptors are not concentrated in caveolae. *J. Cell Sci.* 115, 1331–1340.

- 58 Ringerike T, Blystad FD, Levy FO, Madshus IH and Stang E (2002) Cholesterol is important in control of EGF receptor kinase activity but EGF receptors are not concentrated in caveolae. J. Cell Sci. 115, 1331–1340.
- 59 Labrecque L, Royal I, Surprenant DS, Patterson S, Gingras D and Beliveau R (2003) Regulation of vascular endothelial growth factor receptor-2 activity by caveolin-1 and plasma membrane cholesterol. *Mol. Biol. Cell* 14, 334–347.
- 60 Ikeda S, Ushio-Fukai M, Zuo L, Tojo T, Dikalov S, Patrushev NA and Alexander RW (2005) Novel role for ARF6 in vascular endothelial growth factor-induced signaling and angiogenesis. *Circ. Res.* 96, 467–475.
- **61** Cho CH, Lee CS, Chang M, Jang IH, Kim SJ, Hwang I, Ryu SH, Lee CO and Koh GY (2004) Localization of VEGFR-2 and PLD2 in endothelial caveolae is involved in VEGF-induced phosphorylation of MEK and ERK. *Am. J. Physiol.* 286, H1881–H1888.
- 62 Sanguinetti AR, Caoi H, Mastick CC (2003) Fyn is required for oxidative and hyperosmotic-stress-induced tyrosine phosphorylation of caveolin-1. *Biochem. J.* 376, 159–168.
- 63 Shigematsu S, Watson RT, Khan AH and Pessin JE (2003) The adipocyte plasma membrane caveolin functional/structural organization is necessary for the efficient endocytosis of GLUT-4. J. Biol. Chem. 278, 10683–10690.
- 64 Wang XL, Ye D, Peterson TE, Cao S, Shah VH, Katusic ZS, Sieck GC and Lee HC (2005) Caveolae targeting and regulation of large conductance Ca(2+)-activated K+ channels in vascular endothelial cells. *J. Biol. Chem.* 280, 11656–11664.
- 65 Sampson LJ, Hayabuchi Y, Standen NB and Dart C (2004) Caveolae localize protein kinase A signaling to arterial ATPsensitive potassium channels. *Circ. Res.* 95, 1012–1018.
- **66** Mendez AJ, Lin G, Wade DP, Lawn RM and Oram JF (2001) Membrane domains distinct from cholesterol/sphingomyelinrich rafts are involved in the ABCA1-me-

diated lipid secretory pathway. J. Biol. Chem. 276, 3158–3166.

- 67 Chao WT, Tsai SH, Lin YC, Lin WW and Yang VC (2005) Cellular localization and interaction of ABCA1 and caveolin-1 in aortic endothelial cells after HDL incubation. *Biochem. Biophys. Res. Commun.* 332, 743–749.
- 68 Garrigues A, Escargueil AE and Orlowski S (2002) The multidrug transporter, P-glycoprotein, actively mediates cholesterol redistribution in the cell membrane. *Proc. Natl. Acad. Sci. USA* 99, 10347–10352.
- **69** Demeule M, Jodoin J, Gingras D and Beliveau R (2000) P-glycoprotein is localized in caveolae in resistant cells and in brain capillaries. *FEBS Lett.* 466, 219–224.
- 70 Jodoin J, Demeule M, Fenart L, Cecchelli R, Farmer S, Linton KJ, Higgins CF and Beliveau R (2003) J. Neurochem. 87, 1010–1023.
- 71 Hinrichs JW, Klappe K, Hummel I and Kok JW (2004) ATP-binding cassette transporters are enriched in non-caveolar detergent-insoluble glycosphingolipid-enriched membrane domains (DIGs) in human multidrug-resistant cancer cells. *J. Biol. Chem.* 279, 5734–5738.
- 72 Babitt J, Trigatti B, Rigotti A, Smart EJ, Anderson RGW, Xu S and Krieger M (1997) Murine SR-BI, a high density lipoprotein receptor that mediates selective lipid uptake, is N-glycosylated and fatty acylated and colocalizes with plasma membrane caveolae. J. Biol. Chem. 272, 13242–13249.
- **73** Peng Y, Akmentin W, Connelly MA, Lund-Katz S, Phillips MC and Williams DL (2004) Scavenger receptor BI (SR-BI) clustered on microvillar extensions suggests that this plasma membrane domain is a way station for cholesterol trafficking between cells and high density lipoproteins. *Mol. Biol. Cell* 15, 384–396.
- 74 Azhar S, Nomoto A and Reaven E (2002) Hormonal regulation of adrenal microvillar channel formation. *J. Lipid Res.* 43, 861–871.
- 75 Chatenay-Rivauday C, Cakar ZP, Jeno P, Kuzmenko ES and Fiedler K (2004) Caveolae: biochemical analysis. *Mol. Biol. Rep.* 31, 67–84.
- 76 Garcia-Cardena G, Fan R, Stern DF, Liu J and Sessa WC (1996) Endothelial nitric

oxide synthase is regulated by tyrosine phosphorylation and interacts with caveolin-1. *J. Biol. Chem.* 271, 27237–27240.

- 77 Ghosh S, Gachhui R, Crooks C, Wu C, Lisanti MP and Stuehr DJ (1998) Interaction between caveolin-1 and the reductase domain of endothelial nitric oxide synthase. Consequences for catalysis. J. Biol. Chem. 273, 22267–22271.
- 78 Bucci M, Gratton JP, Rudic RD, Acevedo L, Roviezzo F, Cirino G and Sessa WC (2000) In vivo delivery of the caveolin-1 scaffolding domain inhibits nitric oxide synthesis and reduces inflammation. *Nature Med.* 6, 1362–1267.
- 79 Gratton JP, Fontana J, O'Connor DS, Garcia-Cardena G, McCabe TJ and Sessa WC (2000) Reconstitution of an endothelial nitric oxide synthase (eNOS), hsp90, and caveolin-1 complex in vitro. Evidence that hsp90 facilitates calmodulin-stimulated displacement of eNOS from caveolin-1. J. Biol. Chem. 275, 22268–22272.
- 80 Bernatchez PN, Bauer PM, Yu J, Prendergast JS, He P and Sessa WC (2005) Dissecting the molecular control of endothelial NO synthase by caveolin-1 using cellpermeable peptides. *Proc. Natl. Acad. Sci.* USA 102, 761–766.
- 81 Nohe A, Keating E, Underhill TM, Knaus P and Pedersen NO (2005) Dynamics and interaction of caveolin-1 isoforms with BMP-receptors. J. Cell Sci. 118, 643–650.
- 82 Razandi M, Oh P, Pedram A, Schnitzer J and Levin ER (2002) Ers associate with and regulate the production of caveolin: implications for signaling and cellular actions. *Mol. Endocrinol.* 16, 100–115.
- 83 Huhtakangas JA, Olivera CJ, Bishop JE, Zanello LP and Norman AW (2004) The vitamin D receptor is present in caveolaeenriched plasma membranes and binds 1-alpha,25(OH)2-vitamin D in vivo and in vitro. *Mol. Endocrinol.* 18, 2660–2671.
- 84 Couet J, Li S, Okamoto T, Ikezu T and Lisanti MP (1997) Identification of peptide and protein ligands for the caveolin-scaffolding domain. Implications for the interaction of caveolin with caveolae-associated proteins. J. Biol. Chem. 272, 6525–6533.
- 85 Woodman SE, Schlegel A, Cohen AW and Lisanti MP (2002) Mutational analysis identifies a short atypical membrane

112 References

attachment sequence (KYWFYR) within caveolin-1. *Biochemistry* 41, 3790–3795.

- 86 Jamin N, Neumann JM, Ostuni MA, Vu TK, Yao ZX, Murail S, Robert JC, Giatzakis C, Papadopoulos V and Lacapere JJ (2005) Characterization of the cholesterol recognition amino acid consensus sequence of the peripheral-type benzodiazepine receptor. *Mol. Endocrinol.* 19, 588–594.
- 87 Raman CS, Li H, Martasek P, Kral V, Masters BS and Poulos TL (1998) Crystal structure of constitutive endothelial nitric oxide synthase: a paradigm for pterin function involving a novel metal center. *Cell* 95, 939–950.
- 88 Bornancin F and Parker PJ (1997) Phosphorylation of protein kinase C-alpha on serine 657 controls the accumulation of active enzyme and contributes to its phosphatase-resistant state. J. Biol. Chem. 272, 3544–3549.
- 89 Orry AJ and Wallace BA (2000) Modelling and docking the endothelin G-proteincoupled receptor. *Biophys. J.* 79, 3083–3094.
- **90** Toya Y, Schwencke C, Couet J, Lisanti MP and Ishikawa Y (1998) Inhibition of adenyl cyclase by caveolin peptides. *Endocrinology* 139, 2025–2031.
- 91 Friedland N, Lious HL, Lobel P, Stock AM (2003) Structure of a cholesterolbinding protein deficient in Niemann-Pick type C2 disease. *Proc. Natl. Acad. Sci.* USA 100, 2512–2517.
- 92 Romanowski MJ, Soccio RE, Breslow JL and Burley SK (2002) Crystal structure of the *Mus musculus* cholesterol-regulated START protein 4 (StarD4) containing a StAR-related lipid transfer domain. *Proc. Natl. Acad. Sci. USA* 99, 6949–6954.
- **93** Kallen JA, Schaeppi JM, Bitsch F, Geisse S, Geiser M, Delhon I and Fournier B (2002) X-ray structure of the hROR alpha LBD at 1.63Å: structural and functional data that cholesterol or a cholesterol derivative is the natural ligand of RORalpha. *Structure* 10, 1697–1707.
- 94 Jamin N, Neumann JM, Ostuni MA, Vu TK, Yao ZX, Murail S, Robert JC, Giatzakis C, Papadopoulos V and Lacapere JJ (2005) Characterization of the cholesterol recognition amino acid consensus se-

quence of the peripheral-type benzodiazepine receptor. *Mol. Endocrinol.* 19, 588–594.

- 95 Zheng YH, Plemenitas A, Fielding CJ and Peterlin BM (2003) Nef increases the synthesis of and transports cholesterol to lipid rafts and HIV-1 progeny virus. Proc. Natl. Acad. Sci. USA 100, 8460–8465.
- 96 Epand RM, Sayer BG and Epand RF (2005) Caveolin scaffolding region and cholesterol-rich domains in membranes. J. Mol. Biol. 345, 339–350.
- 97 Harris JS, Epps DE, Davio SR and Kezdy FJ (1995) Evidence for transbilayer, tail-totail cholesterol dimmers in dipalmitoylglycerophosphocholine liposomes. *Biochemistry* 34, 3851–3857.
- 98 Jensen KJ and Brask J (2005) Carbohydrates in peptide and protein design. *Biopolymers*. May 31, e-pub.
- **99** Hailstones D, Sleer LS, Parton RG and Stanley KK (1998) Regulation of caveolin and caveolae by cholesterol in MDCK cells. *J. Lipid Res.* 39, 369–379.
- 100 Sharma DK, Brown JC, Choudhury A, Peterson TE, Holicky E, Marks DL, Simari R, Parton RG and Pagano RE (2004) Selective stimulation of caveolar endocytosis by glycosphingolipids and cholesterol. *Mol. Cell. Biol.* 15, 3114–3122.
- 101 Fielding CJ, Bist A and Fielding PE (1997) Caveolin mRNA levels are up-regulated by free cholesterol and down-regulated by oxysterols in fibroblast monolayers. *Proc. Natl. Acad. Sci. USA* 94, 3753–3758.
- 102 Bist A, Fielding PE and Fielding CJ (1997) Two sterol regulatory element-like sequences mediate up-regulation of caveolin gene transcription in response to low density lipoprotein free cholesterol. *Proc. Natl. Acad. Sci. USA* 94, 10693–10698.
- 103 Zhu Y, Liao HL, Wang N, Yuan Y, Ma KS, Verna L and Stemerman MB (2000) Lipoprotein promotes caveolin-1 and Ras translocation to caveolae: role of cholesterol in endothelial signaling. *Arterioscler. Thromb. Vasc. Biol.* 20, 2465–2470.
- 104 Colonna C and Podesta EJ (2005) ACTHinduced caveolin-1 tyrosine phosphorylation is related to podosome assembly in Y1 adrenal cells. *Exp. Cell Res.* 304, 432–442.

- 105 Lee H, Volonte D, Galbiati F, Iyengar P, Lublin DM, Bregman DB, Wilson MT, Campos-Gonzalez R, Bouzahzah B, Pestell RG, Scherer PE and Lisanti MP (2000) constitutive and growth factor-regulated phosphorylation of caveolin-1 occurs at the same site (Tyr-14) in vivo. *Mol. Endocrinol.* 14, 1750–1775.
- 106 Labrecque L, Nyalendo C, Langlois S, Durocher Y, Roghi C, Murphy G, Gingras D and Beliveau R (2004) Sr-c-mediated tyrosine phosphorylation of caveolin-1 induces its association with membrane type 1 matrix metalloproteinase. *J. Biol. Chem.* 279, 52132–52140.
- 107 Maggi D, Biedi C, Segat D, Barbero D, Panetta D and Cordera R (2002) IGF-1 induces caveolin 1 tyrosine phosphorylation and translocation in the lipid rafts. *Biochem. Biophys. Res. Commun.* 295, 1085–1089.
- 108 Kimura A, Mora S, Shigematsu S, Pessin JE and Saltiel AR (2002) The insulin receptor catalyzes the tyrosine phosphorylation of caveolin-1. J. Biol. Chem. 277, 30153–30158.
- 109 Romiti E, Meacci E, Donati C, Formigli L, Zecchi-Orlandini S, Farnararo M, Ito M and Bruni P (2003) Neutral ceramidase secreted by endothelial cells is released in part associated with caveolin-1. *Arch. Biochem. Biophys.* 417, 27–33.
 receptor substrate-1, but not for mitoger activated protein kinase control. *J. Biol. Chem.* 276, 9670–9678.
 Blair A, Shaul PW, Yuhanna IS, Conrad PA and Smart EJ (1999) Oxidized low density lipoprotein displaces endothelial
- 110 Stehr M, Estrada CR, Khoury J, Danciu TE, Sullivan MP, Peters CA, Solomon KR, Freeman MR and Adam RM (2004) Caveolae are negative regulators of transforming growth factor-beta1 signaling in urethral smooth muscle cells. *J. Urol.* 172, 2451–2455.
- 111 Graziani A, Bricko V, Carmignani M, Graier WF and Groschner K (2004) Cholesterol- and caveolin-rich membrane domains are essential for phospholipase A2dependent EDHF formation. *Cardiovasc. Res.* 64, 234–242.
- 112 Gaudreault SB, Chabot C, Gratton JP and Poirier J (2004) The caveolin scaffolding domain modifies 2-amino-3-hydroxy-5-methyl-4-isoxazole propionate receptor binding properties by inhibiting phospholipase A2 activity. *J. Biol. Chem.* 279, 356–362.
- 113 Tortelote GG, Valverde RH, Lemos T, Guilherme A, Einicker-Lamas M and

Vieyra A (2004) The plasma membrane Ca²⁺ pump from proximal kidney tubules is exclusively localized and active in caveolae. *FEBS Lett.* 576, 31–35.

- 114 Zeidan A, Broman J, Hellstrand P and Sward K (2003) Cholesterol dependence of vascular ERK1/2 activation and growth in response to stretch: role of endothelin-1. Arterioscler. Thromb. Vasc. Biol. 23, 1528–1534.
- 115 Troost J, Lindenmaier H, Haefeli WE and Weiss J (2004) Modulation of cellular cholesterol alters p-glycoprotein activity in multidrug-resistant cells. *Mol. Pharmacol.* 66, 1332–1339.
- 116 Rybin VO, Xu X, Lisanti MP and Steinberg SF (2000) Differential targeting of beta-adrenergic receptor subtypes and adenyl cyclase to cardiomyocyte caveolae. A mechanism to functionally regulate the cAMP signaling pathway. J. Biol. Chem. 275, 41447–41457.
- 117 Parpal S, Karlsson M, Thorn H and Stralfors P (2001) Cholesterol depletion disrupts caveolae and insulin receptor signaling for metabolic control via insulin receptor substrate-1, but not for mitogenactivated protein kinase control. *J. Biol. Chem.* 276, 9670–9678.
- 118 Blair A, Shaul PW, Yuhanna IS, Conrad PA and Smart EJ (1999) Oxidized low density lipoprotein displaces endothelial nitric oxide synthase (eNOS) from plasmalemmal caveolae and impairs eNOS activation. J. Biol. Chem. 274, 32512–32519.
- 119 Niggli V, Meszaros AV, Oppliger C and Tornay S (2004) Impact of cholesterol depletion on shape changes, actin reorganization, and signal transduction in neutrophil-like HL-60 cells. *Exp. Cell Res.* 296, 358–368.
- 120 Taggart MJ, Leavis P, Feron O and Morgan KG (2000) Inhibition of PKCalpha and rhoA translocation in differentiated smooth muscle by a caveolin scaffolding domain peptide. *Exp. Cell Res.* 258, 72–81.
- 121 Caselli A, Taddei ML, Manao G, Camici G and Ramponi G (2001) Tyrosine-phosphorylated caveolin is a physiological substrate of the low M(r) phosphotyrosine phosphatase. J. Biol. Chem. 276, 18849–18854.

6 Phosphorylation of Caveolin and Signaling from Caveolae

Cynthia Corley Mastick, Amy Sanguinetti, Haiming Cao, and Suhani Thakker

6.1 Introduction

Caveolae are specialized invaginated domains of the plasma membrane that act as organizing centers for signaling molecules. Caveolae are formed in membranes by the caveolins, a family of three related gene products (caveolins-1, -2, and -3) [1-4]. Caveolins form homo- and hetero-oligomers that make up the characteristic striated coat of caveolae [5]. Expression of caveolin is necessary and sufficient to induce cell-surface caveolae [6-8]. In addition to their role as coat proteins that drive the invagination of caveolae, the caveolins are also cholesterol-binding proteins, and caveolae are highly enriched in cholesterol, glycolipids and sphingolipids, forming a distinctive domain in the membrane. Lipid-modified signaling molecules, including the tandemly acylated Src-family kinases, are enriched in these structures due to their affinity for the lipid composition of these domains. However, cholesterol-enriched lipid-ordered domains, or "rafts" also form independently of caveolae [9]. The caveolins direct the composition of the signaling complexes organized in caveolae by acting as scaffolding proteins that bind to specific signaling molecules, which include Gaq11, endothelial nitric oxide synthase (eNOS), and Src-family kinases [10-13].

In addition to organizing signaling complexes, the caveolins also participate directly in signaling cascades as substrates for both tyrosine and serine/threonine kinases. Caveolin-1 is phosphorylated on a single tyrosine residue (Tyr14) in the amino-terminus of the protein. This residue is constitutively phosphorylated at low levels in most cell types, and its phosphorylation increases in response to a number of stimuli, including insulin and insulin mimetics (IGF-1, sulfonylureas, phosphoinositolglycan), angiotensin II, endothelin-1, adrenocorticotropin (ACTH), platelet-derived growth factor (PDGF), vascular endothelial growth factor (VEGF), and epidermal growth factor (EGF) (in cells expressing very high levels of the EGF receptor or expressing mutant forms of the EGF receptor) [14–27]. Phosphorylation at Tyr14 is also increased in response to cellular stresses, including shear stress, apoptotic stress, ultraviolet radiation, oxidative stress, and osmotic shock [24,28–34], in response to integrin activation (i. e., during plating and spreading on



Fig. 6.1 Signaling pathways leading to and from caveolin phosphorylation. Hypothesis to be explored: insulin and stress activate a caveolin-directed tyrosine kinase localized in the caveolae. Upon activation, this kinase phosphorylates caveolin-1 on Tyr14 (pY14). This leads to activation of downstream signaling cascades.

fibronectin) [24,34], and when caveolae are triggered to internalize [35–37]. In response to some stimuli, such as insulin, oxidative stress and osmotic shock, the increase in caveolin phosphorylation is robust and sustained, whilst in response to other stimuli, such as shear stress, PDGF and VEGF, the increase is less intense and is transient. Caveolin-1 is also phosphorylated on up to five serine/threonine residues in response to various stimuli [26,31,38,39]. Caveolin-2 is phosphorylated on two tyrosine residues in its amino-terminal domain, Tyr 19 and Tyr27 [40–43]. Tyr27 in caveolin-2 falls within an amino acid sequence (YADP) that is very similar to Tyr14 in caveolin-1 (YTVP), and phosphocaveolin-2 is found in complexes with phosphocaveolin-1. Caveolin-3 lacks a site for tyrosine phosphorylated on either tyrosine or serine.

The goals of our research in this area have been to: 1) trace the signal transduction pathways that lead to caveolin tyrosine phosphorylation; and 2) identify signaling cascades that lie downstream of caveolin phosphorylation. These two areas will form the focus of this chapter (Fig. 6.1). In a very satisfying way, the investigations in these two areas have come together to provide a very clear picture of at least one of the roles of caveolin phosphorylation and caveolae in cells.

6.2

Signaling Pathways Leading to Caveolin Tyrosine Phosphorylation

6.2.1

Caveolins-1 and -2 are Phosphorylated in Response to Insulin in Adipocytes

Whilst caveolae are found in many tissues, they are particularly abundant in lung epithelial cells, endothelial cells, muscle cells, and adipocytes [1,3,4]. In adipocytes,

they cover a significant fraction (ca. 30%) of the total inner cell surface of the plasma membrane [44]. In addition, the expression of caveolins-1 and -2 increases approximately 20-fold upon adipocyte differentiation, with a concomitant 10-fold increase in cell-surface caveolae [15,45,46]. Consistent with an important role for caveolae in adipocyte function, adipose tissue is significantly disrupted in caveolin-1 knockout animals [47–49].

Early studies on isolated caveolin-enriched cell fractions had implicated caveolae in cellular signaling [50-54]. The abundance of caveolae in adipocytes and muscle - two of the major target tissues for insulin - together with their potential role in signaling, suggested that they might play a role in insulin signal transduction. To investigate this, a simple question was asked: Does stimulation of adipocytes with insulin lead to an increase in the phosphorylation of any proteins associated with caveolae? It was found that, in adipocytes, insulin stimulates the tyrosine phosphorylation of three proteins in caveolae: caveolin-1, caveolin-2, and a 29-kDa caveolin-associated protein [14,15]. These three proteins are found in SDS-resistant complexes and can be co-immunoprecipitated. Caveolin phosphorylation shows two additional interesting properties. The first is that it shows specificity for insulin, and does not occur in adipocytes in response to two other growth factors, PDGF and EGF, despite the expression of active receptors for all three growth factors in these cells. This is interesting because stimulation of glucose metabolism also shows specificity for insulin in these cells. Insulin stimulates glucose transport, glycogen synthesis, and lipogenesis ten to hundreds of fold, while EGF and PDGF have no effect on these processes [55]. The second interesting property of insulin-stimulated caveolin phosphorylation is that it is cell type- dependent: it occurs only in the fully differentiated 3T3-L1 adipocytes, not in the preadipocytes, despite the expression of both caveolin and active insulin receptor in both cell types. In fact, caveolin is not phosphorylated in response to insulin in fibroblast cells engineered to express high levels of the insulin receptor. Differentiation dramatically increases the insulin responsiveness of glucose metabolism in these cells [56]. Unlike phosphorylation of the caveolins, the 29-kDa caveolin-associated protein was phosphorylated in response to PDGF as well as in response to insulin, and this phosphorylation occurred in both adipocytes and preadipocytes [14,20]. At present, the identity of this protein is unknown, but this result indicates that other growth factors also signal through tyrosine phosphorylation of caveolar proteins.

6.2.2

The Caveolins are not Direct Substrates of the Insulin Receptor

Insulin binding to its receptor at the cell surface causes a conformational change in the intracellular domain of the receptor, leading to auto-transphosphorylation of the receptor. This autophosphorylation in turn activates the kinase domain, leading to the phosphorylation of direct substrates, including insulin receptor substrate-1 (IRS-1). Since the insulin receptor is associated with caveolae [14,57,58], it was possible that caveolin was a direct substrate of the insulin receptor itself. Several lines of evidence indicate that this is not the case, however [14,15]. First,

118 6 Phosphorylation of Caveolin and Signaling from Caveolae

although caveolin co-fractionates with the insulin receptor, caveolin is phosphorylated *in vitro* under conditions where the insulin receptor in these fractions is completely inactive (i. e., in the absence of insulin). Therefore, caveolin co-purifies with a caveolin-tyrosine kinase that is not the insulin receptor. A second line of evidence comes from the differentiation dependence of caveolin phosphorylation. Both adipocytes and preadipocytes express caveolin-1, and caveolin-1 co-purifies with caveolin-directed tyrosine kinase activity in both cell types. Adipocytes and preadipocytes both express active insulin receptors that phosphorylate direct substrates such as IRS-1 equally well. The insulin receptor co-fractionates with caveolins in both cell types. However, insulin stimulates caveolin phosphorylation only in adipocytes, not in preadipocytes. The cell type dependence of caveolin phosphorylation strongly indicates that the insulin receptor is not the insulinstimulated caveolin tyrosine kinase, and that differentiation leads to the expression of signaling molecules that lie downstream of the insulin receptor and couple activation of the insulin receptor to activation of the caveolin tyrosine kinase.

6.2.3

Src-Family Kinases and Stress-Induced Caveolin Phosphorylation

Signal transduction via transmembrane receptor tyrosine kinases often involves the activation of additional downstream kinases, both serine/threonine-directed and tyrosine-directed. A number of studies have implicated Src-family kinases in tyrosine phosphorylation of caveolin-1. Caveolin-1 was originally identified as a major tyrosine-phosphorylated protein in v-Src-transformed cells [1,39,59,60]. Caveolin is also highly phosphorylated in cells that overexpress c-Src or c-Fyn, and is one of the most prominent phosphoproteins detected in these cells [15,16,31,61]. Caveolin-1 binds to and co-purifies with Src-family kinases [10,52,62,63]. Furthermore, caveolin-1 can be directly phosphorylated by Src and Fyn *in vitro* [16,64]. These data indicated that Src and Fyn are caveolin tyrosine kinases in cells.

Caveolin-1 is phosphorylated in response to a number of cellular stresses, including shear stress, oxidative stress, and osmotic shock [24,28-32,34]. These stresses also activate Src-family kinases, including both Src and Fyn [31,34,65-67]. While overexpression of either kinase is sufficient to cause caveolin tyrosine phosphorylation under basal conditions, overexpression of Fyn also caused significant hyperphosphorylation of caveolin-1 in response to stress [31]. Therefore, whilst overexpression can activate Fyn to some extent, stress clearly activates the kinase. These data indicate that Fyn is part of the signaling cascade leading to caveolin tyrosine phosphorylation in response to stress. Consistent with this, expression of kinase-inactive Src blocked caveolin phosphorylation in response to osmotic shock [29]. Furthermore, the small molecule Src-family kinase inhibitors SU6656, PP2, and PD180970 inhibited both oxidative stress and osmotic shock-induced caveolin phosphorylation, indicating that both require activation of a Src-family kinase [31]. Shear stress-induced caveolin phosphorylation was blocked by the small molecule Src-family kinase inhibitor PP1, and enhanced by inactivation of Csk, a negative regulator of Src-family kinases [34]. Consistent with the inhibitor

data, both oxidative stress and osmotic shock-induced caveolin phosphorylation were blocked in SYF^{-/-} cells, a cell line derived from a knockout animal deficient in three Src-family kinases: Src, Yes, and Fyn [31]. These data show that stress-induced caveolin phosphorylation requires activation of a Src-family kinase.

Using cell lines derived from single kinase knockout mice (Src^{-/-}, Yes^{-/-}, and Fyn^{-/-}), it was found that expression of Fyn is required for both oxidative stress and osmotic shock-induced caveolin phosphorylation, while expression of Src and Yes are not [31]. In fact, phosphorylation in the Src^{-/-} cells was even higher than that observed in the cells from wild-type mice. This correlated with an increase in Fyn expression in these cells relative to wild-type cells. Heterologous expression of Fyn was sufficient to restore oxidative stress-induced caveolin phosphorylation in both the Fyn^{-/-} and SYF^{-/-} cells, indicating that Fyn can promote the phosphorylation of caveolin in the absence of Src or Yes. Therefore, Fyn expression is both necessary and sufficient for stress-induced caveolin phosphorylation.

Both Src and Fyn can directly phosphorylate caveolin, and both are activated in response to stress, but only Fyn is required for stress-induced caveolin phosphorylation. These data suggest that the Src that is activated in response to stress is localized to a compartment that does not contain caveolin [31]. Fyn is targeted to caveolae/lipid rafts due to tandem acylation by myristate and palmitate. Co-localization of dually acylated proteins such as Lyn or Fyn with caveolin has recently been confirmed using fluorescence resonance energy transfer (FRET) [63]. In contrast to Fyn, Src is only singly acylated with myristate. Singly acylated proteins have less affinity for membranes and do not co-purify with caveolae/lipid rafts [52,68]. The behavior of singly acylated proteins, such as Src, was not determined in the FRET experiment. However, Src can bind directly to caveolin [10]. The present data suggest that there are distinct signaling complexes containing caveolin and Fyn or caveolin and Src that are differentially activated in response to stress. In the context of cellular stress, Fyn – but not Src – is a caveolin tyrosine kinase. However, Src may phosphorylate caveolin under other types of stimuli, for example during cell attachment and spreading on fibronectin [24].

6.2.4

Non-Receptor Tyrosine Kinases and Insulin-Induced Caveolin Phosphorylation

Significant data had linked activation of Src-family kinases to caveolin phosphorylation. Therefore, it was initially hypothesized that a Src-family kinase was the insulin-stimulated caveolin-tyrosine kinase in adipocytes. Consistent with this, treatment of isolated caveolar fractions with Src-family kinase inhibitors blocked caveolin phosphorylation *in vitro*, indicating that the caveolin kinase activity that co-purifies with caveolin is a Src-family kinase (CCM, unpublished observation; [16]). (These inhibitors did not block Src-family kinase activation in intact adipocytes as measured by kinase autophosphorylation. In contrast, inhibition of autophosphorylation was readily detectable in fibroblasts. This is a common problem in adipocytes. Compounds that are sufficiently lipophilic to cross membranes are often sequestered within the prominent fat droplets in these cells.) The most

120 6 Phosphorylation of Caveolin and Signaling from Caveolae

abundant Src-family kinase that co-purifies with caveolin-1 in adipocytes is Fyn [14,15,69]. Fyn is activated in response to insulin via Src-homology 2 (SH2) domain-mediated binding to IRS-1 [70]. To determine if Fyn was involved in insulin-stimulated caveolin tyrosine phosphorylation, Fyn was overexpressed in adipocytes [15]. Overexpression of Fyn was sufficient to induce caveolin tyrosine phosphorylation under basal conditions, and hyperphosphorylation of caveolin-1 in response to insulin. These results verify that Fyn is activated in response to insulin, and that it is part of the signaling cascade leading to caveolin tyrosine phosphorylation in response to insulin. However, differentiation does not change the level of expression of Fyn, and Fyn co-localizes with caveolin-1 in both preadipocytes and adipocytes. Furthermore, overexpression of Fyn in fibroblasts did not increase basal caveolin phosphorylation or reconstitute insulin-stimulated phosphorylation. These data indicate that, while Fyn is a part of the insulin-stimulated caveolin phosphorylation pathway in adipocytes, differentiation induces the expression of an additional protein or proteins required for caveolin tyrosine phosphorylation.

An additional line of evidence suggested that a second non-receptor tyrosine kinase may be involved in caveolin phosphorylation. The phosphorylation site on caveolin-1 (Tyr14, L-Y-T-V-P) is not in the context of a consensus Fyn phosphorylation site (I/L-Y-D/E-X-L). Tyr14 lies within the motif I/L-Y-X-X-P, a consensus Abl phosphorylation site [71,72]. Tyr27 in caveolin-2 falls within a similar sequence (E-Y-A-D-P). It has not been determined whether insulin activates Abl in adipocytes, although several known Abl substrates are phosphorylated in response to insulin, including Crk, Cbl, and Dok [73,74]. In addition, the Crk binding partner CAS is dephosphorylated in response to insulin, consistent with Abl activation [75]. Importantly, differentiation of 3T3-L1 fibroblasts into adipocytes leads to a large increase in Abl expression (CCM, unpublished observation), which may account for the cell type dependence of insulin-stimulated caveolin phosphorylation.

6.2.5

Abl is a Caveolin Kinase

To determine initially whether Abl phosphorylates caveolin, a fibroblast cell line expressing a temperature-sensitive form of v-Abl was used [76]. A temperature-sensitive form of v-Abl was necessary because constitutively active v-Abl leads to loss of expression of both caveolin-1 and -2 after only a few days [76,77]. (Expression of v-Src also causes down-regulation of caveolin [60,77]; however, this down-regulation is secondary to cellular transformation, and is not caused by phosphorylation of caveolin [76].) Caveolin-1 was one of the strongest phosphotyrosine signals detected in these cells after activation of the kinase. Caveolin-1 was also one of the most prominent phosphoproteins detected in primary human fibroblast cells overexpressing c-Abl. Caveolin-2 co-immunoprecipitated with caveolin-1, and was also phosphorylated under both of these conditions. Abl can directly phosphorylate caveolin-1 *in vitro* and in a yeast expression system [76,78]. The phosphorylation of caveolin-1 by Abl required Tyr14 and did not occur on a fusion protein in which

this residue was changed to phenylalanine, indicating that the consensus Abl phosphorylation site is the only site in caveolin-1 phosphorylated by Abl.

To verify that Abl phosphorylates caveolin, fibroblast cell lines derived from an Abl knockout mouse (Abl^{-/-}) were utilized. The same cells reconstituted with Abl served as controls (Abl⁺). Although, insulin does not stimulate caveolin phosphorylation in fibroblasts, caveolin is phosphorylated in response to oxidative stress in these cells [28,29]. Oxidative stress activates Abl [79,80]; therefore, oxidative stress-induced phosphorylation was a good system initially to test the requirement for Abl in caveolin phosphorylation [32].

Abl was required for oxidative stress-induced caveolin phosphorylation. Caveolin was not phosphorylated in response to oxidative stress in the $Abl^{-/-}$ cells, but phosphorylation was restored in the Abl^+ cells. In contrast to caveolin, many other proteins were phosphorylated on tyrosine in the $Abl^{-/-}$ cells, indicating that these cells still respond to oxidative stress, but that phosphorylation of Abl substrates is specifically lost. The only site of tyrosine phosphorylation of caveolin-1 in response to oxidative stress is Tyr14, the consensus Abl phosphorylation site. Therefore, expression of Abl is necessary for phosphorylation of caveolin-1 on tyrosine in response to oxidative stress, and the only site of tyrosine phosphorylation is a consensus Abl site. Additionally, overexpression of Abl is sufficient to induce caveolin phosphorylation in some cell types, and purified Abl phosphorylates caveolin at Tyr14 *in vitro*. These data indicate that c-Abl is also a stress-induced caveolin-tyrosine kinase.

6.2.6

Abl and Fyn Cooperate in the Caveolin Phosphorylation Signaling Pathway

The data indicate that expression of both Abl and Fyn is required for stress-induced caveolin phosphorylation. To verify that both Fyn and Abl are required for caveolin phosphorylation in the same cell, the effect of Src-family kinase inhibitors on caveolin phosphorylation in the Abl^{-/-} and Abl⁺ cells was analyzed. Pretreatment of the Abl⁺ cells with SU6656 abolished oxidative stress-induced caveolin phosphorylation, indicating that both Abl and Src-family kinases are required in this pathway in a single cell.

One mechanism to explain the requirement for both Abl and Fyn would be that they act sequentially in a linear pathway. For example, Fyn might activate Abl, and Abl may be the caveolin kinase (Stress \rightarrow Fyn \rightarrow Abl \rightarrow cav-*P*). Consistent with this model, previous studies had shown that Src-family kinase activation leads to Abl activation in a kinase cascade, and that both Fyn and Src can directly phosphorylate Abl *in vitro* [81]. If Fyn does not directly phosphorylate caveolin, then this model predicts that Fyn overexpression-induced caveolin phosphorylation would require Abl expression as well. To test this, Fyn was overexpressed in Abl⁺ or Abl^{-/-} fibroblasts. As was observed in fibroblasts from wild-type mice, overexpression of Fyn was sufficient to induce caveolin phosphorylation in the Abl⁺ cells. However, inconsistent with a linear pathway from Fyn through Abl, Fyn-induced caveolin phosphorylation was even higher in the Abl^{-/-} cells than in the Abl⁺ cells, indicat-

122 6 Phosphorylation of Caveolin and Signaling from Caveolae

ing that Abl actually negatively regulates Fyn under basal conditions. Therefore, overexpression of Fyn can bypass the requirement for Abl in caveolin phosphorylation. These results indicate that Fyn directly phosphorylates caveolin-1 and does not require Abl as an intermediary.

In an alternative linear pathway, Abl could be required for Fyn activation (Stress \rightarrow Abl \rightarrow Fyn \rightarrow cav-*P*). Inconsistent with this model, however, oxidative stressinduced activation of Fyn does not require expression of Abl. While overexpression of Fyn was sufficient to induce caveolin phosphorylation, oxidative stress significantly increased this phosphorylation to very high levels in the Fyn overexpressing cells, indicating further activation of the kinase. This hyperphosphorylation occurred in both the Abl⁺ and Abl^{-/-} cells. To test if Src-family kinase activity is required downstream of Abl, the effect of Src-family kinase inhibitors on Ablinduced caveolin phosphorylation was examined. While SU6656 blocked Fyn overexpression-induced caveolin phosphorylation, Abl-induced phosphorylation was unaffected. These results indicate that Abl, like Fyn, acts directly on caveolin and does not require activation of a downstream Src-family kinase.

6.2.7

Model of the Interaction of Fyn and Abl in Caveolin Phosphorylation

Expression of both Abl and Fyn is necessary for caveolin phosphorylation, and overexpression of either is sufficient to induce caveolin phosphorylation. Therefore, the caveolin phosphorylation pathway represents a newly identified signaling cascade involving both Abl and Fyn. Signaling complexes involving both Src-family kinases and Abl have also been identified in a number of other systems [82-89]. Whilst both Abl and Fyn can phosphorylate caveolin directly, it is believed that Fyn and Abl act synergistically to induce sustained, high-level phosphorylation of caveolin-1. Our model is that Fyn activity is required for the efficient recruitment of Abl to the caveolae, and that Abl is required for sustained phosphorylation of caveolin (Fig. 6.2). In this model, activation of Fyn that is resident in the caveolae leads to phosphorylation of caveolin-1 at Tyr14. This phosphorylation creates a binding site for the SH2 protein binding domain of Abl (consensus binding site, pY-X-X-P), and recruitment of Abl into the complexes (domain structure of Abl: SH3, SH2, kinase domain, tail). Abl then phosphorylates adjacent caveolin molecules in the complex. In this way, the phosphorylation when both kinases are activated is significantly greater than either alone. Overexpression of Abl causes activation of Abl and inefficient phosphorylation of caveolin-1, which leads to the recruitment of Abl into caveolae and more efficient phosphorylation. Overexpression of Fyn bypasses an inherent limitation on the level of caveolin phosphorylation that can normally be catalyzed by Fyn alone (see below). This model may also explain why some stimuli lead to only transient phosphorylation of caveolin, while others lead to sustained phosphorylation of caveolin. Whilst all activate Src-family kinases in caveolae and initiate caveolin phosphorylation, they may differ in their ability to activate and translocate Abl to the caveolae. Our investigations into the signaling cascades that lie downstream of caveolin phosphorylation strongly sup-



Fig. 6.2 Fyn and Abl are caveolin tyrosine kinases. Insulin and stress activate the Src-family kinase Fyn and autophosphorylation of the kinase (pY416). Fyn is localized to lipid rafts via tandem acylation and to caveolae through direct binding to caveolin. Upon

activation, Fyn phosphorylates caveolin-1 on Tyr14. Tyrosine phosphorylation promotes SH2 domain-mediated binding of Abl. This in turn leads to the phosphorylation of adjacent caveolin molecules.

port this model, and give significant insights into the reason that caveolin phosphorylation by Fyn alone is self-limited and requires the activation of two nonreceptor tyrosine kinases, Fyn and Abl, for sustained high level phosphorylation to occur.

6.3 Signaling Pathways Downstream of Caveolin Tyrosine Phosphorylation

With the exception of sites within the activation loops of kinases themselves, the function of tyrosine phosphorylation is to promote protein-protein interactions, particularly through SH2 protein-binding domains. This in turn leads to the activation of downstream signaling cascades. Therefore, phosphorylation of caveolin on tyrosine is likely to be an intermediate step in a signaling cascade occurring within caveolae. Caveolin-1 phosphorylated at Tyr14 would serve as a docking site for SH2 domain-containing proteins and would recruit proteins into caveolae to activate downstream signaling cascades (Fig. 6.3).

In order to identify proteins that bind to phosphorylated caveolin-1, a 3T3-L1 adipocyte cDNA library was screened using a novel yeast dihybrid screen [78]. The Gal4-based yeast two-hybrid system was modified to perform a phosphotyrosine-dependent dihybrid protein interaction screen. Then, a kinase (Abl) was introduced into the two-hybrid system to phosphorylate the bait protein (caveolin-1), after which screening was carried out for phosphorylation-dependent protein interactions. Using this system to screen an adipocyte cDNA library, three proteins were identified that interact with the amino-terminus of caveolin-1: JAB1, TRAF2, and Csk. Of these three proteins, only Csk contains an SH2 domain.



Fig. 6.3 Activation of signaling cascades downstream of caveolin phosphorylation. Extracellular signals (insulin and stress) activate caveolin kinases (Fyn and Abl). This leads to phosphorylation of caveolin-1 on

Tyr14. Tyrosine phosphorylation promotes SH2 domain-mediated protein interactions. This in turn leads to activation of downstream signaling cascades within the caveolae.

6.3.1 Csk Binds to Phosphocaveolin

Based on the initial screen, positive proteins could interact with caveolin-1 in a phosphorylation-dependent or independent manner. In order to counter-screen for phosphorylation-dependent interactions, the positive clones were transformed back into yeast that expressed caveolin-1 alone, caveolin-1 and Abl, or the phosphorylation mutant caveolin-1/Y14F and Abl. Only the interaction of Csk with caveolin-1 was completely dependent on phosphorylation. TRAF2 also bound to non-phosphorylated caveolin-1, although phosphorylation of caveolin-1 increased the binding of TRAF2 to caveolin-1 approximately three-fold. The proteasome sub-unit JAB1 interacted with caveolin-1 in a phosphorylation-independent manner.

Previous studies had shown that TRAF2 and caveolin-1 form a complex that recruits the tumor necrosis factor-alpha (TNFa) receptor after ligand binding [90]. Therefore, focus was centered on Csk. The interaction of Csk with phosphocaveolin in mammalian cells was verified. Phosphocaveolin co-immunoprecipitated with Csk in cells expressing v-Abl, and was one of only two major tyrosine-phosphorylated proteins bound to Csk in these cells [78]. The other phosphoprotein was paxillin, a multi-domain focal adhesion protein known to bind to Csk through an SH2 domain-phosphotyrosine interaction. Phosphocaveolin also co-immunoprecipitated with Csk in untransfected cells and this association increased after induction of oxidative stress or shear stress, and in adipocytes after stimulation with insulin [24,34,78,91]. Csk interacted only with caveolin phosphorylated on Tyr14, and did not bind to non-phosphorylated caveolin. Csk is one of only two proteins that are known to bind specifically to phosphorylated caveolin-1; the SH2 domaincontaining protein Grb7 is the other [16]. Only Csk has been shown to interact with caveolin-1 in a regulated manner in cells.

6.3.2 Regulation of Src-Family Kinases by Csk

Phosphorylation of caveolin is an ideal mechanism to recruit Csk to its substrates (the Src-family kinases) that are highly enriched in the caveolae. Csk is a negative regulator of Src-family kinases [92]. Csk phosphorylates Src-family kinases at an inhibitory tyrosine (Tyr527 in Src). When this residue is phosphorylated, it binds to the SH2 domain in the amino-terminus of the kinase, folding it into an inactive conformation (domain structure of Src-family kinases: SH3, SH2, kinase domain, regulatory tail with Csk phosphorylation site). The Src-family kinases can be reactivated either through dephosphorylation of this residue, or by displacement of the C-terminal tail by another phosphoprotein. However, the Src-family kinases are lipid-modified and localized to the plasma membrane, while Csk is largely free in the cytosol. Csk must be specifically targeted to its substrates in membranes via an SH2-domain-mediated interaction with a tyrosine-phosphorylated protein. In brain and lymphocytes, Csk is recruited to lipid rafts in the plasma membrane through binding to a transmembrane protein called Cbp that is constitutively phosphorylated on tyrosine [93,94]. Binding to phosphorylated Cbp also activates Csk. Phosphocaveolin also targets Csk to its substrates in the plasma membrane. Src-family kinases are highly enriched in caveolae, both through their lipid modifications and through direct binding to the "scaffolding domain" of the caveolins. Therefore, phosphocaveolin is an ideal targeting subunit for Csk. Significantly, caveolin phosphorylation is stimulated in response to extracellular signals. In contrast, Cbp is constitutively phosphorylated, and only transiently dephosphorylated in response to T-cell activation. Regulated phosphorylation of caveolin represents a novel mechanism for the regulation of Src-family kinases by extracellular signals through the recruitment and activation of Csk.

6.3.3

Feedback Inhibition of Fyn Through Activation of Csk

These data support the following model (Fig. 6.4). Activation of Fyn in caveolae by insulin or stress leads to phosphorylation of caveolin-1 on Tyr14, which leads to the recruitment of Csk, and phosphorylation and inhibition of resident Src-family kinases. This feedback mechanism is supported by a number of observations [24,31]. Oxidative stress simultaneously activates Src-family kinases and their negative regulator Csk. Induction of oxidative stress leads to concomitant increases in both active site and Csk inhibitory site phosphorylation of Fyn in many cell types. Basal Csk activity varied from cell type to cell type, but was very low in fibroblasts from wild-type mice and absent from fibroblasts from Csk knockout mice. However, basal Src-family kinase activity was low in all cases. Therefore, Src-family kinases are maintained in an inactive conformation in fibroblasts through a

126 6 Phosphorylation of Caveolin and Signaling from Caveolae



Fig. 6.4 Phosphocaveolin recruits Csk to the caveolae, attenuating Fyn. Csk (c-terminal Src kinase) is a negative regulator of Src-family kinases. Csk must be targeted to its substrates in membranes via an SH2 domain-mediated interaction with a tyrosine-phosphorylated protein. This activates Csk,

leading to phosphorylation of Src-family kinases at their inhibitory site (pY527) and attenuation of their activity. Csk binds specifically to phosphocaveolin only after stimulation with insulin or oxidative stress, and is activated under these conditions.

mechanism that is independent of Csk phosphorylation. In contrast to basal activity, oxidative stress-induced Src-family kinase activity was much greater in Csk^{+/+} cells, indicating that loss of Csk activity leads to dysregulation of the Src-family kinases, but only after activation. Therefore, Csk plays a modulatory, not a strictly regulatory role for Src-family kinases in these cells. This is in marked contrast to the mechanism of regulation of Src-family kinases described in lymphocytes. In these cells, Src-family kinases are constitutively phosphorylated by Csk, and activated by dephosphorylation of this site [95].

6.3.4

Phosphocaveolin in the Loop

Csk must be recruited to its substrates via SH2-mediated binding to a tyrosinephosphorylated targeting protein. A number of observations implicate phosphocaveolin in the feedback inhibition of Src-family kinases [24,31]. Src-family kinases are highly enriched in caveolae, due to both acylation and direct binding to caveolin. Caveolin-1 is phosphorylated in response to stimuli that activate Fyn, including oxidative stress and insulin, and stress-induced phosphorylation of caveolin requires Fyn. Stress-induced caveolin phosphorylation is self-limiting and attenuated by Csk: caveolin is phosphorylated to a much greater extent in Csk^{-/-} cells than in the Csk^{+/+} cells. Csk binds specifically to phosphocaveolin, which is one of only two tyrosine-phosphorylated proteins associated with Csk in fibroblasts and adipocytes. Association with Csk increases significantly in response to either insulin or stress. These data indicate that binding to phosphocaveolin is a major mechanism for the regulation of Csk in response to stress and insulin. Therefore, the role of caveolin phosphorylation in signal transduction becomes a question of the role that regulation of caveolar Src-family kinase activity plays in signal transduction.

6.3.5 Src-Family Kinases, Csk and Actin Remodeling

Src-family kinases are essential for cell proliferation, differentiation, and adhesion [96]. However, uncontrolled Src-family kinase activity leads to constitutive activation of mitogenic pathways, tumor promotion, and loss of cell attachment. Therefore, tight regulation of Src-family kinase activity is essential. The requirement for tight control of Src-family kinase activity is readily apparent in the regulation of actin/cell surface/extracellular matrix adhesions [97-100]. Both Src and Fyn are translocated to newly forming focal contacts during cell spreading. Srcfamily kinase activity is essential for the maintenance of these structures: inhibition of Src-family kinase activity leads to disassembly of adhesions. Paradoxically, constitutively active forms of Src also disrupt cell adhesion, and therefore the activity of Src in focal adhesions must be carefully regulated. During cell spreading there is an initial disruption of plasma membrane-actin attachments to allow membrane extension at the leading edge of the cell, followed by assembly of focal contacts and actin filaments behind the leading edge. Src-family kinases are activated at the cell edge during the initial stage of cell spreading to transiently relieve actin-induced tension at the plasma membrane, but must then be attenuated to allow actin reassembly. Attenuation occurs through phosphorylation by Csk. Cskdeficient cells exhibit unregulated Src-family kinase activity, hyperphosphorylation of actin-associated substrate proteins, impaired stress fiber formation and defects in cell adhesion [97]. Interestingly, anti-sense-mediated suppression of caveolin expression results in similar phenotypes, including elevated Src-family kinase activity and loss of cell adhesion [7,101].

6.3.6

Phosphocaveolin is Enriched at Sites of Attachment of the Actin Cytoskeleton to the Plasma Membrane

Significant data links caveolae and caveolins to the actin cytoskeleton [102,103]. In confluent, quiescent cells and in tissues, caveolin-1 is primary localized to the plasma membrane. The caveolin-1 at cell surface is tethered to the actin cytoskeleton [104,105]. Caveolin binds to actin through filamin, a protein that regulates cortical actin assembly [106]. Agents that disrupt the actin cytoskeleton cause rapid internalization of caveolin [105]. Caveolin-1 also redistributes in response to shear stress and during cell migration [30,32,107–109]. The muscle specific isoform caveolin-3 is also bound to actin, and loss of caveolin-3 causes a form of muscular dystrophy due to defects in the anchoring of the actin cytoskeleton to the plasma membrane [110,111].

Phosphocaveolin is ideally localized to act as the mediator between Src activation and Csk-induced inactivation in the regulation of actin assembly. Phosphocaveolin
128 6 Phosphorylation of Caveolin and Signaling from Caveolae



Fig. 6.5 Caveolin is phosphorylated at actin/plasma membrane attachment sites. Fibroblast cells plated onto fibronectin for 5 min (top) or 15 min. Actin is labeled with FITC-phalloidin (green). Phosphocaveolin is labeled with anti-PY14 antibody and Cy3-labeled secondary antibody (red). The boxed portions in the right-hand panels are shown enlarged on the left.

is highly enriched at or near focal adhesions at the ends of the actin stress fibers [18,24,27,29,32,34,35,109]. In migrating non-confluent cells, caveolin is cleared from the leading edge of the cell and is found predominantly in vesicles at the trailing edge of the cell. However, phosphocaveolin is highly enriched near forming focal adhesions at the cell edge in actively migrating cells, or in cells spreading on fibronectin. Double labeling with phalloidin showed that caveolin phosphorylation occurs at the ends of bundles of actin fibers at the edge of the cell (Fig. 6.5). The enrichment of phosphocaveolin near focal adhesions and the tight association of caveolin with cortical actin indicate that phosphorylation of caveolin may regulate the actin cytoskeleton through Csk-mediated inhibition of Src-family kinases. Consistent with a link between caveolin, Src-family kinases, and the actin cytoskeleton, integrin signaling through Fyn requires expression of caveolin-1 [62,101].

6.3.7 Abl in the Loop

Our data indicate that activation of Fyn in caveolae would be sufficient to induce caveolin tyrosine phosphorylation, although this phosphorylation is expected to be self-limiting and transient due to feedback inhibition through Csk. Both Abl and Fyn are required for robust, sustained caveolin phosphorylation. Our current model is that Fyn and Abl act synergistically in the phosphorylation of caveolin (Fig. 6.6). Extracellular stimuli such as insulin or stress activate Fyn in caveolae, which phosphorylates caveolin. This leads to the recruitment of Csk, attenuates Fyn, and limits the extent of phosphorylation of caveolin. At the same time, these signals activate Abl which translocates into caveolae where it also phosphorylates caveolin. Abl is not regulated by Csk, and remains active in these complexes, main-



Fig. 6.6 Abl kinase is required for stable phosphorylation of caveolin. Both Abl and Fyn are required for caveolin phosphorylation. Both are caveolin tyrosine kinases and directly phosphorylate it *in vitro*. Both are activated by oxidative stress. Fyn is resident in caveolae and may be required for the activa-

tion or recruitment of Abl to caveolae. Recruitment of Csk to phosphocaveolin would attenuate Fyn activity, leading to transient caveolin phosphorylation. Since Abl is not inhibited by Csk, recruitment to or activation of Abl in caveolae would induce stable phosphorylation of caveolin-1.

130 6 Phosphorylation of Caveolin and Signaling from Caveolae

taining a high level of caveolin phosphorylation. In the absence of Abl, caveolin phosphorylation is limited in extent, and is transient. Recruitment/activation of Abl is required for high-level, sustained phosphorylation of caveolin-1, and sustained inhibition of Src-family kinases. This represents a novel mechanism for the attenuation of Src-family kinase activity by Abl: phosphorylation of a scaffolding protein (caveolin) and recruitment of Csk. Paxillin, a substrate of both Abl and Src, is likely to organize a similar regulatory complex by binding to Abl, Src, and Csk [112]. Csk is recruited only after phosphorylation of paxillin.

Interestingly, Fyn and Abl are required for opposing stress-induced pathways in cells. Activation of Fyn is required for the induction of survival pathways [66,113], whereas activation of Abl induces cell death [114]. Activation of Fyn and Abl are also temporally different. Fyn is activated rapidly (within minutes), while Abl activation takes longer (maximum activation after 30 minutes). Low-level exposure to oxidative stress (low concentration or short duration) stimulates survival pathways, while higher exposures induce apoptotic pathways [115]. It is possible that lowlevel exposures stimulate only Fyn, hence survival pathways, while higher exposures are required to activate Abl and apoptotic pathways. Abl-induced recruitment of Csk to complexes containing Src-family kinases would be a mechanism to ensure that survival pathways are turned off, allowing apoptosis to proceed. Counterregulatory effects of Abl and Fyn have been observed in a number of additional signaling pathways [84,88,89], and negative regulatory circuits involving Abl, Fyn, and their substrates have been identified genetically in second-site repressor screens in Drosophila [82,83]. Currently, an Abl-specific inhibitor STI571 (Gleevec) is used in patients to treat chronic myelogenous leukemia. Therefore, counterregulatory complexes containing Src-family kinases and Abl have important therapeutic implications, particularly in the treatment of cancer.

6.3.8

Abl and Actin Remodeling

Abl and Src-family kinases also have opposing effects on the actin cytoskeleton (Abl stabilizes/Src destabilizes) and cell migration (Abl inhibits/ Src promotes). Signaling complexes that regulate focal adhesions are known to contain Abl and Src-family kinases, as well as substrates for both kinases [85–87]. Abl, Fyn and their substrates are associated with signaling complexes that organize N-WASP and Arp2/3 at the cell membrane. As members of this complex, Abl and Fyn play opposing roles in regulating actin assembly and remodeling. Fyn recruits and activates N-WASP and ARP2/3 which initiates actin polymerization. Abl inhibits actin polymerization by phosphorylating and inhibiting enabled (Ena) and activating profillin. Abl also strengthens focal contacts by inhibiting their breakdown through an unidentified mechanism. This is may be due to inhibition of Fyn through recruitment of Csk. Transient Src-family kinase activity is required for focal adhesion turnover and membrane extension during actin remodeling. Caveolin phosphorylation and recruitment of Src-family kinase activity could be

achieved during cell migration to allow transient activation then rapid Csk-induced inactivation of Src-family kinases. Stimulation of caveolin phosphorylation through recruitment/activation of Abl would inhibit Src-family kinase activity at focal contacts, stabilizing these structures.

6.3.9 Insulin-Induced Actin Remodeling, GluT4 Translocation, and Caveolae

Adipocytes are terminally differentiated and are not dividing. They are also not migrating or spreading on the extracellular matrix, and they do not have well-formed stress fibers. Therefore, what is the significance of caveolin phosphorylation and Csk-induced attenuation of Src-family kinase activity in response to insulin in these cells? Insulin increases glucose transport in fat and muscle cells by stimulating the translocation of GluT4 from intracellular vesicles to the plasma membrane. Insulin induces cortical actin remodeling which is required for Glut4 translocation [116,117]. Therefore, the regulation of cortical actin assembly is a key process in the stimulation of glucose transport by insulin. In a manner analogous to cell migration, cortical actin remodeling would require that the tension between the cell surface and the actin cytoskeleton first be transiently relieved, followed by rapid formation of new initiation sites for actin assembly at the membrane. Based on other modes of actin remodeling, it is likely that Src-family kinases and Csk play roles in insulin-induced actin remodeling in adipocytes.

In adipocytes, caveolin is co-localized with actin in unique complexes at the cell surface [118,119]. Disruption of these complexes, either through expression of dominant-negative forms of caveolin or by treatment with cyclodextran, blocks insulin-stimulated cortical actin assembly and GluT4 translocation [120]. Insulin-stimulated caveolin phosphorylation would activate Csk close to sites of cortical actin assembly, which would modulate the insulin-induced actin remodeling. Consistent with this, other stimuli that induce GluT4 translocation also stimulate caveolin phosphorylation, including endothelin-1, angiotensin II, and osmotic stress [14,15,18,28,32].

6.3.10 The Role of Caveolin Phosphorylation in Cells

Our current hypothesis is that caveolin organizes signaling complexes containing Fyn, Csk, and Abl that regulate cortical actin remodeling. Caveolin phosphorylation is required for the recruitment of both Csk and Abl into the complexes, which leads to negative regulation of the Src-family kinases that are resident in the caveolae. When the phosphorylation of caveolin is catalyzed by the Src-kinases themselves, this is a mechanism for the transient activation/rapid Csk-induced attenuation of Src-kinases through feedback inhibition. When the caveolin is phosphorylated by Abl, this regulatory mechanism would inhibit activation of a specific pool of Src family kinases. This regulatory complex involves the following three steps (Fig. 6.7):

- **132** 6 Phosphorylation of Caveolin and Signaling from Caveolae
 - 1) Signals activate Fyn leading to phosphorylation of substrates including caveolin-1.



2) Caveolin phosphorylation recruits Csk and attenuates Fyn activity.



 Prolonged signals activate/recruit Abl leading to prolonged caveolin phosphorylation and attenuation of Fyn.



Fig. 6.7 Caveolin organizes signaling complexes with Fyn, Csk and Abl that regulate cortical actin remodeling.

- 1. Insulin or stress activates Fyn that is resident in the caveolae. This leads to the phosphorylation of specific Fyn substrates involved in focal adhesion turnover, cell migration, and cell survival.
- 2. It also leads to the phosphorylation of caveolin-1 at Tyr14. Caveolin phosphorylation recruits Csk, which attenuates Fyn activity. The transient activation of Fyn allows for the transient release of actin from the plasma membrane. Fyn must be then be inactivated to allow subsequent reformation of actin contact sites.
- 3. Prolonged signals activate/recruit Abl, leading to high-level, sustained phosphorylation of caveolin and sustained attenuation of Fyn. This leads to inhibi-

tion of focal adhesion turnover and inhibition of cell migration, stabilization of actin, and can lead to apoptosis.

This model links caveolin phosphorylation to two pathways known to be impacted by caveolin expression: actin assembly/cell migration and control of cell growth.

6.4 Summary

Two signaling molecules required for caveolin phosphorylation have been identified: Fyn and Abl. The data suggest a novel mechanism for the attenuation of Srckinase activity by Abl: stable tyrosine phosphorylation of a scaffolding protein (caveolin-1) and recruitment of a negative regulator of Src-family kinase activity (Csk). The unexpected complexity of this pathway has important implications for treatment of diseases, including cancer and diabetes.

Three binding partners for phosphocaveolin have been identified: TRAF2, Grb7, and Csk. The identification of Csk as a binding partner for phosphorylated caveolin-1 led to a unique model for the function of caveolae in cells. It also demonstrated the utility of a powerful new technique to identify novel phosphotyrosine-directed protein interactions.

A novel feedback mechanism for Src-family kinase regulation has been identified: recruitment and activation of Csk by Src-family kinase substrates such as caveolin-1. Recruitment and activation of Abl in the complex induces sustained activation of Csk and attenuation of Src-family kinase activity, consistent with the counter-regulatory effects of these kinases. This previously undescribed, but simple, mechanism can explain a number of well-studied observations, including transient activation of Src-family kinases in forming focal contacts, and the counterregulatory effects of Abl and Src-family kinases in cell growth, migration, and actin polymerization.

A novel role for caveolae and caveolin tyrosine phosphorylation in cells was proposed: regulation of cortical actin assembly and/or remodeling. The caveolin-1/Csk/Src family kinase/Abl signaling complex may be involved in transmitting signals to the actin cytoskeleton (i. e., from integrins or oxidative stress) or transmitting signals from the actin cytoskeleton (i. e., from shear or osmotic stress).

Abbreviations

ACTH	adrenocorticotropin
EGF	epidermal growth factor
eNOS	endothelial nitric oxide synthase
FRET	fluorescence resonance energy transfer
IGF-1	insulin-like growth factor-1
IRS-1	insulin receptor substrate-1

134 References

PDGF	platelet-derived growth factor
SDS	sodium dodecyl sulfate
SH2	Src-homology 2
TNFα	tumor necrosis factor-alpha
VEGF	vascular endothelial growth factor

References

- 1 Glenney, J. R., and D. Soppet (1992). Sequence and expression of caveolin, a protein component of caveolae plasma membrane domains phosphorylated on tyrosine in Rous sarcoma virus-transformed fibroblasts. *Proc. Natl. Acad. Sci. USA* 89, 10517–10521.
- 2 Scherer, P. E., Tang, Z., Chun, M., Sargiacomo, M., Lodish, H. F., and Lisanti, M. P. (1995). Caveolin isoforms differ in their N-terminal protein sequence and subcellular distribution. *J. Biol. Chem.* 270, 16395–16401.
- 3 Scherer, P. E., Okamoto, T., Chun, M., Nishimoto, I., Lodish, H. F., and Lisanti, M. P. (1996). Identification, sequence, and expression of caveolin-2 defines a caveolin gene family. *Proc. Natl. Acad. Sci.* USA 93, 131–135.
- 4 Tang, Z., Scherer, P. E., Okamoto, T., Song, K., Chu, C., Kohtz, D. S., Nishimoto, I., Lodish, H. F., and Lisanti, M. P. (1996). Molecular cloning of caveolin-3, a novel member of the caveolin gene family expressed predominantly in muscle. *J. Biol. Chem.* 271, 2255–2261.
- 5 Scheiffele, P., Verkade, Fra, M., Virta, Simons, and Ikonen (1998). Caveolin-1 and -2 in the exocytic pathway of MDCK cells. *J. Cell Biol.* 140, 795–806.
- 6 Fra, A. M., Williamson, E., Simons, K., and Parton, R. G. (1995). De novo formation of caveolae in lymphocytes by expression of VIP21-caveolin. *Proc. Natl. Acad. Sci. USA* 92, 8655–8659.
- 7 Drab, M., Verkade, P., Elger, M., Kasper, M., Lohn, M., Lauterbach, B., Menne, J., Lindschau, C., Mende, F., Luft, F. C., Schedl, A., Haller, H., and Kurzchalia, T. V. (2001). Loss of caveolae, vascular dysfunction, and pulmonary defects in caveolin-1 gene-disrupted mice. *Science* 9, 2449–2452.

- 8 Razani, B., Engelman, J. A., Wang, X. B., Schubert, W., Zhang, X. L., Marks, C. B., Macaluso, F., Russell, R. G., Li, M., Pestell, R. G., Di Vizio, D., Hou, H., Jr., Kneitz, B., Lagaud, G., Christ, G. J., Edelmann, W., and Lisanti, M. P. (2001). Caveolin-1 null mice are viable but show evidence of hyperproliferative and vascular abnormalities. J. Biol. Chem. 276, 38121–38138.
- 9 Kurzchalia, T. V., and Parton, R. G. (1999). Membrane microdomains and caveolae. *Curr. Opin. Cell Biol.* 11, 424–431.
- 10 Li, S., Couet, J., and Lisanti, M. P. (1996). Src tyrosine kinases, Galpha subunits, and H-Ras share a common membraneanchored scaffolding protein, caveolin. Caveolin binding negatively regulates the auto-activation of Src tyrosine kinases. *J. Biol. Chem.* 271, 29182–29190.
- 11 Garcia-Cardena, G., Fan, R., Stern, D. F., Liu, J., and Sessa, W.C. (1996). Endothelial nitric oxide synthase is regulated by tyrosine phosphorylation and interacts with caveolin-1. *J. Biol. Chem.* 271, 27237–27240.
- 12 Okamoto, T., Schlegel, A., Scherer, P. E., and Lisanti, M. P. (1998). Caveolins, a family of scaffolding proteins for organizing "preassembled signaling complexes" at the plasma membrane. *J. Biol. Chem.* 273, 5419–5422.
- 13 Oh, P., and Schnitzer, J. E. (2001). Segregation of heterotrimeric G proteins in cell surface microdomains. G(q) binds caveolin to concentrate in caveolae, whereas G(i) and G(s) target lipid rafts by default. *Mol. Biol. Cell* 12, 685–698.
- 14 Mastick, C. C., Brady, M. J., and Saltiel, A. R. (1995). Insulin stimulates the tyrosine phosphorylation of caveolin. *J. Cell Biol.* 129, 1523–1531.

- 15 Mastick, C. C., and Saltiel, A. R. (1997). Insulin-stimulated tyrosine phosphorylation of caveolin is specific for the differentiated adipocyte phenotype in 3T3-L1 cells. *J. Biol. Chem.* 272, 20706– 20714.
- 16 Lee, H., Volonte, D., Galbiati, F., Iyengar, P., Lublin, D. M., Bregman, D. B., Wilson, M. T., Campos-Gonzalez, R., Bouzahzah, B., Pestell, R. G., Scherer, P. E., and Lisanti, M. P. (2000). Constitutive and growth factor-regulated phosphorylation of caveolin-1 occurs at the same site (Tyr-14) in vivo: identification of a c-Src/Cav-1/Grb7 signaling cassette. *Mol. Endocrinol.* 14, 1750–1775.
- 17 Kim, Y. N., Wiepz, G. J., Guadarrama, A. G., and Bertics, P. J. (2000). Epidermal growth factor-stimulated tyrosine phosphorylation of caveolin-1. Enhanced caveolin-1 tyrosine phosphorylation following aberrant epidermal growth factor receptor status. J. Biol. Chem. 275, 7481–7491.
- 18 Ushio-Fukai, M., Hilenski, L., Santanam, N., Becker, P., Ma, Y., Griendling, K., and RW, A. (2001). Cholesterol depletion inhibits epidermal growth factor receptor transactivation by angiotensin II in vascular smooth muscle cells: Role of cholesterol-rich microdomains and focal adhesions in angiotensin II signaling. J. Biol. Chem. 276, 48269–48275.
- 19 Muller, G., Jung, C., Wied, S., Welte, S., and Frick, W. (2001). Insulin-mimetic signaling by the sulfonylurea glimepiride and phosphoinositolglycans involves distinct mechanisms for redistribution of lipid raft components. *Biochemistry* 40, 14603–14620.
- 20 Newcomb, L. F., and Mastick, C. C. (2002). Src family kinase-dependent phosphorylation of a 29-kDa caveolin-associated protein. *Biochem. Biophys. Res. Commun.* 290, 1447–1453.
- 21 Kim, Y. N., Dam, P., and Bertics, P. J. (2002). Caveolin-1 phosphorylation in human squamous and epidermoid carcinoma cells: dependence on ErbB1 expression and Src activation. *Exp. Cell Res.* 280, 134–147.
- 22 Hua, H., Munk, S., and Whiteside, C. I. (2003). Endothelin-1 activates mesangial cell ERK1/2 via EGF-receptor transactiva-

tion and caveolin-1 interaction. Am. J. Physiol. Renal Physiol. 284, F303-F312.

- 23 Labrecque, L., Royal, I., Surprenant, D. S., Patterson, C., Gingras, D., and Beliveau, R. (2003). Regulation of vascular endothelial growth factor receptor-2 activity by caveolin-1 and plasma membrane cholesterol. *Mol. Biol. Cell* 14, 334–347.
- 24 Cao, H., Sanguinetti, A. R., and Mastick, C. C. (2004). Oxidative stress activates both Src-kinases and their negative regulator Csk and induces phosphorylation of two targeting proteins for Csk: caveolin-1 and paxillin. *Exp. Cell Res.* 294, 159–171.
- 25 Podar, K., Shringarpure, R., Tai, Y. T., Simoncini, M., Sattler, M., Ishitsuka, K., Richardson, P. G., Hideshima, T., Chauhan, D., and Anderson, K.C. (2004). Caveolin-1 is required for vascular endothelial growth factor-triggered multiple myeloma cell migration and is targeted by bortezomib. *Cancer Res.* 64, 7500–7506.
- 26 Fielding, P. E., Chau, P., Liu, D., Spencer, T. A., and Fielding, C. J. (2004). Mechanism of platelet-derived growth factor-dependent caveolin-1 phosphorylation: relationship to sterol binding and the role of serine-80. *Biochemistry* 43, 2578–2586.
- 27 Colonna, C., and Podesta, E. J. (2005). ACTH-induced caveolin-1 tyrosine phosphorylation is related to podosome assembly in Y1 adrenal cells. *Exp. Cell Res.* 304, 432–442.
- 28 Aoki, T., Nomura, R., and Fujimoto, T. (1999). Tyrosine phosphorylation of caveolin-1 in the endothelium. *Exp. Cell Res.* 253, 629–636.
- **29** Volonte, D., Galbiati, F., Pestell, R. G., and Lisanti, M. P. (2001). Cellular stress induces the tyrosine phosphorylation of caveolin-1 (Tyr(14)) via activation of p38 mitogen-activated protein kinase and c-Src kinase. Evidence for caveolae, the actin cytoskeleton, and focal adhesions as mechanical sensors of osmotic stress. *J. Biol. Chem.* 276, 8094–8103.
- 30 Rizzo, V., Morton, C., DePaola, N., Schnitzer, J. E., and Davies, P. F. (2003). Recruitment of endothelial caveolae into mechanotransduction pathways by flow conditioning in vitro. Am. J. Physiol. Heart Circ. Physiol. 285, H1720–H1729.
- 31 Sanguinetti, A. R., Cao, H., and Corley Mastick, C. (2003). Fyn is required for ox-

idative- and hyperosmotic-stress-induced tyrosine phosphorylation of caveolin-1. *Biochem. J.* 376, 159–168.

- 32 Sanguinetti, A. R., and Mastick, C. C. (2003). c-Abl is required for oxidative stress-induced phosphorylation of caveolin-1 on tyrosine 14. *Cell Signal*. 15, 289–298.
- 33 Navakauskiene, R., Treigyte, G., Gineitis, A., and Magnusson, K. E. (2004). Identification of apoptotic tyrosine-phosphorylated proteins after etoposide or retinoic acid treatment. *Proteomics* 4, 1029–1041.
- 34 Radel, C., and Rizzo, V. (2005). Integrin mechanotransduction stimulates caveolin-1 phosphorylation and recruitment of Csk to mediate actin reorganization. Am. J. Physiol. Heart Circ. Physiol. 288, H936– H945.
- 35 Brown, G., Rixon, H. W., and Sugrue, R. J. (2002). Respiratory syncytial virus assembly occurs in GM1-rich regions of the host-cell membrane and alters the cellular distribution of tyrosine phosphorylated caveolin-1. *J. Gen. Virol.* 83, 1841–1850.
- 36 Minshall, R. D., Sessa, W. C., Stan, R. V., Anderson, R. G., and Malik, A. B. (2003). Caveolin regulation of endothelial function. Am. J. Physiol. Lung Cell Mol. Physiol. 285, L1179–L1183.
- 37 Shajahan, A. N., Tiruppathi, C., Smrcka, A. V., Malik, A. B., and Minshall, R. D. (2004). Gbetagamma activation of Src induces caveolae-mediated endocytosis in endothelial cells. *J. Biol. Chem.* 279, 48055–48062.
- 38 Sargiacomo, M., Scherer, P. E., Tang, Z. L., Casanova, J. E., and Lisanti, M. P. (1994). In vitro phosphorylation of caveolin-rich membrane domains: identification of an associated serine kinase activity as a casein kinase II-like enzyme. Oncogene 9, 2589–2595.
- 39 Nomura, R., and Fujimoto, T. (1999). Tyrosine-phosphorylated caveolin-1: immunolocalization and molecular characterization. *Mol. Biol. Cell* 10, 975–986.
- 40 Lee, H., Park, D. S., Wang, X. B., Scherer, P. E., Schwartz, P. E., and Lisanti, M. P. (2002). Src-induced phosphorylation of caveolin-2 on tyrosine 19. Phospho-caveolin-2 (Tyr(P)19) is localized near focal adhesions, remains associated with lipid

rafts/caveolae, but no longer forms a high molecular mass hetero-oligomer with caveolin-1. J. Biol. Chem. 277, 34556–34567.

- 41 Kiss, A. L., Botos, E., Turi, A., and Mullner, N. (2004). Ocadaic acid treatment causes tyrosine phosphorylation of caveolin-2 and induces internalization of caveolae in rat peritoneal macrophages. *Micron* 35, 707–715.
- 42 Wang, X. B., Lee, H., Capozza, F., Marmon, S., Sotgia, F., Brooks, J. W., Campos-Gonzalez, R., and Lisanti, M. P. (2004). Tyrosine phosphorylation of caveolin-2 at residue 27: differences in the spatial and temporal behavior of phospho-Cav-2 (pY19 and pY27). *Biochemistry* 43, 13694–13706.
- 43 Zaas, D. W., Duncan, M. J., Li, G., Wright, J. R., and Abraham, S. N. (2005). Pseudomonas invasion of type I pneumocytes is dependent on the expression and phosphorylation of caveolin-2. *J. Biol. Chem.* 280, 4864–4872.
- 44 Robinson, L. J., Pang, S., Harris, D. S., Heuser, J., and James, D. E. (1992). Translocation of glucose transporter (GLUT4) to the cell surface in permeabilized 3T3-L1 adipocytes: effects of ATP, insulin, and GTPgS and localization of GLUT4 to clathrin lattices. *J. Biol. Chem.* 117, 1181–1196.
- 45 Scherer, P. E., Lisanti, M. P., Baldini, G., Sargiacomo, M., Mastick, C. C., and Lodish, H. F. (1994). Induction of caveolin during adipogenesis and association of GLUT4 with caveolin-rich vesicles. *J. Cell Biol.* 127, 1233–1243.
- 46 Kandror, K. V., Stephens, J. M., and Pilch, P. F. (1995). Expression and compartmentalization of caveolin in adipose cells: coordinate regulation with and structural segregation from GLUT4. J. Cell Biol. 129, 999–1006.
- 47 Razani, B., Combs, T. P., Wang, X. B., Frank, P. G., Park, D. S., Russell, R. G., Li, M., Tang, B., Jelicks, L. A., Scherer, P. E., and Lisanti, M. P. (2002). Caveolin-1-deficient mice are lean, resistant to dietinduced obesity, and show hypertriglyceridemia with adipocyte abnormalities. *J. Biol. Chem.* 277, 8635–8647.
- 48 Cohen, A. W., Razani, B., Wang, X. B., Combs, T. P., Williams, T. M., Scherer, P. E., and Lisanti, M. P. (2003). Caveolin-

1-deficient mice show insulin resistance and defective insulin receptor protein expression in adipose tissue. *Am. J. Physiol. Cell Physiol.* 285, C222–C235.

- 49 Cohen, A. W., Razani, B., Schubert, W., Williams, T. M., Wang, X. B., Iyengar, P., Brasaemle, D. L., Scherer, P. E., and Lisanti, M. P. (2004). Role of caveolin-1 in the modulation of lipolysis and lipid droplet formation. *Diabetes* 53, 1261–1270.
- 50 Lisanti, M. P., Scherer, P. E., Tang, Z., and Sargiacomo, M. (1994). Caveolae, caveolin and caveolin-rich membrane domains: a signalling hypothesis. *Trends Cell Biol.* 4, 231–235.
- 51 Chun, M., Liyanage, U. K., Lisanti, M. P., and Lodish, H. F. (1994). Signal transduction of a G protein-coupled receptor in caveolae: colocalization of endothelin and its receptor with caveolin. *Proc. Natl. Acad. Sci. USA* 91, 11728–11732.
- 52 Shenoy-Scaria, A. M., Dietzen, D. J., Kwong, J., Link, D. C., and Lublin, D.M. (1994). Cysteine3 of Src family protein tyrosine kinase determines palmitoylation and localization in caveolae. *J. Cell Biol.* 126, 353–363.
- 53 Chang, W. J., Ying, Y. S., Rothberg, K. G., Hooper, N. M., Turner, A. J., Gambliel, H. A., De Gunzburg, J., Mumby, S. M., Gilman, A. G., and Anderson, R.G. (1994). Purification and characterization of smooth muscle cell caveolae. *J. Cell Biol.* 126, 127–138.
- 54 Schnitzer, J. E., Oh, P., Jacobson, B. S., and Dvorak, A. M. (1995). Caveolae from luminal plasmalemma of rat lung endothelium: microdomains enriched in caveolin, Ca(2+)-ATPase, and inositol trisphosphate receptor. *Proc. Natl. Acad. Sci.* USA 92, 1759–1763.
- 55 Wiese, R. J., Mastick, C. C., Lazar, D. F., and Saltiel, A. R. (1995). Activation of mitogen-activated protein kinase and phosphatidylinositol 3'-kinase is not sufficient for the hormonal stimulation of glucose uptake, lipogenesis, or glycogen synthesis in 3T3-L1 adipocytes. J. Biol. Chem. 270, 3442–3446.
- 56 Brady, M., Bourbonais, F., and Saltiel, A. (1998). The activation of glycogen synthase by insulin switches from kinase inhibition to phosphatase activation during

adipogenesis in 3T3-L1 cells. J. Biol. Chem. 273, 14063–14066.

- 57 Smith, R. M., Harada, S., Smith, J. A., Zhang, S., and Jarett, L. (1998). Insulininduced protein tyrosine phosphorylation cascade and signalling molecules are localized in a caveolin-enriched cell membrane domain. *Cell Signal.* 10, 355–362.
- 58 Gustavsson, J., Parpal, S., Karlsson, M., Ramsing, C., Thorn, H., Borg, M., Lindroth, M., Peterson, K. H., Magnusson, K. E., and Stralfors, P. (1999). Localization of the insulin receptor in caveolae of adipocyte plasma membrane. *FASEB J*. 13, 1961–1971.
- 59 Glenney, J. R. (1989). Tyrosine Phosphorylation of a 22-kDa protein is correlated with transformation by Rous Sarcoma Virus. J. Biol. Chem. 264, 20163–20166.
- 60 Ko, Y. G., Liu, P., Pathak, R. K., Craig, L. C., and Anderson, R.G. W. (1998). Early effects of pp60(v-src) kinase activation on caveolae. *I. Cell. Biochem.* 71, 524–535.
- 61 Lee, H., Woodman, S. E., Engelman, J. A., Volonte, D., Galbiati, F., Kaufman, H. L., Lublin, D. M., and Lisanti, M. P. (2001). Palmitoylation of caveolin-1 at a single site (Cys-156) controls its coupling to the c-Src tyrosine kinase: targeting of dually acylated molecules (GPI-linked, transmembrane, or cytoplasmic) to caveolae effectively uncouples c-Src and caveolin-1 (TYR-14). *J. Biol. Chem.* 276, 35150–35158.
- 62 Wary, K. K., Mariotti, A., Zurzolo, C., and Giancotti, F. G. (1998). A requirement for caveolin-1 and associated kinase Fyn in integrin signaling and anchorage-dependent cell growth. *Cell* 94, 625–634.
- 63 Zacharias, D. A., Violin, J. D., Newton, A. C., and Tsien, R.Y. (2002). Partitioning of lipid-modified monomeric GFPs into membrane microdomains of live cells. *Science* 296, 913–916.
- 64 Li, S., Seitz, R., and Lisanti, M.P. (1996). Phosphorylation of caveolin by Src tyrosine kinases. J. Biol. Chem. 271, 3863–3868.
- 65 Abe, J., Takahashi, M., Ishida, M., Lee, J. D., and Berk, B.C. (1997). c-Src is required for oxidative stress-mediated activation of big mitogen-activated protein kinase 1. J. Biol. Chem. 272, 20389–20394.

- 138 References
 - 66 Abe, J., Okuda, M., Huang, Q., Yoshizumi, M., and Berk, B. C. (2000). Reactive oxygen species activate p90 ribosomal S6 kinase via Fyn and Ras. J. Biol. Chem. 275, 1739–1748.
 - 67 Kapus, A., Di Ciano, C., Sun, J., Zhan, X., Kim, L., Wong, T. W., and Rotstein, O. D. (2000). Cell volume-dependent phosphorylation of proteins of the cortical cytoskeleton and cell-cell contact sites. The role of Fyn and FER kinases. *J. Biol. Chem.* 275, 32289–32298.
 - 68 Alland, L., Peseckis, S. M., Atherton, R. E., Berthiaume, L., and Resh, M.D. (1994). Dual myristylation and palmitylation of Src family member p59fyn affects subcellular localization. *J. Biol. Chem.* 269, 16701–16705.
 - 69 Muller, G., Jung, C., Wied, S., Welte, S., Jordan, H., and Frick, W. (2001). Redistribution of glycolipid raft domain components induces insulin-mimetic signaling in rat adipocytes. *Mol. Cell. Biol.* 21, 4553–4567.
 - 70 Sun, X. J., Pons, S., Asano, T., Myers, M. G., Glasheen, E., and White, M. F. (1996). The Fyn tyrosine kinase binds IRS-1 and forms a distinct signaling complex during insulin stimulation. *J. Biol. Chem.* 271, 10583–10587.
 - 71 Songyang, Z., and Cantley, L.C. (1995). Recognition and specificity in protein tyrosine kinase-mediated signalling. *Trends Biochem. Sci.* 20, 470–475.
 - 72 Cujec, T. P., Medeiros, P. F., Hammond, P., Rise, C., and Kreider, B. L. (2002). Selection of v-Abl tyrosine kinase substrate sequences from randomized peptide and cellular proteomic libraries using mRNA display. *Chem. Biol.* 9, 253–264.
 - 73 Beitner-Johnson, D., and LeRoith, D. (1995). Insulin-like growth factor-I stimulates tyrosine phosphorylation of endogenous c-Crk. J. Biol. Chem. 270, 5187–5190.
 - 74 Lin, W. H., Huang, C. J., Liu, M. W., Chang, H. M., Chen, Y. J., Tai, T. Y., and Chuang, L. M. (2001). Cloning, mapping, and characterization of the human sorbin and SH3 domain containing 1 (SORBS1) gene: a protein associated with c-Abl during insulin signaling in the hepatoma cell line Hep3B. *Genomics* 74, 12–20.

- 75 Sorokin, A., and Reed, E. (1998). Insulin stimulates the tyrosine dephosphorylation of docking protein p130cas (Crk-associated substrate), promoting the switch of the adaptor protein crk from p130cas to newly phosphorylated insulin receptor substrate-1. *Biochem. J.* 334, 595–600.
- 76 Mastick, C. C., Sanguinetti, A. R., Knesek, J. H., Mastick, G. S., and Newcomb, L. F. (2001). Caveolin-1 and a 29-kDa caveolin-associated protein are phosphorylated on tyrosine in cells expressing a temperature-sensitive v-Abl kinase. *Exp. Cell Res.* 266, 142–154.
- 77 Koleske, A. J., Baltimore, D., and Lisanti, M. P. (1995). Reduction of caveolin and caveolae in oncogenically transformed cells. *Proc. Natl. Acad. Sci. USA* 92, 1381–1385.
- 78 Cao, H., Courchesne, W. E., and Corley Mastick, C. (2002). A phosphotyrosine-dependent dihybrid protein interaction screen reveals a role for phosphorylation of caveolin-1 on tyrosine 14: recruitment of C-terminal Src kinase. J. Biol. Chem. 277, 8771–8774.
- 79 Sun, X., Majumder, P., Shioya, H., Wu, F., Kumar, S., Weichselbaum, R., Kharbanda, S., and Kufe, D. (2000). Activation of the cytoplasmic c-Abl tyrosine kinase by reactive oxygen species. *J. Biol. Chem.* 275, 17237–17240.
- 80 Sun, X., Wu, F., Datta, R., Kharbanda, S., and Kufe, D. (2000). Interaction between protein kinase C delta and the c-Abl tyrosine kinase in the cellular response to oxidative stress. J. Biol. Chem. 275, 7470–7473.
- 81 Plattner, R., Kadlec, L., DeMali, K. A., Kazlauskas, A., and Pendergast, A. M. (1999). c-Abl is activated by growth factors and Src family kinases and has a role in the cellular response to PDGF. *Genes Dev.* 13, 2400–2411.
- 82 Gertler, F. B., Hill, K. K., Clark, M. J., and Hoffmann, F. M. (1993). Dosage-sensitive modifiers of Drosophila abl tyrosine kinase function: prospero, a regulator of axonal outgrowth, and disabled, a novel tyrosine kinase substrate. *Genes Dev.* 7, 441–453.
- 83 Gertler, F. B., Comer, A. R., Juang, J. L., Ahern, S. M., Clark, M. J., Liebl, E. C., and Hoffmann, F.M. (1995). enabled, a

dosage-sensitive suppressor of mutations in the Drosophila Abl tyrosine kinase, encodes an Abl substrate with SH3 domainbinding properties. *Genes Dev.* 9, 521–533.

- 84 Shishido, T., Akagi, T., Ouchi, T., Georgescu, M. M., Langdon, W. Y., and Hanafusa, H. (2000). The kinase-deficient Src acts as a suppressor of the Abl kinase for Cbl phosphorylation. *Proc. Natl. Acad. Sci.* USA 97, 6439–6444.
- 85 Lanier, L. M., and Gertler, F. B. (2000). From Abl to actin: Abl tyrosine kinase and associated proteins in growth cone motility. *Curr. Opin. Neurobiol.* 10, 80–87.
- 86 Krause, M., Sechi, A. S., Konradt, M., Monner, D., Gertler, F. B., and Wehland, J. (2000). Fyn-binding protein (Fyb)/SLP-76-associated protein (SLAP), Ena/vasodilator-stimulated phosphoprotein (VASP) proteins and the Arp2/3 complex link T cell receptor (TCR) signaling to the actin cytoskeleton. J. Cell Biol. 149, 181–194.
- 87 Small, J. V., Stradal, T., Vignal, E., and Rottner, K. (2002). The lamellipodium: where motility begins. *Trends Cell Biol.* 12, 112–120.
- 88 Furstoss, O., Dorey, K., Simon, V., Barila, D., Superti-Furga, G., and Roche, S. (2002). c-Abl is an effector of Src for growth factor-induced c-myc expression and DNA synthesis. *EMBO J.* 21, 514–524.
- 89 Arnaud, L., Ballif, B. A., Forster, E., and Cooper, J. A. (2003). Fyn tyrosine kinase is a critical regulator of Disabled-1 during brain development. *Curr. Biol.* 13, 9–17.
- 90 Feng, X., Gaeta, M. L., Madge, L. A., Yang, J. H., Bradley, J. R., and Pober, J. S. (2001). Caveolin-1 associates with TRAF2 to form a complex that is recruited to tumor necrosis factor receptors. *J. Biol. Chem.* 276, 8341–8349.
- 91 Duxbury, M. S., Ito, H., Ashley, S. W., and Whang, E. E. (2004). CEACAM6 cross-linking induces caveolin-1-dependent, Src-mediated focal adhesion kinase phosphorylation in BxPC3 pancreatic adenocarcinoma cells. *J. Biol. Chem.* 279, 23176–23182.
- 92 Howell, B. W., and Cooper, J.A. (1994). Csk suppression of Src involves movement of Csk to sites of Src activity. *Mol. Cell. Biol.* 14, 5402–5411.

- 93 Kawabuchi, M., Satomi, Y., Takao, T., Shimonishi, Y., Nada, S., Nagai, K., Tarakhovsky, A., and Okada, M. (2000). Transmembrane phosphoprotein Cbp regulates the activities of Src-family tyrosine kinases. *Nature* 404, 999–1003.
- 94 Takeuchi, S., Takayama, Y., Ogawa, A., Tamura, K., and Okada, M. (2000). Transmembrane phosphoprotein Cbp positively regulates the activity of the carboxyl-terminal Src kinase, Csk. J. Biol. Chem. 275, 29183–29186.
- 95 Torgersen, K. M., Vang, T., Abrahamsen, H., Yaqub, S., Horejsi, V., Schraven, B., Rolstad, B., Mustelin, T., and Tasken, K. (2001). Release from tonic inhibition of T cell activation through transient displacement of C-terminal Src kinase (Csk) from lipid rafts. J. Biol. Chem. 276, 29313–29318.
- 96 Thomas, S. M., and Brugge, J. S. (1997). Cellular functions regulated by Src family kinases. Annu. Rev. Cell. Dev. Biol. 13, 513–609.
- 97 Thomas, S. M., Soriano, P., and Imamoto, A. (1995). Specific and redundant roles of Src and Fyn in organizing the cytoskeleton. *Nature* 376, 267–271.
- 98 Fincham, V. J., and Frame, M. C. (1998). The catalytic activity of Src is dispensable for translocation to focal adhesions but controls the turnover of these structures during cell motility. *EMBO J.* 17, 81–92.
- 99 Arthur, W. T., Petch, L. A., and Burridge, K. (2000). Integrin engagement suppresses RhoA activity via a c-Src-dependent mechanism. *Curr. Biol.* 10, 719–722.
- 100 Li, L., Okura, M., and Imamoto, A. (2002). Focal adhesions require catalytic activity of Src family kinases to mediate integrin-matrix adhesion. *Mol. Cell. Biol.* 22, 1203–1217.
- 101 Wei, Y., Yang, X., Liu, Q., Wilkins, J. A., and Chapman, H. A. (1999). A role for caveolin and the urokinase receptor in integrin-mediated adhesion and signaling. *J. Cell Biol.* 144, 1285–1294.
- 102 van Deurs, B., Roepstorff, K., Hommelgaard, A. M., and Sandvig, K. (2003). Caveolae: anchored, multifunctional platforms in the lipid ocean. *Trends Cell Biol.* 13, 92–100.
- 103 Navarro, A., Anand-Apte, B., and Parat, M.O. (2004). A role for caveolae in cell migration. *FASEB J.* 18, 1801–1811.

- 140 References
 - 104 Thomsen, P., Roepstorff, K., Stahlhut, M., and van Deurs, B. (2002). Caveolae are highly immobile plasma membrane microdomains, which are not involved in constitutive endocytic trafficking. *Mol. Biol. Cell* 13, 238–250.
 - 105 Mundy, D. I., Machleidt, T., Ying, Y. S., Anderson, R. G., and Bloom, G. S. (2002). Dual control of caveolar membrane traffic by microtubules and the actin cytoskeleton. J. Cell Sci. 115, 4327–4339.
 - 106 Stahlhut, M., and van Deurs, B. (2000). Identification of filamin as a novel ligand for caveolin-1: evidence for the organization of caveolin-1-associated membrane domains by the actin cytoskeleton. *Mol. Biol. Cell* 11, 325–337.
 - 107 Isshiki, M., Ando, J., Yamamoto, K., Fujita, T., Ying, Y., and Anderson, R.G. (2002). Sites of Ca(2+) wave initiation move with caveolae to the trailing edge of migrating cells. *J. Cell Sci.* 115, 475– 484.
 - 108 Boyd, N. L., Park, H., Yi, H., Boo, Y. C., Sorescu, G. P., Sykes, M., and Jo, H. (2003). Chronic shear induces caveolae formation and alters ERK and Akt responses in endothelial cells. *Am. J. Physiol. Heart Circ. Physiol.* 285, H1113-H1122.
 - 109 Beardsley, A., Fang, K., Mertz, H., Castranova, V., Friend, S., and Liu, J. (2005). Loss of caveolin-1 polarity impedes endothelial cell polarization and directional movement. *J. Biol. Chem.* 280, 3541–3547.
 - 110 Minetti, C., Sotgia, F., Bruno, C., Scartezzini, P., Broda, P., Bado, M., Masetti, E., Mazzocco, M., Egeo, A., Donati, M. A., Volonte, D., Galbiati, F., Cordone, G., Bricarelli, F. D., Lisanti, M. P., and Zara, F. (1998). Mutations in the caveolin-3 gene cause autosomal dominant limb-girdle muscular dystrophy. *Nat. Genet.* 18, 365–368.
 - 111 Hagiwara, Y., Sasaoka, T., Araishi, K., Imamura, M., Yorifuji, H., Nonaka, I., Ozawa, E., and Kikuchi, T. (2000). Caveo-

lin-3 deficiency causes muscle degeneration in mice. *Hum. Mol. Genet.* 9, 3047–3054.

- **112** Turner, C. E. (2000). Paxillin interactions. *J. Cell Sci.* 113, 4139–4140.
- **113** Abe, J., and Berk, B.C. (1999). Fyn and JAK2 mediate Ras activation by reactive oxygen species. *J. Biol. Chem.* 274, 21003–21010.
- 114 Kumar, S., Bharti, A., Mishra, N. C., Raina, D., Kharbanda, S., Saxena, S., and Kufe, D. (2001). Targeting of the c-Abl tyrosine kinase to mitochondria in the necrotic cell death response to oxidative stress. J. Biol. Chem. 276, 17281–17285.
- 115 Martindale, J. L., and Holbrook, N. J. (2002). Cellular response to oxidative stress: signaling for suicide and survival. *J. Cell Physiol.* 192, 1–15.
- 116 Tong, P., Khayat, Z. A., Huang, C., Patel, N., Ueyama, A., and Klip, A. (2001). Insulin-induced cortical actin remodeling promotes GLUT4 insertion at muscle cell membrane ruffles. *J. Clin. Invest.* 108, 371–381.
- 117 Kanzaki, M., and Pessin, J. E. (2001). Insulin-stimulated GLUT4 translocation in adipocytes is dependent upon cortical actin remodeling. J. Biol. Chem. 276, 42436–42444.
- 118 Kanzaki, M., and Pessin, J. E. (2002). Caveolin-associated filamentous actin (Cavactin) defines a novel F-actin structure in adipocytes. J. Biol. Chem. 277, 25867–25869.
- 119 Parton, R. G., Molero, J. C., Floetenmeyer, M., Green, K. M., and James, D. E. (2002). Characterization of a distinct plasma membrane macrodomain in differentiated adipocytes. *J. Biol. Chem.* 277, 46769–46778.
- 120 Watson, R. T., Shigematsu, S., Chiang, S. H., Mora, S., Kanzaki, M., Macara, I. G., Saltiel, A. R., and Pessin, J. E. (2001). Lipid raft microdomain compartmentalization of TC10 is required for insulin signaling and GLUT4 translocation. *J. Cell Biol.* 154, 829–840.

7 Role of Lipid Microdomains in the Formation of Supramolecular Protein Complexes and Transmembrane Signaling

György Vámosi, Andrea Bodnár, György Vereb, János Szöllösi, and Sándor Damjanovich

7.1

Introduction

The plasma membrane is the theater of all kinds of material and information exchange between a cell and its environment - that is, the "outer world". The plasma membrane is basically a lipid bilayer that accommodates a significant number of proteins with diverse structures and tasks necessary for the proper function of cells. Concepts about the architecture of the plasma membrane have changed dramatically through the past decades. The "rigid membrane" concept was replaced by the Singer-Nicolson (S-N) fluid mosaic membrane model in 1972, which postulated the random distribution and free lateral mobility of proteins in the unstructured lipid environment [1]. Construction of the S-N model was greatly advanced by the classical experiment of Frye and Edidin, which demonstrated intermixing of distinct molecular species in the plasma membrane of heterokaryons of human and murine lymphocytes [2]. Although the S-N model was able to explain many phenomena taking place in the cell membrane, it had a limited validity. First of all, it was mostly applicable to the quasi-symmetric circulating blood cells. It has been well known for a long time that cells built into solid tissues are frequently polarized as required by their physical environment and biological function, a most extreme example being the organization of brush border cells, into two discrete regions - the apical and the basolateral - with distinct functions and thus different lipid and protein composition. The picture of "freely moving proteins in the uniform lipid sea" was challenged by experimental observations suggesting the locally restricted mobility of proteins in the lipid bilayer [3,4]. Data indicating the existence of hierarchically built protein complexes even in nonpolarized cells also contradicted the S-N model [5,6]. As for the concept of "uniform lipid sea", an important step forward was the discovery of sphingolipid- and cholesterol-enriched lipid domains - that is, lipid rafts - constituting a proof that phase-separation of lipids observed in model membranes also occurs in the more complex plasma membrane of living cells [7,8]. All these observations led to the "membrane microdomain" concept, namely the compartmentalization/organization of membrane proteins and lipids into nonrandom, well-defined, yet dynamic structures, which exist at different time and size scales.

7.1.1

Lateral organization of membrane proteins

According to our present knowledge, lateral arrangement of membrane components – whether proteins or lipids – is a general phenomenon, which is essential for the proper functioning of both the individual molecules and the whole of the plasma membrane. Processes associated with the plasma membrane (e.g., signal transduction, protein sorting, etc.) demand the cooperation of various membrane proteins and are often accompanied by dynamic rearrangement of the two-dimensional macromolecular patterns at the cell surface. The structured, and at the same time dynamic, nature of the plasma membrane allows accumulation of relevant molecules in particular membrane areas whilst excluding others, thus preventing their interaction [9–11].

The basic organization level of membrane proteins is defined by their molecular association/physical proximity, and is referred to as nanometer- or small-scale clusters [6]. These clusters can be formed either in a homologous or heterologous fashion; that is, their molecular components can be either identical or distinct. The generalized occurrence of such protein complexes was initially proposed in the early 1980s, based on the preferential accommodation of the genetically determined membrane-spanning α -helices of proteins into distinct membrane microdomains [12]. This assumption has been supported by a considerable amount of experimental data. A major asset in studying small-scale protein assemblies was the adaptation of fluorescence resonance energy transfer (FRET) to cellular systems (see Section 7.2).

Different types of small-scale protein patterns can be distinguished in the plasma membrane [5]:

- In many cases, a given "functional unit" comprises several components/subunits (e.g., in multi-chain immune recognition receptor complexes [13]).
- External stimuli can also induce cluster formation through reorganization of membrane proteins in the plane of the plasma membrane (e.g., ligand-evoked aggregation of the epidermal growth factor receptor [14]).
- Apart from the aforesaid examples, where the individual components are either preassembled or come together upon ligand binding/external stimuli, co-localization of apparently unrelated proteins can also be observed in many cases (e.g., association of the insulin receptor and major histocompatability complex (MHC) I glycoproteins [15,16]). Revealing the co-localization of such proteins may call our attention to their potential functional relationship.

Beyond the small-scale protein associations (i.e., colocalization on the 1- to 10-nm scale), clustering of membrane proteins at a second hierarchical level can also be observed in many cases [5,11,17]. These so-called large-scale clusters can be several hundred nanometers in diameter and could contain tens to thousands of proteins. Some of the biophysical methods applicable for studying large-scale protein clusters are summarized in Section 7.2.

7.1.2 Factors controlling the organization of membrane proteins

The mosaic-like organization of lipid structures – that is, the assembly of lipids with similar physico-chemical character (saturation, length, etc.) – into distinct domains [4] plays an important role in the formation of both small- and large-scale protein assemblies. The lipid domain structure of the plasma membrane may cause selective accumulation of membrane proteins (or their exclusion from distinct membrane areas) through preferential interaction of the membrane-interacting region of a given protein with a select class of lipids [18]. Lipid rafts, enriched in cholesterol and glycosphingolipids, are a special type of lipid domains [4,7]. Lipid rafts were shown to accumulate a set of membrane proteins as well as cytosolic signaling elements, and were proposed to act as specialized signaling compartments [13,19–23]. There are several vehicles that may help target proteins into lipid rafts:

- a shell of annular lipids encasing the transmembrane segment of proteins [24];
- the addition of a GPI-anchor or saturated acyl chains via post-translational modification [25–28]; and
- interactions with proteins having a high affinity to the lipid raft environment.

Rafts are complex and dynamic structures which vary both in size and protein content [10]. By enabling the dynamic association/reassociation of proteins residing within the same domain, lipid rafts provide a platform for their functional cooperation. The dynamic exchange of components between rafts with different composition (or between raft and non-raft regions) as well as the aggregation of smaller rafts, raft "microdomains" into "macrodomains" also play an important role in the spatiotemporal organization of plasma membrane-associated processes taking place in lipid rafts [29].

In addition to lipid domains, cells have other means by which they can control the formation and maintenance of specific assemblies of membrane proteins (for reviews, see [3,5,11,30]). Protein-protein interactions - for example, between the transmembrane α -helices – may also contribute to the stability of protein clusters and membrane microdomains [31]. Vesicular transport mechanisms can produce selective accumulation of membrane proteins by means of "directed" transport of vesicular components to a given membrane region [32,33]. The cytoskeleton may act either by actively directing redistribution of proteins in the plasma membrane, or by restricting their motion and trapping them in a given membrane area by barriers formed from joint structures of the membrane and the cytoskeleton [34]. The assembly of lipid rafts into macrodomains is also governed by the actin cytoskeleton (see [29] and references therein). The free diffusion of membrane proteins can also be hindered by interaction with elements of different cytosolic signaling elements (e.g., G-proteins, kinases). These factors are not independent of each other and may act in concert to generate supramolecular structures in the plasma membrane.

144 7 Role of Lipid Microdomains in the Formation of Supramolecular Protein Complexes

In this chapter selected examples for the existence of hierarchically built protein complexes are provided. Their regulation by the lipid domain structure of the plasma membrane and the functional consequences of the formation of protein clusters in transmembrane signaling will also be discussed.

7.2

Biophysical Strategies for Studying the Lateral Organization of Membrane Proteins

Cell-surface proteins display non-random distribution patterns ranging in size from a few nanometers to microns. Here, we describe some widely used fluorescence-based biophysical methods that can be used to study the different hierarchical levels of the lateral organization and interactions of membrane proteins. Classical biochemical and immunological methods (chemical cross-linking, co-immunoprecipitation, detergent resistance analysis, etc.) provide valuable information on the interaction of membrane proteins, but have several drawbacks. First, with these methods proteins cannot be studied in their native environment. Second, the use of conventional extraction and isolation procedures inherent to these techniques may also disrupt protein-protein interactions, or may induce the formation of artificial protein aggregates. These methods provide information only on the bulk properties of interactions, without resolving cell-to-cell or subcellular variations. On the other hand, fluorescence-based observations of appropriately labeled molecules allow the detection of interactions with a subcellular resolution. Fluorescence detection is sensitive (down to the level of single molecules under select conditions), relatively non-destructive, and specific due to the high affinity of interaction between commonly used markers (specific antibodies, toxins, etc.) and their targets. Genes tailored to code for proteins carrying a fluorescent tag (GFP and its spectral variants) are routinely used to report on the subcellular distribution of proteins.

7.2.1

Determination of Domain Size and Overlap between Fluorescence Distributions using Fluorescence Microscopy

The lateral distribution of cell-surface proteins and their co-localization in membrane microdomains can be studied by digital imaging microscopy with a resolution of 200–300 nm, as set by Abbe's law [35]. The application of confocal laser scanning microscopy or multiphoton excitation [36] greatly improves image sharpness and contrast by excluding photons arising from out-of-focus planes. This resolution is not sufficient to directly indicate molecular associations; however, it allows the observation of overlap/segregation between different protein clusters or lipid microdomains. Quantitative analysis of the size of clusters/microdomains can be derived from the spatial autocorrelation function of the fluorescence intensity distribution [39]. Using this approach, cluster sizes of predominantly raftlocalized proteins IL-2R α , HLA I and II, as well as GPI-anchored proteins CD48, were determined to be ~600–700 nm, which coincided well with cluster sizes determined from electron microscopic analysis of immunogold-labeled samples. These domain sizes correlated well with the mean barrier-free path measured for the MHC I glycoprotein and its truncation mutants [37], as well as dimensions of confinement zones derived from single particle tracking in which E-cadherin or epidermal growth factor receptor (EGFR) molecules could freely diffuse [38]. Disruption of lipid rafts by cholesterol extraction blurred the boundaries of protein clusters and extended their diameter to ~1000 nm [39].

The extent of overlap between clusters of different membrane proteins and lipid microdomains can be characterized by the cross-correlation coefficient of the pixel intensities of individual fluorescence distributions [11,39,40]. For a pair of images x and y, the cross-correlation coefficient is calculated as:

$$C = \frac{\sum_{i,j} (\mathbf{x}_{ij} - \langle \mathbf{x} \rangle) (\mathbf{y}_{ij} - \langle \mathbf{y} \rangle)}{\sqrt{\sum_{i,j} (\mathbf{x}_{ij} - \langle \mathbf{x} \rangle)^2 \sum_{i,j} (\mathbf{y}_{ij} - \langle \mathbf{y} \rangle)^2}}$$
(7.1)

where x_{ij} and y_{ij} are fluorescence intensities at pixel coordinates i,j in images x and y, and $\langle x \rangle$, $\langle y \rangle$ are the mean intensities. The theoretical maximum is C = 1 for identical images, and a value of 0 implies independent localization of the labeled molecules.

7.2.2 Fluorescence Resonance Energy Transfer (FRET)

Techniques utilizing Förster-type FRET offer convenient tools for mapping the spatial distribution and molecular vicinity relations of membrane proteins on live cells *in situ*, without any major interference with the physiological condition of the cells. Several techniques have been developed to measure FRET on cell surfaces [5,41–43].

FRET is a process wherein energy is transferred non-radiatively from an excited donor fluorophor to a nearby acceptor via dipole-dipole coupling [44]. In order for FRET to occur, it is necessary that the dipole moments of the dyes have a proper relative orientation and the emission spectrum of the donor and the absorption spectrum of the acceptor molecule overlap with each other [45]. From the point of view of biological applications, the most important property of FRET is that its rate is inversely proportional to the sixth power of the donor-acceptor distance; hence, it is a sensitive tool for the determination of inter- or intramolecular distances in the 1- to 10-nm range, and can be applied as a "spectroscopic ruler" [46]. The efficiency of FRET, *E*, is defined as the fraction of excitation quanta transferred from the donor to the acceptor, which can be expressed as:

$$E = \frac{k_{FRET}}{k_f + k_{nf} + k_{FRET}} = \frac{R_o^6}{R_o^6 + R^6}$$
(7.2)

146 7 Role of Lipid Microdomains in the Formation of Supramolecular Protein Complexes

where *R* is the donor-acceptor distance, R_o is the so-called Förster distance, at which the FRET efficiency is 50% for the given donor-acceptor pair, and k_{FRET} , k_f and k_{nf} are the rate constants of de-excitation by FRET, fluorescence emission and non-radiative processes other than FRET, respectively. R_o is usually 5–10 nm, which defines the distance range for which FRET is applicable. In practice, the detection of FRET is based on the measurement of one or more of the following physical parameters: a) the decreased intensity of the donor (donor quenching); b) the enhanced emission of the acceptor (sensitized emission); c) the decreased fluorescence lifetime of the donor; d) the increased fluorescence anisotropy of the donor; or e) the decreased photobleaching rate of the donor.

Measurement of FRET by microscopy provides subcellular mapping of proteinprotein interactions, allowing the visual identification of compartments/organelles where the interactions of interest take place (for a review, see [47]). Without going into detail, the most commonly used FRET microscopic methods are described below.

Based on the simultaneous detection of three (or in the case of using autofluorescence correction, four) fluorescence intensities (autofluorescence, donor and acceptor channels and FRET channel) at each pixel, and using appropriate controls for determining the spectral spillover between the channels, FRET efficiency can be measured even at rather low expression levels [48].

With the acceptor photobleaching technique, the extent of donor quenching due to FRET is detected in a fairly simple way [49,50]. If the acceptor is irreversibly photodestroyed by selective illumination at the acceptor's absorption wavelength, the intensity of the donor increases, and from the extent of increase E can be calculated:

$$E = 1 - \frac{I_{DA}}{I_D} \tag{7.3}$$

where I_{DA} and I_D are the background-corrected donor fluorescence intensities with and without acceptor – that is, before and after photobleaching. The advantage of the method is its simplicity and its requirement for only donor-acceptor double-labeled samples.

The donor photobleaching (pbFRET) method exploits the increased resistivity of the donor to photobleaching in the presence of acceptor. Photobleaching is initiated from the excited state. FRET reduces the fluorescence lifetime of the donor; that is, the dye spends less time in the excited state, resulting in an elongated photobleaching time constant:

$$E = 1 - \frac{T_D}{T_{DA}} \tag{7.4}$$

where T_D and T_{DA} are the photobleaching time constants of the donor in the absence and presence of acceptor, respectively [50,51].

The fluorescence lifetime of the donor can be directly measured by using the imaging version of phase fluorimetry, fluorescence lifetime imaging microscopy (FLIM) [52]. Lifetime is a fairly robust parameter, which makes FLIM an attractive method in spite of its relatively complex instrumentation.

Although FRET microscopy has a clear advantage in providing subcellular information on interactions at microscopic resolution, it requires large numbers of cells to be evaluated to provide statistically valid data, and this is both time-consuming and labor-intensive. To determine small quantitative changes in proteinprotein interactions, at least several hundreds or thousands of cells must be evaluated. A suitable alternative is that of flow cytometric cell-by-cell FRET measurement, which provides the mean FRET efficiency for each cell in the analyzed population [53,54]. The high number of cells that can be analyzed by flow cytometry provides excellent statistics, high accuracy, and reproducibility.

7.2.3

Fluorescence Cross-Correlation Spectroscopy: Analysis of Protein Co-Mobility

Most FRET methods are limited to recording a static picture – a "snapshot" of molecular proximities of the labeled biomolecules – without being able to report on the temporal stability of molecular interactions. Co-localization at the nanometer scale as revealed by FRET does not necessarily mean that the studied proteins form stable complexes with one another. The co-mobility – that is, the joint diffusive motion of proteins – is an evidence for their stable interaction, and this can be investigated by using the dynamic method of fluorescence cross-correlation spectroscopy (FCCS), the two-channel version of fluorescence correlation spectroscopy (FCS). A recent compilation on the theory and applications of FCS can be found in a publication edited by two pioneers of the technique, Rigler and Elson [55].

In FCS, a laser beam focused to a subfemtoliter volume element excites fluorophores diffusing across the sensitive volume (a prolate ellipsoid defined by the surface at which the detection efficiency drops to e^{-2} times the value at the center), giving rise to a fluctuating fluorescence signal. The rate of fluctuations is related to the mobility of the fluorescing molecules (i. e., the diffusion coefficient). Typically, the intensity autocorrelation function is calculated from the intensity versus time signal (either on-line by a dedicated correlator card or off-line), and this function is fitted according to model functions assuming different mechanisms of diffusion and accounting for various photophysical processes (singlet-triplet transition, dark state formation, photobleaching) or chemical reactions (e.g., protonation) taking place in the system. For a system with *n* different species diffusing in 2D (the plane of the plasma membrane) labeled with dyes undergoing singlet-triplet transition [56], the autocorrelation function is as follows:

$$G(\tau) = \frac{\left\langle \delta F(t) \cdot \delta F(t+\tau) \right\rangle}{\left\langle F \right\rangle^2} = \frac{\int_0^{\infty} \delta F(t) \cdot \delta F(t+\tau) dt}{\left\langle F \right\rangle^2} = \frac{1}{N} \cdot \frac{1 - \Theta_{tr} + \Theta_{\mu} e^{-\tau/\tau_{tr}}}{1 - \Theta_{tr}} \cdot \sum_i \phi_i \left(1 + \frac{\tau}{\tau_i} \right)^{-1}$$
(7.5)

The left part of the equation is the definition of the autocorrelation function: the square brackets refer to averaging the expression over the duration T of the meas-

148 7 Role of Lipid Microdomains in the Formation of Supramolecular Protein Complexes

urement, the variable τ is the lag time (the time difference between the samples taken from the F(t) curve), $\langle F \rangle$ is the mean fluorescence intensity over the studied time interval, and $\delta F(t) = F(t) - \langle F \rangle$ is the deviation of the actual fluorescence intensity from the mean. *N* is the mean number of molecules in the sensitive volume, Θ_{tr} is the fraction of molecules in the triplet state, τ_{tr} is the phosphorescence lifetime, τ_i is the diffusion time (the mean dwell time of a molecule in the sensitive volume) and φ_i is the weight of the *i*th species. The diffusion time is inversely proportional to the diffusion coefficient D_i :

$$D_i = \omega_{xy}^2 / 4\tau_i \tag{7.6}$$

with ω_{xy} indicating the lateral radius of the sensitive volume.

Thus, the parameters of major importance derived from the autocorrelation function regarding protein-protein interactions are the diffusion coefficient and the local concentration of the labeled molecules. The value of *D* decreases if the labeled proteins form aggregates or interact with the cytoskeleton. Another good measure of the aggregation (homo-association) state of the studied proteins is the fluorescence intensity of the jointly diffusing units, which is simply the ratio $\langle F \rangle / N$.

The diffusion coefficient has only a weak dependence on the molecular weight $(D \propto MW^{-1/3})$; thus, in a relatively heterogeneous system such as the plasma membrane of a live cell it is not always possible to distinguish between monomeric and dimeric states.

In FCCS, two molecular species are labeled with preferably distinctly excitable and detectable fluorophores. In case the two molecular species are associated, their joint diffusive motion will result in parallel fluctuations of the fluorescence intensities $F_a(t)$ and $F_b(t)$ in the two detection channels. In this case, the crosscorrelation function $G^x(\tau)$ (see Eq. 7.7) has a nonzero amplitude $G^x(0)$, which is proportional to the concentration c_{ab} of the complexed fraction of molecular species "a" and "b" (see Eq. 7.8):

$$G^{\times}(\tau) = \frac{\left\langle \delta F_a(t) \cdot \delta F_b(t+\tau) \right\rangle}{\left\langle F_a \right\rangle \left\langle F_b \right\rangle}$$
(7.7)

$$G^{x}(0) \propto \frac{C_{ab}}{V_{eff}\left(c_{a,tot}\right)\left(c_{b,tot}\right)}$$
(7.8)

where $c_{a,tot}$ and $c_{b,tot}$ are the total concentrations of molecules *a* and *b* in free state or in complex, and V_{eff} is the so-called effective volume. The actual form of the cross-correlation function also depends on the geometrical parameters of the laser foci and the diffusion properties of the different molecular species [57,58].

7.2.4 Atomic Force Microscopy (AFM)

In the version of this technique applied to cellular imaging, a needle with a sharp tip (radius of curvature 10–50 nm) is scanned over the cell surface at a pressing force on the order of 0.1 to 10 pN, and the height of the needle is recorded at each position, thus generating a topographic image of the soft cell surface. The major advantage of AFM as compared to other microscopic techniques is that it provides a better resolution than optical microscopy and yet allows the observation of living cells, which is impossible with electron microscopy. In order to study the cellsurface distribution of specific membrane proteins by AFM, immunogold labeling was applied to provide specificity. This approach was successfully applied to study distinct hierarchical levels of the cell surface distribution of class I and class II MHC molecules in immunocompetent cells [59,60].

7.2.5

Scanning Near-Field Optical Microscopy (SNOM)

This technique combines the enhanced lateral and vertical resolution characteristic of AFM with simultaneous measurements of topographic and optical signals. SNOM can achieve a spatial resolution beyond the diffraction limit by scanning a sub-wavelength-sized aperture, confining the excitation beam to the close proximity of the sample. The resolution depends on the size of the aperture and the distance between the probe and the sample, rather than on the wavelength. The most commonly used SNOM probe is an aluminum-coated tapered optical fiber with an aperture <100 nm, positioned very close to the surface (~10 nm). The position of the probe is controlled by a piezo-electric stand, and the light emitted by the sample is collected either by an objective or through the same optical fiber. SNOM can simultaneously map topographic and optical properties with high spatial resolution, has single molecule sensitivity, and is relatively noninvasive. In addition, by using fluorescent labels it can provide a high level of molecular specificity. SNOM detected a highly patchy distribution of some fluorescent lipid analogues reflecting a lipid domain structure in fixed, dried cells [61]. The sizes of these patches were consistent with the sizes of domains implied by measurements of lateral diffusion with fluorescence photobleaching recovery. SNOM was also used to observe clustering of fluorescently labeled membrane receptors; clusters of platelet-derived growth factor (PDGF) receptors down to 80-100 nm diameter were observed on glioblastoma cells [50], and activation-dependent clustering of ErbB2 was revealed on breast tumor cells [62].

7.3

The Immunological Synapse

One of the best-characterized examples of specialized signaling domains where the dynamic assembly of both small- and large-scale protein associations have a crucial role, is the contact region formed between a T cell and a target or an antigen-presenting cell (APC), termed the immunological synapse (IS) [63,64]. Formation of the IS is initiated by recognition of the peptide antigen presented in complex with MHC glycoproteins on the APC or target cell by the T cell receptor complex (TCR) on the appropriate T cell [65,66]. Recognition of the antigen causes the redistribution of numerous molecules in both cells and leads to the formation of well-defined junctional structures at the interface of the two cells. Participation of the TCR along with adhesion and co-stimulatory molecules, co-receptors and associated cytosolic signaling elements on T cells as well as recruitment of MHCpeptide complexes and adhesion molecules on APCs can be observed in all ISs [67-69]. Supramolecular organization of these molecules is generated by their clustering/segregation at both the micrometer and submicrometer (nanometer) scale. ISs mature through discrete stages characterized by high-order temporal and spatial cooperation of multiple elements (membrane proteins, signaling molecules) required for appropriate function [66-68].

Involvement of lipid rafts in T cell signaling in general, and in the IS formation in particular, has been extensively studied and firmly established [19,29,70,71]. It is hypothesized that upon synapse formation small, individual rafts are assembled into raft macrodomains in a process regulated by the actin cytoskeleton [29]. Assembly of raft macrodomains occurs in three stages. Engagement of TCR initiates signals for actin polymerization followed by actin- and myosin-dependent migration of rafts to the site of cell signaling. Raft microdomains coalesce and, as a consequence, signaling proteins residing in discrete domains are now proximal within the same microenvironment, which enhances their interactions and amplifies the initial signals resulting in further raft assembly and signal amplification. The cascade continues until arrested by stop signals, which might include protein tyrosine phosphatases [29].

Despite the common features, ISs display diversity both in function and architecture, depending on the nature of interacting cells and as a consequence, that of the interacting molecular partners [67–69,72]. In the prototypic "bull-eye" structure, a central supramolecular activation cluster (cSMAC) containing TCR and MHC-peptide is surrounded by a pSMAC (peripheral SMAC) of adhesion molecules [63,64,73,74]. The "secretory synapse" formed between cytotoxic T cells (CTLs) and their targets has a double cSMAC containing the secretory apparatus juxtaposed with the TCR cluster [75]. The formation of several small, distinct TCR clusters instead of a large, single one (multifocal synapses) or nonclustered, diffuse distribution of TCR (nonsegregated IS) in the contact zone was also detected in many cases [69,72]. Dynamic contacts allowing migration (amoeboid movement) of T cells across the surface of the counterpart cells are presumably nonsegregated but at the same time mobile structures (migratory synapses) [76].



Fig. 7.1 Measurement of MHC I homoassociation on OCM-3 uveal melanoma cells. The solid line denotes the distribution of FRET efficiency values between MHC I molecules measured by flow cytometry on a cell-by-cell basis. MHC I was targeted by Cy3- and Cy5conjugated L368 mAbs specific for β 2-microglobulin (β 2m, light chain of MHC I). The

positive FRET value (peaking at ~10%) indicates homoassociation of MHC I. As a positive control, the intramolecular FRET efficiency between the heavy chain of MHC I and β 2m targeted by Cy3-W6/32 and Cy5-L368 mAbs, respectively, was also determined (dotted line).

The IS may fulfill numerous functions with varying importance for a particular cell-cell interaction [67,68]. For example, it may enhance and/or prolong signaling, integrate different signaling pathways, direct granule release and cytokine secretion, terminate signaling processes, and balance enhancing and terminating signals.

Whereas antigen-induced redistribution of the relevant molecules on T cells has been extensively characterized (for reviews, see [63,67–69,72]), much less is known about the behavior of MHC and adhesion molecules or the mechanisms controlling their accumulation on target cells or APCs. Therefore, in the following section focus is centered on data regarding the distribution of MHC as well as intracellular adhesion molecules (ICAM-1) on APCs and target cells.

By using FRET and lateral diffusion (single particle tracking) experiments, clustering (self-association) of MHC I glycoproteins was observed at the surface of various human cell types [77–80] (see Fig. 7.1). A recent detergent-solubility analysis of their self-association properties also confirmed that homotypic association is an inherent property of MHC I (and MHC II) molecules [81], in accordance with earlier observations of their spontaneous clustering after reconstitution into liposome model systems [82]. Electron and scanning force microscopic experiments also disclosed the nonrandom (clusterized) organization of MHC I molecules at a higher hierarchical level: immunogold-labeled MHC I molecules were observed to

152 7 Role of Lipid Microdomains in the Formation of Supramolecular Protein Complexes

form domains of several hundred nanometers diameter [59]. The degree of MHC I oligomerization showed good correlation with the expression of free MHC I heavy chains [lacking β 2-microglobulin (β 2m); "FHC"]: both of these were significant on activated or transformed/tumor cells [83]. The two forms of heavy chains participated in common small- and large-scale clusters at the surface of human B cells, as revealed by FRET and SNOM experiments [84]. Culturing cells with β 2m resulted in a decreased homotypic association of intact MHC I heterodimers and their reduced co-clustering with free heavy chains [84]. According to these data, FHCs likely have an important contribution to MHC I clustering: their involvement seems to stabilize MHC I clusters and *vice versa*, their functionally active conformation, which is still capable of rebinding β 2m, may also be stabilized by participation in these clusters [83,84]. Otherwise, FHCs would undergo irreversible denaturation and become functionally inactive and/or would be either internalized or released in a soluble form from the cell surface [85,86].

Clustered cell-surface distribution and anomalous diffusion of MHC II glycoproteins as well as their heteroassociation with MHC I were also reported in numerous cell types, including APCs [77,87–89]. Atomic force and electron microscopic data showed that MHC II molecules form homoclusters not only on the nanometer scale attainable by FRET, but also at a higher hierarchical level, in the micrometer distance range [60]. Electron microscopy revealed that a fraction of MHC II molecules was heteroclustered with MHC I at the same hierarchical level [60]. Molecular associations were detected between the ICAM-1 adhesion molecules and MHC glycoproteins at the surface of human T and B lymphoma cell lines [77,90]. In addition, a high degree of ICAM-1 self-association was found on HUT102 B2 human T lymphoma cells [77]. The above-mentioned association motifs of ICAM-1 and MHC glycoproteins were also observed on uveal melanoma and colon carcinoma cells. Interferon (IFN)- γ changed the expression levels of MHC and ICAM-1 molecules as well as inducing the re-arrangement of their spatial distribution/association patterns on these cells [78,91].

Since clustering of MHC and ICAM-1 molecules could be observed in the IS [64,74], it is reasonable to assume that in vivo formation of the aforesaid association patterns of MHC I, MHC II and ICAM-1 proteins may promote IS formation, and their high local concentration can significantly increase the avidity of APC-T cell interaction [63,92]. Indeed, diminishing MHC I oligomerization on target cells by β 2m treatment considerably reduced the efficiency of activation and effector function of allospecific cytotoxic T lymphocytes [84]. This hypothesis is also supported by studies with soluble MHC:peptide multimers (dimers, trimers or tetramers) showing that aggregation of MHC molecules may significantly increase the efficiency of activation/immune response of T cells [93–95].

Although to a different extent, constitutive or inducible association of MHC glycoproteins and ICAM-1 molecules with lipid rafts could be observed in various cell types [39,78,90,96–101]. Disruption of lipid raft integrity with filipin or methyl- β -cyclodextrin caused dispersion of large-scale MHC clusters [39,97] and translocation of MHC to the soluble membrane fractions [90,96]. Dissociation of small-scale MHC II clusters was also detected upon cholesterol depletion [97]. These data



Fig. 7.2 Triple co-localization of lipid rafts, ICAM-1 and MHC I on OCM-3 human uveal melanoma cells, as detected by CLSM. Lipid rafts were labeled by Alexa Fluor 488-conjugated cholera toxin B subunit (panel A). ICAM-1 (panel B) and MHC I (panel C) molecules were targeted by Alexa 546-MEM111 and Cy5-W6/32 mAbs, respectively. Membrane areas where two or three membrane

species co-localize are indicated with mixed colors in the overlay image (panel D). The pairwise cross-correlation coefficients between the fluorescence distributions indicate a high level of co-localization between the observed markers: $C_{AB} \sim 0.7$, $C_{AC} \sim 0.7$, $C_{BC} = 0.65$. The applied colors are pseudocolors. (Scale bar = 2 μ m.)

indicate that lipid raft association is one of the underlying mechanisms responsible for MHC clustering.

Co-localization of MHC I and ICAM-1 could be observed within lipid rafts of colon carcinoma and uveal melanoma cells [78,91] (Fig. 7.2). On human B lymphoblastoid cells, physical association of MHC I and ICAM-1 could be detected both in detergent-insoluble and in detergent-soluble membranes [90]. The disruption of raft integrity resulted in a significant loss of MHC I and ICAM-1 from the raft fraction, but their association was still detectable, implying that this interaction does not critically depend on the structure of rafts.

The accommodation of MHC and ICAM-1 in lipid rafts has important functional consequences in the process of antigen presentation. Lipid raft-assisted compartmentation of MHC II was shown to enhance the efficiency of antigen presentation to CD4⁺ T cells [96,97,102]. This effect was more prominent at low

154 7 Role of Lipid Microdomains in the Formation of Supramolecular Protein Complexes

antigen doses, suggesting that rafting MHC domains are critical for T cell activation by rare MHC II-peptide complexes and less important when the antigen density is high [96,97,102,103]. It was also demonstrated that APC lipid rafts, raftassociated relevant MHC II-peptide complexes and even immunologically irrelevant MHC II molecules accumulate at the IS [103]. Upon maturation of the IS, relevant MHC II-peptide complexes were sorted to the central region of the interface, while irrelevant MHC II molecules were excluded from this site [103]. Similar to T cells, remodeling of the APC surface after the initial TCR signal seems to be also cytoskeleton-dependent [97].

Co-clustering of ICAM-1 with MHC I in lipid rafts of B lymphoblastoid cells was also shown to facilitate efficient presentation of viral peptides to CTLs [90]. Raft accommodation of MHC I-ICAM-1 assemblies enabled specific recruitment of Src kinases harbored in lipid rafts to these complexes. Since activity of Src kinases along with preserved integrity of rafts was critical for the CTL response, it can be suggested that engagement of raft-included MHC I and ICAM-1 initiates intracellular signaling, leading to the concomitant migration of rafts and MHC I-ICAM-1 assemblies to the area of the initial target cell-CTL contact [90]. This mechanism could provide the linkage between antigen recognition and early immunological synapse formation [90,92].

7.4

Voltage-Gated K⁺ Channels in Lipid Rafts: Possible Involvement in Local Regulatory Processes

Kv1.3 channels [104,105], the dominant voltage-gated potassium channels of T lymphocytes, play key roles in the control of membrane potential and calcium signaling, thereby affecting signal transduction pathways leading to the antigeninduced activation of these cells [106,107].

Activation of T cells through the TCR generates an oscillatory Ca²⁺ signal, which is created by a release from the Ca²⁺-storage compartments of the endoplasmic reticulum (ER) triggered by IP₃ and consequential Ca²⁺ influx from the extracellular space [108]. The depolarizing Ca²⁺ influx must be counterbalanced by the activation of K⁺ channels, clamping the membrane potential at negative values and thereby providing a sufficient driving force for further Ca²⁺ entry. Voltage-gated K⁺ (Kv) channels form highly K⁺-selective pores that are conformationally switched open or closed by changes in membrane voltage [109]. In addition to the classical synapse-forming molecules, recently we described the recruitment of Kv1.3 to the immunological synapse [110], which may be mediated by its association with lipid rafts [111,112]. Elevation of the cholesterol content of the membrane modulated the gating properties of these channels [111]. By using confocal laser scanning microscopy (CLSM) and FRET, we have demonstrated clustering of Kv1.3 channels and their co-localization with the TCR/CD3 complex in Jurkat cells and human peripheral CTLs [110,113], indicating a possible crosstalk between the TCR signaling complex and Kv1.3. In nonengaged CTLs, small patches containing Kv1.3 and TCR/CD3 complexes were evenly distributed on the cell surface, whereas they were both enriched at the IS formed at the CTL-target cell interface while maintaining their molecular proximity [110]. This is in accordance with biochemical studies suggesting the interaction of Kv1.3 channels with protein kinases, Lck and PKC, as well as various adaptor molecules, such as hDlg, PSD-95 (95 kDa postsvnaptic density protein), Kvβ2, ZIP1 (PKC-ζ-interacting protein 1) and ZIP2, and the co-receptor CD4 (for reviews, see [107,114,115]). Physical association of Kv1.3 channels with the TCR might underline the importance of previous findings, in which regulation of Kv1.3 function by protein tyrosine kinase- and PKC-dependent phosphorylation was described. Besides the possibility of Kv1.3 channel regulation during formation of the IS, and the consequent modulation of Ca²⁺ signaling, localization of Kv1.3 in the IS raises the possibility of reciprocal regulation of IS function by nearby Kv1.3 [115]. For example, conformational changes in membrane proteins driven by the membrane potential might affect the antigen-recognition process [11,116], or local K⁺ efflux through Kv1.3 might activate the function of β_1 -integrin, a possible determinant of the stability of the immunological synapse [117].

In addition to Kv1.3 channels, a whole class of voltage-gated Kv K⁺ channels are targeted to lipid rafts, a finding which bridges a gap between classical molecular immunology and electrophysiology (for a review, see [112]).

7.5 Cell Fusion as a Tool for Studying Dynamic Behavior of Protein Clusters

Whereas the Frye-Edidin experiment proved the lateral redistribution of membrane proteins on the micrometer scale [2], due to technical limitations, it could not resolve whether intermixing of proteins results from the movement of largescale clusters with constant protein composition or instead, proteins could be exchanged between clusters at the molecular level by dynamic association-dissociation events (dynamically changing composition). In order to address this question, the classical experiment was repeated with some modifications [118]. Intermixing of membrane proteins was studied on homokaryons of human lymphoblasts where, before fusion, the two cells were labeled separately with antibodies (or their Fab fragments) carrying spectrally distinct fluorophores targeting the proteins of interest. The process of intermixing was monitored by simultaneous application of photobleaching FRET and SNOM techniques, which provided a resolution power well beyond that of the original experiment and allowed us to explore the dynamic behavior of both hierarchical levels of protein clusters (see Section 7.2).

As noted in Section 7.3, clustering of MHC glycoproteins is well-characterized and has an important role in antigen presentation [59,60,83,84,102]. Both smalland large-scale clusters of MHC I were found to be dynamic: dissociation and reassociation of small-scale protein complexes and reformation of large-scale associations took place after the fusion of cells that were labeled with fluorescein- and rhodamine-conjugated Fab fragments against MHC I heavy chains. The redistribution of micrometer-scale clusters preceded that of the small-scale clusters, corroborating the hierarchical organization of MHC I [118].

Large-scale homoclusters of MHC II were as dynamic as those of MHC I. Smalland large-scale heteroclusters of MHC I and MHC II also showed dynamic behavior in cell fusion experiments. At the same time, small-scale associations of MHC II were static as revealed by FRET: intermixing of components did not take place between nanometer-scale clusters of MHC II even 80 minutes after fusion [118]. This may be explained by the molecular structure of MHC II which, in its functional dimer form, has two membrane-spanning α -helical transmembrane domains. It can be hypothesized, that hydrophobic interactions between these domains are strong enough to prevent the mixing of small-scale clusters of MHC II [17].

One may picture this situation as overlapping large-scale clusters of MHC I and MHC II containing dynamic small-scale homoclusters of MHC I and static small-scale clusters of MHC II. Although the small-scale homoclusters of MHC II are not dynamic, they engage in dynamic small-scale association with MHC I [118].

There was no difference in the behavior of transferrin receptors and the GPIanchored CD48 protein, representatives of nonraft- and raft-associated proteins, respectively. Therefore, it could be concluded that the lipid microenvironment, on its own, does not determine the dynamic properties of protein associations. Forces resulting in protein clustering, whether they are related to membrane trafficking or to microdomain formation, do not generally inhibit the dynamic exchange of protein components between protein clusters, if the clusters reside in the same type of microdomain [118].

As a conclusion, one can say that in the composition of both small- and largescale membrane protein clusters dynamism is the rule, rather than the exception. Whereas preformed large-scale homoclusters of proteins are generally not associated in a static manner, and exchange of components between like clusters is usually observed, tight interactions in the small-scale associations of some proteins could completely block the exchange of these molecules between nanometersized clusters [118].

7.6

Lipid Rafts as Platforms for Cytokine Receptor Assembly and Signaling

Interleukin-2 (IL-2) and -15 (IL-15) are substantially involved in controlling T cell homeostasis and function [119,120,121]. Their receptors comprise three distinct components: while the α -chains are cytokine-specific, the β - and γ_c -subunits are employed by both IL-2 and IL-15. In addition, the so-called common γ_c -chain is the component of a series of other cytokine receptors – that is, the members of the γ_c cytokine receptor family (IL-4, IL-7, IL-9, and IL-21). As a result of combining various subunits, several forms of receptor complexes with different affinities may exist at the cell surface. Heterodimerization of the intracellular domains of the β - and γ_c -chains was found critical for one set of signaling events shared by both

cytokines. In this case the two cytokines activate similar signaling pathways involving Janus kinase (JAK1 and JAK3)-assisted tyrosine phosphorylation of downstream signaling molecules (e.g., STAT3 and STAT5). Sharing of common receptor subunits explains the redundancy in the biological actions of IL-2 and IL-15 (e.g., stimulation of T cell proliferation) [119,120]. In addition to the shared functions, they can also exhibit opposing contributions to T-cell-mediated immunity [119,120]. Since IL-2 plays a pivotal role in activation-induced cell death (AICD), it is critical in the elimination of self-reactive T cells and so in peripheral self-tolerance. At the same time, IL-15 manifests anti-apoptotic actions and inhibits IL-2-mediated AICD. IL-2 and IL-15 play opposing roles in the control of the homeostasis of CD8⁺ memory phenotype T cells. IL-15 provides potent and selective stimulation of memory phenotype CD8⁺ T cells *in vivo*, whereas IL-2 inhibits the persistence of these cells.

FRET data suggested that, in contrast to the earlier "sequential subunit-organization" (affinity conversion) model proposing ligand-induced association of initially separate subunits [122], the three chains of the high-affinity IL-2R (IL- $2R\alpha\beta\gamma_c$ complex are preassembled in the plasma membrane of human T lymphoma cells even in the absence of IL-2 [123]. Binding of IL-2 or IL-15 was reported to differentially modulate the conformation of the receptor complex: addition of IL-2 made the heterotrimer more compact, while IL-15 loosened the interaction/ proximity between the IL-2R α and γ_c -chains [123]. A similar preassembly of the heterotrimeric IL-15R (IL-15R $\alpha\beta\gamma_c$), as well as the molecular proximity of IL-15R α and IL-2Ra was demonstrated on human T cells expressing all the elements of the IL-2/IL-15 receptor system [98] (Fig. 7.3). Whereas neither IL-2, nor IL-15 affected the molecular association of the two α -chains, the interaction between the β - and IL-15Rα subunits became tighter upon IL-15 treatment, as indicated by FRET measurements. Based on these data, a heterotetrameric model of the IL-2/IL-15R complex can be envisaged: binding of IL-2 or IL-15 rearranges the subunits to form the appropriate $\alpha\beta\gamma_c$ high-affinity receptor complex, while the other α -chain rotates or moves away from the site of cytokine-receptor interaction. FRET experiments also indicated the homodimeric/oligomeric molecular association of IL-15R α [98]. Homoassociation of IL-2Ra may also occur in T lymphoma cells, although in a cell type-dependent manner [124].

It was shown by using biophysical (FRET and CLSM) and biochemical approaches that IL-2/IL-15R subunits are mainly partitioned into lipid rafts in the plasma membrane of human T lymphoma cells [39,79,98] (see Fig. 7.3). These rafts contained, among others, GM1 gangliosides, the GPI-anchored CD48 protein and a major fraction of MHC glycoproteins. At the same time the domains containing transferrin receptors (coated pits) were clearly distinct from GM1-containing lipid rafts (Fig. 7.4).

In addition, the high-affinity IL-2R was shown to be resistant to cold detergent extraction both in murine and human T lymphoma cells [79,125]. It was also demonstrated that the integrity of lipid rafts has a crucial role in organizing the lateral distribution of IL-2R and it is also essential for IL-2-mediated signaling [39,79]. Disruption of the native structure of lipid rafts by cholesterol extraction resulted in

158 7 Role of Lipid Microdomains in the Formation of Supramolecular Protein Complexes



the dispersion of the above raft components and simultaneously abrogated STAT3/ STAT5 tyrosine phosphorylation related to IL-2 signaling [39,79]. This suggests that raft integrity is critical in keeping the intracellular domains of the β - and γ_{c} chains together with the docked JAKs and STATs in a proper, juxtaposed position. Upon cholesterol depletion, the size of lipid rafts increased and the boundaries of the microdomains became fuzzier – that is, the compactness and cohesion within the microdomain declined [39]. This implies that the lipid microenvironment, in particular cholesterol, might be an important factor in maintaining the integrity of signaling complexes. Lipid rafts may promote the formation and cytokine-specific modulation of IL-2R/IL-15R complexes and β - and γ_c -subunit "switching" between IL-2 and IL-15 receptors as well.

Marmor and Julius proposed a different mode of IL-2R signal regulation by rafts, based on observations on murine T cells [126]. Whereas IL-2R α was also found constitutively enriched in lipid rafts, the β - and γ_c -chains, along with JAK1 and JAK3 kinases, were found mostly in the detergent-soluble membrane fractions. IL-2-mediated assembly of the high-affinity receptor complex as well as activation of JAKs occurred exclusively in the soluble fractions. As a consequence, disruption of lipid raft integrity did not impair IL-2R-induced signaling. It was proposed that sequestration of IL-2R α within raft domains of murine T cells hampers its interactions and regulates IL-2 signaling through impeding its interaction with the signaling $\beta\gamma_c$ heterodimer.

A third mechanism for raft-assisted IL-2 signaling is outlined by Goebel et al. [127]. Using biochemical approaches, these authors demonstrated selective enrichment of IL-2/15R β chains, but not cytokine-specific α -chains or γ_c -chains, in lipid rafts of phytohemagglutinin (PHA)-activated human peripheral T cells. IL-2 stimulation was accompanied by a partial translocation of the β -chains along with the associated signaling molecules (JAK1, lck, Grb-2) to the soluble membrane fraction. Furthermore, disruption of lipid raft integrity attenuated IL-2 signaling.

 Fig. 7.3 Co-localization of IL-15Rα and IL-2Rα in lipid rafts of FT7.10 T lymphoma cells. (Panels A-D): Confocal microscopic images of the distribution of IL-15R α (panel A, Cy3anti-FLAG), IL-2R α (panel B, Cy5–7G7 B6) and GM1 ganglioside, a lipid raft marker (panel C, labeled by Alexa Fluor 488-cholera toxin B). In the overlay image (panel D) those pixels appearing in white represent colocalization of all three molecules. Confocal sections were recorded from the cover slipproximal surface of the cell. Pairwise correlation coefficients between the different channels were fairly high ($C_{AB} = 0.79$, $C_{AC} = 0.59$, $C_{BC} = 0.67$), referring to substantial overlap between the lateral distributions of the studied molecules. (Scale bar = 2 μ m.) Panels E-H: Acceptor photobleaching FRET measurement: Confocal images of IL-15R α (panel E) and IL-2R α (panel F) recorded before acceptor photobleaching (see Section 7.2 for a description of FRET microscopic methods). Receptor subunits were targeted by Cy3-7A4 24 (donor) and Cy5-anti-Tac (acceptor) mAbs. Panel G shows the pixel-by-pixel FRET efficiency map between IL-15R α and IL-2R α as determined from the donor images taken before and after photobleaching. The color code ranges between FRET efficiency values of 0 (purple) to 100% (red). Pixels with fluorescence intensities lower than the background threshold are displayed in black. Panel H shows the frequency distribution histogram of FRET efficiencies in the individual pixels. The mean FRET efficiency value was 17%.

160 7 Role of Lipid Microdomains in the Formation of Supramolecular Protein Complexes



These authors suggested that sequestration of the β -chains in rafts prior to IL-2 stimulation controls the specific cytokine responsiveness, since it ensures that pairing with the more "promiscuous" γ_c -chains only takes place upon binding of the appropriate ligand [127].

Association of IL-2 receptors with detergent-resistant membrane microdomains was also reported to define a clathrin-independent endocytotic pathway [128].

Incongruence observed in the raft association of the IL-2/IL-15R system could indicate that composition of rafts as well as partition of a given protein between the raft and non-raft membrane regions can exhibit cell- or species-specificity [79,127]. Whether the observed discrepancies are only due to the different experimental approaches or they are caused by real differences in the observed systems (and if yes, what is the cause of distinct localization: post-translational modification, lipid composition or protein-protein interactions) is yet to be determined.

Flow cytometric FRET experiments revealed the co-localization of the IL-2/IL-15R system and MHC I molecules in the plasma membrane of cells of human T lymphoma/leukemia origin [39,79,98]. Although the exact role of this co-localiza-



Fig. 7.5 Mobility and co-mobility measurements by FCS and FCCS. (A) Normalized autocorrelation curves detected from Cy5tagged anti-Tac Fabs free in solution or bound to IL-2R α subunits on Kit 225 K6 T lymphoma cells. The diffusion time of the antibody decreased by an order of magnitude upon receptor binding. (B) Cross-correlation curve measured on Kit 225 FT 7.10 cells between IL-2R α and IL-15R α (labeled with Cy5anti-Tac Fab and Alexa 488 anti-FLAG mAbs,

respectively). The nonzero cross-correlation amplitude suggests that at least a certain fraction of the proteins form stable complexes for at least the duration of the diffusion time. As a negative control, cross-correlation between transferrin receptors (Trfr) and IL-2R α was determined on K6 cells yielding a flat correlation curve, which is indicative of no interaction. Receptors were labeled by Alexa 488-MEM75 mAb and Cy5-anti-Tac Fab, respectively.

162 7 Role of Lipid Microdomains in the Formation of Supramolecular Protein Complexes

tion has not yet been elucidated, a regulatory tyrosine phosphorylation cross-talk, as suggested earlier for the class I MHC-insulin receptor interaction [15], cannot be excluded [39,79]. Association of the IL-2/IL-15R with MHC II glycoproteins was also demonstrated. Co-localization of the elements of the IL-2/IL-15R system with MHC glycoproteins also takes place in lipid rafts, as revealed by confocal microscopy [39,98].

FRET assays do not report on the dynamics and stability of protein-protein interactions. By using FCCS, it was shown that α subunits of IL-2R and IL-15R diffused together at least for several tens of milliseconds – the time window of an FCCS experiment (Fig. 7.5). Similar stable association was detected between MHC I and IL-2R α or IL-15R α chains [98]. On the other hand, no cross-correlation could be detected between IL-2R α and coated pit-located transferrin receptor molecules (Fig. 7.5).

An interesting consequence of these results follows from the relative expression levels of the studied molecules. The ratio of the amount of IL-15Ra, IL-2Ra and MHC I is ~1:10:50–100 on the Kit 225 FT7.10 T lymphoma cells used in the FCCS experiments. If complexes of 1:1 stoichiometry were formed between IL-15Ra and MHC I, then the out-of-complex fraction of MHC I molecules would suppress the cross-correlation amplitude below the detection level (see Eq. 7.8). This suggests that higher-order aggregates of MHC class I molecules float together with IL-2 and IL-15 receptors in large supramolecular complexes in the plasma membrane. These results are in accordance with previous data on the homoassociation of MHC I molecules detected by FRET and electron microscopy/atomic force microscopy (see Section 7.3) [59,83].

Our observations suggest the possibility of a supramolecular complex of MHC, ICAM-1 molecules and cytokine receptor subunits that could include all members of the γ_c cytokine receptor family in addition to IL-2R α and IL-15R α , in particular, IL-4R α , IL-7R α , IL-9R α and IL-21R α . Such an association in a lipid raft-accommodated supercomplex could provide one explanation for the functional competition among cytokines that has been observed on the simultaneous addition of IL-2 and IL-4 to lymphocytes. Furthermore, the definition of such a supercomplex of cytokine receptors would also add to our understanding of the regulation of lymphocyte proliferation and effector immune responses that are mediated by these pivotal γ_c -associated cytokines.

7.7

Organization and Function of Receptor Tyrosine Kinases is Linked to Lipid Microdomains

Receptor tyrosine kinases are prominent examples of the proteins involved in cell signaling that form molecular associations and superstructures in the cell membrane. Cooperative interactions between or among receptor tyrosine kinases play a pivotal role in signal transduction. This includes homo- and hetero-dimerizations as well as potentially higher-order associations, which may be either dependent or independent of ligands. In addition to protein-protein interactions, lipid microdomains also play an important role in the organization of superstructures. The EGF receptor, the insulin receptor, the PDGF receptor, vascular endothelial growth factor (VEGF) and nerve growth factor (NGF) receptors have been shown to be localized to low density, cholesterol-rich membrane domains. In all cases, signaling by these receptors can be modulated by changes in cellular cholesterol content. Thus, raft localization appears to be of functional importance to these receptors (for a review, see [129]).

One of the best-studied receptors in this respect is the EGF receptor which belongs to the type I family of transmembrane receptor tyrosine kinases. In addition to EGFR or ErbB1, the other members of the family are ErbB2 (HER2 or Neu), ErbB3 and ErbB4 [130–133]. Within a given tissue, these receptors are rarely expressed alone, but are found in various combinations. Members of the family display various degrees and combinations of homo- and hetero-associations at the cell surface, depending on their relative expression levels. ErbB2 is an orphan receptor: no soluble physiological ligand, specific to ErbB2, has been detected so far. Despite this fact, ErbB2 participates actively in ErbB receptor combinations, and receptor complexes including the ErbB2-ErbB3 dimer which appears to be more potent in mitogenic activity than any other combination [130-134]. Recently, a molecular model was built for the nearly full-length ErbB2 dimer based on the Xray or nuclear magnetic resonance (NMR) structures of extracellular, transmembrane and intracellular domains, and intramolecular distances determined by FRET. Favorable dimerization interactions were predicted for the extracellular, transmembrane and protein kinase domains which may act in a coordinated fashion in ErbB2 homodimerization, or alternatively in ErbB heterodimerizations [135].

Molecular-scale physical associations among ErbB family members have been studied using classical biochemical [136,137], molecular biological and biophysical methods [48,62,138,139]. When isolated from cells, members of the ErbB family self-associate (homoassociate) and associate with other family members (heteroassociate) [136]. However, experiments on isolated proteins are inherently unable to detect interactions in cellular environments *in vivo* and *in situ*, and cannot detect heterogeneity within or among cells. FRET measurements detected dimerization of ErbB1 receptors in fixed [138,140] and living cells [140]. FRET was also applied to monitor the association pattern of ErbB2 in breast tumor cells [48,62,138,139]. There was considerable homoassociation of ErbB2 and heteroassociation of ErbB2 with EGFR in quiescent breast tumor cells. ErbB2 homoassociation was enhanced by EGF treatment in SKBR-3 cells and in the BT474 subline BT474M1 with high tumorigenic potential, whereas the original BT474 line was resistant to this effect. These differences correlated well with EGFR expression. Since these measurements were performed with flow cytometry, one single FRET efficiency value was obtained for each cell analyzed. In order to reveal heterogeneity in the homoassociation pattern of ErbB2 within a single cell, one of the microscopic FRET approaches had to be utilized. Donor photobleaching FRET microscopy was used to visualize FRET efficiency within single cells with spatial resolution limited only
164 7 Role of Lipid Microdomains in the Formation of Supramolecular Protein Complexes

by diffraction in the optical microscope [48,50,141]. This allows detailed analysis of the spatial heterogeneity of molecular interactions. Donor pbFRET measurements showed that ErbB2 homoassociation was also heterogeneous in unstimulated breast tumor cells; and membrane domains with erbB2 homoassociation had mean diameters of less than 1 μ m [62,142]. It was not clear whether the domain size was imposed by the optical resolution limit of wide-field microscopy in the X-Y plane or whether it originated from the actual size of ErbB2 aggregates.

In order to refine the size estimate of domains containing ErbB2 molecules, it was necessary to use SNOM [50,143,144]. This technique is not limited by diffraction optics, and can readily image objects in the 0.1 to 1 µm range, including subµm lipid and protein clusters in the plasma membrane [4] (see Section 7.2). ErbB2 was concentrated in irregular membrane patches with a mean diameter of approximately 500 nm, containing up to 1000 ErbB2 molecules in nonactivated SKBR-3 and MDA453 human breast tumor cells. The mean cluster diameter increased to 600–900 nm when SKBR-3 cells were treated with EGF, heregulin or a partially agonistic anti-erbB2 antibody. The increase in cluster size was inhibited by an EGFR-specific tyrosine kinase inhibitor, suggesting that EGFR was somehow involved in organizing this ErbB2 superstructure. Since the domain size was larger than the resolution limit of confocal microscopy (200–300 nm in the X-Y plane), we were able to confirm the SNOM results with confocal laser scanning microscopy (Fig. 7.6) [142].

The role of lipid rafts in the organization of these EGFR and ErbB2-containing superstructures has also been revealed in further experiments. The size of lipid rafts was investigated with dye-labeled cholera toxin B (CTX-B) subunit, which binds to the glycosphingolipid GM1 ganglioside. In order to study the role of lipid rafts in the homoassociation patterns of ErbB2, confocal microscopic studies were performed. The signal from one laser beam was used to monitor lipid rafts, and signals excited by the other two laser beams to reveal the homoassociation pattern of ErbB2. Observations suggest that similarly to ErbB1 [129], ErbB2 is localized mostly in GM1-enriched membrane domains, that are distinct from caveolae. However, there is a negative correlation between ErbB2 homoassociation and the local density of the lipid raft marker CTX-B. This environment could alter the association properties of ErbB2. Since stimulating ErbB2 increases the size of ErbB2 clusters [142] and lipid rafts [139], the amount of ErbB2 concentrated in rafts is very likely related to the function of the protein. Localization of ErbB2 in lipid rafts is dynamic, since it can be dislodged from rafts by cholera toxin-induced raft crosslinking [139]. Upon crosslinking with CTX-B, GM1 leaves ErbB protein clusters behind and migrates into caveolae. The association properties and biological activity of ErbB2 excluded from rafts differ from those inside rafts. For example, internalization of ErbB2 mediated by 4D5 (the parent murine version of trastuzumab, a monoclonal anti-ErbB2 antibody used in breast cancer therapy) is blocked in CTX-B-pretreated cells, while its antiproliferative effect is not. A role of lipid rafts in limiting autoactivation of the ErbB signaling system is supported by the increased tyrosine phosphorylation of Shc after removing ErbB proteins from rafts by CTX-B treatment. On the other hand, lipid rafts are also responsible for



Fig. 7.6 Scanning near-field optical microscopy (SNOM) and confocal laser scanning microscopy (CLSM) images of TAMRA-4D5labeled ErbB2 receptors on SKBR-3 cells. SKBR-3 cells grown on glass coverslips were labeled on ice using TAMRA-conjugated 4D5 monoclonal anti-ErbB2 antibody Fab fragments. (A) For SNOM, a custom-built shared



aperture instrument was used on formaldehyde-fixed, dehydrated, air-dried samples [142]. (B) 1-mm optical sections of live cells were obtained in CLSM with a Zeiss LSM 410 instrument. Fluorescence intensity is displayed in pseudocolor. (Scale bar = $2 \mu m$.)

maintaining ErbB proteins in a growth factor-responsive state. This is supported by the following results:

- Neither EGF, nor heregulin are able to activate Shc if ErbB proteins are removed from lipid rafts.
- The formation of heregulin-responsive ErbB2/ErbB3 heterodimers and heregulin-induced ErbB2 tyrosine phosphorylation decrease if ErbB proteins are removed from lipid rafts.

These results emphasize that alterations in the local environment of ErbB2 strongly influence its association properties, which are reflected in its biological activity and in its behavior as a target for therapy [139].

The results of several studies have implied that integrins and growth factor receptors cooperate in tumor formation, and have demonstrated the existence of integrin-growth factor receptor complexes leading to a decreased threshold of transmembrane signaling [145–147]. Similar to ErbB2, integrins were shown to be raft-associated [148]. Cooperative signaling between ErbB proteins and integrins [146,149] is a common feature of invasive cancer cells, and association of β_1 -integrin with ErbB2 proteins could provide a framework in which tumor cell metastasis might be better understood. FRET and confocal microscopic measurements demonstrated an association between ErbB2 and β_1 -integrin in the nanometer range and on the scale of membrane microdomains, respectively. Lipid rafts showed a substantial overlap with both ErbB2- and β_1 -integrin-rich microdo-

166 7 Role of Lipid Microdomains in the Formation of Supramolecular Protein Complexes

mains [150]. These results corroborated the existence of molecular interactions between ErbB2 and β_1 -integrins and their association with lipid rafts. Interestingly, although ErbB2-positive but trastuzumab-resistant breast and gastric cancer cell lines were found to express a substantially higher amount of β_1 -integrin than corresponding trastuzumab lines, there was no significant difference among them in the trastuzumab-induced tyrosine phosphorylation of ErbB2, which was probably caused by the weak functional interaction between the two proteins.

In summary, there is ample evidence for the role of local factors, for example, the lipid environment, other receptor tyrosine kinases and integrins, in influencing receptor tyrosine kinase activity and consequential modulation of the ability of these molecules in driving cellular functions. In addition to a more thorough understanding of the complexity of the receptor tyrosine kinase signaling networks, experiments linking lipid rafts and receptor function may facilitate the development of more efficient therapeutic strategies targeting receptor tyrosine kinases [150].

Acknowledgments

G. Vámosi and A. Bodnár contributed equally to this manuscript. These studies were supported by the following grants: OTKA TS40773, T48745, T42618, F46497; T38037; T43061; ETT 602/2003, 603/2003; 524/2004; 532/2004; NATO Life Science and Technology Collaborative Linkage Grant 980200, EU FP6 LSHB-CT-2004–503467; Bolyai János Research Fellowships (to G. Vámosi and A. Bodnár) and Békési Fellowship (to G. Vereb).

Abbreviations

β2m	β2-microglobulin
AFM	atomic force microscopy
AICD	activation-induced cell death
APC	antigen-presenting cell
CLSM	confocal laser scanning microscopy
cSMAC	central supramolecular activation cluster
CTL	cytotoxic T cell
CTX-B	cholera toxin B
EGFR	epidermal growth factor receptor
ER	endoplasmic reticulum
FCCS	fluorescence cross-correlation spectroscopy
FCS	fluorescence correlation spectroscopy
FHC	free heavy chain
FLIM	fluorescence lifetime imaging microscopy
FRET	fluorescence resonance energy transfer
ICAM	intracellular adhesion molecule

IFN	interferon
IL	interleukin
IS	immunological synapse
JAK	janus kinase
MHC	major histocompatibility complex
NGF	nerve growth factor
NMR	nuclear magnetic resonance
pbFRET	photobleaching FRET
PDGF	platelet-derived growth factor
PHA	phytohemagglutinin
pSMAC	peripheral supramolecular activation cluster
SNOM	scanning near-field optical microscopy
TCR	T-cell receptor
VEGF	vascular endothelial growth factor

References

- S.J. Singer, G.L. Nicolson. The fluid mosaic model of the structure of cell membranes. *Science* 1972, *175*, 720–731.
- 2 L. D. Frye, M. Edidin. The rapid intermixing of cell surface antigens after formation of mouse-human heterokaryons. *J. Cell Sci.* 1970, 7, 319–335.
- 3 M. Edidin. Patches and fences: probing for plasma membrane domains. J. Cell Sci. Suppl. 1993, 17, 165–169.
- 4 M. Edidin. Lipid microdomains in cell surface membranes. *Curr. Opin. Struct. Biol.* 1997, 7, 528–532.
- 5 S. Damjanovich, R. J. Gáspár, C. Pieri. Dynamic receptor superstructures at the plasma membrane. *Q. Rev. Biophys.* 1997, 30, 67–106.
- 6 S. Damjanovich, J. Matkó, L. Mátyus, G. Szabò, Jr., J. Szöllősi, J.C. Pieri, T. Farkas, R. Gáspár, Jr. Supramolecular receptor structures in the plasma membrane of lymphocytes revealed by flow cytometric energy transfer, scanning force- and transmission electron-microscopic analyses. *Cytometry* 1998, 33, 225–233.
- 7 K. Simons, E. Ikonen. Functional rafts in cell membranes. *Nature* 1997, *387*, 569–572.
- 8 K. Simons, W. L. Vaz. Model systems, lipid rafts, and cell membranes. *Annu. Rev. Biophys. Biomol. Struct.* 2004, *33*, 269–295.

- 9 G. Vereb, L. Mátyus, L. Bene, G. Panyi, Z. Bacsó, M. Balázs, J. Matkó, J. Szöllősi, R. Jr. Gáspár, S. Damjanovich. Plasma membrane bound macromolecules are dynamically aggregated to form nonrandom co-distribution patterns of selected functional elements. Do pattern recognition processes govern antigen presentation and intercellular interactions? J. Mol. Recognit. 1995, 8, 237–246.
- **10** M. Edidin. Shrinking patches and slippery rafts: scales of domains in the plasma membrane. *Trends Cell Biol.* 2001, *11*, 492–496.
- 11 G. Vereb, J. Szöllősi, J. Matkó, P. Nagy, T. Farkas, L. Vígh, L. Mátyus, T.A. Waldmann, S. Damjanovich. Dynamic, yet structured: The cell membrane three decades after the Singer-Nicolson model. *Proc. Natl. Acad. Sci. USA* 2003, 100, 8053–8058.
- 12 S. Damjanovich, B. Somogyi, L. Trón. Macromolecular dynamics and information transfer. *Adv. Physiol. Sci.* 1981, 30, 9–15.
- 13 J. Matkó, J. Szöllősi. Landing of immune receptors and signal proteins on lipid rafts: a safe way to be spatio-temporally coordinated? *Immunol. Lett.* 2002, *82*, 3–15.
- 14 R. Zidovetzki, Y. Yarden, J. Schlessinger, T.M. Jovin. Rotational diffusion of epider-

mal growth factor complexed to cell surface receptors reflects rapid microaggregation and endocytosis of occupied receptors. *Proc. Natl. Acad. Sci. USA* 1981, 78, 6981–6985.

- 15 T. S. Ramalingam, A. Chakrabarti, M. Edidin. Interaction of class I human leukocyte antigen (HLA-I) molecules with insulin receptors and its effect on the insulinsignaling cascade. *Mol. Biol. Cell.* 1997, 8, 2463–2474.
- 16 J. Reiland, M. Edidin. Chemical crosslinking detects association of insulin receptors with four different class I human leukocyte antigen molecules on cell surfaces. *Diabetes* 1993, 42, 619–625.
- 17 S. Damjanovich, L. Mátyus, L. Damjanovich, L. Bene, A. Jenei, J. Matkó, R. Gáspár, J. Szöllősi. Does mosaicism of the plasma membrane at molecular and higher hierarchical levels in human lymphocytes carry information on the immediate history of cells? *Immunol. Lett.* 2002, *82*, 93–99.
- 18 J. F. Tocanne, L. Cezanne, A. Lopez, B. Piknova, V. Schram, J. F. Tournier, M. Welby. Lipid domains and lipid/protein interactions in biological membranes. *Chem. Phys. Lipids* 1994, 73, 139–158.
- 19 V. Horejsi. The roles of membrane microdomains (rafts) in T cell activation. *Immunol. Rev.* 2003, 191, 148–164.
- 20 V. Horejsi. Membrane rafts in immunoreceptor signaling: new doubts, new proofs? *Trends Immunol.* 2002, 23, 562–564.
- 21 D. C. Hoessli, S. Ilangumaran, A. Soltermann, P. J. Robinson, B. Borisch, U. D. Nasir. Signaling through sphingolipid microdomains of the plasma membrane: the concept of signaling platform. *Glycoconj. J.* 2000, *17*, 191–197.
- 22 H.A. Lucero, P.W. Robbins. Lipid raftsprotein association and the regulation of protein activity. *Arch. Biochem. Biophys.* 2004, 426, 208–224.
- 23 X. L. Yang, W. C. Xiong, L. Mei. Lipid rafts in neuregulin signaling at synapses. *Life Sci.* 2004, 75, 2495–2504.
- 24 R.G. Anderson, K. Jacobson. A role for lipid shells in targeting proteins to caveolae, rafts, and other lipid domains. *Science* 2002, 296, 1821–1825.

- 25 T. Harder, K. Simons. Caveolae, DIGs, and the dynamics of sphingolipid-cholesterol microdomains. *Curr. Opin. Cell Biol.* 1997, 9, 534–542.
- **26** D.A. Brown. Interactions between GPIanchored proteins and membrane lipids. *Trends Cell Biol.* 1992, *2*, 338–343.
- 27 M. D. Resh. Membrane targeting of lipid modified signal transduction proteins. *Subcell. Biochem.* 2004, *37*, 217–232.
- 28 M. D. Resh. Fatty acylation of proteins: new insights into membrane targeting of myristoylated and palmitoylated proteins. *Biochim. Biophys. Acta* 1999, 1451, 1–16.
- 29 W. Rodgers, D. Farris, S. Mishra. Merging complexes: properties of membrane raft assembly during lymphocyte signaling. *Trends Immunol.* 2005, 26, 97–103.
- **30** A. Kusumi, Y. Sako. Cell surface organization by the membrane skeleton. *Curr. Opin. Cell Biol.* 1996, *8*, 566–574.
- 31 M.A. Lemmon, D.M. Engelman. Specificity and promiscuity in membrane helix interactions. Q. Rev. Biophys. 1994, 27, 157–218.
- 32 Q. Tang, M. Edidin. Vesicle trafficking and cell surface membrane patchiness. *Biophys. J.* 2001, *81*, 196–203.
- 33 L.A. Gheber, M. Edidin. A model for membrane patchiness: lateral diffusion in the presence of barriers and vesicle traffic. *Biophys. J.* 1999, 77, 3163–3175.
- 34 K. Ritchie, A. Kusumi. Role of the membrane skeleton in creation of microdomains. *Subcell. Biochem.* 2004, *37*, 233–245.
- 35 J. Pawley. Handbook of biological confocal microscopy. Plenum Press, New York, 1995.
- 36 W. R. Zipfel, R. M. Williams, W. W. Webb. Nonlinear magic: multiphoton microscopy in the biosciences. *Nat. Biotechnol.* 2003, *21*, 1369–1377.
- 37 M. Edidin, M.C. Zuniga, M.P. Sheetz. Truncation mutants define and locate cytoplasmic barriers to lateral mobility of membrane glycoproteins. *Proc. Natl. Acad. Sci. USA* 1994, 91, 3378–3382.
- 38 A. Kusumi, Y. Sako, M. Yamamoto. Confined lateral diffusion of membrane receptors as studied by single particle tracking (nanovid microscopy). Effects of calcium-induced differentiation in cultured

epithelial cells. *Biophys. J.* 1993, 65, 2021–2040.

- 39 G. Vereb, J. Matkó, G. Vámosi, S. M. Ibrahim, E. Magyar, S. Varga, J. Szöllősi, A. Jenei, R. J. Gáspár, T. A. Waldmann, S. Damjanovich. Cholesterol-dependent clustering of IL-2Ralpha and its colocalization with HLA and CD48 on T lymphoma cells suggest their functional association with lipid rafts. *Proc. Natl. Acad. Sci. USA* 2000, *97*, 6013–6018.
- 40 G. Vereb, J. Szöllősi, S. Damjanovich, J. Matkó. Exploring membrane microdomains and functional protein clustering in live cells with flow and image cytometric methods. *Reviews in Fluorescence*, Kluwer Academic/Plenum Publishers, New York, 2004, pp. 105–152.
- 41 S. Damjanovich, L. Bene, J. Matkó, L. Mátyus, Z. Krasznai, G. Szabo, C. Pieri, R. J. Gáspár, J. Szöllősi. Two-dimensional receptor patterns in the plasma membrane of cells. A critical evaluation of their identification, origin and information content. *Biophys. Chem.* 1999, *82*, 99–108.
- 42 J. Szöllősi, S. Damjanovich, L. Mátyus. Application of fluorescence resonance energy transfer in the clinical laboratory: routine and research. *Cytometry* 1998, 34, 159–179.
- 43 G. Vereb, J. Matkó, J. Szöllősi. Cytometry of fluorescence resonance energy transfer. *Methods Cell Biol.* 2004, 75, 105–152.
- 44 T. Förster. Zwischenmolekulare Energiewanderung und Fluoreszenz. Ann. Phys. 1948, 2, 55–75.
- 45 R. M. Clegg. Fluorescence resonance energy transfer (FRET). In: *Fluorescence Imaging Spectroscopy and Microscopy*, Wiley, John & Sons, Inc., New York, 1996, pp. 179–252.
- 46 L. Stryer, R. P. Haugland. Energy transfer: a spectroscopic ruler. *Proc. Natl. Acad. Sci.* USA 1967, 58, 719–726.
- 47 E.A. Jares-Erijman, T.M. Jovin. FRET imaging. Nat. Biotechnol. 2003, 21, 1387–1395.
- 48 P. Nagy, G. Vámosi, A. Bodnár, S. J. Lockett, J. Szöllősi. Intensity-based energy transfer measurements in digital imaging microscopy. *Eur. Biophys. J.* 1998, 27, 377–389.
- **49** P. I. Bastiaens, I. V. Majoul, P. J. Verveer, H. D. Soling, T. M. Jovin. Imaging the in-

tracellular trafficking and state of the AB5 quaternary structure of cholera toxin. *EMBO J.* 1996, *15*, 4246–4253.

- 50 G. Vereb, C. K. Meyer, T. M. Jovin, Novel microscope-based approaches for the investigation of protein-protein interactions in signal transduction. In: *Interacting Protein Domains, their Role in Signal and Energy Transduction*. Springer-Verlag, New York, 1997, pp. 49–52.
- 51 T.M. Jovin, D.J. Arndt-Jovin, FRET microscopy: digital imaging of fluorescence resonance energy transfer. Application in cell biology. In: *Cell Structure and Function by Microspectrofluorometry*. Academic Press, San Diego, 1989, pp. 99–117.
- 52 R. M. Clegg, O. Holub, C. Gohlke. Fluorescence lifetime-resolved imaging: measuring lifetimes in an image. *Methods Enzymol.* 2003, 360, 509–542.
- 53 L. Trón, J. Szöllősi, S. Damjanovich, S. H. Helliwell, D. J. Arndt-Jovin, T. M. Jovin. Flow cytometric measurements of fluorescence resonance energy transfer on cell surfaces. Quantitative evaluation of the transfer efficiency on a cell by cell basis. *Biophys. J.* 1984, 45, 939–946.
- 54 Z. Sebestyén, P. Nagy, G. Horváth, G. Vámosi, R. Debets, J. W. Gratama, D. R. Alexander, J. Szöllősi. Long wavelength fluorophores and cell-by-cell correction for autofluorescence significantly improves the accuracy of flow cytometric energy transfer measurements on a dual-laser benchtop flow cytometer. *Cytometry* 2002, 48, 124–135.
- 55 R. Rigler, E. S. Elson. Fluorescence Correlation Spectroscopy. Theory and Applications. Springer Verlag, Berlin, Heidelberg, New York, Barcelona, Hong Kong, London, Milan, Paris, Singapore, Tokyo, 2001.
- 56 J. Widengren, Ü. Mets, R. Rigler. Fluorescence correlation spectroscopy of triplet state in solution: A theoretical and experimental study. J. Phys. Chem. 1995, 99, 13368–13379.
- 57 K. Bacia, I.V. Majoul, P. Schwille. Probing the endocytic pathway in live cells using dual-color fluorescence cross-correlation analysis. *Biophys. J.* 2002, *83*, 1184–1193.
- 58 T. Weidemann, M. Wachsmuth, M. Tewes, K. Rippe, J. Langowski. Analysis of ligand binding by two-colour fluores-

170 References

cence cross-correlation spectroscopy. Single Molecules 2002, 3, 49–61.

- 59 S. Damjanovich, G. Vereb, A. Schaper, A. Jenei, J. Matkó, J. P. Starink, G. Q. Fox, D. J. Arndt-Jovin, T. M. Jovin. Structural hierarchy in the clustering of HLA class I molecules in the plasma membrane of human lymphoblastoid cells. *Proc. Natl. Acad. Sci. USA* 1995, *92*, 1122–1126.
- 60 A. Jenei, S. Varga, L. Bene, L. Mátyus, A. Bodnár, Z. Bacsó, C. Pieri, R. J. Gáspár, T. Farkas, S. Damjanovich. HLA class I and II antigens are partially co-clustered in the plasma membrane of human lymphoblastoid cells. *Proc. Natl. Acad. Sci.* USA 1997, 94, 7269–7274.
- 61 J. Hwang, L.A. Gheber, L. Margolis, M. Edidin. Domains in cell plasma membranes investigated by near-field scanning optical microscopy. *Biophys. J.* 1998, 74, 2184–2190.
- 62 P. Nagy, L. Bene, M. Balázs, W.C. Hyun, S.J. Lockett, N.Y. Chiang, F. Waldman, B.G. Feuerstein, S. Damjanovich, J. Szöllősi. EGF-induced redistribution of erbB2 on breast tumor cells: flow and image cytometric energy transfer measurements. *Cytometry* 1998, 32, 120–131.
- 63 S. K. Bromley, W. R. Burack, K. G. Johnson, K. Somersalo, T. N. Sims, C. Sumen, M. M. Davis, A. S. Shaw, P. M. Allen, M. L. Dustin. The immunological synapse. *Annu. Rev. Immunol.* 2001, 19, 375–396.
- 64 A. Grakoui, S. K. Bromley, C. Sumen, M. M. Davis, A. S. Shaw, P. M. Allen, M. L. Dustin. The immunological synapse: a molecular machine controlling T cell activation. *Science* 1999, 285, 221–227.
- 65 K. H. Lee, A. D. Holdorf, M. L. Dustin, A. C. Chan, P. M. Allen, A. S. Shaw. T cell receptor signaling precedes immunological synapse formation. *Science* 2002, 295, 1539–1542.
- **66** M. F. Krummel, M. M. Davis. Dynamics of the immunological synapse: finding, establishing and solidifying a connection. *Curr. Opin. Immunol.* 2002, *14*, 66–74.
- **67** D. M. Davis, M. L. Dustin. What is the importance of the immunological synapse? *Trends Immunol.* 2004, *25*, 323–327.
- **68** J. Jacobelli, P. G. Andres, J. Boisvert, M. F. Krummel. New views of the immunological synapse: variations in assembly and

function. Curr. Opin. Immunol. 2004, 16, 345–352.

- 69 A. Trautmann, S. Valitutti. The diversity of immunological synapses. *Curr. Opin. Immunol.* 2003, 15, 249–254.
- 70 T. Harder. Lipid raft domains and protein networks in T-cell receptor signal transduction. *Curr. Opin. Immunol.* 2004, 16, 353–359.
- 71 H. T. He, A. Lellouch, D. Marguet. Lipid rafts and the initiation of T cell receptor signaling. *Semin. Immunol.* 2005, *17*, 23–33.
- 72 P. Friedl, J. Storim. Diversity in immunecell interactions: states and functions of the immunological synapse. *Trends Cell Biol.* 2004, 14, 557–567.
- 73 A. Kupfer, S.J. Singer. The specific interaction of helper T cells and antigen-presenting B cells. IV. Membrane and cytoskeletal reorganizations in the bound T cell as a function of antigen dose. *J. Exp. Med.* 1989, *170*, 1697–1713.
- 74 C. R. Monks, B. A. Freiberg, H. Kupfer, N. Sciaky, A. Kupfer. Three-dimensional segregation of supramolecular activation clusters in T cells. *Nature* 1998, 395, 82–86.
- 75 J.C. Stinchcombe, G. Bossi, S. Booth, G.M. Griffiths. The immunological synapse of CTL contains a secretory domain and membrane bridges. *Immunity* 2001, 15, 751–761.
- 76 P. Friedl, E. B. Brocker. TCR triggering on the move: diversity of T-cell interactions with antigen-presenting cells. *Immunol. Rev.* 2002, *186*, 83–89.
- 77 L. Bene, M. Balázs, J. Matkó, J. Most, M. P. Dierich, J. Szöllősi, S. Damjanovich. Lateral organization of the ICAM-1 molecule at the surface of human lymphoblasts: a possible model for its co-distribution with the IL-2 receptor, class I and class II HLA molecules. *Eur. J. Immunol.* 1994, 24, 2115–2123.
- 78 L. Bene, A. Bodnár, S. Damjanovich, G. Vámosi, Z. Bacsó, J. Aradi, A. Berta, J. Damjanovich. Membrane topography of HLA I, HLA II, and ICAM-1 is affected by IFN-gamma in lipid rafts of uveal melanomas. *Biochem. Biophys. Res. Commun.* 2004, *322*, 678–683.
- 79 J. Matkó, A. Bodnár, G. Vereb, L. Bene, G. Vámosi, G. Szentesi, J. Szöllősi, R.

Gáspár, V. Horejsi, T.A. Waldmann, S. Damjanovich. GPI-microdomains (membrane rafts) and signaling of the multichain interleukin-2 receptor in human lymphoma/leukemia T cell lines. *Eur. J. Biochem.* 2002, *269*, 1199–1208.

- 80 P. Smith, I. Morrison, K. Wilson, N. Fernandez, R. Cherry. Anomalous diffusion of major histocompatibility complex class I molecules on HeLa cells determined by single particle tracking. *Biophys. J.* 1999, 76, 3331–3344.
- 81 K. Triantafilou, M. Triantafilou, K. M. Wilson, N. Fernandez. Human major histocompatibility molecules have the intrinsic ability to form homotypic associations. *Hum. Immunol.* 2000, *61*, 585– 598.
- 82 A. Chakrabarti, J. Matkó, N.A. Rahman, B.G. Barisas, M. Edidin. Self-association of class I major histocompatibility complex molecules in liposome and cell surface membranes. *Biochemistry* 1992, 31, 7182–7189.
- 83 J. Matkó, Y. Bushkin, T. Wei, M. Edidin. Clustering of class I HLA molecules on the surfaces of activated and transformed human cells. *J. Immunol.* 1994, *152*, 3353–3360.
- 84 A. Bodnár, Z. Bacsó, A. Jenei, T. M. Jovin, M. Edidin, S. Damjanovich, J. Matkó. Class I HLA oligomerization at the surface of B cells is controlled by exogenous β₂-microglobulin: implications in activation of cytotoxic T lymphocytes. *Int. Immunol.* 2003, *15*, 331–339.
- 85 S. Demaria, R. Schwab, S. Gottesman, Y. Bushkin. Soluble β₂-microglobulin-free class I heavy chains are released from the surface of activated and leukemia cells by a metalloprotease. *J. Biol. Chem.* 1994, 269, 6689–6694.
- 86 W. Pickl, W. Holter, J. Stöckl, O. Majdic, W. Knapp. Expression of LA45 reactive β₂-microglobulin free HLA class I achains on activated T cells is regulated by internalization, constitutive and protein kinase C inducible release. *Tissue Antigens* 1996, 48, 15–21.
- 87 W. F. Wade, J. H. Freed, M. Edidin. Translational diffusion of class II major histo-compatibility complex molecules is constrained by their cytoplasmic domains. *J. Cell Biol.* 1989, *109*, 3325–3331.

- 88 K. M. Wilson, I. E. Morrison, P. R. Smith, N. Fernandez, R. J. Cherry. Single particle tracking of cell-surface HLA-DR molecules using R-phycoerythrin labeled monoclonal antibodies and fluorescence digital imaging. J. Cell Sci. 1996, 109 (Pt 8), 2101–2109.
- 89 J. Szöllősi, S. Damjanovich, M. Balázs, P. Nagy, L. Trón, M. J. Fulwyler, F. M. Brodsky. Physical association between MHC class I and class II molecules detected on the cell surface by flow cytometric energy transfer. J. Immunol. 1989, 143, 208–213.
- 90 T. Lebedeva, N. Anikeeva, S.A. Kalams, B.D. Walker, I. Gaidarov, J. H. Keen, Y. Sykulev. Major histocompatibility complex class I-intercellular adhesion molecule-1 association on the surface of target cells: implications for antigen presentation to cytotoxic T lymphocytes. *Immunol*ogy 2004, 113, 460–471.
- 91 Z. Bacsó, L. Bene, L. Damjanovich, S. Damjanovich. INF-γ rearranges membrane topography of MHC-I and ICAM-1 in colon carcinoma cells. *Biochem. Biophys. Res. Commun.* 2002, 290, 635– 640.
- 92 T. Lebedeva, M. L. Dustin, Y. Sykulev. ICAM-1 co-stimulates target cells to facilitate antigen presentation. *Curr. Opin. Immunol.* 2005, 17, 251–258.
- 93 J. Boniface, J. Rabinowitz, C. Wülfing, J. Hampl, Z. Reich, J. Altman, R. Kantor, C. Beeson, H. McConnell, M. Davis. Initiation of signal transduction through the T cell receptor requires the multivalent engagement of peptide/MHC ligands. *Immunity* 1998, 9, 459–466.
- **94** J. Cochran, T. Cameron, L. Stern. The relationship of MHC-peptide binding and T cell activation probed using chemically defined MHC class II oligomers. *Immunity* 2000, *12*, 241–250.
- 95 M.A. Daniels, S.C. Jameson. Critical role for CD8 in T cell receptor binding and activation by peptide/major histocompatibility complex multimers. *J. Exp. Med.* 2000, 191, 335–345.
- 96 H.A. Anderson, E.M. Hiltbold, P.A. Roche. Concentration of MHC class II molecules in lipid rafts facilitates antigen presentation. *Nat. Immunol.* 2000, 1, 156–162.

- 172 References
 - 97 I. Gombos, C. Detre, G. Vámosi, J. Matkó. Rafting MHC-II domains in the APC (presynaptic) plasma membrane and the thresholds for T-cell activation and immunological synapse formation. *Immunol. Lett.* 2004, 92, 117–124.
 - 98 G. Vámosi, A. Bodnár, G. Vereb, A. Jenei, C. K. Goldman, J. Langowski, K. Tóth, L. Mátyus, J. Szöllősi, T. A. Waldmann, S. Damjanovich. IL-2 and IL-15 receptor α-subunits are coexpressed in a supramolecular receptor cluster in lipid rafts of T cells. *Proc. Natl. Acad. Sci. USA* 2004, 101, 11082–11087.
 - 99 M. Wadehra, H. Su, L. K. Gordon, L. Goodglick, J. Braun. The tetraspan protein EMP2 increases surface expression of class I major histocompatibility complex proteins and susceptibility to CTLmediated cell death. *Clin. Immunol.* 2003, 107, 129–136.
 - 100 R. W. Tilghman, R. L. Hoover. E-selectin and ICAM-1 are incorporated into detergent-insoluble membrane domains following clustering in endothelial cells. *FEBS Lett.* 2002, 525, 83–87.
 - 101 J. Goebel, K. Forrest, D. Flynn, R. Rao, T. L. Roszman. Lipid rafts, major histocompatibility complex molecules, and immune regulation. *Hum. Immunol.* 2002, 63, 813–820.
 - 102 N. J. Poloso, P. A. Roche. Association of MHC class II-peptide complexes with plasma membrane lipid microdomains. *Curr. Opin. Immunol.* 2004, 16, 103–107.
 - 103 E. M. Hiltbold, N. J. Poloso, P. A. Roche. MHC class II-peptide complexes and APC lipid rafts accumulate at the immunological synapse. *J. Immunol.* 2003, 170, 1329–1338.
 - 104 G. Panyi, Z.-F. Sheng, L.-W. Tu, C. Deutsch. C-type inactivation of a voltagegated K⁺ channel occurs by a cooperative mechanism. *Biophys. J.* 1995, 69, 896–904.
 - 105 G. Panyi, C. Deutsch. Assembly and suppression of endogenous Kv1.3 channels in human T cells. J. Gen. Physiol. 1996, 107, 409–420.
 - 106 G. Panyi, G. Vámosi, A. Bodnár, R. Gáspár, S. Damjanovich. Looking through ion channels: recharged concepts in T-cell signaling. *Trends Immunol.* 2004, 25, 565–569.

- 107 G. Panyi, Z. Varga, R. Gáspár. Ion channels and lymphocyte activation. *Immunol. Lett.* 2004, 92, 55–66.
- 108 R. S. Lewis. Calcium signaling mechanisms in T lymphocytes. Annu. Rev. Immunol. 2001, 19, 497–521.
- 109 P. C. Biggin, T. Roosild, S. Choe. Potassium channel structure: domain by domain. *Curr. Opin. Struct. Biol.* 2000, 10, 456–461.
- 110 G. Panyi, G. Vámosi, Z. Bacsó, M. Bagdany, A. Bodnár, Z. Varga, R. Gáspár, L. Mátyus, S. Damjanovich. Kv1.3 potassium channels are localized in the immunological synapse formed between cytotoxic and target cells. *Proc. Natl. Acad. Sci. USA* 2004, *101*, 1285–1290.
- 111 P. Hajdu, Z. Varga, C. Pieri, G. Panyi, R. Gáspár, Jr. Cholesterol modifies the gating of Kv1.3 in human T lymphocytes. *Pflugers Arch.* 2003, 445, 674–682.
- 112 J. R. Martens, K. O'Connell, M. Tamkun. Targeting of ion channels to membrane microdomains: localization of KV channels to lipid rafts. *Trends Pharmacol. Sci.* 2004, 25, 16–21.
- 113 G. Panyi, M. Bagdany, A. Bodnár, G. Vámosi, G. Szentesi, A. Jenei, L. Mátyus, S. Varga, T. A. Waldmann, R. Gáspár, S. Damjanovich. Colocalization and nonrandom distribution of Kv1.3 potassium channels and CD3 molecules in the plasma membrane of human T lymphocytes. *Proc. Natl. Acad. Sci. USA* 2003, 100, 2592–2597.
- 114 K.G. Chandy, H. Wulff, C. Beeton, M. Pennington, G.A. Gutman, M. D. Cahalan. K⁺ channels as targets for specific immunomodulation. *Trends Pharmacol. Sci.* 2004, 25, 280–289.
- 115 J. Matkó. K⁺ channels and T-cell synapses: the molecular background for efficient immunomodulation is shaping up. *Trends Pharmacol. Sci.* 2003, 24, 385–389.
- 116 L. Bene, J. Szöllősi, M. Balázs, L. Mátyus, R. Gáspár, M. Ameloot, R. E. Dale, S. Damjanovich. Major histocompatibility complex class I protein conformation altered by transmembrane potential changes. *Cytometry* 1997, 27, 353–357.
- 117 M. Levite, L. Cahalon, A. Peretz, R. Hershkoviz, A. Sobko, A. Ariel, R. Desai, B. Attali, O. Lider. Extracellular K⁺ and opening of voltage-gated potassium chan-

nels activate T cell integrin function: physical and functional association between Kv1.3 channels and β1 integrins. J. Exp. Med. 2000, 191, 1167-1176.

- 118 P. Nagy, L. Mátyus, A. Jenei, G. Panyi, S. Varga, I. Matkó, I. Szöllősi, R. Gáspár, T.M. Jovin, S. Damjanovich. Cell fusion experiments reveal distinctly different association characteristics of cell-surface receptors. J. Cell Sci. 2001, 114, 4063-4071.
- 119 T.A. Waldmann, S. Dubois, Y. Tagaya. Contrasting roles of IL-2 and IL-15 in the life and death of lymphocytes: implications for immunotherapy. Immunity 2001, 14, 105-110.
- 120 T.A. Fehniger, M.A. Cooper, M.A. Caligiuri. Interleukin-2 and interleukin-15: immunotherapy for cancer. Cytokine Growth Factor Rev. 2002, 13, 169-183.
- 121 T.A. Fehniger, M.A. Caligiuri. Interleukin 133 G. Vereb, P. Nagy, J.W. Park, J. Szöllősi. 15: biology and relevance to human disease. Blood 2001, 97, 14-32.
- 122 S. Kondo, A. Shimizu, Y. Saito, M. Kinoshita. T. Honio. Molecular basis for two different affinity states of the interleukin 2 receptor: affinity conversion model. Proc. Natl. Acad. Sci. USA 1986, 83, 9026-9029.
- 123 S. Damjanovich, L. Bene, J. Matkó, A. Alileche, C.K. Goldman, S. Sharrow, T.A. Waldmann. Preassembly of interleukin 2 (IL-2) receptor subunits on resting Kit 225 K6 T cells and their modulation by IL-2, IL-7, and IL-15: a fluorescence resonance energy transfer study. Proc. Natl. Acad. Sci USA 1997, 94, 13134-13139.
- 124 D.M. Eicher, S. Damjanovich, T.A. Waldmann. Oligomerization of IL-2Ra. Cytokine 2002, 17, 82-90.
- 125 D.C. Hoessli, M. Poincelet, E. Rungger-Brandle. Isolation of high-affinity murine interleukin 2 receptors as detergent-resistant membrane complexes. Eur. J. Immunol. 1990, 20, 1497-1503.
- 126 M.D. Marmor, M. Julius. Role for lipid rafts in regulating interleukin-2 receptor signaling. Blood 2001, 98, 1489-1497.
- 127 J. Goebel, K. Forrest, L. Morford, T.L. Roszman. Differential localization of IL-2and -15 receptor chains in membrane rafts of human T cells. J. Leukoc. Biol. 2002, 72, 199-206.
- 128 C. Lamaze, A. Dujeancourt, T. Baba, C.G. Lo, A. Benmerah, A. Dautry-Varsat. Inter-

leukin 2 receptors and detergent-resistant membrane domains define a clathrin-independent endocytic pathway. Mol. Cell 2001, 7, 661–671.

- 129 L.J. Pike. Growth factor receptors, lipid rafts and caveolae: An evolving story. Biochim. Biophys. Acta 2005.
- 130 P. Nagy, A. Jenei, S. Damjanovich, T.M. Jovin, J. Szöllősi. Complexity of signal transduction mediated by ErbB2: clues to the potential of receptor-targeted cancer therapy. Pathol. Oncol. Res. 1999, 5, 255-271.
- 131 A. Citri, K. B. Skaria, Y. Yarden. The deaf and the dumb: the biology of ErbB-2 and ErbB-3. Exp. Cell Res. 2003, 284, 54-65.
- 132 Y. Yarden, M.X. Sliwkowski. Untangling the ErbB signalling network. Nat. Rev. Mol. Cell. Biol. 2001, 2, 127-137.
- Signaling revealed by mapping molecular interactions: implications for ErbB-targeted cancer immunotherapies. Clin. Appl. Immunol. Rev. 2002, 2, 169-186.
- 134 M.X. Sliwkowski, J.A. Lofgren, G.D. Lewis, T.E. Hotaling, B.M. Fendly, J.A. Fox. Nonclinical studies addressing the mechanism of action of trastuzumab (Herceptin). Semin. Oncol. 1999, 26, 60-70.
- 135 P. Bagossi, G. Horváth, G. Vereb, J. Szöllősi, J. Tőzsér. Molecular modeling of nearly full-length ErbB2 receptor. Biophys. J. 2005, 88, 1354–1363.
- 136 E. Tzahar, H. Waterman, X. Chen, G. Levkowitz, D. Karunagaran, S. Lavi, B.J. Ratzkin, Y. Yarden. A hierarchical network of interreceptor interactions determines signal transduction by Neu differentiation factor/neuregulin and epidermal growth factor. Mol. Cell. Biol. 1996, 16, 5276-5287.
- 137 M.X. Sliwkowski, G. Schaefer, R.W. Akita, J.A. Lofgren, V.D. Fitzpatrick, A. Nuijens, B.M. Fendly, R.A. Cerione, R.L. Vandlen, K.L. Carraway, III. Coexpression of erbB2 and erbB3 proteins reconstitutes a high affinity receptor for heregulin. J. Biol. Chem. 1994, 269, 14661-14665.
- 138 T.W. Gadella, Jr., T.M. Jovin. Oligomerization of epidermal growth factor receptors on A431 cells studied by time-resolved fluorescence imaging microscopy. A stereochemical model for tyrosine ki-

174 References

nase receptor activation. J. Cell Biol. 1995, 129, 1543–1558.

- 139 P. Nagy, G. Vereb, Z. Sebestyén, G. Horváth, S. J. Lockett, S. Damjanovich, J. W. Park, T. M. Jovin, J. Szöllősi. Lipid rafts and the local density of ErbB proteins influence the biological role of homo- and heteroassociations of ErbB2. *J. Cell Sci.* 2002, 115, 4251–4262.
- 140 T. W. Gadella, Jr., R. M. Clegg, T. M. Jovin. Fluorescence lifetime imaging microscopy: Pixel-by-pixel analysis of phasemodulation data. *Bioimaging* 1994, 2, 139–159.
- 141 P. I. Bastiaens, T. M. Jovin. Microspectroscopic imaging tracks the intracellular processing of a signal transduction protein: fluorescent-labeled protein kinase C beta I. Proc. Natl. Acad. Sci. USA 1996, 93, 8407–8412.
- 142 P. Nagy, A. Jenei, A. K. Kirsch, J. Szöllősi, S. Damjanovich, T. M. Jovin. Activationdependent clustering of the erbB2 receptor tyrosine kinase detected by scanning near-field optical microscopy. J. Cell Sci. 1999, 112 (Pt 11), 1733–1741.
- 143 E. Monson, G. Merritt, S. Smith, J. P. Langmore, R. Kopelman. Implementation of an NSOM system for fluorescence microscopy. *Ultramicroscopy* 1995, 57, 257–262.
- 144 A. K. Kirsch, C. K. Meyer, T. M. Jovin, Integration of optical techniques in scanning probe microscopes: the scanning near-field optical microscope (SNOM). In: *Proceedings of NATO Advanced Research Workshop: Analytical Use of Fluorescent*

Probes in Oncology, Plenum Press, New York, 1996, pp. 317–323.

- 145 F. G. Giancotti, E. Ruoslahti. Integrin signaling. Science 1999, 285, 1028–1032.
- 146 S. Miyamoto, H. Teramoto, J. S. Gutkind, K. M. Yamada. Integrins can collaborate with growth factors for phosphorylation of receptor tyrosine kinases and MAP kinase activation: roles of integrin aggregation and occupancy of receptors. J. Cell Biol. 1996, 135, 1633–1642.
- 147 F. Wang, V.M. Weaver, O.W. Petersen, C.A. Larabell, S. Dedhar, P. Briand, R. Lupu, M.J. Bissell. Reciprocal interactions between β1-integrin and epidermal growth factor receptor in three-dimensional basement membrane breast cultures: a different perspective in epithelial biology. *Proc. Natl. Acad. Sci. USA* 1998, 95, 14821–14826.
- 148 B. Leitinger, N. Hogg. The involvement of lipid rafts in the regulation of integrin function. J. Cell Sci. 2002, 115, 963–972.
- 149 M. A. Adelsman, J. B. McCarthy, Y. Shimizu. Stimulation of β1-integrin function by epidermal growth factor and heregulin-β has distinct requirements for erbB2 but a similar dependence on phosphoinositide 3-OH kinase. *Mol. Biol. Cell* 1999, 10, 2861–2878.
- 150 M. Mocanu, Z. Fazekas, M. Petrás, P. Nagy, Z. Sebestyén, J. Isola, J. Timar, J.W. Park, G. Vereb, J. Szöllősi. Associations of ErbB2, β1-integrin and lipid rafts on Herceptin (Trastuzumab) resistant and sensitive tumor cell lines. *Cancer Lett.* 2005, 227, 201–212.

8 Caveolin and its Role in Intracellular Chaperone Complexes

William V. Everson and Eric J. Smart

8.1 Caveolae and Caveolin-1

Our understanding of plasma membrane microdomains in general, and of caveolae and lipid rafts in particular, has grown tremendously during the past few years. One important paradigm shift occurred with the clear demonstration that two very similar domains exist on the plasma membrane called caveolae and lipid rafts [1,2]. Caveolae in most cells and tissues are defined by the presence of a 22-kDa protein called caveolin-1, whereas lipid rafts are defined by the presence of glycosyl-phosphatidylinositol (GPI)-anchored proteins such as CD55. Caveolae and lipid rafts are similar in that both types of domain are enriched in cholesterol, sphingomyelin, and signaling proteins [3]. However, caveolae contain significantly more cholesterol than lipid rafts, and it is this high relative concentration of cholesterol that is thought to be responsible for the invaginated morphology of caveolae in some cell types [4,5]. In contrast, caveolae that are depleted of cholesterol can lose their invaginated morphology and flatten within the plane of the plasma membrane [6]. Thus, the contents of caveolin and cholesterol in cells appear to be critical elements in the formation, number and morphology of caveolae. The localization of caveolin-1 in caveolae is critical in the organization of caveolae, and this is a combined result of the effect of caveolin on the cholesterol content, organization of the membrane lipid components, and the effect of caveolin as it forms oligomers and is able to bind to and scaffold other proteins.

Few studies have been conducted in which the levels and distribution of caveolin-1 and the absolute content of cholesterol within caveolae have been analyzed. Variation has been observed in the content of caveolin in caveolin-expressing cells and tissues that differ by more than two orders of magnitude. Quite clearly, a "caveolin-containing" cell that has a relative 100-fold enrichment of caveolin compared to another "caveolin- containing" cell is quite likely to have a dramatically different organization of caveolae, a different proportion of caveolae within the plasma membrane, and a different distribution of caveolin-organized domains within other intracellular membranes causing variation in caveolin's role in the regulation of trafficking and signaling pathways. However, few studies have taken

176 8 Caveolin and its Role in Intracellular Chaperone Complexes

into account the heterogeneity and functional differences between distinct cell types. In a parallel argument, cells in which caveolin expression has been lost or ablated in culture will respond differently, reflecting the alteration of steady-state equilibria between caveolae domains, lipid rafts, and phospholipids regions of the plasma membrane and intracellular membranes. Thus, some signaling or trafficking pathways normally regulated by caveolin may show little impact of caveolin disruption due to compensatory activation of secondary pathways localized to lipid rafts or even phospholipids domains. A number of ligands that initiate signaling in caveolae can be internalized by clathrin-mediated endocytosis, although in caveolin-expressing cells the rapid caveolae-mediated endocytic pathway dominates under normal physiologic conditions. This also holds true for membrane components. Overexpression of caveolin-1 changed the uptake of specific glycosphingolipids and shifted GM1 from the clathrin endocytic pathway to a caveolae-endocytic pathway [7]. The abolition of GPI synthesis to cause cell depletion of GPI-anchored proteins resulted in an increase in caveolae at the cell surface, indicating that some type of inverse relationship existed between the amount of lipid rafts and caveolae expressed on the cell surface [8]. The manipulation of total cell cholesterol content with cyclodextrin has been shown to have profound effects on caveolae organization and localization of caveolae-associated proteins, including caveolin-1. The effects of cyclodextrin require careful measurement of caveolae cholesterol content before and after treatment to determine whether such effects either disrupt caveolae or deplete the entire plasma membrane surface of cholesterol [9]. Physiological manipulations, such as the addition of lipoproteins which bind to receptors and alter cholesterol distribution in caveolae, result in changes in caveolae cholesterol and caveolin content, and change caveolae signaling and trafficking as a consequence [10,11]. These studies document the importance of careful quantitative analysis of the level of expression of caveolin protein and the mass of cholesterol in the cell as a whole, and within the caveolae compartment specifically, to demonstrate the physiological relevance of altered caveolae cholesterol levels to the change that occurs in a specific pathway or activity. Otherwise, the change could be due to alterations in cholesterol levels in noncaveolae membrane domains which cause a change in a pathway associated with these domains.

Because of the steady-state interactions between caveolae and other membrane domains, the merging of endocytic pathways and capacity for redundancy in endocytosis, and redundancy in signal transduction that has been identified since the first null mouse without an obvious phenotype was produced, care must be taken in interpreting the results of gross manipulations which alter protein and lipid content of cells and caveolae. In particular, studies in cells in which caveolin is absent, or in which the expression has been ablated, must be carefully evaluated. Several studies have been published from which conclusions have been drawn using negative data about the role (or lack of role) of caveolae and caveolin in a particular pathway, without having incorporated controls to ensure that the model system being employed is physiologically relevant or evaluation of changes in the activity of compensatory, overlapping, or redundant pathways.

8.2 Caveolin Protein Structure, Domains, and Membrane Interactions

Three caveolins have been identified in mammalian cells, termed caveolin-1, -2, and -3. Caveolin-1 and -2 are co-expressed in a wide range of tissues, exhibiting high expression in lung, vascular tissue, fibroblasts and adipocytes, whereas caveolin-3 shows limited expression and is the dominant caveolin in striated muscle. Caveolin-1, a 21- to 24-kDa protein, is a major integral membrane component of caveola membranes (Fig. 8.1). Caveolin-1 contains an additional N-terminal 27 amino acids (aa) that are not found in caveolin-3, and is also 16 amino acids longer than caveolin-2. Thus, all three caveolins resolve at distinct molecular weights by sodium dodecylsulfate-polyacryamide gel electrophoresis (SDS-PAGE). The predicted secondary structure of caveolin contains an N-terminal cytoplasmic region, a membrane-associated domain, and a cytosolic C-terminal tail. All three caveolins exhibit a similar structure, with the N- and C-terminal ends of the protein in the cytoplasm flanking a ~20 aa hydrophobic domain that forms a hairpin in the membrane. Several regions of caveolin-1 have been identified that mediate interactions



Fig. 8.1 Caveolin protein structure and domains. The distinct functional domains in caveolin-1 (Cav-1) are shown graphically and labeled as described in the text. Both caveolin-1 and caveolin-2 have alternative splice sites and exist as full-length alpha and truncated beta isomers. Domains showing potential conservation based on primary sequence homology are aligned for caveolin-3 (below Cav-1) and caveolin-2 (above Cav-1). Caveolin-3 shows high sequence homology in the hairpin, oligomerization, scaffolding and C-

terminal cytoplasmic domains, including conservation of the position of three cysteines that are acylated in caveolin-1, as discussed in the text. Caveolin-2 shows homology in the hairpin domain and the N-terminal half of the oligomerization domain, but differs significantly in the primary sequence of the scaffolding domain and the C-terminal cytoplasmic domain. All three caveolins contain unique sequences in the N-terminal region that aligns with the first 50 amino acids of caveolin-1.

178 8 Caveolin and its Role in Intracellular Chaperone Complexes

with itself and other proteins. The N-terminal membrane proximal region (residues 61–101) of caveolin-1 is sufficient to mediate the formation of caveolin homooligomers, and a portion of this controls exit from the Golgi [12]. A part of this region (residues 82–101) has been termed the "scaffolding domain", and mediates the putative interaction of caveolin-1 with signaling molecules such as small GTPbinding proteins, Src family tyrosine kinases, and endothelial nitric oxide synthase (eNOS) [13,14]. Other domains within caveolin have also been reported to interact with and serve as scaffolds to bind signaling molecules to caveolae [15].

Caveolin-1 has three cysteines, caveolin-2 has five cysteines, and caveolin-3 has eight cysteines. Caveolin-1 and -3 are both highly homologous at the protein level, with conservation of the location of the three cysteines C-terminal to the hairpin domain that we have shown are functionally acylated in caveolin-1 (discussed below). Caveolin-2 shows much less homology, and none of its cysteines aligns with the cysteines in caveolin-1. It is not known whether caveolin-2 or caveolin-3 are acylated proteins. The potential role of acylation in the control of membrane interactions of caveolin-1 are discussed below, but the acylation and potential functions associated with acylation in caveolin-2 or caveolin-3 remain to be elucidated.

8.3

Caveolin Expression and Localization in the Cell

Subcellular fractionation, immunofluorescence microscopy and electron microscopy with immunogold labeling methods have each been used to examine the localization of caveolin-1 in different tissues. Early studies isolated caveolae and lipid rafts based on the resistance of these cholesterol- and sphingolipid-rich domains to solubilization in cold buffers containing Triton X-100 detergent and the light buoyancy of the fractions on sucrose gradients. A nondetergent methods was developed that is based on sequential density gradient centrifugation, in which the plasma membrane is first isolated, after which the caveolae and noncaveolae domains are separated on a second density gradient. An analysis of the purity and enrichment of the final caveolae fraction showed that this method yields a highpurity fraction, enriched in caveolae, and containing minimal contaminants (< 5%) from other organelle fractions [16]. In contrast, when detergents such as 0.1% Triton X-100 are used with density gradient centrifugation to isolate membranes enriched in cholesterol and sphingolipids, the buoyant fractions have been shown to be a mixture of membranes from multiple organelles. Moreover, during the isolation procedure mixing of the lipids and artifactual associations of proteins has been shown to occur. A rigorous technique to isolate caveolae from lung tissue by using cationic colloidal silica particles showed that starting with the plasma membrane and avoiding the use of detergent eliminates many of the contaminants originating from intracellular membranes when detergent-based fractionation methods are used [17].

The analysis of caveolae distribution within cells has been carried out using subcellular fractionation with calculation of relative recoveries of material in differ-

ent fractions, and by immunomicroscopic methods and gene manipulation techniques. When the overall distribution of caveolin was examined to determine where caveolin is localized by either confocal microscopy or subcellular fractionation by these more rigorous methods, a substantial quantity of caveolin was found to be associated with the intracellular membranes, and caveolin-1 has been shown to be present at some level within most subcellular fractions.

Several studies have identified technical issues that must be considered in the immunolocalization of caveolin and caveolae-associated proteins. It is well known that epitopes in proteins can be masked by interactions of the protein with other proteins, macromolecules such as lipids and membrane surfaces. It appears that this technical issue is very important in studies on proteins that associate with cholesterol-/sphingolipid-rich membranes or surfaces rich in neutral lipids. Several published studies have reported apparently conflicting results with regard to the expression or localization of caveolae in cells or tissues. Some of the first commercially available antibodies against caveolin-1 were known to recognize epitopes that were easily hidden when caveolin localized to different intracellular sites. Methods such as brief treatment with SDS have been employed to allow for antigen retrieval [18], giving antibodies access to epitopes in caveolin on intracellular membranes in leukocytes. Some apparent discrepancies in these relative expression or localization studies can be explained by the differences in epitope accessibility dependent on fixation, permeabilization, and the antibody detection methods employed. A recent study expanded this point to show that a number of current commercially available antibodies that are in wide use differ in their abilities to bind to caveolin at distinct sites, and that individual antibodies show distinct localization patterns under different fixation and permeabilization conditions [19]. Again, this study highlighted the importance of careful evaluation of negative data and the danger of making sweeping conclusions based on negative evidence. Confirmation of data interpretation by employment of two or more alternative approaches – for example, subcellular fractionation, combined with either immunomicroscopic methods or genetic studies with expression of mutant proteins in cells or animals - are critical when studies are carried out to evaluate the localization of caveolin and its relevance to regulation of signaling and trafficking pathways.

Because of the close association between caveolin and cholesterol- and sphingolipid-rich domains, the presence of caveolin in most subcellular fractions and organelles suggests that specialized membrane domains may be important functional units in many organelles and subcellular fractions. This is in addition to the prominent roles that have been identified for caveolae in the plasma membrane in the regulation of signal transduction and rapid endocytosis. Much remains to be investigated to define the functions of these domains within distinct organelles, or those involved in specialized trafficking between intracellular organelles based on either protein or lipid organization of these domains. The results of these studies clarify the point that negative data alone may be of little value, and that a combination of antibodies with several methods for fixation, antigen unmasking and detection of caveolins may be required to evaluate the total distribution of caveolin-1 within the cell.

8.4

Caveolin Expression and Localization Varies Depending on the Physiological State of Cells in Culture

The localization of caveolin in cells has been shown to be dependent on culture conditions, growth state, and the degree of confluence. In studies in which caveolin has been expressed in cells lacking caveolae, such as lymphocytes [20] or cancer cells [21], caveolin-1 expression resulted in its localization to the plasma membrane, the formation of caveolae, and the redistribution of other caveolaespecific proteins to these newly formed caveolae. The level of expression and distribution of caveolin-1 in isolated versus confluent endothelial cells is dramatically different. As cells approach confluence, caveolin and cholesterol accumulate at the lateral borders of endothelial cells, along with other caveolin-associated proteins and signaling molecules, including annexin II and other caveolae-localized proteins [22]. The organization of caveolae and associated proteins in these lateral membranes has been well defined in venular endothelial cells (Fig. 8.2). In confluent venular endothelial cells and intact veins, a large tubular vesicular network forms from endosomes originating from caveolae. This system has been extensively characterized and identified as being critical in transcytosis and leukocyte extravasation, and is termed the vesicular vacuolar organelle (VVO). The VVO represents an extensive set of tubules and vesicles that are connected, dynamic, and extend to take up approximately 18% of the total volume of the cytoplasm [23-25]. Several studies have linked caveolin and caveolae to macromolecule transport across the endothelium [26,27]. The structure and regulation of caveolae and its associated structures such as the VVO may be important in the functions of caveolin in vivo in control of vessel permeability and leukocyte migration. There is much that remains to be investigated using appropriate models, as well as studies in intact vessels and animals.

Both caveolin-1 and caveolin-2 exhibit widespread tissue distribution and extensive co-localization within the cell [28–31]. Both are found distributed in association with the plasma membrane and with intracellular membranes associated with several organelles in cells. Caveolin-1 and caveolin-2 also exhibit coordinate changes in their tissue-specific expression. For example, ablation of caveolin-1 in the mouse resulted in the widespread down-regulation of caveolin-2 in all of the tissues in which caveolin-1 was expressed [31]. Mutations in either caveolin-1 or caveolin-2 cause both proteins to mis-localize and accumulate with newly formed lipid bodies localized in the perinuclear region of the cell [32,33]. An intriguing study in cells lacking both caveolin-1 and -2 (LnCaP cells) showed that the formation of caveolae required expression of both caveolin-1 and caveolin-2, and that the localization of caveolin-1 to the cell surface required phosphorylation of caveolin-2 at specific sites [21]. However, the caveolin-2 null mouse shows normal caveolin-1 expression and normal caveolae in many tissues [34]. These data are intriguing, and suggest that much remains to be defined in the specific function, role and interactions between caveolin-1 and -2 in the regulation of caveolae organization and associated functions.



Fig. 8.2 Caveolin-1 is present in multiple subcellular locations. Top panel: a typical immunoblot from subcellular fractions isolated from confluent human umbilical vein endothelial cells (HUVEC). Twelve fractions were collected and an aliquot, matched for recovery to compare the amount of caveolae in different fractions, loaded into lanes 1-12 (light to dense fractions). The first two fractions represent the cytosol (the fraction loaded). Caveolin was present in fractions 1-2 corresponding to the cytosol (Cyto), fractions 3-5 which contain the plasma membrane and caveolae (Cav), fractions 6-8 which contain the Golgi (Golgi), fractions 9-11 which contain the endoplasmic reticulum and microsomes (ER/Mc), fraction 12 which contains the nuclear pellet (not labeled), and the lane containing the homogenate (H). Bottom panel: a schematic of caveolin-1-positive membranes and vesicles in a "typical" HUVEC. Caveolin localizes to the apical plasma membrane in small clusters and large clusters of caveolae (1, 2), to vesicles near

clusters of caveolae (3), to multiple clusters of vesicles within the cytosol which may correspond to early endosomes (4), a sorting or recycling compartment(s) (5), vesicles clustered in the perinuclear region (11), to the vesicular vacuolar organelle (VVO) which comprises a large volume in the cytosol adjacent to the lateral membrane, to regions within lateral membranes (6), regions of the ER (7) and Golgi (8), and domains within the basal membrane including focal adhesions (10). Inset: the inset in the bottom panel shows a single confocal "slice" obtained from confluent HUVEC following immunostaining with caveolin-1 antibody followed by Cy3 fluorophore conjugated to a second antibody. The location of the nucleus is indicated (N), and regions with intense staining are marked which correspond to caveolin-positive membranes and vesicles in large clusters adjacent to the plasma membrane (2,3,4), the lateral membrane and VVO (6,9), and the perinuclear vesicles (11).

8.5

Caveolin-1 Expression Confers a New Level of Regulation

Caveolin-1 association with a number of proteins can be inhibitory. Likewise, caveolin expression in tissues in which its expression is normally low results in the formation of caveolae. The presence of caveolin then causes docking of signaling proteins, inhibition where proteins were previously constitutively active, and induction of regulatory pathways where they were absent, as a result. For example, expression of caveolin in hepatocytes causes down-regulation of lipid uptake in the liver [35], a tissue that is normally active in the uptake of high-density lipoprotein (HDL), leading to higher circulating levels of lipoproteins as a result. Loss of caveolin in cancer removes a regulatory structure, and the inhibitory activity that caveolin-1 confers to a number of signaling proteins, resulting in overactivation of growth signaling. Dysregulation of caveolin-1 is associated with an overactivation of growth signals in breast [36,37], and prostate cancers [38-40]. Several of the growth-signaling pathways which are constitutively turned on in prostate cancer, are found to be associated, and regulated, within caveolae in normal cells. A number of other proteins have also been suggested to be inhibited by the direct binding of caveolin-1.

8.6

Caveolae Cholesterol and Caveolin Localization to Caveolae

Soon after caveolin was first cloned, studies with filipin - an antibiotic that specifically binds cholesterol [41-43] showed that filipin-cholesterol complexes were enriched in nonclathrin-coated endosomes and regions of membrane corresponding to caveolae. This raised the question of how and why cholesterol was enriched in caveolae. Experimental manipulation of caveolae cholesterol by the addition of bacterial cholesterol oxidase to the media of cells was used to investigate this question [6,44]. Cholesterol oxidase caused rapid oxidation of cholesterol in caveolae, leading to cholesterol dispersal and the rapid internalization of caveolin-1 [6]. The removal of cholesterol oxidase from the media allowed the re-formation of caveolae, and both caveolin-1 and cholesterol were restored to caveolae in the plasma membrane over a period of about 90 minutes. Immunolocalization studies showed that caveolin-1 moved rapidly out of the caveolae, and was subsequently found on vesicles in the endoplasmic reticulum-Golgi-intermediate-compartment (ER-GIC), a region between the Golgi and the endoplasmic reticulum (ER) [6]. This identified a very rapid pathway connecting the plasma membrane to a region adjacent to the ER, as caveolin totally redistributed within 30 seconds following the addition of bacterial cholesterol oxidase. Temperature shifting and inhibitors were employed to define a cycle of caveolin-1 movement from cell-surface caveolae into the cell and back to the cell surface. Caveolin-1 rapidly translocated first to the rough ER (within 30 seconds), then slowly over 15–20 minutes to the ER-GIC, to the Golgi, before finally returning to surface caveolae. An investigation of cholesterol changes during the same time course showed that cholesterol followed a similar pattern of movement out of the caveolae, followed by a slow return to the caveolae, suggesting that caveolin-1 and cholesterol were linked together in a cycle. These studies were among the first to reveal a caveolin-trafficking cycle between cell-surface caveolae and the ER-Golgi membranes, for the transport of both caveolin and cholesterol between the plasma membrane and these perinuclear organelles.

8.7 Caveolin and Cholesterol Cross Membranes During Trafficking

During the process of vesicle trafficking, membrane topology is normally conserved – that is, material found in the extracellular (exofacial) leaflet of the plasma membrane is topologically conserved. For example, it remains associated with the exofacial leaflet during endocytosis and after fusion with organelles. The exofacial leaflet is thus the inner leaflet of vesicles and the interior leaflet of organelles such as the ER. Proteins and lipids associated with the cytofacial leaflet (cytoplasmic side) remain associated with the cytofacial leaflet of endocytic vesicles, and upon fusion remain localized with the cytoplasmic leaflet. Surprisingly, during translocation of caveolin and cholesterol between the plasma membrane and the ER, caveolin-1 "flips" across the membrane from the cytofacial leaflet and becomes associated with the exofacial leaflet of the ER [6,45] rather than the cytosolic leaflet.

Early studies which identified caveolin as a transmembrane protein may have been based on this same process of flipping, as this appears to occur when caveolin is newly synthesized. Initially, caveolin is transported to the cell surface with cholesterol, but then appears outside the cell, and is subsequently found localized in the cytofacial leaflet of the plasma membrane in caveolae. This suggests that caveolin-1 has a role in cholesterol trafficking across membranes from one leaflet to another, and that this flipping function is active at multiple membranes where cholesterol is transported. Another protein family has been identified that exhibits this same flipping behavior; this family - the engrailed family of homeoproteins associates with caveolae when expressed in COS cells [46,47]. The engrailed protein, as well as other homeodomain proteins, contains a short domain that has been identified as a potential flip domain, which may allow their crossing from one layer to the other in membranes rich in cholesterol [46,47]. Robenek et al. have recently further characterized the crossing of caveolin-1 from one to another during endocytosis using ultrathin cryosectioning, freeze fracture, and immunogold labeling [45]. During the cycling of caveolin between the plasma membrane and intracellular sites, caveolin switches from the cytofacial leaflet to the exofacial leaflet, and is exclusively localized within the exofacial leaflet in the ER [45]. As mentioned above, overexpression of caveolins or expression of specific caveolin mutants causes formation of lipid droplets marked by caveolin localized to the ER, as lipids accumulated and budded off from the cytofacial leaflet due to the accumula-

184 8 Caveolin and its Role in Intracellular Chaperone Complexes

tion of caveolins at the ER [32,33]. Another study extended these observations to examine the relationship between caveolin-1 and lipid droplet biogenesis originating in association with the ER [48]. Caveolin-1 was found localized to the surface of droplets, and also extensively localized to the interior of the droplets, which exhibited a highly organized lamellar structure. However, caveolin-1 was localized strictly in association with the exofacial leaflet of the ER. The model of lipid budding as lipids accumulate and aggregate within the cytofacial leaflet would result in droplets lacking caveolin. There may be differences due to cell type, and droplets produced for different functional reasons may involve different pathways in their formation. Lipid droplets that are transient, rapidly induced and critical to inflammatory prostaglandin production are likely to differ from lipid droplets in adipocytes. The specific proteins and mechanisms regulating their formation and breakdown are also likely to differ. Much remains to be defined before we understand the basic mechanisms regulating lipid storage and specialized lipid-metabolizing organelles, and caveolin has been localized to both types of droplets in specific cell types.

8.8

Two Chaperone Complexes Regulate a Caveola-Cholesterol Trafficking Cycle

In addition to organizing cholesterol in caveolae, caveolin has been found to play a specific role in nonvesicular cholesterol trafficking in cells. A direct role for caveolin in trafficking newly synthesized cholesterol from the ER to the plasma membrane was shown in cells which lacked caveolin and in which caveolin was reexpressed. Radiolabeled tracers and blockers of cholesterol transport were used to show that caveolin and cholesterol trafficked together from intracellular sites to cell-surface caveolae [49]. A subsequent study revealed that a chaperone complex ferries newly synthesized cholesterol from the ER through the cytosol to plasma membrane caveolae [50]. This chaperone complex was isolated from the cytosol, distinct from any caveola membrane, plasma membrane, intracellular membrane or vesicle fraction. The complex was shown to be comprised of cholesterol, caveolin-1 and three additional proteins – cyclophilin A (cypA), cyclophilin 40 (cyp40), and heat shock protein 56 (hsp56). The activity of this complex could be inhibited by compounds which bind to and inhibit the function of hsp56 (rapamycin) or interact with cyclophilins (cyclosporine). Either of these inhibitors prevented the formation of the chaperone complex and blocked cholesterol movement from the Golgi to plasma membrane caveolae.

The linkage between caveolin movement in and out of caveolae to caveolae cholesterol homeostasis led to the subsequent identification of a second chaperone complex which regulates transport of newly delivered cholesterol into the cell. This second complex consists of cholesteryl ester, caveolin-1, cypA, cyp40, and annexin II [10]. HDL can deliver cholesteryl ester to caveolae via scavenger receptor BI (SR-BI). Cholesteryl ester is then transported via this complex into the cell, where it can be converted to cholesterol and transported back to the cell surface to caveolae via the first chaperone complex. These two complexes provide a nonvesicle pathway which can regulate cholesterol uptake and delivery from caveolae in fibroblasts and other caveolin-1-containing cells, but not in cells such as hepatocytes, which normally contain minimal caveolin-1.

8.9 Caveolae Linked to Nongenomic Actions and Uptake of Estrogen

The initial paradigm of sex hormone action in the cell begins with the delivery of sex hormones to the cell surface, passage across the plasma membrane, interaction with a receptor localized in the cytosol, and transport of the hormone-receptor complex to the nucleus, where the sex hormone-bound receptor acts to alter specific gene transcription. During the early 1990s, a few studies showed that the initial model proposing that the steroid receptor was the cytosolic chaperone for these hydrophobic compounds (which do not diffuse readily through an aqueous environment) was in fact incorrect, since the majority of the estrogen receptors were found exclusively in the nucleus. The initial studies which observed estrogen receptors within the cytosol were due to an artifact that occurred during processing [51,52]. In the absence of ligand, the receptor had a lowered binding affinity for DNA, causing the receptor to leak out of the nucleus. Thus, the initial steps that were posited to explain uptake and transportation across the cytosol to the nucleus no longer had a mechanistic explanation. Additionally, to date, there is still no mechanism in place to account for the initial uptake of sex hormone by cells, or to explain how sex hormones cross the plasma membrane to enter the cell, or what regulates delivery to the nucleus or sites of nongenomic action. A number of recent studies have begun to fill in this gap, however. The first point to be made was that caveolae mark cholesterol-rich membranes which are organized into a network in the cell, providing an organized set of membranes through which other sterols can "diffuse". The second point was the discovery that the estrogen receptor is found associated with caveolae at the plasma membrane [53-55]. The third point is the recent finding that activation of nitric oxide by sex hormones requires a specific pool of estrogen which is tightly associated with HDL [14]. Estrogen is tightly bound specifically with HDL, and HDL from female mice - but not male mice - activates eNOS. The activation of eNOS is linked to HDL binding to SR-BI and SR-BI-mediated delivery of estrogen to caveolae. This study was the first to identify caveolae as a putative site of uptake of estrogen into endothelial cells. The results of these studies, taken together, define an emerging new paradigm of sex hormone delivery to cells and transport of sex hormones from the plasma membrane to specific intracellular organelles which initiate in caveolae. The role of intracellular caveolin-associated membranes in estrogen trafficking within the cell remains to be elucidated. This emerging paradigm of the first steps in estrogen delivery to cells may represent a new paradigm for the uptake of some other sterols. Recently, vitamin D was shown to bind to a pool of the nuclear vitamin D receptor, which localizes to caveolae in the intestine [56,57]. Thus, the initial inter-

186 8 Caveolin and its Role in Intracellular Chaperone Complexes

action of vitamin D with cells and many of its nongenomic actions appears to extend this emerging model of sex hormone uptake to a model that may be important in a broader context of other sterols. In addition, it might also provide novel insights into the uptake and action of other lipophilic hormones, vitamins, and nutrients.

8.10

Protein Acylation and Caveolae

Acylation of proteins has been linked to caveolae localization, presumably due to the direct interaction of acyl chains by intercalation into the organized lipid of caveolae membranes. Site-directed mutational analysis of eNOS acylation shows the role of two-site acylation in caveolae association and regulation of eNOS activation in caveolae [58]. eNOS is myristoylated during post-translational processing and acylated by covalent attachment of fatty acids at the N-terminus and at a specific cysteine. Silencing of each of these acylation sites inhibited eNOS activity to one-tenth of the activity of wild-type eNOS, and silencing of both acylations is synergistic, resulting in one-hundredth of the activity. The loss of acylation resulted in a loss of association of eNOS with caveolae to the same extent as the loss of activity.

Caveolin itself is acylated at three cysteines, and acylation has been shown to be important in the interaction and organization of caveolin and other proteins with caveolae [59,60]. Acylation is also important in the interaction of caveolin-1 with cholesterol in chaperone complexes. Site-directed mutagenesis studies have revealed that two of the three acylation sites – Cys143 and Cys156 – are required for formation and activity of the chaperone complex that delivers cholesterol from the Golgi to caveolae [61]. The third acylation, at Cys133, is essential for the formation and activity of the chaperone complex that internalizes cholesterol ester into cells [10].

A number of studies have shown that acylation of proteins is a factor in raft association of proteins and in raft signaling. The Src family of protein tyrosine kinases has been found to contain a number of different acyl groups, with stearate and oleate derivatives showing decreased affinity for rafts compared to the myristoylated forms [62]. Enzymes have been identified which can de-acylate lipid raftassociated proteins, such as a recently cloned lysophospholipase/acyl thioesterase [63]. We have observed heterogeneity in the acylation of several caveolin-associated proteins (unpublished observations). The mechanisms and roles of acylation in the overall organization and dynamics of caveolae are fruitful areas for further exploration.

8.11 Scavenger Receptors Localize to Caveolae

CD36 has been localized to lipid rafts in several tissues, and we and others have shown CD36 to be localized within endothelial caveolae. CD36 has a putative role in the control of hypertension, and in angiogenesis [64,65]. CD36 in striated muscle caveolae appears to be important in the uptake and accumulation of fatty acids, in both skeletal muscle and heart [66,67]. Scavenger receptor class B, Type I is an 82-kDa glycosylated plasma membrane protein that is closely related to CD36 [68-72]. SR-BI is a transmembrane protein with short cytoplasmic N-terminal and Cterminal cytoplasmic domains, two transmembrane domains, and a large extracellular domain comprising the majority of the protein [69,71,72]. SR-BI is a receptor for HDLs and has broad ligand-binding specificity. In addition to HDL, SR-BI binds native low-density lipoprotein (LDL), oxidized LDL, acetylated LDL, and anionic phospholipids, but not the many polyanions that are bound by the type-A scavenger receptors [73-75]). SR-BI is a physiological receptor for HDL and also plays a role in regulating plasma cholesterol levels; however, the function of SR-BI in cells other than hepatic or steroidogenic cells is not well understood. We have recently shown that SR-BI has additional roles in endothelial cells, regulating HDL-mediated delivery of estrogen to cells [14], and regulating HDL-mediated inhibition of a cell apoptosis pathway involving eNOS [76].

8.12

Cholesterol Homeostasis Regulates Caveolin Localization and Organization of other Proteins in Caveolae

The cycle of cholesterol movement between cell-surface caveolae and intracellular pools rich in cholesterol contributes to caveolae cholesterol homeostasis. This cycle is controlled in a complex fashion that can involve scavenger receptors (SR-BI and CD-36), lipoproteins, and even sex hormones [11,14,65,77,78]. Many studies using methyl-beta-cyclodextrin have documented that cholesterol depletion and repletion cause major changes in the organization of caveolae, caveolin-1 localization to caveolae, or redistribution to intracellular sites, and resultant effects on numerous signaling proteins that are found in association with caveolae or bound to caveolae by caveolin-1. A recent report has shown that shear stress induction of eNOS activity is also susceptible to disruption of caveolae cholesterol, either by methylbeta cyclodextrin depletion, or by cyclosporine A inhibition of caveolin-1-cypA interaction [79]. This indicates that disruption of this chaperone complex cycle contributes to the endothelial pathogenesis which leads to hypertension induced by cyclosporine.

8.13

Chaperone Complexes Involved in Cholesterol Transport in Specialized Tissues

The intestine has a major role in the overall regulation of systemic cholesterol levels. We have identified a third chaperone complex that is involved in the net uptake and transport of cholesterol in the intestine [80]. Annexin II and caveolin-1 form a chaperone complex with cyp40 and cypA that binds cholesterol in the intestine. Ablation of annexin II or caveolin-1 in the zebrafish disrupted cholesterol uptake. Further, the drug ezetimibe decreased cholesterol levels in the circulation by disrupting intestinal cholesterol transport. Ezetimibe also disrupted this novel annexin II-caveolin-cholesterol chaperone complex. However, the interaction of annexin II with caveolin-1 in this intestinal complex is guite different from the two chaperone complexes described previously. Annexin II and caveolin-1 appear to form a heterodimeric band which is stable to heat, SDS and reducing agents, and was also immunoreactive against both annexin II and caveolin-1 and resolved as a complex at 55-58 kDa on SDS gels. This band, after digestion and analysis, yielded several peptides corresponding to regions of each individual protein, and its size was consistent with that of a heterodimeric complex. This novel complex is not limited to zebrafish, however. In mouse intestine, a band that corresponds to this heterodimeric complex, along with immunoreactive bands corresponding to the monomeric forms of each protein were also found. Recently, another protein – NPC1 like 1 protein, has been found in intestine, and shown to regulate intestinal cholesterol and sitosterol uptake [81,82]. NPC-1 fibroblasts (which lack functional NPC-1, the first protein in this family) show a block in cholesterol transport causing accumulation of cholesterol-rich late endosomes [83]. In these cells, both caveolin and annexin II exhibit elevated expression and accumulation with cholesterol in late endosomes. Further studies will be needed to determine the relationship between these two overlapping pathways and their roles in intestinal cholesterol trafficking.

8.14

Caveolin is Linked to Additional Sterol and Lipid Uptake and Trafficking Pathways

Several additional studies have indicated new roles for caveolin and associated proteins in the intracellular transport of cholesterol and the specific uptake of lipids. Caveolin-1 and SCP-2 directly interact in L-type fibroblasts, and co-localize both at the plasma membrane and in the cytosol [84]. The interplay between these nonvesicle trafficking pathways and vesicle-mediated transport in control of cholesterol trafficking and caveolae structure and organization will provide fruitful ground for further exploration.

8.15 Conclusions

Over a decade ago, caveolin was identified as a marker of caveolae and was found to be present in most cell types. Investigations of caveolae and lipid rafts over the past decade have resulted in more than 3500 publications which have brought tremendous change to our understanding of the organization of membranes and the functional organization of signaling and trafficking pathways within the cell. A new paradigm of the cell membrane consisting of multiple functional and structural domains has replaced the simple phospholipid bilayer model. This has also brought a conceptual paradigm shift of examining the integration of pathways linked through caveolae across the whole cell. The next frontier will involve the progression to models that are sufficiently sophisticated to provide a mechanistic understanding of the integration of pathways across tissues and organ systems. These investigations will be aided by use of cell- and tissue-specific knockout and knockin models which will, in turn, lead to a new level of integration in our understanding of the molecular events underlying diseases such as diabetes, inflammation, vascular disease, and cancer.

Abbreviations

cyp40	cyclophilin 40
сурА	cyclophilin A
eNOS	endothelial nitric oxide synthase
ER	endoplasmic reticulum
ER-GIC	endoplasmic reticulum-Golgi-intermediate-compartment
GPI	glycosyl-phosphatidylinositol
HDL	high-density lipoprotein
hsp	heat shock protein
HUVEC	human umbilical vein endothelial cells
LDL	low-density lipoprotein
РМ	plasma membrane
SDS-	sodium dodecylsulfate-polyacryamide gel electrophoresis
PAGE	
SR-BI	scavenger receptor BI
VVO	vesicular vacuolar organelle

References

Hooper NM. Detergent-insoluble glycosphingolipid/cholesterol-rich membrane domains, lipid rafts and caveolae (review).
Mol. Membr. Biol. 16: 145–156, 1999.
Iwabuchi K, Handa K, Hakomori S. Separation of glycosphingolipid-enriched microdomains from caveolar membrane characterized by presence of caveolin. Methods Enzymol. 312: 488–494, 2000.

- 190 References
 - 3 Smart EJ, Graf GA, McNiven MA, Sessa WC, Engelman JA, Scherer PE, Okamoto T, Lisanti MP. Caveolins, liquid-ordered domains, and signal transduction. *Mol. Cell. Biol.* 19: 7289–7304, 1999.
 - 4 Rothberg KG, Heuser JE, Donzell WC, Ying Y, Glenney JR, Anderson RGW. Caveolin, a protein component of caveolae membrane coats. *Cell* 68: 673–682, 1992.
 - 5 Rothberg KG, Ying Y, Kamen BA, Anderson RGW. Cholesterol controls the clustering of the glycophospholipid-anchored membrane receptor for 5-methyltetrahydrofolate. *J. Cell Biol.* 111: 2931–2938, 1990.
 - 6 Smart EJ, Ying YS, Conrad PA, Anderson RG. Caveolin moves from caveolae to the Golgi apparatus in response to cholesterol oxidation. J. Cell Biol. 127: 1185–1197, 1994.
 - 7 Singh RD, Puri V, Valiyaveettil JT, Marks DL, Bittman R, Pagano RE. Selective caveolin-1-dependent endocytosis of glycosphingolipids. *Mol. Biol. Cell* 14: 3254–3265, 2003.
 - 8 Abrami L, Fivaz M, Kobayashi T, Kinoshita T, Parton RG, van der Goot FG. Cross-talk between caveolae and glycosylphosphatidylinositol-rich domains. *J. Biol. Chem.* 276: 30729–30736, 2001.
 - 9 Smart EJ, Anderson RG. Alterations in membrane cholesterol that affect structure and function of caveolae. *Methods Enzymol.* 353: 131–139, 2002.
 - 10 Uittenbogaard A, Everson WV, Matveev SV, Smart EJ. Cholesteryl ester is transported from caveolae to internal membranes as part of a caveolin-annexin II lipid-protein complex. J. Biol. Chem. 277: 4925–4931, 2002.
 - 11 Uittenbogaard A, Shaul PW, Yuhanna IS, Blair A, Smart EJ. High density lipoprotein prevents oxidized low density lipoproteininduced inhibition of endothelial nitric-oxide synthase localization and activation in caveolae. J. Biol. Chem. 275: 11278–11283, 2000.
 - 12 Machleidt T, Li WP, Liu P, Anderson RG. Multiple domains in caveolin-1 control its intracellular traffic. *J. Cell Biol.* 148: 17–28, 2000.
 - 13 Ghosh S, Gachhui R, Crooks C, Wu C, Lisanti MP, Stuehr DJ. Interaction between caveolin-1 and the reductase domain of en-

dothelial nitric-oxide synthase. Consequences for catalysis. *J. Biol. Chem.* 273: 22267–22271, 1998.

- 14 Gong MC, Wilson ME, Kelly T, Su W, Dressman J, Kincer JF, Matveev S, Guo L, Guerin TM, Li X-A, Zhu W, Uittenbogaard A, Smart EJ. HDL-associated estradiol stimulates endothelial NO synthase and vasodilation in an SR-BI-dependent manner. J. Clin. Invest. 111: 1579–1587, 2003.
- 15 Yao Q, Chen J, Cao H, Orth JD, McCaffery JM, Stan RV, McNiven MA. Caveolin-1 interacts directly with dynamin-2. *J. Mol. Biol.* 348: 491–501, 2005.
- 16 Smart EJ, Ying YS, Mineo C, Anderson RG. A detergent-free method for purifying caveolae membrane from tissue culture cells. *Proc. Natl. Acad. Sci. USA* 92: 10104–10108, 1995.
- 17 Schnitzer JE, McIntosh DP, Dvorak AM, Liu J, Oh P. Separation of caveolae from associated microdomains of GPI-anchored proteins. *Science* 269: 1435–1439, 1995.
- 18 Robinson JM, Vandre DD. Antigen retrieval in cells and tissues: enhancement with sodium dodecyl sulfate. *Histochem. Cell. Biol.* 116: 119–130, 2001.
- 19 Pol A, Martin S, Fernandez MA, Ingelmo-Torres M, Ferguson C, Enrich C, Parton RG. Cholesterol and fatty acids regulate dynamic caveolin trafficking through the Golgi complex and between the cell surface and lipid bodies. *Mol. Biol. Cell* 16: 2091–2105, 2005.
- 20 Fra AM, Williamson E, Simons K, Parton RG. De novo formation of caveolae in lymphocytes by expression of VIP21-caveolin. *Proc. Natl. Acad. Sci. USA* 92: 8655–8659, 1995.
- 21 Sowa G, Pypaert M, Fulton D, Sessa WC. The phosphorylation of caveolin-2 on serines 23 and 36 modulates caveolin-1-dependent caveolae formation. *Proc. Natl. Acad. Sci. USA* 100: 6511–6516, 2003.
- 22 Corvera S, DiBonaventura C, Shpetner HS. Cell confluence-dependent remodeling of endothelial membranes mediated by cholesterol. J. Biol. Chem. 275: 31414–31421, 2000.
- 23 Dvorak AM, Feng D. The vesiculo-vacuolar organelle (VVO). A new endothelial cell permeability organelle. J. Histochem. Cytochem. 49: 419–432, 2001.

- 24 Dvorak AM, Kohn S, Morgan ES, Fox P, Nagy JA, Dvorak HF. The vesiculo-vacuolar organelle (VVO): a distinct endothelial cell structure that provides a transcellular I. Leukoc. Biol. 59: 100-115, 1996.
- 25 Feng D, Nagy JA, Dvorak HF, Dvorak AM. Ultrastructural studies define soluble macromolecular, particulate, and cellular transendothelial cell pathways in venules, lymphatic vessels, and tumor-associated microvessels in man and animals. Microsc. Res. Tech. 57: 289-326, 2002.
- 26 Bauer PM, Yu J, Chen Y, Hickev R, Bernatchez PN, Looft-Wilson R, Huang Y, Giordano F, Stan RV, Sessa WC. Endothelial-specific expression of caveolin-1 impairs microvascular permeability and angiogenesis. Proc. Natl. Acad. Sci. USA 102: 204-209, 2005.
- 27 Stan RV. Structure and function of endothelial caveolae. Microsc. Res. Tech. 57: 350-364, 2002.
- 28 Mora R. Bonilha VL. Marmorstein A. Scherer PE, Brown D, Lisanti MP, Rodriguez-Boulan E. Caveolin-2 localizes to the Golgi complex but redistributes to plasma membrane, caveolae, and rafts when co-expressed with caveolin-1. J. Biol. Chem. 274: 25708-25717, 1999.
- 29 Parolini I, Sargiacomo M, Galbiati F, Rizzo G. Grignani F. Engelman IA, Okamoto T. Ikezu T, Scherer PE, Mora R, Rodriguez-Boulan E, Peschle C, Lisanti MP. Expression of caveolin-1 is required for the transport of caveolin-2 to the plasma membrane. Retention of caveolin-2 at the level of the Golgi complex. J. Biol. Chem. 274: 25718-25725, 1999.
- 30 Scherer PE, Lewis RY, Volonte D, Engelman JA, Galbiati F, Couet J, Kohtz DS, van Donselaar E, Peters P, Lisanti MP. Celltype and tissue-specific expression of caveolin-2. Caveolins 1 and 2 co-localize and form a stable hetero-oligomeric complex in 40 Thompson TC, Timme TL, Li L, Goltsov vivo. J. Biol. Chem. 272: 29337-29346, 1997.
- 31 Scherer PE, Okamoto T, Chun M, Nishimoto I, Lodish HF, Lisanti MP. Identification, sequence, and expression of caveolin-2 defines a caveolin gene family. Proc. Natl. Acad. Sci. USA 93: 131-135, 1996.
- 32 Ostermeyer AG, Paci JM, Zeng Y, Lublin DM, Munro S, Brown DA. Accumulation

of caveolin in the endoplasmic reticulum redirects the protein to lipid storage droplets. J. Cell Biol. 152: 1071-1078, 2001.

- pathway for macromolecular extravasation. 33 Pol A, Luetterforst R, Lindsay M, Heino S, Ikonen E. Parton RG. A caveolin dominant negative mutant associates with lipid bodies and induces intracellular cholesterol imbalance. J. Cell Biol. 152: 1057-1070, 2001.
 - 34 Razani B, Wang XB, Engelman JA, Battista M, Lagaud G, Zhang XL, Kneitz B, Hou H. Christ GI. Edelmann W. Lisanti MP. Caveolin-2 deficient mice show evidence of severe pulmonary dysfunction without disruption of caveolae. Mol. Biol. Cell 22: 2329-2344, 2002.
 - 35 Frank PG, Pedraza A, Cohen DE, Lisanti MP. Adenovirus-mediated expression of caveolin-1 in mouse liver increases plasma high-density lipoprotein levels. Biochemistry 40: 10892-10900, 2001.
 - 36 Bouras T, Lisanti MP, Pestell RG. Caveolin-1 in breast cancer. Cancer Biol. Ther. 3: 931-941, 2004.
 - 37 Yang G, Truong LD, Timme TL, Ren C, Wheeler TM, Park SH, Nasu Y, Bangma CH, Kattan MW, Scardino PT, Thompson TC. Elevated expression of caveolin is associated with prostate and breast cancer. Clin. Cancer Res. 4: 1873-1880, 1998.
 - 38 Mouraviev V, Li L, Tahir SA, Yang G, Timme TM, Goltsov A, Ren C, Satoh T, Wheeler TM, Ittmann MM, Miles BJ, Amato RJ, Kadmon D, Thompson TC. The role of caveolin-1 in androgen insensitive prostate cancer. J. Urol. 168: 1589-1596, 2002.
 - 39 Pramudji C, Shimura S, Ebara S, Yang G, Wang J, Ren C, Yuan Y, Tahir SA, Timme TL, Thompson TC. In situ prostate cancer gene therapy using a novel adenoviral vector regulated by the caveolin-1 promoter. Clin. Cancer Res. 7: 4272-4279, 2001.
 - A. Caveolin-1, a metastasis-related gene that promotes cell survival in prostate cancer. Apoptosis 4: 233-237, 1999.
 - 41 Montesano R, Mossaz A, Vassalli P, Orci L. Specialization of the macrophage plasma membrane at sites of interaction with opsonized erythrocytes. J. Cell Biol. 96: 1227-1233, 1983.

- 192 References
 - 42 Montesano R, Vassalli P, Orci L. Structural heterogeneity of endocytic membranes in macrophages as revealed by the cholesterol 53 Chambliss KL, Shaul PW. Rapid activation probe, filipin. J. Cell Sci. 51: 95-107, 1981.
 - 43 Orci L, Montesano R, Meda P, Malaisse-Lagae F, Brown D, Perrelet A, Vassalli P. Heterogeneous distribution of filipin-cholesterol complexes across the cisternae of the Golgi apparatus. Proc. Natl. Acad. Sci. USA 78: 293-297, 1981.
 - 44 Conrad PA, Smart EJ, Ying Y, Anderson RWG, Bloom GS. Caveolin cycles between plasma membrane caveolae and the Golgi complex by microtubule-dependent and microtubule-independent steps. J. Cell Biol. 55 Kim HP, Lee JY, Jeong JK, Bae SW, Lee 131: 1421-1433, 1995.
 - 45 Robenek MJ, Schlattmann K, Zimmer KP, Plenz G, Troyer D, Robenek H. Cholesterol transporter caveolin-1 transits the lipid bilayer during intracellular cycling. FASEB J. 17: 1940-1942, 2003.
 - 46 Joliot A, Maizel A, Rosenberg D, Trembleau A, Dupas S, Volovitch M, Prochiantz A. Identification of a signal sequence necessary for the unconventional secretion of Engrailed homeoprotein. Curr. Biol. 8: 856-863, 1998.
 - 47 Joliot A, Trembleau A, Raposo G, Calvet S, Volovitch M, Prochiantz A. Association of Engrailed homeoproteins with vesicles presenting caveolae-like properties. Development 124: 1865-1875, 1997.
 - 48 Robenek MJ, Severs NJ, Schlattmann K, Plenz G, Zimmer KP, Troyer D, Robenek H. Lipids partition caveolin-1 from ER membranes into lipid droplets: updating the model of lipid droplet biogenesis. FASEB J. 18: 866-868, 2004.
 - 49 Smart EJ, Ying Y, Donzell WC, Anderson RG. A role for caveolin in transport of cholesterol from endoplasmic reticulum to plasma membrane. J. Biol. Chem. 271: 29427-29435, 1996.
 - 50 Uittenbogaard A, Ying Y, Smart EJ. Characterization of a cytosolic heat-shock protein-caveolin chaperone complex. Involvement in cholesterol trafficking. J. Biol. Chem. 273: 6525-6532, 1998.
 - 51 King WJ, Greene GL. Monoclonal antibodies localize oestrogen receptor in the nuclei of target cells. Nature 307: 745-747, 1984.
 - 52 Welshons WV, Lieberman ME, Gorski J. Nuclear localization of unoccupied oestro-

gen receptors. Nature 307: 747-749, 1984.

- of endothelial NO synthase by estrogen: evidence for a steroid receptor fast-action complex (SRFC) in caveolae. Steroids 67: 413-419, 2002.
- 54 Chambliss KL, Yuhanna IS, Mineo C, Liu P, German Z, Sherman TS, Mendelsohn ME, Anderson RG, Shaul PW. Estrogen receptor alpha and endothelial nitric oxide synthase are organized into a functional signaling module in caveolae. Circ. Res. 87: E44-E52, 2000.
- HK, Jo I. Nongenomic stimulation of nitric oxide release by estrogen is mediated by estrogen receptor alpha localized in caveolae. Biochem. Biophys. Res. Commun. 263: 257-262, 1999.
- 56 Huhtakangas JA, Olivera CJ, Bishop JE, Zanello LP, Norman AW. The vitamin D receptor is present in caveolae-enriched plasma membranes and binds 1 alpha,25(OH)2-vitamin D3 in vivo and in vitro. Mol. Endocrinol. 18: 2660-2671, 2004
- 57 Norman AW, Olivera CJ, Barreto Silva FR, and Bishop JE. A specific binding protein/ receptor for 1alpha,25-dihydroxyvitamin D(3) is present in an intestinal caveolae membrane fraction. Biochem. Biophys. Res. Commun. 298: 414-419, 2002.
- 58 Shaul PW, Smart EJ, Robinson LJ, German Z, Yuhanna IS, Ying Y, Anderson RG, Michel T. Acylation targets emdothelial nitric-oxide synthase to plasmalemmal caveolae. J. Biol. Chem. 271: 6518-6522, 1996.
- 59 Galbiati F, Volonte D, Meani D, Milligan G, Lublin DM, Lisanti MP, Parenti M. The dually acylated NH2-terminal domain of gi1alpha is sufficient to target a green fluorescent protein reporter to caveolin-enriched plasma membrane domains. Palmitoylation of caveolin-1 is required for the recognition of dually acylated g-protein alpha subunits in vivo. J. Biol. Chem. 274: 5843-5850, 1999.
- 60 Shenoy-Scaria AM, Dietzen DJ, Kwong J, Link DC, Lublin DM. Cysteine3 of Src family protein tyrosine kinase determines palmitoylation and localization in caveolae. J. Cell Biol. 126: 353-363, 1994.

- 61 Uittenbogaard A, Smart EJ. Palmitoylation of caveolin-1 is required for cholesterol binding, chaperone complex formation, and rapid transport of cholesterol to caveolae. J. Biol. Chem. 275: 25595–25599, 2000.
- 62 Liang JS, Distler O, Cooper DA, Jamil H, Deckelbaum RJ, Ginsberg HN, Sturley SL. HIV protease inhibitors protect apolipoprotein B from degradation by the proteasome: A potential mechanism for protease inhibitor-induced hyperlipidemia. *Nature Med.* 7: 1327–1331, 2001.
- **63** Wang A, Johnson CA, Jones Y, Ellisman MH, Dennis EA. Subcellular localization and PKC-dependent regulation of the human lysophospholipase A/acyl-protein thioesterase in WISH cells. *Biochim. Biophys. Acta* **1484**: 207–214, 2000.
- 64 Febbraio M, Hajjar DP, Silverstein RL. CD36: a class B scavenger receptor involved in angiogenesis, atherosclerosis, inflammation, and lipid metabolism. J. Clin. Invest. 108: 785–791, 2001.
- 65 Kincer JF, Uittenbogaard A, Dressman J, Guerin TM, Febbraio M, Guo L, Smart EJ. Hypercholesterolemia promotes a CD36dependent and endothelial nitric oxide synthase-mediated vascular dysfunction. *J. Biol. Chem.* 277: 23525–23533, 2002.
- **66** Bastie CC, Hajri T, Drover VA, Grimaldi PA, Abumrad NA. CD36 in myocytes channels fatty acids to a lipase-accessible triglyceride pool that is related to cell lipid and insulin responsiveness. *Diabetes* **53**: 2209–2216, 2004.
- 67 Nozaki S, Tanaka T, Yamashita S, Sohmiya K, Yoshizumi T, Okamoto F, Kitaura Y, Kotake C, Nishida H, Nakata A, Nakagawa T, Matsumoto K, Kameda-Takemura K, Tadokoro S, Kurata Y, Tomiyama Y, Kawamura K, Matsuzawa Y. CD36 mediates longchain fatty acid transport in human myocardium: complete myocardial accumulation defect of radiolabeled long-chain fatty acid analog in subjects with CD36 deficiency. *Mol. Cell. Biochem.* 192: 129–135, 1999.
- 68 Acton S, Rigotti A, Landschulz KT, Xu S, Hobbs HH, Krieger M. Identification of scavenger receptor SR-BI as a high density lipoprotein receptor. *Science* 271: 518–520, 1996.
- 69 Acton SL, Scherer PE, Lodish HF, Krieger M. Expression cloning of SR-BI, a CD36-

related class B scavenger receptor. J. Biol. Chem. 269: 21003–21009, 1994.

- 70 Babitt J, Trigatti B, Rigotti A, Smart EJ, Anderson RG, Xu S, Krieger M. Murine SR-BI, a high density lipoprotein receptor that mediates selective lipid uptake, is Nglycosylated and fatty acylated and colocalizes with plasma membrane caveolae. J. Biol. Chem. 272: 13242–13249, 1997.
- 71 Connelly MA, Klein SM, Azhar S, Abumrad NA, Williams DL. Comparison of class B scavenger receptors, CD36 and scavenger receptor BI (SR-BI), shows that both receptors mediate high density lipoproteincholesteryl ester selective uptake but SR-BI exhibits a unique enhancement of cholesteryl ester uptake. *J. Biol. Chem.* 274: 41–47, 1999.
- 72 Trigatti B, Rigotti A, Krieger M. The role of the high-density lipoprotein receptor SR-BI in cholesterol metabolism. *Curr. Opin. Lipidol.* 11: 123–131, 2000.
- 73 Azhar S, Leers-Sucheta S, Reaven E. Cholesterol uptake in adrenal and gonadal tissues: the SR-BI and 'selective' pathway connection. *Front. Biosci.* 8: 998–1029, 2003.
- 74 Krieger M. Charting the fate of the 'good cholesterol': identification and characterization of the high-density lipoprotein receptor SR-BI. Annu. Rev. Biochem. 68: 523–558, 1999.
- 75 Varban ML, Rinninger F, Wang N, Fairchild-Huntress V, Dunmore JH, Fang Q, Gosselin ML, Dixon KL, Deeds JD, Acton SL, Tall AR, Huszar D. Targeted mutation reveals a central role for SR-BI in hepatic selective uptake of high density lipoprotein cholesterol. *Proc. Natl. Acad. Sci. USA* 95: 4619–4624, 1998.
- 76 Li XA, Guo L, Dressman JL, Asmis R, Smart EJ. A novel ligand-independent apoptotic pathway induced by scavenger receptor class B, type I and suppressed by endothelial nitric-oxide synthase and high density lipoprotein. J. Biol. Chem. 280: 19087–19096, 2005.
- 77 Chikani G, Zhu W, Smart EJ. Lipids: potential regulators of nitric oxide generation. Am. J. Physiol. Endocrinol. Metab. 287: E386-E389, 2004.
- 78 Everson WV, Smart EJ. Influence of caveolin, cholesterol, and lipoproteins on nitric oxide synthase. implications for vascular

194 References

disease. Trends Cardiovasc. Med. 11: 246–250, 2001.

- 79 Lungu AO, Jin ZG, Yamawaki H, Tanimoto T, Wong C, Berk BC. Cyclosporin A inhibits flow-mediated activation of endothelial nitric oxide synthase via altering cholesterol content in caveolae. J. Biol. Chem. e-pub, 2004.
- 80 Smart EJ, deRose RA, Farber SA. Annexin 2-caveolin 1 complex is a target of ezetimibe and regulates intestinal cholesterol transport. *Proc. Natl. Acad. Sci. USA* 101: 3450–3455, 2004.
- 81 Davis HR, Jr., Zhu LJ, Hoos LM, Tetzloff G, Maguire M, Liu J, Yao X, Iyer SP, Lam MH, Lund EG, Detmers PA, Graziano MP, Altmann SW. Niemann-Pick C1 Like 1 (NPC1L1) is the intestinal phytosterol and cholesterol transporter and a key modulator of whole-body cholesterol homeostasis. *J. Biol. Chem.* 279: 33586–33592, 2004.
- 82 Garcia-Calvo M, Lisnock J, Bull HG, Hawes BE, Burnett DA, Braun MP, Crona

JH, Davis HR, Jr., Dean DC, Detmers PA, Graziano MP, Hughes M, Macintyre DE, Ogawa A, O'Neill K A, Iyer SP, Shevell DE, Smith MM, Tang YS, Makarewicz AM, Ujjainwalla F, Altmann SW, Chapman KT, Thornberry NA. The target of ezetimibe is Niemann-Pick C1-Like 1 (NPC1L1). *Proc. Natl. Acad. Sci. USA* 102: 8132–8137, 2005.

- 83 Garver WS, Hossain GS, Winscott MM, and Heidenreich RA. The Npc1 mutation causes an altered expression of caveolin-1, annexin II and protein kinases and phosphorylation of caveolin-1 and annexin II in murine livers. *Biochim. Biophys. Acta* 1453: 193–206, 1999.
- 84 Zhou M, Parr RD, Petrescu AD, Payne HR, Atshaves BP, Kier AB, Ball JM, Schroeder F. Sterol carrier protein-2 directly interacts with caveolin-1 in vitro and in vivo. *Biochemistry* 43: 7288–7306, 2004.

The Roles of Caveolae and Caveolin in Cell Shape, Locomotion, and Stress Fiber Formation

Sang Chul Park and Kyung A. Cho

9.1 Introduction

9

Molecular mechanistic studies of senescent phenotypes such as morphological alteration and deterioration in physiological function have led to the finding that the caveolin molecule might play a prime determinant role in the bimodal nature of the aging process [1,2]. The defective endocytosis system in the aged organism, which is due to an imbalance or dysregulation of the signal transduction system for the entry and relay of signals, was seen to be the basis of the gate theory of aging, with special emphasis being placed on the role of prime modulator molecules to determine the senescent phenotype [3]. Although having been proposed previously as a candidate for a prime modulator molecule [4], caveolin has more recently been suggested as being a possible determinant of the morphological adjustment of senescent cells via modulation of the cytoskeleton, as well as having a role as a functional modulator [5].

Various cellular processes such as differentiation, apoptosis, migration, transformation and cellular senescence require morphological changes with cytoskeletal rearrangement. In this context, a new proposal has been made that caveolins might play a role in morphological determination via actin remodeling through the modulation of polarity and Rho GTPase activity [6,7]. Thus, the activity of caveolin as a modulator of polarity and Rho GTPase activity might provide a novel – albeit somewhat intriguing – approach to study the crosstalk of cellular function and structure relationships.

9.2 Caveolin and Polarity

Animal cells adopt a vast diversity of shapes, ranging from the relatively simplelooking columnar epithelial cell to the highly complex branched structure of a neuron. Shape and migration are highly dependent on the external environmental conditions for the directional cues that drive intracellular polarity. All eukaryotic



Fig. 9.1 Cellular motility. Cells move through the polarized and dynamic re-organization of the actin cytoskeleton, involving a protruding force at the front (blue arrows), combined with a contractile force in the cell body (green double-headed arrows). Caveolin-1 (red) accumulates at the leading edge and involves polarization during cell movement. N = nucleus.

cells are able to polarize in response to cues at the plasma membrane, for example during the budding of yeast, asymmetrical cell division, and mammalian cell differentiation [8].

Caveolin-1 may play an important role in cell motility because it exhibits anterior-posterior polarization during cell migration. Caveolin-1 was found to accumulate at the leading edge of cultured fibroblasts [9] and human umbilical vein smooth muscle cells [10]. Caveolae and lipid rafts also exhibit pronounced polarity during cell migration (Fig. 9.1), with raft-associated ganglioside GM1 being found at the leading edge of human adenocarcinoma MCF-7 cells, when stimulated with insulin growth factor-1 (IGF-1) [11]. Endothelial cells (ECs) exhibit a polarized distribution of caveolin-1 when traversing a filter pore [12]. In these cells, caveolin-1 seems to be released from the caveolar structure in the cell rear and to be relocalized at the cell front; this is in contrast to the situation that occurs during planar movement, when caveolin-1 is concentrated at the rear of endothelial cells, co-localized with caveolae. Phosphorylation of the Tyr14 residue of caveolin-1 is also required for polarization of the protein during transmigration [12]. Therefore, the localization and phosphorylation of caveolin might play a crucial role in the polarity of cellular morphological changes, although the molecular mechanism involved is, as yet, unclear.

9.3 Caveolin and Rho-family GTPases

Rho GTPases participate in the regulation of polarity, microtubule dynamics and cell shape, the latter property being regulated by the Rho GTPases adjusting the organization of the actin cytoskeleton. For example, Cdc42 induces filopodia, Rac induces lamellipodia, and Rho induces focal adhesion and associated stress fibers [13] (Fig. 9.2).

Rho GTPases are a group of molecular switches which control complex cellular processes. They cycle between two conformational states: one state is bound to GTP (the "active" state), and the other state is bound to GDP (the "inactive" state), although both states hydrolyze GTP to GDP. The active GTPases recognize target proteins and generate a response until GTP hydrolysis returns the switch to the GDP state.

In order to drive the processes in precise fashion, the activities of Rho GTPases



morphological changes via the formation of filopodia, lamellipodia, focal adhesion and actin stress fiber. Rho GTPases are recruited into the caveolae, where their activities are regulated by caveolin-1. Caveolin-1 can interact with the integrin α subunit and Fyn, leading to stable focal adhesion formation.

198 9 The Roles of Caveolae and Caveolin in Cell Shape, Locomotion, and Stress Fiber Formation

must be tightly controlled spatiotemporally within the cells. Recent studies have suggested that the function of Rho GTPases may be deeply related with caveolae or lipid rafts. In endothelial cells, sucrose gradient density centrifugation studies have revealed that a significant proportion of RhoA and Cdc42 are localized within caveolae-enriched membrane domains. Moreover, caveolin-1 is directly bound with RhoA but not with Cdc42 [6]. In neonatal rat cardiomyocytes, the initiation and transduction of stretch-induced RhoA and Rac1 activation requires caveolar compartment [7]. However, in unstretched cardiomyocytes RhoA and Rac1 were detected in both the caveolae and noncaveolar fractions. RhoA and Rac1 was activated within 4 minutes by stretching, then inactivated after 15 minutes, and subsequently became dissociated from the caveolae. In addition, treatment with methyl- β -cyclodextrin (m β CD), a caveolae-disrupting agent, inhibits the stretching-induced RhoA and Rac1 activation [7]. Moreover, integrins influence the targeting of Rac and Rho GTPases to the plasma membrane via lipid raft or caveolae and their coupling to downstream effector molecules [14]. In migrating cells, integrins interact with caveolae or lipid rafts to mobilize the activated GTP-bound Rac and Rho to the plasma membrane.

As illustrated in Figure 9.2, caveolin-1 is involved in the regulation of cellular morphological changes via the modulation of Rho GTPases activity. Senescent human diploid fibroblasts (HDFs) show an altered cellular morphology of flattened and enlarged cell shape, in contrast to the small, spindle shape of young cells. In senescent HDFs, the activities of Rac1 and Cdc42 are significantly increased, and overexpression of active Rac1 and Cdc42 in young HDFs resulted in senescence-like morphological changes. We have observed that the active forms of Rac1 and Cdc42 are localized in caveolae and interact directly with caveolin-1 in senescent HDFs [5]. Interestingly, it has been suggested that caveolin-1 is required for filopodia formation, which may enhance the invasive ability of lung adenocarcinoma cells, though the level of caveolin itself is relatively low [15]. These results suggest that caveolin might involve the determination of cell shape and migration through polarity arrangement and the regulation of Rho GTPases activity.

9.4

Caveolin and Focal Adhesion Complex

When cells come into contact with the extracellular matrix (ECM), they usually extend filopodia. Then, integrins located at the tip of filopodia bind to the ECM and initiate the formation of focal adhesions. Actin-rich lamellipodia are then generated as the cell spreads on the ECM [16]. As the integrins bind to the ECM they become clustered in the plane of the cell membrane and associate with the cytoskeletal and signaling complex that promote the assembly of actin filaments. The reorganization of actin filaments into larger stress fibers in turn causes more clustering of integrins, thus enhancing the matrix binding and organization by integrins in a positive feedback system. As a consequence, the ECM proteins, integrins and cytoskeletal proteins assemble into aggregates on each side of the membrane, these aggregates being termed "focal adhesions". Several integrins have been found to associate laterally with caveolin-1, at least in primary cells (see Fig. 9.2) [17,18]. Although the biochemical nature of this interaction is not yet clear, inhibition of caveolin expression suppresses the formation of focal adhesions and actin stress fibers [5,17].

Integrins activate a variety of protein tyrosine kinases, including focal adhesion kinase (FAK), which may be recruited and acquire tyrosine kinase activity by its interaction with integrins. Subsequently, FAK can phosphorylate other cytoskeletal proteins. In addition to the activation of FAK, some β_1 and α_v integrins can activate the tyrosine kinase Fyn and the adapter protein Shc. In this pathway, caveolin-1 appears to function as a membrane adaptor, which couples the integrin α -subunit to Fyn [19]. In senescent HDFs, the inhibition of caveolin-1 expression can disrupt focal adhesion and actin stress fiber, most likely due to the de-phosphorylation of FAK [5]. In NIH-3T3 cells, caveolin-2 undergoes Src-induced phosphorylation on Tyr19, which is localized near to the focal adhesion, and remains associated with the caveolae [20].

9.5 The Dynamics of Caveolin and Actin

Ultrastructural and biochemical analyses have implicated the actin cytoskeleton as having a role in caveolae function [21–24]. Caveolin is also involved in dynamic actin remodeling in endocytosis or molecular trafficking.

In adipocytes, the stimulation of glucose uptake by insulin is achieved via the translocation of intracellular glucose transporter 4 (GLUT4) to the cell-surface membrane. Insulin-induced GLUT4 translocation in adipocytes requires dynamic actin remodeling at the inner surface of the plasma membrane (cortical actin) and in the perinuclear region [25]. In particular, the treatment of adipocytes with the actin-depolymerizing agents cytochalasin D and latrunculin A or B, and the actin-stabilizing agent jasplakinolide, all lead to the inhibition of insulin-stimulated GLUT4 translocation [25–29].

Caveolin-1 mediates the translocation of GLUT4 through direct binding with actin filaments in adipocytes. In differentiated adipocytes, the stress fiber F-actin becomes small patches of punctate actin that are co-localized with the caveolin-positive clusters. In adipocytes, caveolin-actin structures are disrupted by m β CD (which in caveolae causes cholesterol depletion) [30]. Taken together, these results suggest that, in adipocytes, caveolin-1 may play an important role in the translocation of GLUT4 through direct binding with actin.

The F-actin cross-linking protein filamin was identified as being a ligand for caveolin-1 by using a yeast two-hybrid screen [31]. The N-terminus of caveolin-1 binds to both nonmuscle and muscle filamin, indicating that such interaction might not be cell type-specific. Caveolin-1 is present in filamin-positive patches at the plasma membrane, and is co-aligned with actin stress fibers upon Rho stimulation. Filamin plays a role in cell locomotion and mechanoprotection, and also as a
200 9 The Roles of Caveolae and Caveolin in Cell Shape, Locomotion, and Stress Fiber Formation

cytoskeletal linker for transmembrane proteins such as integrins and glycoprotein I complex in platelets. Therefore, the interaction of caveolin-1 with filamin supports one of the possibilities that caveolin-1 is linked to the cortical actin cytoskeleton via filamin.

Caveolin-1, in addition, can indirectly regulate actin remodeling by the activation of Rho family small GTP-binding protein TC10 upon insulin-stimulation in adipocytes. The translocation of GLUT4 requires both phosphatidylinositol (PI) 3-kinase-dependent pathways for activation of Akt [32-34], and a PI3-kinase-independent pathway for the activation of TC10 [35-37]. Recently, both PI3-kinase and TC10 have been reported as being involved in the regulation of actin cytoskeleton rearrangement in various cell types [25,38,39]. TC10 is mainly localized to the caveolae, and subsequently caveolin-1-enriched caveolae are required for the activation of TC10 through a CAP-Cbl signaling pathway [35,36]. Insulin stimulates the phosphorylation of another insulin receptor substrate, the proto-oncogene c-Cbl [40]. Once phosphorylated, Cbl can recruit the SH2-containing adapter protein CrkII to caveolae along with the guanine nucleotide exchange factor C3G. The latter then appears to activate TC10 [35]. Although the action of TC10 remains unknown, a potential clue towards understanding its physiological function comes from the role of Rho proteins in modulation of the actin cytoskeleton. The inhibition of Rho function with toxin B, or expression of the amino-terminal domain of TC10, disrupts the adipocyte actin cytoskeleton and inhibits GLUT4 translocation. Moreover, the dominant-negative mutant of TC10 disrupts caveolae-associated Factin structures in adipocytes [41]. The disruption of caveolae with cholesteroldepleting drugs or by overexpression of the inhibitory forms of caveolin-1 completely blocks TC10 activation, as well as the stimulation of glucose transport by insulin [36]. However, direct evidence for an interaction between TC10 and caveolin-1 remains vague.

9.6

Caveolae-Dependent Endocytosis via Actin Stress Fiber

Recently, caveolae-mediated endocytosis has been implicated in the internalization of certain membrane components, extracellular ligands, bacterial toxins, and several nonenveloped viruses [42,43]. Simian virus 40 (SV40) utilizes endocytosis through caveolae to achieve infectious entry into host cells [44,45]. After binding to the caveolae, the virus particles induce the transient breakdown of actin stress fibers. Actin is then recruited to virus-loaded caveolae as actin patches are responsible sites for actin "tail" formation. Latrunculin A (an actin monomer-sequestering drug) and jasplakinolide (an actin polymer-stabilizing drug) both reduced virus internalization by 60–65% during the later stages of the entry process [46]. This result suggests that actin might function as a scaffold to restrict the lateral mobility of the caveolae. These events are dependent upon the presence of cholesterol and the activation of tyrosine kinases, which phosphorylate proteins in caveolae. SV40 particles induce tyrosine phosphorylation of proteins at virus-loaded caveolae, and

thereby inhibition of tyrosine kinases inhibits the caveolae- mediated internalization of the virus. These findings indicate that the phosphorylation of caveolae proteins is essential for the formation of caveolae-derived endocytic vesicles and for viral infection into the cell [46]. Therefore, it is possible that tyrosine phosphorylation of caveolae-associated proteins might trigger the recruitment of components for F-actin assembly [46,47].

9.7 Summary

The roles of caveolin and caveolae in a variety of cellular processes have been reassessed. In particular, responsibility for the regulation of cellular shape determination, migration and stress fiber formation has been reviewed. The mode of cellular polarity can be partially explained by the recruitment and polarization of intracellular caveolin, which might be responsible in turn for cellular motility (Fig. 9.3). Likewise, the interaction of caveolin with Rho GTPases might affect the



Fig. 9.3 Role of caveolin-1 via actin remodeling. Caveolin-1 may involve in morphological changes or migration via regulation of Rho GTPases, focal adhesion formation or actin remodeling. For molecular trafficking, caveolin can interact with F-actin or filamin. These interactions may play a crucial role in the translocation of GLUT4 in adipocytes or endocytosis of viruses.

202 References

relevant activities for microtubule dynamics, actin remodeling and migration, resulting in cellular morphological changes as well as locomotive activity. Moreover, caveolin's interaction with the focal adhesion kinase complex and actin cytoskeletal structures resulted in the modulation of molecular trafficking for many biologically active molecules. These findings of caveolin function led us to explain, in part, the bimodal nature of changes in structure and function in senescent phenotypes. In this context, the role of caveolin as a determinant of structural change in the senescent cell highlighted the novel modality for the modulation of age-related morphological alteration as well as functional deterioration through the simple adjustment of caveolin status.

In conclusion, it is possible that the status of caveolin – via its interaction with actin cytoskeletal structures, focal adhesion kinase complex and Rho GTPases – may affect a variety of cellular processes such as signaling modulation, viral infectivity, hormone sensitivity, transformability, invasiveness and neuronal capacity, among a broad spectrum of cellular phenomena.

Abbreviations

EC	endothelial cell
ECM	extracellular matrix
FAK	focal adhesion kinase
GLUT4	glucose transporter 4
HDF	human diploid fibroblast
IGF-1	insulin growth factor-1
mβCD	methyl-β-cyclodextrin
PI	phosphatidylinositol
SV40	Simian virus 40

Acknowledgements

This work has been supported by the grants from the Korea Science and Engineering Foundation (RII-2002-097-05001-0 and RII-2002-097-00001-0) and the Korea Research Foundation for Health Science.

References

- 1 W.Y. Park, K.A. Cho, J.S. Park, D.I. Kim, S.C. Park, J. Biol. Chem. 2000, 275(27), 20847–20852.
- 2 K.A. Cho, S.C. Park, Mech. Ageing. Dev. 2005, 126(1), 105–110.
- 3 E. J. Yeo, S. C. Park, Mech. Ageing. Dev. 2002, 123(12), 1563–1578.
- 4 K.A. Cho, S.J. Ryu, J.S. Park, I.S. Jang, J.S. Ahn, K.T. Kim, S.C. Park, J. Biol. Chem. 2003, 278(30), 27789–27795.
- 5 K.A. Cho, S. J. Ryu, Y.S. Oh, J. H. Park, J.W. Lee, H. P. Kim, K.T. Kim, I.S. Jang, S.C Park, J. Biol. Chem. 2004, 279(40), 42270–42278.

- 6 D. Gingras, F. Gauthier, S. Lamy, R. R. Desrosiers, R. Beliveau, *Biochem. Biophys. Res. Commun.* 1998, 247, 888–893.
- **7** S. Kawamura, S. Miyamoto, J. H. Brown, *J. Biol. Chem.* 2003, 278, 31111–31117.
- 8 M. Vicente-Manzanares, F. Sanchez-Madrid, Dev. Immunol. 2000, 7(2–4), 51–65.
- 9 K.G. Rothberg, J.E. Heuser, W.C. Donzell, Y.S. Ying, J.R. Glenney, R.G. Anderson, *Cell* 1992, 68, 673–682.
- 10 S. S. Okada, J. E. Tomaszewski, E. S. Barnathan, *Exp. Cell. Res.* 1995, 217, 180–187.
- 11 S. Manes, E. Mira, C. Gomez-Mouton, R. A. Lacalle, P. Keller, J. P. Labrador, A. C. Martinez, *EMBO J.* 1999, 18, 6211–6220.
- 12 M.O. Parat, B. Anand-Apte, P.L. Fox, *Mol. Biol. Cell.* 2003, 14, 3156–3168.
- 13 A. Hall, Science 1998, 279, 509-514.
- 14 A. F. Palazzo, C. H. Eng, D. D. Schlaepfer, E. E. Marcantonio, G. G. Gundersen, *Science* 2004, 303, 836–839.
- 15 C.C. Ho, P.H. Huang, H.Y. Huang, Y.H. Chen, P.C. Yang, S.M. Hsu, *Am. J. Pathol.* 2002, 161, 1647–1656.
- 16 F.G. Giancotti, E. Ruoslahti, Science 1999, 285(5430), 1028–1032.
- 17 K. K. Wary, F. Mainiero, S. J. Isakoff, E. E. Marcantonio, F. G. Giancotti, *Cell* 1996, 87, 733–743.
- 18 Y. Wei, X. Yang, Q. Liu, J.A. Wilkins, H.A. Chapman, J. Cell. Biol. 1999, 144, 1285–1294.
- 19 K. K. Wary, A. Mariotti, C. Zurzolo, F.G. Giancotti, *Cell* 1998, 94, 625–634.
- 20 H. Lee, D. S. Park, X. B. Wang, P. E. Scherer, P. E. Schwartz, M. P. Lisanti, *J. Biol. Chem.* 2002, 277, 34556–34567.
- 21 P. Rohlich, A.C. Allison, J. Ultrastruct. Res. 1976, 57, 94–103.
- 22 B. van Deurs, K. Nilausen, J. Cell. Biol. 1982, 94, 279–286.
- 23 R.G. Parton, B. Joggerst, K. Simons, J. Cell. Biol. 1994, 127, 1199–1215.
- **24** T. Fujimoto, *Seikagaku* 1995, 67, 1396–1401.
- **25** M. Kanzaki, J. E. Pessin, *J. Biol. Chem.* 2001, 276, 42436–42444.
- 26 T. Tsakiridis, P. Tong, B. Matthews, E. Tsiani, P. J. Bilan, A. Klip, G. P. Downey, *Microsc. Res. Tech.* 1999, 47, 79–92.
- 27 Z. A. Khayat, P. Tong, K. Yaworsky, R. J. Bloch, A. Klip, *J. Cell. Sci.* 2000, 113 (Pt 2), 279–290.

- **28** W. Omata, H. Shibata, L. Li, K. Takata, I. Kojima, *Biochem. J.* 2000, 346 (Pt 2), 321–328.
- 29 P. Tong, Z. A. Khayat, C. Huang, N. Patel, A. Ueyama, A. Klip, J. Clin. Invest. 2001, 108, 371–381.
- 30 M. Kanzaki, M. Furukawa, W. Raab, J. E. Pessin, J. Biol. Chem. 2004, 279, 30622–30633.
- **31** M. Stahlhut, B. van Deurs, *Mol. Biol. Cell.* 2000, 11, 325–337.
- 32 K. Kotani, W. Ogawa, M. Matsumoto, T. Kitamura, H. Sakaue, Y. Hino, K. Miyake, W. Sano, K. Akimoto, S. Ohno, M. Kasuga, *Mol. Cell. Biol.* 1998, 18, 6971– 6982.
- 33 Q. Wang, R. Somwar, P. J. Bilan, Z. Liu, J. Jin, J. R. Woodgett, A. Klip, *Mol. Cell. Biol.* 1999, 19, 4008–4018.
- 34 M. L. Standaert, G. Bandyopadhyay, L. Perez, D. Price, L. Galloway, A. Poklepovic, M. P. Sajan, V. Cenni, A. Sirri, J. Moscat, A. Toker, R. V. Farese, J. Biol. Chem. 1999, 274, 25308–25316.
- 35 S. H. Chiang, C. A. Baumann, M. Kanzaki, D. C. Thurmond, R.T. Watson, C. L. Neudauer, I. G. Macara, J. E. Pessin, A. R. Saltiel, *Nature* 2001, 410, 944–948.
- 36 R. T. Watson, S. Shigematsu, S. H. Chiang, S. Mora, M. Kanzaki, I. G. Macara, A. R. Saltiel, J. E. Pessin, *J. Cell. Biol.* 2001, 154, 829–840.
- 37 A. Baumann, V. Ribon, M. Kanzaki, D. C. Thurmond, S. Mora, S. Shigematsu, P. E. Bickel, J. E. Pessin, A. R. Saltiel, *Nature* 2000, 407, 202–207.
- 38 G.A. Murphy, P.A. Solski, S.A. Jillian, P. Perez de la Ossa, P. D. Eustachio, C. J. Der, M.G. Rush, *Oncogene* 1999, 18, 3831–3845.
- 39 C. Vignal, M. De Toledo, F. Comunale, A. Ladopoulou, C. Gauthier-Rouviere, A. Blangy, P. Fort, J. Biol. Chem. 2000, 275, 36457–36464.
- 40 V. Ribon, A. R. Saltiel, *Biochem. J.* 1997, 324 (Pt 3), 839–845.
- 41 M. Kanzaki, J. E. Pessin, J. Biol. Chem. 2002, 277, 25867–25869.
- 42 B. Razani, M. P. Lisanti, *Exp. Cell. Res.* 2001, 271, 36–44.
- **43** L. Pelkmans, A. Helenius, *Traffic* 2002, 3, 311–320.
- 44 H. A. Anderson, Y. Chen, L. C. Norkin, Mol. Biol. Cell. 1996, 7, 1825–1834.

- 204 References
 - 45 D. Stang, J. Kartenbeck, R.G. Parton, Mol. 47 A.E. Engqvist-Goldstein, D.G. Drubin, Biol. Cell. 1997, 8, 47-57.
 - 46 L. Pelkmans, D. Puntener, A. Helenius, Science 2002, 296, 535-539.
- Annu. Rev. Cell. Dev. Biol. 2003, 19, 287-332.

10 Lipid Rafts in Trafficking and Processing of Prion Protein and Amyloid Precursor Protein

Daniela Sarnataro, Vincenza Campana, and Chiara Zurzolo

10.1 Introduction

Extensive evidence has accumulated in recent years that several human disorders have the same molecular basis, namely, a change in the conformation of a protein. These disorders have been named "conformational diseases" [1], and the most important among them are prion and Alzheimer diseases. Prion diseases have been proposed to be caused by the transconformation of the cellular prion protein PrP^c into a protease-resistant and readily aggregated isoform (PrP^{Sc}) which accumulates in the brain. However, controversy exists regarding exactly where and how in the cell this conversion occurs. Conversely, in Alzheimer's disease (AD) mutations in genes encoding three different proteins, amyloid precursor protein (APP) and presenilins 1 and 2 have been shown to lead to the overproduction and accumulation in the brain of the amyloid peptide A β that converts from a soluble form into amyloid fibrils. During its biogenesis, APP undergoes sequential enzymatic cleavages, and a critical question for understanding the pathogenesis of Alzheimer's disease is where, and at what rate, A β is produced in neurons. In the case of prion diseases, it has been postulated that missorting of APP or altered intracellular trafficking of A β could play a key role in the pathogenesis of AD (2,3).

Lipid rafts, which are small membrane patches of highly ordered saturated lipids and cholesterol, have been proposed to play a role in the biogenesis of both these neurodegenerative disorders because they might represent a preferential site for the formation of the pathological forms of the prion protein and amyloidogenic A β . Thus, for both conformational diseases it appears that the intracellular trafficking of PrP^c and APP, and their association with rafts, have a predominant role in their related pathogenesis. Here, we review the latest findings on these two fundamental aspects.

10.2

Lipid Rafts and Caveolae

10.2.1

Biochemical Properties and Functions

Lipid rafts have been defined as islands of highly ordered saturated lipids and cholesterol that are laterally mobile in the plane of a more disordered fluid bilayer of largely unsaturated lipids [4–6]. Because of their ability to segregate functional proteins, lipid rafts have been proposed to play a central role in many cellular processes, including intracellular signaling and protein and lipid sorting [4,5]. In particular, Simons and Ikonen in 1997 postulated that, in polarized epithelial cells, rafts can act as sorting platforms for inclusion of proteins into apical post-trans-Golgi network (TGN) sorting vesicles. Later studies subsequently provided evidence for the role of lipid rafts in signaling, as in the case of IgE receptor (FcRI) and T-cell antigen receptor (TCR) [7,8].

Association with detergent-resistant membranes (DRM) is a useful criterion to estimate whether a protein associates with lipid rafts [6]. After solubilization of membranes or cells with Triton X-100 at 4 °C, raft-associated proteins and lipids remain insoluble and can then be floated to low density by sucrose density gradient centrifugation. If cholesterol is extracted by using methyl- β -cyclodextrin (m β CD) or is complexed by saponin, the raft proteins usually (but not always) become detergent-soluble [9].

Constitutive raft residents include glycophosphatidylinositol (GPI)-anchored proteins (e. g., the prion protein), double acylated proteins (e. g., tyrosine kinases of the Src family), palmitate-anchored proteins and transmembrane proteins (e. g., β -secretase; BACE) [5].

One source of confusion in the field of rafts has been the inter-relationship between rafts and caveolae. Indeed, for a long time these two terms have been used interchangeably. However, this issue has now been clarified by the analysis of caveolin knockout mice [10,11].

Caveolae appear as "smooth" uncoated-pits of 50- to 100-nm flask-shaped invaginations of the plasma membrane, originally identified by electron microscopy in a wide variety of tissues and cell types [12,13]. They represent a morphologically identifiable subset of lipid rafts identified by the coat protein, caveolin. Whilst the biochemical composition of lipid rafts and caveolae is thought to overlap, these microdomains are not equivalent [14].

Caveolar invagination is possibly driven by the polymerization of caveolins, of which there are three types: caveolin-1, -2, and -3. Caveolin-1 appears to have a central role in the formation of caveolae, because it was shown that cells without caveolin-1 lacked morphological caveolae, and that reintroduction of the protein was sufficient to generate caveolae [15,16].

Caveolae usually remain attached to the cell surface, but their internalization can be stimulated under certain conditions; for example, by Simian virus-40 (SV40) [17] or by treatment with the phosphatase inhibitor okadaic acid [18,19]. Both caveolae and rafts mediate the internalization of sphingolipids and sphingolipid-binding toxins, GPI-anchored proteins [20], and the autocrine motility factor (AMF).

Internalizations via caveolae or via lipid rafts are fundamentally similar processes, defined by their clathrin independence and sensitivity to cholesterol depletion. However, the cholesterol-dependent invagination of rafts occurs independently of caveolin-1 and of dynamin 2, a GTPase, localized at the neck of the caveolae [21] which regulates their internalization. Interestingly, in some cases caveolin-1 acts as a negative regulator of the budding of caveolar invaginations but caveolae become competent for endocytosis after specific signalling stimuli [22].

Caveolae/raft dysfunction has been implicated recently in the pathogenesis of different human diseases. Several groups of pathogens, including bacteria, prions, viruses, and parasites appear to hijack lipid rafts during internalization [23–25]. In this chapter, we illustrate the proposed role of lipid rafts in the trafficking and processing of PrP^c and APP and in the pathogenesis of their related diseases.

10.3 PrP^c and Prion Diseases

Transmissible spongiform encephalopathies (also known as prion diseases) have become an interesting example of how lipid rafts are involved in regulating protein trafficking and processing (for a review, see [26]). Prion diseases include a large group of neurodegenerative disorders including scrapie in sheep and goats, bovine spongiform encephalopathies, chronic wasting disease in deer and elk, and Creutzfeldt-Jakob disease (CJD) in humans [27–29]. These pathologies are caused by an agent consisting of a protein which is usually referred to as a prion ("proteinaceous infectious only") [27]. The prion diseases are often (but not always) characterized by the cerebral accumulation of a protease-resistant, misfolded isoform of the prion protein (PrP), the so-called PrP^{Sc} (for scrapie PrP), which is derived from the normal isoform of the cellular glycoprotein PrP^c (for cellular PrP) [27,30]. Compared to PrP^c, the PrP^{Sc} form has an increased content of β -sheet structures and is partially resistant to proteinase K digestion; therefore, it aggregates and accumulates in the brain [27]. Thus, this conformational change of PrP has been proposed to be the cause of the disease.

Prion diseases can be of infectious, sporadic and genetic origin. In the case of infectious diseases, a direct interaction between the pathogenic PrP^{Sc} template and the endogenous PrP^c substrate is proposed to drive the formation of nascent infectious misfolded prions [27]. In contrast, in the genetic forms of the diseases, mutations present in the gene of PrP^c, *Prpn* [27,31], can directly and spontaneously induce alterations in PrP^c conformation. The exact mechanism by which PrP^c is converted into PrP^{Sc} is not yet known, but several lines of evidence suggest that lipid rafts are involved [32,33].

PrP^c is a cell-surface GPI-anchored protein which is normally expressed in neurons and in a range of non-neuronal tissues, as well as leukocytes [34]. Human PrP

208 10 Lipid Rafts in Trafficking and Processing of Prion Protein and Amyloid Precursor Protein

is the product of a single gene which directs the synthesis of a protein containing 253 amino acids, with five octapeptide repeats near the N-terminus, two glycosylation sites and one disulfide bridge. The GPI-anchor, which attaches the protein to the outer surface of the cell membrane, allows its association with rafts.

The role of PrP^c remains unknown. In 1992, Weissmann's group [35] generated PrP^c-null mice and reported that PrP^c expression is dispensable for normal development and behavior. However, its conservation in different species infers some relevance in basic physiological processes. PrP^c has been proposed to be involved in several functions:

- copper/zinc transport or metabolism [36-38];
- protection from oxidative stress [38];
- cellular signalling [39,40];
- membrane excitability and synaptic transmission [41,42];
- neuritogenesis [43]; and
- apoptosis [44].

Strong evidence for PrP^c function derives from its interaction with different partners, ligands and/or effectors, such as laminin, BIP, GFAP, and Bcl-2 (for a review, see [45]), each of which can be relevant for the different proposed function of PrP^c.

Several studies support the possibility that PrP^c misfolding and/or misfunction correlate with defects in its trafficking and with its raft association. In particular, the intracellular localization of some pathological mutants of PrP^c, which cause inherited prion diseases, has been shown to be altered [31]. Therefore, in order to understand the mechanism of the disease it is important to understand the relationships between raft association, intracellular trafficking, proper protein folding, and the function of PrP^c.

10.3.1

The Site of PrP^c Conversion: The Role of Rafts in the Different Intracellular Compartments

A major unresolved question is where PrP^c-PrP^{Sc} transconformation occurs. Both PrP^c and PrP^{Sc} have been localized at the plasma membrane and have been shown to undergo endocytosis [46,47], so it is likely that the plasma membrane and/or the endolysosomal compartment participate in PrP^{Sc} formation. In addition, in the case of inherited diseases, pulse-chase experiments indicate that the pathological conversion of mutant PrP to the PrP^{Sc}-like conformer proceeds in a stepwise manner, via a series of identifiable biochemical intermediates, and that one of the earliest properties of PrP^{Sc} (i.e., resistance to cleavage by phosphatidylinositol phospholipase C) is acquired in the endoplasmic reticulum (ER) [31]. Moreover, the ER could also be implicated in the degradation of misfolded PrP mutants via the ER-associated degradation (ERAD) pathway [48–50].

Hence, the available data indicate that in the case of genetic prion diseases originating from PrP mutants (for a review, see [31]), the ER might be directly involved in protein transconformation and consequent PrP^{Sc} formation. Conversely, because in infected cells the stimulation of PrP^c retrograde transport results in an increase in the PrP^{Sc} form [51], we recently proposed [26] that in infectious diseases the ER could represent an amplification compartment for PrP^{Sc} produced earlier at other subcellular sites.

Detergent-resistant microdomains in the ER (ER-rafts) could also be involved in the folding of PrP^c [52]. Like neuronal cell lines, epithelial cells synthesize four different isoforms of PrP^c corresponding to the unglycosylated, monoglycosylated, immature diglycosylated, and mature diglycosylated isoforms. Interestingly, it was found that the immature diglycosylated precursor form of PrP^c was recovered in a DRM fraction in the ER. Using cholesterol- and sphingolipid-depleting drugs in concomitance with pulse-chase experiments, we were able to show that this earlier raft association was crucial for the correct folding of PrP. Indeed, a portion of the protein became misfolded in the ER when cholesterol was extracted [52].

The mechanism that underlies the specificity of cholesterol depletion for PrP^c misfolding could be explained in two ways:

- either cholesterol depletion could perturb the formation of specific DRMs present in the ER, leading to an interference of the association of the immature PrP^c isoform with rafts; or
- cholesterol itself could be directly involved in the folding of PrP^c by acting as a lipochaperone in the ER [53,54].

In contrast with the proposed role for the ER in prion conversion, the results of several studies have indicated that the conversion of PrP^c to a protease- and phospholipase-resistant state is an event that occurs after the protein has reached the cell surface [47,55,56]. Hence, the release of PrP^c from the cell surface by different methods [47,55,57] and exposure of PrP^c to different antibodies [57] prevents PrP^{sc} formation in infected cells.

Therefore, the current evidence suggests that both the ER and the plasma membrane might be important sites for prion formation, and that they are differently involved in the genetic and infectious diseases. Whilst in the case of genetic prion diseases, the ER could be involved in the generation of PrP^{Sc} or PrP^{Sc}-like conformers from PrP mutants [31], in the case of infectious diseases, the first contact between physiological and pathological PrP isoforms could occur at the plasma membrane, although the transconformation might occur later after internalization.

After its transport to the plasma membrane, PrP^c is constitutively internalized and recycles back to the surface [31,46]. The endocytic recycling pathway is of interest from the standpoint of prion generation, since there is evidence that initial steps in the conversion of PrP^c into PrP^{Sc} may take place on the plasma membrane or following the internalization of PrP^c [47,51,55]. The route and mechanism of internalization of PrP^c are controversial because both caveolae and clathrin-coated pits have been shown to be involved (for a review, see [31,46]). Clathrin-coated pits appear to be primarily responsible for endocytic uptake of PrP^c. This conclusion is based on immunogold localization of PrP^c in these organelles by electron microscopy [58,59] and inhibition of PrP^c internalization by incubating cells in hypertonic sucrose, which disrupts clathrin lattices [58].

Because PrP^c lacks a cytoplasmic tail that could interact directly with adaptor proteins and clathrin [60], several candidate proteins have been proposed to be PrP-interacting partners mediating its internalization via the clathrin pathway. Specifically, a basic amino acid motif found in the N-terminal region of PrP^c [31,60,61] has been shown to be essential for coated pit localization and internalization. In contrast to these studies, the presence of PrP^c in caveolar-like domains (CLDs) has also been extensively reported [62,63]. In Chinese hamster ovary (CHO) cells, which express caveolin-1, PrP^c is enriched in caveolae both at the TGN and at the plasma membrane and in interconnecting chains of endocytic caveolae, but it is apparently absent in clathrin-coated pits and vesicles [64].

The initial recruitment of PrP^c to pre-endocytic membranes may therefore be a complex event which occurs by more than one mechanism. It is possible that PrP^c is internalized by default by clathrin-coated vesicles, and that caveolae or CLDs provide alternative internalization pathways occurring in particular cells or conditions.

These different mechanisms may provide a range of possibilities for protein conversion and pathological spreading. Indeed, both CLDs and clathrin-coated pits have been suggested to be involved in PrP^c to PrP^{Sc} transconformation [60,63], but until the internalization pathway of PrP^c is clarified it will be difficult to establish the involvement of one or the other pathway in transconformation.

The fact that most PrP^c molecules reside in raft domains does not argue against an association of the protein with coated-pits, because only a small fraction of the PrP^c molecules are undergoing endocytosis at any one time, and this fraction has probably left the raft domain to enter the coated pits. In this context, it has been demonstrated [60] that PrP^c, prior to endocytosis, leaves the detergent-insoluble raft environment to cluster, along with TfR (the prototypical transmembrane protein endocytosed by coated pits and excluded by lipid rafts) [65] in non-raft membranes. Thus, PrP^c on the cell surface rapidly traffics through two very different membrane environments, probably with different consequences for its conformational stability.

10.3.2 Role of Lipid Rafts in PrP^{sc} Formation

Like other pathogens, infectious prions might use rafts to enter the cells and possibly to initiate and propagate the PrP^c to PrP^{Sc} conversion process that allows prion amplification [56,66–68]. Several lines of evidences suggest that rafts are candidate sites for the generation of PrP^{Sc} in infected cells:

- Both PrP^c and PrP^{Sc} are recovered in lipid rafts [68].
- Impairment of raft association by drugs that reduce intracellular levels of cholesterol decrease PrP^{Sc} formation in infected cells [33]; similarly, removal of PrP^c from rafts by exchanging its GPI anchor with a transmembrane domain prevents PrP^{Sc} formation [63].

• Infectious prion rods were found to contain two sphingolipids, GalCer and sphingomyelin [69], which are enriched in rafts, suggesting that selected raft lipids might interact with normal and/or pathogenic prion proteins. In particular, PrP^c and PrP^{Sc}-associated rafts appear to have different characteristics, being separated from each other by solubilization and flotation on density gradients [68].

These data suggest either the presence of different kinds of rafts or different characteristics of membrane association of the wild-type and pathological isoforms with the same raft-type. The exact mechanisms by which rafts could control the transconformation of PrP^c to PrP^{Sc} remains unknown, although different scenarios can be envisaged [26]:

- Rafts could function as a platform by which PrP^c is transported to a specific compartment where the conformational change into PrP^{Sc} occurs (Fig. 10.1A). We have shown previously that in transfected cells, cholesterol- and sphingolipid-depleting drugs which impair PrP^c raft association do not impair its exocytic trafficking to the plasma membrane [70], but rather slow down PrP^c endocytic trafficking (D. Sarnataro, personal communication), suggesting that rafts could represent the cellular environment that regulates PrP^{Sc} biosynthesis during endocytosis.
- Rafts might represent a container of indispensable machinery required for PrP^{Sc} formation, such as proteins/lipids, that facilitate the conversion process (Fig. 10.1B).
- 3. Rafts could represent a favorable environment for transconformation by facilitating a close encounter between the substrate (PrP^c) and the seed (PrP^{Sc}), and could act by concentrating PrP^c and PrP^{Sc} molecules within confined regions of the plasma membrane (Fig. 10.1C). The role of rafts as a "meeting place" between PrP^c and PrP^{Sc} is supported by the studies of Baron et al., who have shown that PrP^c to PrP^{Sc} transconformation occurs only when the two protagonists of the reaction are inserted into contiguous membranes [32].
- 4. Rafts might allow a direct interaction of sphingolipids and/or cholesterol with PrP^c and/or with PrP^{Sc}, which could affect the conformational stability of PrP^C (Fig. 10.1D). In this case, it is possible that different raft-resident lipids could act as molecular chaperones to facilitate the unfolding of one or more α -helices or the refolding into β -sheets of PrP^c. Alternatively, a change in the local environment (in terms of enrichment in specific lipids and proteins) could mediate this process.

10.3.3

Mechanism of Raft Action in Prion Conversion

It has been shown that direct cell-to-cell contact between infected cells and uninfected cells (i.e., with PrP^c and PrP^{Sc} being on opposite cell surfaces [71]), can efficiently induce the passage of prion infection between cells. The importance of this membrane environment in the conversion reaction has been underscored by 212 10 Lipid Rafts in Trafficking and Processing of Prion Protein and Amyloid Precursor Protein



Fig. 10.1 Different models for the role of rafts in prion formation, based on the apparent implication of rafts in many features of prion protein trafficking. (A) Rafts may represent the vehicle of prion transport to the intracellular compartment (e.g., endoplasmic reticulum (ER), plasma membrane, endosomes-lysosomes, caveolae) where the conformational changes could occur. (B) Rafts may embody the cofactors (protein-X or lipid chaperones) of the transconformation machinery. (C) Rafts may represent: (1) the membrane location where PrP^c accumulates and where it encounters PrP^{sc} ; thus, the prion conversion reaction would be enhanced in these microdomains. Alternatively (2), rafts may represent a membrane environment accumulating both PrP^{c} and PrP^{Sc} ; coalescence of these two specific prion rafts could favor and initiate prion conversion. (D) PrP^{c} conformation could be favored and stabilized by association with specific raft domains, so that when PrP^{c} exits them it is misfolded and might better interact with PrP^{Sc} and mediate transconformation.

several studies where specific lipids have been shown to play direct roles as chaperones in protein folding [72,73]. In particular, both PrP^c and PrP^{Sc} can bind to raftlike membranes enriched in cholesterol and sphingolipids [74]. In order to analyze the role of lipid rafts in this conformational transition, binding of both the α helical-enriched structure (α -PrP) and the β -sheet-enriched form (β -PrP) to model lipid membranes was recently investigated [75]. The result of these studies indicate that binding to raft membranes results in a stabilization of α -helical structures,



in lipid rafts. PrP^c associates with specific rafts early during its biosynthesis. Perturbation of this membrane environment promotes PrP^c misfolding. Misfolded PrP, in the presence of PrP^{Sc} or of pathological mutants, can acquire the pathological conformation. Once formed, PrP^{Sc} can associate with new specific types of raft in which aggregation is favored.

while interaction with negatively charged lipid (non-raft) membranes increases β -sheet content [74].

This evidence for a "protective role" of rafts in the transconformation process is further reinforced by the fact that, in a non-cellular assay, PrP^c within isolated lipid rafts is highly resistant to conversion to PrP^{sc} [32], and inhibition of sphingolipid synthesis in cells increases the rate of conversion of PrP^c [76].

Our own data on the destabilization of PrP^c folding in cholesterol depletion also support the hypothesis that the raft environment is necessary to stabilize the proper PrP^c conformation, therefore suggesting that transconformation occurs outside of the rafts, where PrP^c folding is destabilized and where misfolded PrPintermediates might be more prone to interact with PrP^{Sc} and to transconform (Fig. 10.2). However, the fact that recombinant β -PrP also has a high affinity for raft-like membranes [75] and that PrP^{Sc} is also found enriched in rafts [68] leads us to propose that, together with the protective role in prion transconformation, rafts could have a second role in promoting aggregation of PrP^{Sc} . In this scenario, PrP^{Sc} would form outside rafts in a non-protective environment but, once formed, PrP^{Sc}

214 10 Lipid Rafts in Trafficking and Processing of Prion Protein and Amyloid Precursor Protein

would be able to reassociate with the rafts (perhaps different to the rafts where PrP^c is found), and this would favor its aggregation (Fig. 10.2).

This hypothesis is also supported by the data of Fantini et al. [77], who proposed that PrP^c can maintain a non-pathological conformation by interacting with lipid rafts through a sphingolipid-binding domain (V3-like domain). Interestingly, in the E200K PrP mutant, which undergoes PrP transconformation in familial CJD, this mutation abrogates sphingomyelin recognition. A similar sphingolipid-binding motif has also been identified in gp120 glycoprotein of HIV and in the β -amyloid peptide in Alzheimer's disease, suggesting a role of lipid rafts in the pathogenesis of these different diseases.

10.3.4

Role of Rafts in Proteolytic Attack on PrP^c

PrP^c is a complex multi-domain protein that contains potential toxic sequences (PrP^c 106–126) [78]. It has been shown that the PrP^c 106–126 peptide could trigger neurotoxicity and lead to an apoptotic response by insertion into membranes [79]. Like APP, PrP^c undergoes protein kinase C (PKC)-regulated proteolysis by identical proteases of the disintegrin family, which also inactivate the peptide.

Indeed, the normal cleavage of PrP^c, directed by ADAM10 (a disintegrin and metalloprotease) occurs inside the 106–126 "toxic" core of the protein, leading to the formation of a 11- to 12-kDa fragment referred to as N1. This processing also has a PKC-regulated counterpart regulated by the ADAM17. Interestingly, this cleavage site is inaccessible in PrP^{Sc}, where the toxic core of the protein remains uncleaved and might have a function in the production of the PrP^{Sc} aggregates [80].

PrP^c is the first example of a GPI-anchored protein that is a substrate of ADAM10 and 17, which are the β-secretase enzymes mediating APP cleavage [81,82]. The participation of these enzymes in the processing of PrP^c is quite unexpected because ADAM proteases are ectoenzymes (i.e., they have transmembrane activities), and substrates of disintegrins are generally transmembrane proteins (e.g., βAPP). Nevertheless, cleavage of both PrP^c and APP by ADAM10 and 17 takes place inside the toxic domains of these two proteins. Thus, this proteolytic attack could be viewed as an inactivating mechanism since it prevents the proteinaceous accumulation of Aβ and prion proteins often detected in affected brains.

To our knowledge, it has not yet been demonstrated whether the proteolytic attack on PrP^c occurs in lipid rafts, as has been postulated for APP (see below). However, this is an interesting possibility because in human embryonic kidney (HEK) cells a minor fraction of the ADAM10 immature proform was raft-associated [83] and another member of the metalloprotease family, ADAM19, has been found to be active in lipid rafts [84].

10.4 Alzheimer's Disease: The Role of Rafts in APP Trafficking and Processing

10.4.1 The "History" of APP Cleavage

The amyloid β -peptide (A β) is one of the hallmarks of AD. A β derives from a transmembrane protein, the amyloid precursor protein (APP) [85], through two sequential cleavages. APP is first cleaved by β -secretase BACE 1 (the β -site APP cleavage enzyme), in its luminal domain to generate a secreted ectodomain (sAPP β) and a membrane-bound APP carboxyl terminal fragment (CTF β). This 10-kDa C-terminal stub of APP is subsequently the substrate for β -secretase or presenilin 1 [86], which cleaves the transmembrane domain of APP to release two A β peptides of 40 and 42 amino acids [85].

In an alternative non pathogenic pathway, APP is cleaved within the A β sequence by α -secretase (belonging to the ADAM family of metalloproteases), which generates other secreted derivates known as sAPP α and APP CTF α . Like CTF β , CTF α can be cleaved by γ -secretase yielding CTF γ and a fragment designated p3 [87]. Since α -cleavage cuts APP within the A β region it prevents A β formation and, as α and β -cleavage compete for their substrate APP, it is crucial to understand the mechanism which regulates and controls the access of these enzymes to APP.

10.4.2

Intracellular Compartments and Aß Generation: Involvement of Lipid Rafts

The broad intracellular distribution of the key proteins involved in A β generation has made it difficult to identify the intracellular site where this event occurs. APP is distributed predominantly in the TGN, but significant amounts of the protein are found at the cell surface and within endosomes [88]. BACE has been placed within the early endosomes and/or throughout the endosomal-lysosomal system [88,89], although a predominant TGN localization has also been reported [90,91]. Finally, recent evidence suggests that presenilins are located primarily not only in the ER and Golgi apparatus but also on the plasma membrane and in the endosomes [90–92]. In addition to these different localizations, many reports indicate that APP, β -secretase and presenilin 1 all reside in rafts [86,93–95]. These findings led to the proposal that lipid rafts may be a site for the proteolytic processing of APP [96,97] (Fig. 10.3A). However, the observation that APP is present in DRMs from brain tissue at levels no higher than several other non-DRM proteins questions the involvement of lipid rafts in APP proteolytic processing.

The conflicting reports about the presence of APP in DRMs could be explained by reasoning that APP in DRMs is rapidly processed such that the steady-state amount of raft-associated protein is lower than it would otherwise be expected. However, it is possible that APP cleavage occurs outside rafts but that, after its generation, the A β peptide rapidly translocates to the DRM after it is cleaved from APP because it has a high affinity for cholesterol [98] and ganglioside GM1



Fig. 10.3 Two different models for β -secretase processing of APP. (A) In neuronal APP-overexpressing cells, two cellular pools of both APP and BACE1 are present at the plasma membrane: one raft-associated and another outside rafts. Rafts are small and highly dispersed at the cell surface, and because they contain only a few proteins it is likely that APP and BACE1 are localized in separate rafts at the plasma membrane. Endocytosis is an essential step for APP- and BACE1-rafts to cluster and allow β -cleavage. In this model strong cholesterol depletion alters APP and BACE1 raft-association causing inhibition of A β formation. (B) In an alternative model, the β -cleavage of APP should occur outside lipid rafts, as APP is mainly distributed in non-raft domains (a). In this model, the amyloid- β is physiologically produced, but under mild cholesterol reduction (b) raft disorganization occurs and this process facilitates the close contact between APP and BACE1, leading to β -cleavage and a higher production of A β . Under strong cholesterol depletion (c), although BACE1 can more easily encounter APP, its activity is repressed because of raft disruption, and consequently A β production is inhibited. [99,100], both of which are enriched in DRM [62,101] (Fig. 10.3B). Although depleting cells of cholesterol with m β CD has been shown to decrease the production of A β [102], this effect may not be entirely due to the disruption of lipid rafts, because such treatment also disrupts clathrin-coated pits [103] in which APP has been localized [104] (see below).

Rafts could also be involved in the formation of amyloid fibrils, that are one of the pathological hallmarks of AD [105]. The molecular mechanism of amyloid fibril formation involves a major conformational transition of A β , from α -helix to β sheet [3], thus resembling the PrP^c-PrP^{Sc} transconformation. In the case of AD, the conformational change required for the conversion of soluble peptide into amyloid fibrils is modulated by pH, A β concentration and alteration in the primary sequence of A β . In addition, two raft lipids (GM1 and cholesterol) bind to A β and promote fibril formation [86,106]. Furthermore, synthetic lipid vesicles from bovine brain containing gangliosides such as GM1 bind to A β , inducing an increased amount of α -helix at pH 7 and β -sheet at pH 6 [107]. It has also been demonstrated that a conformationally altered form of A β , which acts as a "seed" for amyloid fibril formation, is generated in cholesterol-rich microdomains [108]. These findings support the view that lipid rafts could participate in the generation of A β , and that raft lipids could modulate its conformation.

Moreover, since a sphingolipid-binding domain similar to the V3-like domain of PrP has been identified in A β , APP and prion proteins might interact with lipid rafts by a common mechanism [109]. The molecular model proposed in Figure 10.1 to explain the role of lipid rafts in the PrP^c to PrP^{Sc} conversion might also apply for A β : in particular, evidence from biochemical, genetic and *in-vivo* studies has indicated that apolipoprotein E (apoE) seems to act as a pathological chaperone in AD amyloidogenesis, promoting fibril formation by inducing or stabilizing β -sheet conformation [3].

10.4.3

The Role of Rafts in β-Secretase Activity

A fraction of APP and BACE1 were shown to be associated with a raft population distinct from caveolae in a cholesterol-dependent manner. In particular, BACE1 activity decreases in cholesterol-depleted cells [93,95,102], while α -secretase activity increases after inhibition of β -secretase under these conditions. The decreased β -secretase activity after cholesterol depletion could be a result of the reduced amount of the substrate available for β -secretase cleavage. On the other hand, the increase in α -cleavage activity may derive from the fact that APP is present in two cellular pools, in equilibrium with each other – one that is raft-associated and leads to A β generation, and another that is outside rafts where the major amount of α -cleavage could take place. Cholesterol depletion would shift the partitioning of APP from the lipid rafts to the surrounding lipid bilayer, where it would become more accessible to α -cleavage (see below).

Interestingly, β -cleavage does not appear to be induced by partitioning of APP and BACE1 into rafts alone, but to be dependent on endocytosis [95]. However, the

218 10 Lipid Rafts in Trafficking and Processing of Prion Protein and Amyloid Precursor Protein



Fig. 10.4 APP cleaving pathways and cellular compartments involved in its processing. APP is a large type I transmembrane protein which is cleaved out sequentially by α -, β and γ -secretase enzymes. Once synthesized in the ER, APP can follow distinct destinations in the cells. It can directly reach the plasma membrane ① from which it could be internalized via clathrin-coated pits 2 or caveolae 3. After internalization from the plasma membrane, APP can be rapidly recycled back to the cell surface ④ or reach the lysosomes for degradation (5). Plasma mem*brane*: APP can be processed by α - (a), β - (b) and γ -secretase (c) enzymes on the plasma membrane. Cell-surface lipid rafts seem to be involved in the α - and β -cleavage of APP, while γ -secretase activity seems to operate

outside plasma membrane rafts (see text for discussion and Fig. 10.3A and B). *Caveolae*: APP has been localized in caveolae together with the CTF α fragment. Caveolae at the plasma membrane, intracellular form of caveolae or an unidentified intracellular compartment (UIC) could be the site for α -processing of APP.

Early/recycling endosomes: the early and/or recycling endosomes could represent the intracellular sites for both β - and γ -secretase cleavage of APP. Lipid rafts might be involved in these processing events. Trans Golgi Network: β - and γ -secretase enzymes are both present in this organelle. The TGN could represent the intracellular site for their activity on APP. The possible involvement of lipid rafts is discussed in the text.

exact cellular site of β -cleavage is not totally clear as it may occur during or after delivery to the cell surface, and/or during endocytosis [104,110–112] (Fig. 10.4).

In pulse-chase experiments, perturbation of clathrin-dependent and -independent endocytosis led to a decreased secretion of newly generated A β , while the α - cleavage was not affected [113,114]; this suggested that "most but not all" of the β cleavage occurs after internalization. Furthermore, since cholesterol depletion is also known to decrease the rate of endocytosis [103], this also likely contributes to the decreased β -cleavage. The inhibition of β -cleavage by cholesterol depletion suggests that the processing of APP by BACE1 might critically depend on the lipid raft environment. Ehehalt et al. [95] proposed that rafts would have to cluster together to bring APP and BACE1 into the same raft environment in order for β -cleavage to occur. Hence, APP and BACE1 would meet after endocytosis by coalescence of BACE1- or APP-containing rafts within endosomes (see Fig. 10.3A). This hypothesis is supported by the fact that the induction of raft clustering at the cell surface with antibody cross-linking would allow β -cleavage to occur in patches containing both APP and BACE1 at the plasma membrane, and this could increase A β formation (Fig. 10.3) [95].

In contrast with these findings, Chyung and Selkoe [115] reported that when dynamin function was inhibited by the K44A mutant in HELA cells, the α -secretase cleavage products of APP increased in the membrane, as well as the *total* amount of secreted A β . This could be due to the fact that α -cleavage occurs on the cell surface [116,117].

To investigate further the significance of lipid rafts in APP β -processing, a GPI anchor has been added to BACE, replacing the transmembrane and C-terminal domains in order to target the enzyme exclusively to lipid rafts [118]. Expression of GPI-BACE substantially up-regulates the secretion of both sAPP β and A β over the levels observed in cells overexpressing wild-type BACE. Furthermore, when lipid rafts were disrupted by depleting cellular cholesterol levels this effect was reversed. These results suggest that the processing of APP to the A β occurs mainly in lipid rafts, and that cholesterol levels are critically involved in regulating the access of α - and β -secretase to APP.

In another system (and with contrasting results) it has been reported that in neurons, APP does not seem to have the capacity to be incorporated into cholesterol-rich environments, even when overexpressed [119] (Fig. 10.3B, panel a). Furthermore, treatment leading to moderate reductions of neuronal cholesterol (Fig. 10.3B, panel b) in hippocampal membranes in culture resulted in increased APP β -cleavage, which is inconsistent with the occurrence of BACE1 cleavage of APP in DRMs. Conversely, a mild reduction of membrane cholesterol resulted in more BACE1 in the soluble fractions, higher BACE1-APP co-localization, and enhanced β -processing, whereas a strong cholesterol reduction (Fig. 10.3B, panel c) resulted in a significant fall in A β generation, as observed previously [95,102] (see also figure legend for description).

The discrepancy of these results with previous findings [95] could be explained by postulating that the latter data were obtained in cells that overexpressed APP and thus contained larger amounts of the protein in DRMs. Alternatively, in the overexpressing cells, more cleavage may have occurred as a result of excess APP in non-DRM domains, as suggested by Dotti et al. [119]. Consequently, further studies are required to understand the role of rafts in APP β -cleavage and the mechanism regulating the trafficking and clustering of APP and BACE1.

10.4.4

The Role of Caveolae/Lipid Rafts in α-Secretase Activity

Several reports support the view that α -secretase-mediated cleavage of APP occurs on the cell surface [116,117,120–122]. Furthermore, a recent report showed that in non-neuronal cells APP is enriched within caveolae and is physically associated by its cytoplasmic domain with caveolin-1 [97]. The C-terminal fragment resulting from APP processing by α-secretase is also localized within caveolae-enriched fractions. Importantly, in AD caveolae dysfunction may cause reduced activity of asecretase and accumulation of toxic AB, and caveolin depletion by antisense oligonucleotides prevented α -cleavage. On the other hand, caveolin overexpression increased the α -secretase-mediated proteolysis of APP [97], strongly suggesting that α -cleavage could occur in caveolae (Fig. 10.4). However, proteinase inhibitors added to the cell surface had no effect on APP cleavage, indicating that the bulk of the processing takes place intracellularly [123-125]. Therefore, a second mechanism should exist that involves an intracellular compartment which may be independent of the plasma membrane caveolae. Alternatively, this phenomenon could be explained by the presence of an intracellular form of caveolae - that is, plasmalemmal vesicles. Thus, the α -secretase processing of APP could be regulated by the cycle of caveolae internalization and recycling. Because α -secretase cleavage occurs at both Leu17 and Val18 in the amyloid peptide [126], it is also possible to speculate on the existence of multiple α -secretases which cleave APP at distinct α -sites.

APP is known to transit through clathrin-coated pits and vesicles on its way to endosomes and lysosomes [127]. It is therefore conceivable that a pool of APP is localized in caveolae where α -secretase processing takes place, whereas the remaining intact APP may be cleared from the cell surface via clathrin-coated pits and targeted to endosomes and lysosomes for proteolysis (Fig. 10.4).

Furthermore, Kojro et al. [83] found that small amounts of the ADAM10 (with α secretase activity on APP) immature proform were associated with rafts in human embryonic kidney (HEK) cells and that cholesterol depletion by m β CD increased α -secretase activity. Similarly, filipin treatment, which causes the destruction of caveolar structures, also led to a substantial increase in α -secretase activity.

These data indicate that only a "minor part" of APP could be cleaved by the α secretase within lipid rafts or caveolae microdomains. Moreover, fluorescence anisotropy studies and biochemical assays [83] indicate that increased membrane fluidity and impaired APP internalization are responsible for the increased α -secretase activity after acute cholesterol depletion by treatment with m β CD. Specifically, increased membrane fluidity could increase the lateral movement of APP and the α -secretase activity within the membrane.

Ledesma et al. [128] have shown that plasmin (a serine protease), which is present exclusively in lipid rafts of hippocampal neurons in culture, participates in APP α -processing directly or through the activation of other proteases (i. e., ADAM 10 [129]). Reduced brain plasmin could be one cause of amyloid plaque formation, since first, plasmin levels are low in brains affected by AD and some aged humans, and second, activation of plasmin increases the α -processing of APP and decreases the levels of A β peptide [128]. Thus, the formation of amyloid plaques during senescence can be a consequence of a natural decrease in levels and/or activity of plasmin-mediated α -cleavage of APP. Genetic predisposition and environmental factors would determine who suffers down-regulation of plasmin throughout life.

10.4.4.1 The Role of Lipid Rafts in this Event

Ledesma et al. [130] have shown that AD hippocampi show a large disorganization of membrane lipid rafts and reduced plasmin activity. Raft disorganization could lead to AD due to a failure in the activation of a plasmin-mediated amyloid clearance pathway. Thus, these findings highlight the importance of correct membrane organization in the maintenance of amyloid clearance (see Fig. 10.3B).

Such hypotheses are in apparent contradiction with the idea that raft perturbation by cholesterol depletion may prevent AD by reducing the production of A β [95,102]. The amount of cholesterol removal and the different cell types used may explain the diverse results of rafts disruption. Hence, plasmin activity was lost when cholesterol reduction was no higher than 36% in human samples. In contrast, inhibition of A β production was observed in one case after acute removal of membrane cholesterol (60%) [102] (see Fig. 10.3B), and also in undifferentiated N2A cells [95]. From these data it is clear that the lack of proper rafts organization produces changes in different raft-mediated events that might be related to AD pathogenesis.

10.4.5

The Role of Caveolae/Rafts in γ -Secretase Activity

The γ -secretase complex was also shown to be associated with rafts [131,132], mainly in the late-Golgi and post-Golgi-derived vesicles as well as in recycling endosomes, but not in plasma membrane rafts [95,132,133].

A recent report showed that despite γ -secretase cleavage occurring in lipid rafts, γ -secretase catalytic activity is independent of the presence of cholesterol [133,134]. Indeed, although biochemical assays in neuronal cells showed mature γ -secretase to be associated with lipid rafts in a cholesterol-dependent manner when cholesterol was acutely depleted using m β CD, the quantities of A β produced were not altered.

In addition, it has been proposed that γ -cleavage of APP could also occur following endocytosis, resulting in the intracellular generation of A β that could be either retained in intracellular compartments or released from recycling endosomes [115]. This balance of intracellular versus cell-surface A β generation could be regulated by the half-life of APP at the cell surface.

The current data therefore support two different possibilities for APP γ -cleavage:

- γ-cleavage occurs within cholesterol-rich lipid rafts in an intracellular compartment such as early/recycling endosomes, late-Golgi and post-Golgi-derived vesicles; or
- γ -cleavage of APP occurs at the cell surface outside rafts (see Fig. 10.4).

10.4.6

The Contribution of Cholesterol and Sphingolipids in APP Processing

The cholesterol content of lipid rafts has been shown to contribute to the integrity of raft structure and the function of lipid rafts in signaling and membrane trafficking [4,135]. In addition, the results of several studies have shown that sphingolipids modulate raft functions [136,137].

A very recent report [138] examined the changes in APP processing and A β generation in sphingolipid-deficient cells, thus demonstrating the importance of sphingolipid levels for modulating APP processing. In particular, it has been shown that sphingolipid deficiency enhances sAPP α secretion via activation of the MAPK/ERK pathway, though the mechanism by which sphingolipid deficiency enhances ERK activity is as yet unknown. Interestingly, cholesterol depletion is reported to increase sAPP α secretion rates, although it is not known whether the MAPK/ERK pathway is involved in this cholesterol depletion-mediated pathway [83].

Furthermore, the possibility that cholesterol and sphingolipids depletion enhances APP α -cleavage in different ways cannot be excluded. Previous reports have shown that cholesterol depletion causes a decrease in β -cleavage activity and an increase in α -cleavage activity of APP [102]. In contrast, under conditions in which cholesterol levels are unchanged and lipid raft dysfunction is caused by sphingolipid depletion, the α -cleavage of APP is enhanced without affecting the β -cleavage activity of APP or the APP level in lipid rafts. These findings could suggest that sphingolipid depletion may enhance α -cleavage of APP without shifting the intracellular trafficking of APP from the "A β generation site" (lipid rafts) to the "A β nongeneration site" (outside rafts).

Therefore, cholesterol and sphingolipids may play different roles in affecting α and β -cleavage of APP. Interestingly, also in the case of prion diseases, the effect of sphingolipid depletion on the formation of scrapie prion protein is opposite to that of cholesterol depletion [33,76].

10.5 Conclusions

In this chapter we have examined the intracellular routing and processing of PrP^c and APP, and have discussed the possible intracellular sites of prion conversion and A β generation. Because the intracellular trafficking and processing of prion proteins have crucial roles in prion conversion, it is believed that investigations of the different stages of intracellular trafficking and processing of prion proteins, together with an analysis of the cellular site involved in the conformational changes, will help to clarify the mechanisms which regulate prion formation.

Current data suggest that the ER might play a major role in the conversion of mutant PrP, whilst in the infectious diseases it is the transport of PrP^c to the plasma membrane and its subsequent internalization that appear to be require-

ments for conversion. Furthermore, lipid rafts with which both PrP^c and PrP^{Sc} are associated also appear to be fundamental for the conversion process. There are clear indications that rafts play an important role in stabilizing PrP^c conformation, thereby exerting a protective role for the disease (see Fig. 10.2). They also appear to play a crucial role in the pathogenesis of Alzheimer's disease.

In particular, current evidence indicates that APP α -cleavage could occur both at the cell surface and intracellularly. Although some of the α -secretases have been localized in caveolae, the majority of this cleavage seems to occur outside rafts. Nonetheless, it is clear that membrane and raft integrity play major roles in controlling the balance between α - and β -cleavage which compete for the same substrate. Indeed, a moderate or drastic cholesterol depletion could affect the levels of these cleavages in different ways, leading to an increment of one or the other secretase processing, and thereby either incrementing or reducing A β production (see Fig. 10.3B). The manner in which this occurs and the nature of the major players is not completely clear, however.

A more meticulous analysis of lipid raft composition, together with the application of new methods to investigate the specific location of PrP and APP/A β in living cells and to reveal conformational changes within the molecules should provide a better understanding of prion and A β generation. This will lead to a better understanding of the pathogenesis of prion and Alzheimer's diseases, and possibly also to the development of new drugs for the prevention and therapy of these conditions.

Acknowledgments

One of the authors (D.S.) is recipient of a fellowship from FIRB 2001, No. NE01S29H. Prion research in the Zurzolo laboratory is supported by EU grant QLK-CT-2002–81628, the Weizmann-Pasteur Foundation, the FRM grant and from Telethon (No. GGP04147).

Abbreviations

Αβ	amyloid β-peptide
AD	Alzheimer's disease
AMF	autocrine motility factor
ароЕ	apolipoprotein E
APP	amyloid precursor protein
BIP	immunoglobulin heavy chain binding protein
BACE	BETA-SITE APP cleavage enzyme
СНО	Chinese hamster ovary
CJD	Creutzfeldt-Jakob disease
CLD	caveolar-like domain
CTF	carboxyl terminal fragment

224 References

DRM	detergent-resistant membrane
ER	endoplasmic reticulum
ERAD	ER-associated degradation
GFAP	glial fibrillary ecidic protein
GPI	glycophosphatidylinositol
HEK	human embryonic kidney
HEK	human embryonic kidney
mβCD	methyl-β-cyclodextrin
PKC	protein kinase C
SV40	Simian virus-40
TCR	T-cell receptor
TGN	trans-Golgi network

References

- Mann, D.M.A., Iwatsubo, T., Nochlin, D., et al. (1997) Amyloid deposition in chromosome 1-linked Alzheimers disease: the Volga German families. *Ann. Neurol.* 41: 52–57.
- **2** Hsiao, K. K., Chapman, P., Nilsen, S., et al. (1996) Correlative memory deficits, Aβ elevation, and amyloid plaques in transgenic mice. *Science* 274: 99–102.
- 3 Soto, C. (1999) Alzheimer's and prion diseases as disorders of protein conformation: implication for the design of novel therapeutic approaches. J. Mol. Med. 77: 412–418.
- 4 Simons, K., Ikonen, E. (1997) Functional rafts in cell membranes. *Nature* 387: 569–572.
- 5 Simons, K., Toomre, D. (2000) Lipid rafts and signal transduction. *Nat. Rev. Mol. Cell. Biol.* 1(1): 31–39.
- 6 Brown, D. A., London, E. (1998) Functions of lipid rafts in biological membranes. *Annu. Rev. Cell. Dev. Biol.* 14: 111–136.
- 7 Janes, P. W., Ley, S. C., Magee, A. I., Kabouridis, P. S. (2000) The role of lipid rafts in T cell antigen receptor (TCR) signalling. *Semin. Immunol.* 12: 23–34.
- 8 Langlet, C., Bernard, A. M., Drevot, P., He, H. T. (2000) Membrane rafts and signaling by the multichain immune recognition receptors. *Curr. Opin. Immunol.* 12: 250–255.
- 9 London, E., Brown, D.A. (2000) Insolubility of lipids in Triton X-100: physical ori-

gin and relationship to sphingolipid/cholesterol membrane domains (rafts). *Biochim. Biophys. Acta* 1508: 182–195.

- 10 Drab, M., Verkade, P., Elger, M., Kasper, M., Lohn, M., Lauterbach, B., Menne, J., Lindschau, C., Mende, F., Luft, F. C., Schedl, A., Haller, H., Kurzchalia, T. V.(2001) Loss of caveolae, vascular dysfunction and pulmonary defects in caveolin-1 gene disrupted mice. *Science* 293: 2449–2452.
- Capozza, F., Cohen, A. W., Cheung, M. W., Sotgia, F., Schubert, W., Battista, M., Lee, H., Frank, P. G., Lisanti, M. P. (2005) Muscle-specific interaction of caveolin isoforms: Differential complex formation between caveolins in fibroblastic vs. muscle cells. *Am. J. Physiol. Cell. Physiol.* 288: C677–91 EPUB 2004 Nov 17.
- 12 Palade, G. E. (1953) Fine structure of blood capillaries. J. Appl. Physiol. 24: 1424.
- 13 Parton, R. G. (2003) Caveolae from ultrastructure to molecular mechanisms. *Nat. Rev. Mol. Cell. Biol.* 4: 162–167.
- 14 Liu, J., Oh, P., Horner, T., Rogers, R.A., Schnitzer, J. E. (1997) Organized endothelial cell surface signal transduction in caveolae distinct from glycosylphosphatidylinositol-anchored protein microdomains. J. Biol. Chem. 272: 7211–7222.
- 15 Fra, A. M., Williamson, E., Simons, K., Parton, R. G. (1995) De novo formation of caveolae in lymphocytes by expression of

VIP21-caveolin. Proc. Natl. Acad. Sci. USA 92: 8655–8659.

- 16 Lipardi, C., Mora, R., Colomer, V., Paladino, S., Nitsch, L., Rodriguez-Boulan, E., Zurzolo, C. (1998) Caveolin transfection results in caveolae formation but not apical sorting of glycosylphosphatidylinositol (GPI)-anchored proteins in epithelial cells. J. Cell Biol. 140(3): 617–626.
- 17 Pelkmans, L., Helenius, A. (2002) Endocytosis via caveolae. *Traffic* 3: 311–320.
- 18 Thomsen, P., Roepstorff, K., Stahlhut, M., van Deurs, B. (2002) Caveolae are highly immobile plasma membrane microdomains, which are not involved in constitutive endocytic trafficking. *Mol. Biol. Cell* 13: 238–250.
- 19 Parton, R.G., Joggerstam, B., Simons, K. (1994) Regulated internalization of caveolae. J. Cell Biol. 127: 1199–1215.
- 20 Mayor, S., Riezman, H. (2004) Sorting GPI-anchored proteins. Nat. Rev. Mol. Cell. Biol. 5(2): 110–120.
- 21 Oh, P., McIntosh, D. P., Schnitzer, J. E. (1998) Dynamin at the neck of caveolae mediates their budding to form transport vesicles by GTP-driven fission from the plasma membrane of endothelium. *J. Cell Biol.* 141: 101–114.
- 22 Nabi, I. R., Le, P. U. (2003) Caveolae/raftsdependent endocytosis. J. Cell Biol. 161: 673–677.
- 23 van der Goot, F.G., Harder, T. (2001) Raft membrane domains: from a liquid-ordered membrane phase to a site of pathogen attack. *Semin. Immunol.* 13(2): 89–97.
- 24 Nichols, B. J., Lippincott-Schwartz, J. (2001) Endocytosis without clathrin-coats. *Trends Cell Biol.* 11: 406–412.
- **25** Conner, S. D., Schmid, S. L. (2003) Regulated portals of entry into the cell. *Nature* 422: 37–44.
- 26 Campana, V., Sarnataro, D., Zurzolo, C. (2005) The highways and byways of prion protein trafficking. *Trends Cell Biol.* 15(2): 102–111.
- 27 Prusiner, S. B. (1998) Prions. Proc. Natl. Acad. Sci. USA 95(23): 13363–13383.
- 28 Harris, D.A. (1999) Cellular biology of prion diseases. *Clin. Microbiol. Rev.* 12: 429–444.
- 29 Aguzzi, A., Heppner, F. L. (2000) Pathogenesis of prion diseases: a progress report. *Cell Death Differ*. 7: 889–902.

- 30 Chiesa, R., Harris, D. A. (2001) Prion disease: what is the neurotoxic molecule? *Neurobiol. Dis.* 8: 743–763.
- 31 Harris, D.A. (2003) Trafficking, turnover and membrane topology of PrP. Br. Med. Bull. 66: 71–85.
- 32 Baron, G. S., Wehrly, D., Dorward, D. W., Chesebro, B., Caughey, B. (2002) Conversion of raft associated prion protein to the protease-resistant state requires insertion of PrP-res (PrP(Sc)) into contiguous membranes. *EMBO J.* 21(5): 1031–1040.
- 33 Taraboulos, A., Scott, M.R.D., Semenov, A., Avraham, D., Laszlo, L., Prusiner, S. B. (1995) Cholesterol depletion and modification of COOH-terminal targeting sequence of the prion protein inhibit formation of the scrapie isoform. *J. Cell Biol.* 129(1): 121–132.
- 34 Aguzzi, A., Polymenidou, M. (2004) Mammalian prion biology: one century of evolving concepts. *Cell* 116(2): 313–327.
- 35 Bueler, H., Fischer, M., Lang, Y., Bluethmann, H., Lipp, H. P., DeArmond, S. J., Prusiner, S. B., Aevet, M., Weissmann, C. (1992) Normal development and behaviour of nice lacking the neuronal cell-surface PrP protein. Nature 356: 577–582.
- 36 Pauly, P. C., Harris, D. A. (1998) Copper stimulates endocytosis of the prion protein. J. Biol. Chem. 273(50): 33107– 33110.
- 37 Watt, N.T., Hooper, N.M. (2003) The prion protein and neuronal zinc homeostasis. *Trends Biochem. Sci.* 28(8): 406–410.
- 38 Brown, D. R. (2001) Copper and prion disease. Brain Res. Bull. 55(2): 165–173.
- 39 Chiarini, L. B., Freitas, A. R., Zanata, S. M., Brentani, R. R., Martins, V. R., Linden, R. (2002) Cellular prion protein transduces neuroprotective signals. *EMBO J.* 21(13): 3317–3326.
- 40 Mouillet-Richard, S., Ermonval, M., Chebassier, C., Laplanche, JL., Lehmann, S., Launay, J.M., Kellermann, O. (2000) Signal transduction through prion protein. *Science* 289(5486): 1925–1928.
- 41 Mallucci, G. R., Ratte, S., Asante, E.A., Linehan, J., Gowland, I., Jefferys, J.G., Collinge, J. (2002) Post-natal knockout of prion protein alters hippocampal CA1 properties, but does not result in neurodegeneration. *EMBO J.* 21(3): 202–210.

- 226 References
 - 42 Collinge, J., Whittington, M.A., Sidle, K.C., Smith, C.J., Palmer, M.S., Clarke, A.R., Jefferys, J.G. (1994) Prion protein is necessary for normal synaptic function. *Nature* 370(6487): 295–297.
 - 43 Graner, E., Mercadante, A. F., Zanata, S. M., Forlenza, O. V., Cabral, A. L., Veiga, S. S., Juliano, M. A., Roesler, R., Walz, R., Minetti, A., Izquierdo, I., Martins, V. R., Brentani, R. R. (2000) Cellular prion protein binds laminin and mediates neuritogenesis. *Brain Res. Mol. Brain Res.* 76(1): 85–92.
 - 44 Solforosi, L., Criado, J. R., McGavern, D. B., Wirz, S., Sanchez-Alavez, M., Sugama, S., DeGiorgio, L. A., Volpe, B. T., Wiseman, E., Abalos, G., Masliah, E., Gilden, D., Oldstone, M. B., Conti, B., Williamson, R. A. (2004) Cross-linking cellular prion protein triggers neuronal apoptosis in vivo. *Science* 303(5663): 1514–1516 (e-pub 2004 January 29).
 - 45 Lee, K.S., Linden, R., Prado, M.A., Brentani, R.R., Martins, V.R. (2003) Towards cellular receptors for prions. *Rev. Med. Virol.* 13(6): 399–408.
 - 46 Prado, M.A., Alves-Silva, J., Magalhaes, A.C., Prado, V.F., Linden, R., Martins, V.R., Brentani, R.R. (2004) PrPc on the road: trafficking of the cellular prion protein. J. Neurochem. 88(4): 769–781.
 - 47 Borchelt, D. R., Taraboulos, A., Prusiner S. B. (1992) Evidence for synthesis of scrapie prion proteins in the endocytic pathway. *J. Biol. Chem.* 267(23): 16188–16199.
 - 48 Zanusso, G., Petersen, R.B., Jin, T., Jing, Y., Kanoush, R., Ferrari, S., Gambetti, P., Singh, N. (1999) Proteasomal degradation and N-terminal protease resistance of the codon 145 mutant prion protein. *J. Biol. Chem.* 274(33): 23396–23404.
 - 49 Nunziante, M., Gilch, S., Schatzl, H.M. (2003) Prion diseases: from molecular biology to intervention strategies. *Chembiochem* 4(12): 1268–1284.
 - 50 Hegde, R. S., Rane, N. S. (2003) Prion protein trafficking and the development of neurodegeneration. *Trends Neurosci.* 26(7): 337–339.
 - 51 Beranger, F., Mange, A., Goud, B., Lehmann, S. (2002) Stimulation of PrP(C) retrograde transport toward the endoplasmic reticulum increases accumulation of

PrP(Sc) in prion-infected cells. J. Biol. Chem. 277(41): 38972–38977.

- 52 Sarnataro, D., Campana, V., Paladino, S., Stornaiuolo, M., Nitsch, L., Zurzolo, C. (2004) PrP(C) association with lipid rafts in the early secretory pathway stabilizes its cellular conformation. *Mol. Biol. Cell* 15(9): 4031–4042 (e-pub 2004, June 30).
- 53 Bogdanov, M., Dowhan, W. (1999) Lipidassisted protein folding. J. Biol. Chem. 274(52): 36827–36830.
- 54 Sanders, C. R., Nagy, J. K. (2000) Misfolding of membrane proteins in health and disease: the lady or the tiger? *Curr. Opin. Struct. Biol.* 10(4): 438–442.
- 55 Caughey, B., Raymond, G. J. (1991) The scrapie-associated form of PrP is made from a cell surface precursor that is both protease- and phospholipase-sensitive. *J. Biol. Chem.* 266(27): 18217–18223.
- 56 Taraboulos, A., Raeber, A. J., Borchelt, D. R., Serban, D., Prusiner, S. B. (1992) Synthesis and trafficking of prion proteins in cultured cells. *Mol. Biol. Cell* 3(8): 851–863.
- 57 Supattapone, S. Nishina K., Rees J. R. (2002) Pharmacological approaches to prion research. *Biochem. Pharmacol.* 63(8): 1383–1388.
- 58 Shyng, S. L., Heuser, J. E., Harris, D.A. (1994) A glycolipid-anchored prion protein is endocytosed via clathrin-coated pits. J. Cell Biol. 125(6): 1239–1250.
- 59 Madore, N., Smith, K. L., Graham, C. H., Jen, A., Brady, K., Hall, S., Morris, R. (1999) Functionally different GPI proteins are organized in different domains on the neuronal surface. *EMBO J.* 18(24): 6917–6926.
- 60 Sunyach, C., Jen, A., Deng, J., Fitzgerald, K.T., Frobert, Y., Grassi, J., McCaffrey, M. W., Morris, R. (2003). The mechanism of internalization of glycosylphosphatidylinositol-anchored prion protein. *EMBO J.* 22(14): 3591–3601.
- 61 Lee, K. S., Magalhaes, A. C., Zanata, S. M., Brentani, R. R., Martins, V. R., Prado, M. A. (2001) Internalization of mammalian fluorescent cellular prion protein and N-terminal deletion mutants in living cells. *J. Neurochem.* 79(1): 79–87.
- 62 Vey, M., Pilkuhn, S., Wille, H., Nixon, R., DeArmond, S. J., Smart, E. J., Anderson, R. G., Taraboulos, A., Prusiner, S.B.

(1996) Subcellular colocalization of the cellular and scrapie prion proteins in caveolae-like membranous domains. *Proc. Natl. Acad. Sci. USA* 93(25): 14945–14949.

- 63 Kaneko, K., Vey, M., Scott, M., Pilkuhn, S., Cohen, F. E., Prusiner, S. B. (1997) COOH-terminal sequence of the cellular prion protein directs subcellular trafficking and controls conversion into the scrapie isoform. *Proc. Natl. Acad. Sci. USA* 94(6): 2333–2338.
- 64 Peters, P. J., Mironov, A., Jr., Peretz, D., van Donselaar, E., Leclerc, E., Erpel, S., DeArmond, S. J., Burton, D. R., Williamson, R. A., Vey, M., Prusiner, S. B. (2003) Trafficking of prion proteins through a caveolae-mediated endosomal pathway. *J. Cell Biol.* 162(4): 703–717.
- 65 Gacescu, R., Demaurex, N., Parton, R. G., Hunziker, W., Huber, L. A., Gruenberg, J. (2000) The recycling endosome of Madin-Darby canine kidney cells is a mildly acidic compartment rich in rafts component. *Mol. Biol. Cell* 11: 2775–2791.
- 66 Baron, G. S., Caughey, B. (2003) Effect of glycosylphosphatidylinositol anchor-dependent and -independent prion protein association with model raft membranes on conversion to the protease-resistant isoform. J. Biol. Chem. 278(17): 14883–14892 (e-pub 2003 February 19).
- 67 Botto, L., Masserini, M., Cassetti, A., Palestini, P. (2004) Immunoseparation of prion protein enriched domains from other detergent-resistant membrane fractions, isolated from neuronal cells. *FEBS Lett.* 557(1–3): 143–147.
- 68 Naslavsky, N., Stein, R., Yanai, A., Friedlander, G., Taraboulos, A. (1997) Characterization of detergent-insoluble complexes containing the cellular prion protein and its scrapie isoform. *J. Biol. Chem.* 272(10): 6324–6331.
- 69 Klein, T. R., Kirsch, D., Kaufmann, R., Riesner, D. (1998) Prion rods contain small amount of two host sphingolipids as revealed by thin-layer chromatography and mass spectrometry. *Biol. Chem.* 379: 655–666.
- 70 Sarnataro, D., Paladino, S., Campana, V., Grassi, J., Nitsch, L., Zurzolo, C. (2002) PrPC is sorted to the basolateral membrane of epithelial cells independently of

its association with rafts. *Traffic* 3(11): 810–821.

- 71 Kanu, N., Imokawua, Y., Drechsel, D. N., Williamson, R. A., Birkett, C. R., Bostock, C. J., Brockes, J. P. (2002) Transfer of scrapie prion infectivity by cell contact in culture. *Curr. Biol.* 12: 523–530.
- 72 Bogdanov, M., Dowhan, W. (1999) Lipidassisted protein folding. J. Biol. Chem. 274(52): 36827–36830.
- 73 Sanders, C. R., Nagy, J. K. (2000) Misfolding of membrane proteins in health and disease: the lady or the tiger? *Curr. Opin. Struct. Biol.* 10(4): 438–442.
- 74 Sanghera, N., Pinheiro, T. J. (2002) Binding of prion protein to lipid membranes and implications for prion conversion. *J. Mol. Biol.* 315(5): 1241–1256.
- 75 Critchley, P., Kazlauskaite, J., Eason, R., Pinheiro, T. J. (2004) Binding of prion proteins to lipid membranes. *Biochem. Biophys. Res. Commun.* 313(3): 559–567.
- 76 Naslavsky, N., Shmeeda, H., Friedlander, G., Yanai, A., Futerman, A. H., Barenholz, Y., Taraboulos, A. (1999) Sphingolipid depletion increases formation of the scrapie prion protein in neuroblastoma cells infected with prions. *J. Biol. Chem.* 274(30): 20763–20771.
- 77 Fantini, J., Garmy, N., Mahfoud, R., Yahi, N. (2002) Lipid rafts: structure, function and role in HIV, Alzheimer's and prion diseases. *Expert Rev. Mol. Med.* 2002: 1–22.
- 78 Forloni, G., Angeretti, N., Chiesa, R., Monzani, E., Salmona, M., Bugiani, O., Tagliavini, F. (1993) Neurotoxicity of a prion protein fragment. *Nature* 362: 543–546.
- 79 Ettaiche, M., Pichot, R., Vincent, J. P., Chabry, J. (2000) In vivo cytotoxicity of the prion protein fragment 106–126. *J. Biol. Chem.* 275: 36487–36490.
- 80 Jobling, M. F., Stewart, L. R., White, A. R., McLean, C., Friedhuber, A., Maher, F., Beyreuther, K., Masters, C. L., Barrow, C. J., Collins, S. J., Cappai, R. (1999) The hydrophobic core sequence modulates the neurotoxic and secondary structure properties of the prion peptide 106–126. *J. Neurochem.* 73(4): 1557–1565.
- **81** Turner, A. J., Hooper, N. (1999) Role for ADAM-family proteinases as membrane

228 References

protein secretases. Biochem. Soc. Trans. 27: 255–259.

- 82 Nunan, J., Small, D. H. (2000) Regulation of APP cleavage by α-, β- and •-secretase. *FEBS Lett.* 483: 6–10.
- 83 Kojro, E., Gimpl, G., Lammich, S., Marz, W., Fahrenholz, F. (2001) Low cholesterol stimulates the nonamyloidogenic pathway by its effect on the α-secretase ADAM 10. *Proc. Natl. Acad. Sci. USA* 98(10): 5815–5820.
- 84 Wakatsuki, S., Kurisaki, T., Sehara-Fujisawa, A. (2004) Lipid rafts identified as locations of ectodomain shedding mediated by Meltrin beta/ADAM19. *J. Neurochem.* 89(1): 119–123.
- 85 Selkoe, D. J. (2001) Alzheimer's disease: genes, proteins, and therapy. *Physiol. Rev.* 81: 741–766.
- 86 Golde, T. E., Eckman, C. B. (2001) Cholesterol modulation as an emerging strategy for the treatment of Alzheimer's disease. *Drug Discov. Today* 6: 1049–1055.
- 87 Sisodia, S. S., St. George-Hyslop, P. H. (2002) gamma-Secretase, Notch, Abeta and Alzheimer's disease: where do the presenilins fit in? *Nat. Rev. Neurosci.* 3(4): 281–290.
- 88 Vassar, R., Bennett, B. D., Babu-Khan, S., Kahn, S., Mendiaz, E. A., Denis, P., Teplow, D. B., Ross, S., Amarante, P., Loeloff, R., Luo, Y., Fisher, S., Fuller, J., Edenson, S., Lile, J., Jarosinski, M. A., Biere, A. L., Curran, E., Burgess, T., Louis, J. C., Collins, F., Treanor, J, Rogers, G., Citron, M. (1999) Beta-secretase cleavage of Alzheimer's amyloid precursor protein by the transmembrane aspartic protease BACE. *Science* 286(5440): 735–741.
- 89 Capell, A., Steiner, H., Willem, M., Kaiser, H., Meyer, C., Walter, J., Lammich, S., Multhaup, G., Haass, C. (2000) Maturation and pro-peptide cleavage of betasecretase. *J. Biol. Chem.* 275: 30849–30854.
- 90 Annaert, W.G., Levesque, L., Craessaerts, K., Dierinck, I., Snellings, G., Westaway, D., St. George-Hyslop, P., Cordell, B., Fraser, P., de Strooper, B. (1999) Presenilin 1 controls gamma-secretase processing of amyloid precursor protein in pre-Golgi compartments of hippocampal neurons. J. Cell Biol. 147: 277–294.

- 91 Zhang, J., Kang, D. E., Xia, W., Okochi, M., Mori, H., Selkoe, D. J., Koo, E. H. (1998) Subcellular distribution and turnover of presenilins in transfected cells. *J. Biol. Chem.* 273: 12436–12442.
- **92** Kaether, C., Lammich, S., Edbauer, D., Ertl, M., Rietdorf, J., Capell, A., Steiner, H., Haass, C. (2002) Presenilin-1 affects trafficking and processing of beta APP and is targeted in a complex with nicastrin to the plasma membrane. *J. Cell Biol.* 158: 551–561.
- 93 Riddell, D. R., Christie, G., Hussain, I., Dingwall, C.(2001) Compartmentalization of β-secretase (Asp2) into low-buoyant density, noncaveolar lipid rafts. *Curr. Biol.* 11: 1288–1293.
- 94 Burns, M., Duff, K. (2002) Cholesterol in Alzheimer's disease and tauopathy. Ann. N. Y. Acad. Sci. 977: 367–375.
- 95 Ehehalt, R., Keller, P., Haass, C., Thiele, C., Simons, K. (2003) Amyloidogenic processing of the Alzheimer beta-amyloid precursor protein depends on lipid rafts. *J. Cell Biol.* 160(1): 113–123.
- 96 Lee, S. J., Liyanage, U., Bickel, P. E., Xia, W., Lansbury, P. T., Kosik, K. S. (1998) A detergent-insoluble membrane compartment contains A beta in vivo. *Nat. Med.* 4: 730–734.
- 97 Ikezu, T., Trapp, B. D., Song, K. S., Schlegel, A., Lisanti, M. P., Okamoto, T. (1998) Caveolae, plasma membrane microdomains for α-secretase-mediated processing of the amyloid precursor protein. *J. Biol. Chem.* 273(17): 10485–10495.
- 98 Avdulov, N. A., Chochina, S. V., Igbavboa, U., Warden, C. S., Vassiliev, A. V., Wood, W. G. (1997) Lipid binding to amyloid beta-peptide aggregates: preferential binding of cholesterol as compared with phosphatidylcholine and fatty acids. *J. Neurochem.* 69: 1746–1752.
- 99 Choo-Smith, L. P., Garzon-Rodriguez, W., Glabe, C. G., Surewicz ,W. K. (1997) Acceleration of amyloid fibril formation by specific binding of Abeta-(1–40) peptide to ganglioside-containing membrane vesicles. J. Biol. Chem. 272: 22987–22990.
- 100 Matsuzaki, K., Horikiri, C. (1999) Interactions of amyloid beta-peptide (1–40) with ganglioside-containing membranes. *Biochemistry* 38: 4137–4142.

- 101 Brown, D.A., Rose, J.K. (1992) Sorting of GPI-anchored proteins to glycolipid-enriched membrane subdomains during transport to the apical cell surface. *Cell* 68: 533–544.
- 102 Simons, M., Keller, P., De Strooper, B., Beyreuther, K., Dotti, C. G., Simons, K. (1998) Cholesterol depletion inhibits the generation of beta-amyloid in hippocampal neurons. *Proc. Natl. Acad. Sci. USA* 95: 6460–6464.
- 103 Rodal, S. K., Skretting, G., Garred, O., Vilhardt, F., van Deurs, B., Sandvig, K. (1999) Extraction of cholesterol with methyl-beta-cyclodextrin perturbs formation of clathrin-coated endocytic vesicles. *Mol. Biol. Cell* 10: 961–974.
- 104 Koo, E. H., Squazzo, S. L. (1994) Evidence that production and release of amyloid beta protein involves the endocytic pathway. *J. Biol. Chem.* 269: 17386–17389.
- 105 Dumery, L., Bourdel, F., Soussan, Y., Fialkowsky, A., Viale, S., Nicolas, P., Reboud-Ravaux, M. (2001) Beta-amyloid protein aggregation: its implication in the physiopathology of Alzheimer's disease. *Pathol. Biol. (Paris)* 49: 72–85.
- 106 Kakio, A. (2002) Interactions of amyloid beta-protein with various gangliosides in raft-like membranes: importance of GM1 ganglioside-bound form as an endogenous seed for Alzheimer amyloid. *Biochemistry* 41: 7385–7390.
- 107 McLaurin, J. (1998) Structural transitions associated with the interaction of Alzheimer beta-amyloid peptides with gangliosides. J. Biol. Chem. 273: 4506–4515.
- 108 Mizuno, T., Nakata, M., Naiki, H., Michikawa, M., Wang, R., Haass, C., Yanagisawa, K. (1999) Cholesterol-dependent generation of a seeding amyloid β-protein in cell culture. *J. Biol. Chem.* 274: 15110–15114.
- 109 Mahfoud, R., Garmy, N., Maresca, M., Yahi, N., Puigserver, A., Fantini, J. (2002) Identification of a common sphingolipidbinding domain in Alzheimer, prion and HIV proteins. *J. Biol. Chem.* 277: 11292–11296.
- 110 Perez, R.G., Soriano, S., Hayes, D.J., Ostaszewski, B., Xia, W., Selkoe, D.J., Chen, X., Stokin, G.B., Koo, E.H. (1999) Mutagenesis identifies new signals for betaamyloid precursor protein endocytosis,

turnover and the generation of secreted fragments, including Abeta 42. *J. Biol. Chem.* 274: 18851–18856.

- 111 Huse, J.T., Pijak, D.S., Leslie, G.J., Lee, V.M., Doms, R.W. (2000) Maturation and endosomal targeting of beta-site amyloid precursor protein-cleaving enzyme. The Alzheimer's disease beta secretase. *J. Biol. Chem.* 275: 33729–33737.
- 112 Kamal, A., Almenar-Queralt, A., LeBlanc, J. F., Roberts, E. A., Goldstein, L. S. (2001) Kinesin-mediated axonal transport of a membrane compartment containing betasecretase and presenilin-1 requires APP. *Nature* 414: 643–648.
- 113 Damke, H.T., Baba, T., Warnock, D.E., Shmid, S.L. (2004) Induction of mutant dynamin specifically blocks endocytic coated vesicle formation. *J. Cell Biol.* 127: 915–934.
- 114 Lanzetti, L., Rybin, V., Malabarba, M.G., Christoforidis, S., Scita, G., Zerial, M., Di Fiore, P.P. (2000) The Eps8 protein coordinates EGF receptor signalling through Rac and trafficking through Rab5. *Nature* 408: 374–377.
- 115 Chyung, J. H., Selkoe, D. J. (2003) Inhibition of receptor-mediated endocytosis demonstrates generation of amyloid betaprotein at the cell surface. *J. Biol. Chem.* 278(51): 51035–51043 (e-pub 2003 October 02).
- 116 Haass, C., Koo, E. H., Mellon, A., Hung, A. Y., Selkoe, D. J. (1992) Targeting of cell surface beta-amyloid precursor protein to lysosomes: alternative processing into amyloid-bearing fragments. *Nature* 357: 500–503.
- 117 Parvarthy, S., Hussain, I., Karran, E. H., Turner, A. J., Hooper, N. M. (1999) Cleavage of Alzheimer's amyloid precursor protein by alpha-secretase occurs at surface of neuronal cells. *Biochemistry* 38: 9728–9734.
- 118 Cordy, J. M., Hussain, I., Dingwall, C., Hooper, N. M., Turner, A. J. (2003) Exclusively targeting beta-secretase to lipid rafts by GPI-anchor addition up-regulates beta-site processing of the amyloid precursor protein. *Proc. Natl. Acad. Sci. USA* 100: 11735–11740.
- 119 Abad-Rodriguez, J., Ledesma, M. D., Craessaerts, K., Perga, S., Medina, M., Delacourte, A., Dingwall, C., De Strooper,

B., Dotti, C.G. (2004) Neuronal membrane cholesterol loss enhances amyloid peptide generation. J. Cell Biol. 167: 953-960.

- 120 Koo, E. H., Park, L., Selkoe, D. J. (1993) Amyloid beta-protein as a substrate interacts with extracellular matrix to promote neurite outgrowth. Proc. Natl. Acad. Sci. USA 90: 4748-4752.
- 121 Sisodia, S.S. (1992) Beta-amyloid precursor protein cleavage by a membranebound protease. Proc. Natl. Acad. Sci. USA 89: 6075-6079.
- 122 Arribas, J., Lopez-Casillas, F., Massague, J. (1997) Role of the juxtamembrane domains of the transforming growth factoralpha precursor and the beta-amyloid pre- 131 Li, Y.M., Xu, M., Lai M.T., Huang, Q., cursor protein in regulated ectodomain shedding. J. Biol. Chem. 272: 17161-17165.
- 123 De Strooper, B., Umans, L., Van Leuven, F., Van Den Berghe, H. (1993) Study of the synthesis and secretion of normal and artificial mutants of murine amyloid precursor protein (APP): cleavage of APP occurs in a late compartment of the default secretion pathway. J. Cell Biol. 121: 295-304.
- 124 Haass, C., Koo, E. H., Capell, A., Teplow, D. B., Selkoe, D. J. (1995) Polarized sorting of beta-amyloid precursor protein and its proteolytic products in MDCK cells is regulated by two independent signals. J. Cell Biol. 128: 537-547.
- 125 De Strooper, B., Van Leuven, F., Van den Berghe, H. (1992) Alpha 2-macroglobulin and other proteinase inhibitors do not interfere with the secretion of amyloid precursor protein in mouse neuroblastoma cells. FEBS Lett. 308(1): 50-53.
- 126 Selkoe, D.J. (1994) Cell biology of the amyloid beta-protein precursor and the mechanism of Alzheimer's disease. Annu. Rev. Cell Biol. 10: 373-403.
- 127 Nordstedt, C., Caporaso, G.L., Thyberg, J., Gandy, S.E., Greengard, P. (1993) Identification of the Alzheimer beta/A4 amyloid precursor protein in clathrincoated vesicles purified from PC12 cells. I. Biol. Chem. 268(1): 608-612.
- 128 Ledesma, M.D., Da Silva, G.S., Crassaerts, K., Delacourte, A., De Strooper, B., Dotti, C.G. (2000) Brain plasmin enhances APP α -cleavage and is reduced in

Alzheimer's disease brains. EMBO Rep. 1(6): 530-535.

- 129 Koike, H., Tomioka, S., Sorimachi, H., Saido, T.C., Maruyama, K., Okuyama, A., Fujisawa-Sehara, A., Ohno, S., Suzuki, K., Ishiura, S. (1999) Membrane-anchored metalloprotease MDC9 has an alpha-secretase activity responsible for processing the amyloid precursor protein. Biochem. J. 343: 371-375.
- 130 Ledesma, M. D., Abad-Rodriguez, J., Galvan C., Biondi E., Navarro, P., Delacourte, A., Dingwall, C., Dotti, C.G. (2003) Rafts disorganization leads to reduced plasmin activity in Alzheimer's disease brains. EMBO Rep. 4(12): 1190-1196.
- Castro, J. L., DiMuzio-Mower, J., Harrison, T., Lellis, C., Nadin, A., Neduvelil J.G., et al. (2000) Photoactivated gammasecretase inhibitors directed to the active site covalently label presenilin 1. Nature 405: 689-694.
- 132 Wahrle, S., Das, P., Nyborg, A.C., McLendon, C., Shoji, M., Kawarabayashi, T., Younkin, L.H., Younkin, S.G., Golde, T.E. (2002) Cholesterol-dependent gamma-secretase activity in buoyant cholesterol-rich membrane microdomains. Neurobiol. Dis. 9(1): 11-23.
- 133 Vetrivel, K.S., Cheng, H., Lin, W., Sakurai, T., Li, T., Nukina, N., Wong, P.C., Xu, H., Thinakaran, G. (2004) Association of --secretase with lipid rafts in post-Golgi endosomes membranes. J. Biol. Chem. 279: 44945-44954.
- 134 Wada, S., Morishima-Kawashima, M., Qi, Y., Misono, H., Shimada, Y., Ohno-Iwashita, Y., Ihara, Y. (2003) Gamma-secretase activity is present in rafts but is not cholesterol-dependent. Biochemistry 42(47): 13977-13986.
- 135 Anderson, R.G., Jacobson, K. (2002) A role for lipid shells in targeting proteins to caveolae, rafts, and other lipid domains. Science 296(5574): 1821-1825.
- 136 Hanada, K., Izawa, K., Nishijima, M., Akamatsu, Y. (1993) Sphingolipid deficiency induces hypersensitivity of CD14, a glycosylphosphatidylinositol-anchored protein, to phosphatidylinositol-specific phospholipase C. J. Biol. Chem. 268: 13820-13823.

- 137 Hanada, K., Nishijima, M., Akamatsu, Y., Pagano, R. E. (1995) Both sphingolipids and cholesterol participate in the detergent insolubility of alkaline phosphatase, a glycosylphosphatidylinositol-anchored protein, in mammalian membranes. J. Biol. Chem. 270: 6254–6260.
- 138 Sawamura, N., Ko, M., Yu, W., Zou, K., Hanada, K., Suzuki, T., Gong, J. S., Yanagisawa, K., Michikawa, M. (2004) Modulation of amyloid precursor protein cleavage by cellular sphingolipids. *J. Biol. Chem.* 279(12): 11984–11991.

11 Caveolae and the Endothelial Nitric Oxide Synthase

Olivier Feron

11.1 Introduction

Nitric oxide (NO) is a multifaceted molecule which plays key roles in many biological situations [1]. Endothelial nitric oxide synthase (eNOS) is the major NOS isoform responsible for cardiovascular homeostasis. eNOS is a calmodulin-activated enzyme which consists of an oxygenase and a reductase domain containing binding sites for a variety of cofactors that promote electron transfer from one domain to the other, leading ultimately to the conversion of L-arginine to citrulline and NO [2] (Fig. 11.1).



Fig. 11.1 Reciprocal regulation of the eNOS catalytic activity by caveolin (Cav) and calcium-bound calmodulin (CaM). eNOS is a two-domain enzyme consisting of a reductase domain containing binding sites for flavin mononucleotide (FMN), flavin adenine dinucleotide (FAD) and nicotinamide adenine dinucleotide phosphate hydrogen (NADPH) and an N-terminal oxygenase domain with binding sites for heme, L-arginine (Arg) and tetrahydrobiopterin (BH4). eNOS is active as a dimer: NADPH-derived electrons are transferred to the flavins and then to the heme (located on the vis-à-vis monomer) where O_2 can be recruited and catalyzes NO synthesis from L-arginine. Caveolin interaction with the eNOS oxygenase domain stabilizes the complex and prevents L-arginine binding, whereas caveolin interaction with the reductase domain antagonizes CaM binding and slows down electron transfer to the oxygenase domain, thereby inhibiting NO production.



Fig. 11.2 The multiple roles of nitric oxide (NO) in the healthy heart. Endothelial cells in coronary arteries (as in other vascular beds) constitutively express endothelial nitric oxide synthase (eNOS) that regulates critical vascular functions such as contractility, permeability, leukocyte adhesion, platelet aggregation

and angiogenesis through basal and stimulated NO production. In cardiac myocytes, both eNOS and nNOS are constitutively expressed and modulate key processes including excitation-contraction coupling, mitochondrial respiration, and receptor-mediated autonomic stimulation.

The principal source of NO within the normal myocardium is the endothelium of the coronary vasculature. Endothelial cells constitutively express the eNOS isoform, which generates NO in response to specific extracellular signals to regulate vascular smooth muscle tone and thrombogenicity, among other actions [3] (Fig. 11.2). In addition to coronary vascular (and endocardial) endothelium, both atrial and ventricular myocytes - including specialized pacemaker tissue - also express eNOS. Excitation-contraction coupling, modulation of autonomic signaling and mitochondrial respiration are more directly regulated by myocyte-produced NO [4] (see Fig. 11.2). Amazingly, the same molecule, NO, has been involved in many cardiovascular diseases [5]. A shift from finely regulated NO production by eNOS (and neuronal NOS, also expressed in cardiac myocytes) to a deregulated NO production by the inducible NOS isoform (iNOS) is, indeed, associated with a variety of heart diseases. The extent of NO release (e.g., limited for the two constitutive NOS and in large excess for iNOS) is thought to account for the differential effects on cardiac function. Recognition of the threshold above which NO becomes toxic is, however, unclear and certainly appears elusive when considering, for instance, the potential benefits of nitrates - medications which deliver large amounts of NO. A safer way of addressing the question of cytotoxic versus protective effects of NO is to emphasize the qualitative characteristics of NO release. Consequently, the following paragraphs will emphasize, by a series of examples, the developing consensual view according to which (in nonpathological

states at least) NO exerts its regulatory roles by being produced at the right time in the right place.

11.2 Caveolin: A Scaffold for eNOS

Since NO is a very labile and highly reactive messenger molecule with autocrine and paracrine functions, the site of NO production should logically have a major influence on its biological activity. The discovery in 1996 of the location of eNOS in caveolae [6,7] was therefore viewed as the proof of concept that compartmentation of the enzyme is critical to fine-tune NO synthesis. This specific locale of eNOS had been suspected based on the double acylation process that characterizes eNOS: myristoylation on glycine (at position 2) and palmitoylation on two cysteines (at positions 15 and 26). In fact, by using cultured endothelial cells, eNOS was shown to be preferentially located in caveolae (versus the rest of the plasma membrane), with each acylation process enhancing the caveolar enrichment some 10-fold [6]. This discovery was rapidly followed by the identification of a tight regulation between eNOS and caveolin, the structural protein of caveolae. It has been reported that, in endothelial cells and cardiac myocytes, eNOS was quantitatively associated with caveolin-1 and caveolin-3, respectively [8]. The determinants of this interaction subsequently became the focus of studies conducted by several independent groups. It was shown, by exploring the differential effects of detergents, that although eNOS thiopalmitoylation is not absolutely required to induce formation of the caveolin-eNOS complex, acylation largely facilitates the interaction between both proteins [9]. The sequences involved in this mutual interaction were also identified, based on the studies of Lisanti et al., who found a region within the caveolin sequence that could act as a scaffold for many caveolar proteins [10].

This so-called "caveolin scaffolding domain" (CSD) is a region spanning 20 residues in the caveolin sequence (mapping to residues 81-101 in the human caveolin-1 sequence) [10]. Using a glutathione S-transferase (GST)-CSD fusion protein as a bait to select peptide ligands from a bacteriophage display library, Lisanti and colleagues identified a "caveolin binding motif" (CBM) that appeared to be present in whole or in part in many proteins located in caveolae ($\Phi X \Phi X X X X$ Φ XX Φ , where Φ represents an aromatic amino acid) [11]. That the interaction with caveolin involves these consensus sequences is, however, only documented for a few caveolar residents but among them, stands eNOS [12,13]. Several laboratories have indeed investigated the molecular determinants of the caveolin-eNOS interaction using in-vitro binding assay systems with GST fusion proteins (including deletion mutants) and an in-vivo yeast two-hybrid system [9,13-15]. The major conclusions from these studies are that eNOS and caveolin-1 interact directly rather than indirectly, and that this interaction involves multiple sites: the oxygenase and reductase domains of eNOS and the two cytoplasmic domains of caveolin-1. The CBM (sequence 350–358 FPAAPFSGW) that recognizes the CSD is located in the
236 11 Caveolae and the Endothelial Nitric Oxide Synthase

oxygenase domain of eNOS, between the heme and the calmodulin binding domains. This location is adjacent to a glutamate residue (Glu361) necessary for the binding of L-arginine, which suggests that caveolin may interfere with heme iron reduction, similarly to L-arginine-based NOS inhibitors (see Fig. 11.1).

This latter observation led us and others to investigate whether the caveolineNOS interaction was inhibitory. Like all known NO synthases, eNOS enzyme activity is dependent on calmodulin binding, the activation of which requires an increase in intracellular calcium. With the discovery of the caveolin-eNOS interaction, it appeared that calmodulin acts, in fact, as a direct allosteric competitor promoting the disruption of the heteromeric complex formed between eNOS and caveolin in a Ca²⁺-dependent fashion [13,15,16] (Fig. 11.1). Both the CSD and the CBM domains were shown to be involved. Accordingly, peptides corresponding to the CSD domain were shown to interact directly with the enzyme and markedly inhibit NOS activity in endothelial cells [13]. Likewise, site-directed mutagenesis of the CBM was found to block the ability of caveolin-1 to suppress NO release in transfection experiments [12].

Using full-length eNOS or truncated enzyme which only expresses the oxygenase domain, studies conducted by Ghosh et al. [17] led to the establishment of a model, according to which caveolin interaction with the oxygenase domain helps to target the eNOS-caveolin complex to caveolae. In contrast, caveolin interaction with the reductase domain is primarily responsible for antagonizing calmodulin binding and for slowing electron transfer from the reductase, thus inhibiting heme iron reduction and NO synthesis (see Fig. 11.1).

11.3

The Caveolin-eNOS Regulatory Cycle

Consecutive to disruption of the caveolin-eNOS complex induced by agonist stimulation or shear stress, eNOS has been proposed to traffic intracellularly (Fig. 11.3). Prabhakar et al. showed that eNOS is de-palmitoylated after prolonged agonist stimulation, and is no longer selectively sequestered in the caveolae [18]. The acyl-protein thioesterase 1 (APT1) has, indeed, been shown specifically to promote eNOS depalmitoylation through a Ca²⁺-calmodulin-dependent pathway [19]. The translocated enzyme then partitions into noncaveolar plasma/intracellular membranes, and also in the cytosol. The interaction of eNOS with Hsp90 and consecutive (de)phosphorylation (see below) is probably also involved in trafficking of the enzyme, but the molecular mechanisms underlying these processes are still not clearly delineated. By contrast, several lines of evidence have indicated that, subsequent to the translocation of the enzyme and after the decline in $[Ca^{2+}]_i$ to basal levels, eNOS may once again interact with caveolin and is then re-targeted to the caveolae, the process being accelerated (or stabilized) by enzyme palmitoylation [14]. The long-chain fatty acyl CoA synthetase was recently identified as a key modulator of eNOS re-palmitoylation [20]. Re-association of eNOS with caveolin could occur either at the plasma (caveolar) or perinuclear membrane levels, or



Fig. 11.3 The caveolin-eNOS regulatory cycle (see text for details). NOSIP = eNOS interacting protein; NOSTRIN = eNOS traffic inducer.

even in the cytosol through which caveolin complexes may shuttle between the caveolae and an internalized caveolar vesicle/trans-Golgi network (see Fig. 11.3).

The lag time between the agonist-induced disruption of the caveolin-eNOS interaction and the heterocomplex re-formation is thought to be nonexclusively associated with eNOS activation. Indeed, the arginine transporter CAT1 being located in the caveolae [21], the dissociation of the enzyme from its proximity as well as from several receptors/effectors localized in the caveolae [22,23], is likely to serve as a feedback mechanism for eNOS activation. Also, because NO activates molecular targets outside the endothelial cell, it seems likely that the intracellular locale of eNOS could affect the signaling roles of its product (e.g., the paracrine effects of NO), and thereby modulate the response to extracellular signals. NO finds, indeed, most of its targets in the proximal myocyte layers or circulating blood cells such as platelets and red cells. Similarly, it appears logical that the luminal surface of the endothelium, which is directly exposed to the blood flow and therefore expected to be sensitive to hemodynamic forces, would be a primary site for the documented flow-responsive eNOS activity.

The group of Müller-Esterl has identified, by using yeast two-hybrid screening, two new proteins named NOSIP ("eNOS interacting protein") and NOSTRIN ("eNOS traffic inducer") that specifically bind to the human eNOS oxygenase do-

238 11 Caveolae and the Endothelial Nitric Oxide Synthase

main [24,25] (see Fig. 11.3). Although structurally unrelated, overexpression of both proteins in eNOS-expressing cells has similar effects: net dissociation of eNOS from the plasma membrane and inhibition of agonist-induced NO production. Some observations indicate that both effects are very likely to be related (e.g., NOSIP and NOSTRIN modulate eNOS enzyme activity by uncoupling eNOS from plasma membrane caveolae). For instance, NOSIP overexpression does not impact on eNOS activity measured *in vitro* using a citrulline assay. Also, eNOS binding sites for caveolin and NOSIP do overlap. Nonetheless, whether NOSTRIN and NOSIP promote eNOS translocation from the plasma membrane or inhibit the reverse transport (usually observed after prolonged stimulation) remains unknown.

Of note, although NOSIP and NOSTRIN share some of their functional features and are both particularly abundant in vascularized organs, they differ in other respects [24,25]. Accordingly, while NOSIP overexpression induces eNOS translocation to intracellular sites that co-localize with Golgi and cytoskeletal marker proteins (β -COP and tubulin, respectively), NOSTRIN overexpression largely targets the enzyme to vesicle-like structures spread all over the cytosol. Also, in endothelial cells, NOSIP is mostly found in the cytosol and the nucleus, whereas NOS-TRIN is found exclusively in extranuclear locations and at the plasma membrane; in this context, a positive effect of NOSTRIN to address eNOS to the plasma membrane (caveolae) cannot be excluded.

Importantly, although this chapter is focused on the caveolin-related mode of regulation of eNOS activity, it should be emphasized that besides the proteinprotein interactions detailed above, eNOS is also regulated by phosphorylations [26]. Among the putative phosphorylation sites within the eNOS sequence, two (Ser1177 and Thr495) are currently considered to be critical for eNOS activation. Ser1177 phosphorylation is proposed to improve electron flux through the enzyme and to increase its affinity for calmodulin, whereas dephosphorylation of Thr495 is thought to suppress the steric inhibition for calmodulin association to its binding site [26] (Fig. 11.3). Accordingly, the phosphorylation of eNOS on Ser1177 has been extensively documented as a major kinase-dependent mode of eNOS activation, whereas Thr495 has been reported to participate, when phosphorylated, in the tonic inhibition of eNOS activity. In addition, by directly examining changes in native eNOS post-translational regulation, we recently found that the scaffolding function of Hsp90 previously identified for Akt [27,28] also applied to the phosphatase calcineurin [29,30]. These Hsp90-driven protein-protein associations provide an explanation for the reciprocal regulation of eNOS on distant phosphorylation sites (e.g., Akt-dependent Ser1177 phosphorylation and calcineurin-mediated Thr495 dephosphorylation).

The regulation of eNOS activity/trafficking is very likely to result from the combination of its dynamic interaction with the different partners identified to date (including caveolin, NOSIP, NOSTRIN and Hsp90) and the phosphorylation pattern of the enzyme. In the following sections, the caveolin-eNOS interaction will be retained as the main thread, and reference will be made to other modes of regulation when necessary.

11.4 Lipoproteins and Caveolin-eNOS Interaction

The observations by Fielding et al. that, in human fibroblasts, high levels of cellular free cholesterol (FC) induce caveolin gene transcription [31,32] led us to examine whether a similar increase in caveolin abundance in endothelial cells could account for a reduction in NO production. Accordingly, we exposed endothelial cells to low-density lipoprotein (LDL)-cholesterol and found that both caveolin abundance and caveolin-eNOS complex formation were increased [33] (Fig. 11.4). Furthermore, NO release (under basal and stimulated conditions) was inhibited, thereby providing some insights for a new pathogenic mechanism linking hypercholesterolemia and endothelial dysfunction. A defect in the NO pathway is, indeed, viewed as a hallmark of endothelial dysfunction characterized by an impaired endothelium-dependent vasodilation and the unopposed influence of thrombogenic and proliferative factors on the vessel wall.

In parallel to these observations that cholesterol could lead to eNOS inhibition through the induction of caveolin expression, Blair et al. [34] documented that



Fig. 11.4 Reversible lipoprotein-dependent regulation of caveolin-eNOS interaction. High levels of native LDL stimulate caveolin transcription and thereby promote the interaction of eNOS with the increased caveolin pool (left). The resulting inhibition of eNOS activity can be reversed by statins that, by inhibiting the endogenous synthesis of cholesterol in endothelial cells (as well as indirectly by reducing circulatory LDL-cholesterol), negatively impact on caveolin expression. The interaction of oxidized LDL (oxLDL) with the CD36 receptor located in caveolae leads to a marked depletion of caveolar cholesterol and to the translocation of caveolin and eNOS to intracellular compartments (see right), wherein basal and agonist-stimulated NO production are dramatically reduced. Conversely, HDL, through caveolar SRBI binding, provisions cholesterol esters to the cell, thereby reversing the deleterious effects of oxLDL.

240 11 Caveolae and the Endothelial Nitric Oxide Synthase

oxidized LDL (oxLDL) caused the translocation of both eNOS and caveolin from caveolae to intracellular membranes (Fig. 11.4). These authors further documented that oxLDL, through class B CD36 receptor binding [35], act as acceptors of cholesterol, leading to marked depletion of caveolae cholesterol and redistribution of caveolin and eNOS (but not of other caveolar residents such as PKC α and gang-lioside GM1). They also showed that when examining the pattern of eNOS activation upon exposure to acetylcholine, the dose-response curve was shifted to the right by 100-fold. A recent study by the Lisanti's group also documented that the loss of caveolin-1 (as observed in caveolin knockout mice) resulted in a dramatic down-regulation of CD36 and, importantly, conferred a significant protection against atherosclerosis in double apoE/caveolin knockout mice [36]. This latter study confirmed the findings of Kincer et al., who showed that in apoE/CD36 double knockout mice – in contrast to apoE knockout mice – the acetylcholine-evoked reduction in blood pressure is conserved (as well as the eNOS localization to caveolae) [37].

Whether such processes of native and oxLDL-dependent regulation of the caveolin-eNOS interaction are reversible was also addressed. Our group showed, both *in vitro* and *in vivo*, that hydroxymethylglutaryl coenzyme A (HMGCoA) reductase inhibitors (statins) could reduce caveolin abundance [38,39]. In cultured endothelial cells, the reduction in caveolin abundance obtained with atorvastatin was associated with a restoration of basal and agonist-stimulated eNOS activity (see Fig. 11.4) [38]. In dyslipidemic, apolipoprotein (apo) $E^{-/-}$ mice, the alterations in heart rate and blood pressure variabilities were corrected by chronic treatment with rosuvastatin [39]. Our findings also highlighted the therapeutic potential of statins in diseases other than hypercholesterolemia, such as hypertension or heart failure. Indeed, we showed that statins could decrease caveolin abundance in endothelial cells and in apo $E^{-/-}$ mice, independently of the extracellular and plasma load in LDL-cholesterol, respectively [38,39].

As for the oxLDL-mediated CD36-dependent alteration in caveolin-eNOS biology, the reversibility of the phenomenon was documented with high-density lipoprotein (HDL) that prevented both the depletion of caveolar cholesterol and the eNOS displacement from caveolae [37]. Amazingly, the provision of cholesterol esters by HDL binding to the scavenger receptor BI (and not the inhibition of cholesterol transfer from caveolae to oxLDL) was found to account for the correction of eNOS mislocalization and the restoration of the acetylcholine-induced activation of eNOS [40] (see Fig. 11.4). Of note, the eNOS stimulation by HDL was recently shown to involve Src-mediated signaling [41].

The conclusion to be drawn from the above studies is that both native LDL in excess and oxLDL contribute to the change in NO production, and therefore account for the many functional defects associated with NO deficiency. Although the oxidative stress paradigm is well established as a trigger of the atherosclerotic process, endothelial dysfunction occurs before the appearance of any ultrastructural change in the vessel wall. It may therefore be postulated that chronologically, chronic elevations in serum LDL-cholesterol and then lipoprotein oxidation contribute to caveolin-dependent alteration in NO signaling. Importantly, both phenomena appear reversible. Physiologically, the beneficial effects of HDL are clearly emphasized by several studies described herein. Moreover, the pharmacologically pleiotropic effects of statins can be expected (interestingly) at doses that do not necessarily require any reduction in LDL correction.

11.5 Angiogenesis and Caveolin-eNOS Interaction

In 1999, Lisanti and colleagues documented that caveolin-1 was down-regulated by angiogenic growth factors in subconfluent endothelial cells, but not when these cells were confluent [42]. These authors further showed that, in contrast, caveolin-1 protein levels were up-regulated during endothelial cell differentiation and that expression of caveolin-1 in confluent endothelial cells stimulated endothelial tube formation [43]. Although, the original observation that growth factor exposure can down-regulate caveolin expression in endothelial cells remains a matter of debate, the second set of data (e.g., the role of caveolin in endothelial differentiation/tube reorganization) was verified by other investigators (but only for the link between a reduction in caveolin abundance and the inhibition of angiogenesis). Indeed, Griffoni et al. found, using caveolin antisense technology, that a reduction in caveolin abundance reduced vessel formation in the chorioallantoic membrane assay [44].

Consecutively, we found that statins could stimulate tube formation from macrovascular endothelial cells cultured on Matrigel® through a decrease in caveolin abundance (and in its inhibitory interaction with eNOS) [27]. In microvascular endothelial cells, statins only marginally decreased the abundance of caveolin protein (that amounted to almost 10-fold the pool of caveolin in macrovascular endothelial cells), and therefore did not impact on the regulation of eNOS activity. In those endothelial cells, however, statins were shown to stimulate angiogenesis through eNOS phosphorylation on Ser1177 – a process which is facilitated by the recruitment of the chaperone protein hsp90 that acts as an adaptor between eNOS and the kinase. Still, caveolin overexpression or cell loading with caveolin scaffolding domain-derived peptides prevented the ability not only of statins but also of vascular endothelial growth factor (VEGF) to stimulate in-vitro angiogenesis (Fig. 11.5). These observations were further confirmed in a mouse model of angiogenesis (in tumors). In-vivo lipofection of a caveolin plasmid led, indeed, to the inhibition of both NO-mediated vasodilation and angiogenesis in tumors [45] (Fig. 11.5).

Paradoxically, we found that in aortic endothelial cells isolated from caveolin-1-deficient mice $(Cav^{-/-})$ (as well as $Cav^{+/-}$ mice), VEGF-induced NO production and endothelial tube formation were dramatically decreased when compared with $Cav^{+/+}$ endothelial cells (see Fig. 11.5). The VEGFR-2 mistargeting (due to the lack of caveolin in $Cav^{-/-}$ mice) was identified as the cause of the incapacity of VEGF to activate the downstream signaling cascades, including eNOS and ERK activation. This led to dramatic consequences in a model of adaptive angiogenesis obtained after femoral artery resection [46]. In fact, contrary to $Cav^{+/+}$ mice, both $Cav^{-/-}$ and 242 11 Caveolae and the Endothelial Nitric Oxide Synthase



Caveolin abundance

Fig. 11.5 Model of the regulation of angiogenesis by caveolin-eNOS interaction. Experimentally based relationship between the abundance of caveolin and the NO-dependent angiogenesis. Note the position of the "normal" or "physiological" (100% expression) caveolin phenotype within the bell-shaped pattern, emphasizing the antiangiogenic effects of increasing caveolin

abundance by recombinant (rec.) caveolin expression, caveolin scaffolding domain (CSD) peptides or LDL-cholesterol exposure as well as the pro-angiogenic effects of reducing caveolin abundance (upon statins) until a given threshold from which anti-angiogenic effects will be observed (similar to those obtained with caveolin overexpression).

Cav^{+/-} mice failed to recover a functional vasculature, as authenticated by laser Doppler evaluation of the ischemic tissue perfusion and histochemical analyses. These data recapitulate the findings of Woodman et al., who found in Cav^{-/-} mice a dramatic reduction in both vessel infiltration and density in tumor models of angiogenesis [47]. Interestingly, in aortic endothelial cells isolated from Cav^{-/-} mice, we further documented that recombinant caveolin expression in endothelial cells helped to redirect the VEGFR-2 in caveolar membranes and to restore the VEGF/NO and VEGF/ERK signaling cascades. Amazingly, however, when (too-) elevated levels of recombinant caveolin were reached, VEGF exposure failed to activate ERK and eNOS [46], these findings being in good agreement with the experiments of caveolin transfection described above [45] (Fig. 11.5).

A model integrating the "compartmentalizing" effect of caveolin (e.g., receptoreffector coupling is either prevented or promoted when/where caveolin is down- or up-regulated) and the "inhibitory" hypothesis (e.g., inhibition proportional to caveolin levels) is described in Figure 11.5.

11.6

Vasodilation, Endothelial Permeability and Caveolin-eNOS Interaction

The role of caveolae in endothelium-dependent and NO-mediated vascular relaxation was documented in the original papers reporting the phenotype of caveolindeficient mice by the groups of Kurzchalia and Lisanti [48,49]. These authors have, indeed, evaluated the NO-mediated vasorelaxing effects of acetylcholine on aortic rings precontracted with phenylephrine (an α 1-adrenergic vasoconstrictor). Both groups reported a significantly greater relaxation in Cav-1 null aortic rings at all acetylcholine concentrations examined. Drab et al. further documented that, in primary culture of aortic vascular smooth muscle cells, the basal release of NO was one-third higher than in wild-type cells and the content of cyclic guanosine monophosphate (cGMP) was about three-fold higher in knockout animals [48]. Razzani et al. also found that, in the presence of the NOS inhibitor L-NAME, the increase in contractile response to phenylephrine was significantly greater in the Cav-1 null mice [49].

That eNOS becomes hyperactivated in the absence of caveolin-1 formed the basis for further studies exploring this paradigm in a variety of biological contexts, including agonist- and shear stress-induced vasoresponse as well as disease states. For example, Omura et al. reported that eicosapentaenoic acid stimulated NO production and the associated endothelium-dependent relaxation through stimulation of the dissociation of the caveolin-eNOS complex [50]. Similarly, increasing vascular flow (which is by far the main in-vivo trigger for vasodilation) was shown rapidly to activate caveolar eNOS by inducing calmodulin-dependent eNOS dissociation from caveolin [51]. By contrast, in a model of portal hypertension, Shah et al. [52] found that caveolin expression was significantly increased in liver sinusoids and venules, thereby leading to a significant reduction in NO production. An abnormal tight coupling between eNOS and caveolin was also identified by Murata et al. in a rat model of hypoxia-induced pulmonary hypertension (another disease state which is in part due to impaired bioactivity of vascular NO) [53]. These authors documented that, in the hypoxic pulmonary artery, the increased caveolineNOS interaction accounted for the impaired eNOS activity in either the presence or absence of carbachol. In yet another study, Pelligrino et al. attributed the defect in acetylcholine-induced vasodilation of ovariectomized rat pial arteries to caveolin. More exactly, they found that the endothelial dysfunction observed in these operated rats could be reversed by estradiol treatment through a reduction in endothelial caveolin-1 expression [54]. Finally, using isolated tumor arterioles mounted on a pressure myograph, we documented that local tumor irradiation induced NOmediated vasorelaxation through not only an increase in the abundance of eNOS but also a decrease in caveolin-1 expression [55].

Besides vasodilation, caveolin-eNOS interaction has also been proposed to impact on vascular permeability. Caldwell et al. initially reported that VEGF increased endothelial cell permeability by an eNOS-dependent mechanism of transcytosis in caveolae and also, interestingly, that VEGF-R2 and eNOS co-localized with caveolin-1 in plasma membrane caveolae in retinal microvascular endothelial cells [56]. The same authors then documented that VEGF stimulated the translocation of eNOS, caveolin-1 and the VEGF receptor into the nucleus [57]. In the context of inflammation and tumor angiogenesis, Sessa's group also identified a role for caveolin in endothelial cell permeability. Accordingly, Bucci et al. used two experimental inflammatory models (subplantar administration of carrageenan and ear application of mustard oil) to examine the effects of the CSD on induced

244 11 Caveolae and the Endothelial Nitric Oxide Synthase

permeability [58]. These authors showed that the systemic administration of CSDderived peptides fused with a cellular internalization sequence (derived from the Antennapedia homeodomain) suppressed acute inflammation and vascular leak to the same extent as the NOS inhibitor L-NAME. Gratton et al. showed that the same fusion peptide inhibited eNOS-dependent vascular leakage in tumors and consecutively delayed tumor progression in mice [59]; extravasation of plasma proteins is, indeed, known to contribute to the formation of a provisional matrix required to initiate neoangiogenesis.

Taken together, these data emphasize the key role of caveolin and caveolae in regulating smooth muscle relaxation and endothelial permeability. The results of several studies have also suggested that activated eNOS, as observed in inflammation, tumors or hypoxia (versus healthy tissues), is particularly sensitive to alterations in the caveolin pool, thereby opening new perspectives of treatments (using, for example, CSD-derived peptides).

11.7 Caveolin-3-eNOS Interaction in Cardiac Myocytes

To explore the dual roles of eNOS caveolar targeting in cardiac myocytes (e.g., compartmentation to facilitate activation upon agonist stimulation and inhibition of the eNOS catalytic activity in basal conditions), we refer to the paradigm of the muscarinic cholinergic NO-mediated regulation of heart rate. eNOS activated by muscarinic cholinergic agonists contributes to the so-called accentuated antagonism – that is, the ability of muscarinic cholinergic stimulation to attenuate βadrenergic signaling in various models [4]. In order to examine the impact of eNOS compartmentation in caveolae of cardiac myocytes, we first used neonatal myocytes isolated from eNOS-deficient mouse, that were transfected with cDNA constructs encoding either the wild-type eNOS or a myristoylation-deficient eNOS mutant [23] (Fig. 11.6). These knock-in experiments provided us with myocytes expressing eNOS protein in either the caveolae or in the cytosolic compartment. In myocytes expressing wild-type eNOS, a muscarinic cholinergic agonist dramatically reduced the spontaneous heart beat rate (in a cGMP-dependent manner), whereas in the myr-eNOS myocytes the agonist failed to exert its negative chronotropic effect (Fig. 11.6). The second arm of the caveolae regulation (i.e., that caveolin-3 exerts inhibitory effects on myocyte eNOS) was demonstrated by documenting the blockade of muscarinic cholinergic agonist-induced negative chronotropic effects in cardiac myocytes loaded with caveolin-3 scaffolding domain-derived peptides.

Interestingly, a large fraction of sarcolemmal m2 muscarinic Cholinergic receptor (mAchR) was found to be targeted to cardiac myocyte caveolae upon agonist stimulation [60], thereby reinforcing the role of compartmentation in regulating NO signaling in myocyte. In a further study, we examined the impact of this translocation on the mAchR internalization process and the consecutive alteration in downstream NO signaling [22]. It was found that mAChR stimulation led to the



Fig. 11.6 The key roles of caveolin-3 and caveolae in the muscarinic cholinergic (mAchR) regulation of heart rate. Top left: Under basal conditions, caveolin-3 maintains eNOS in its inactivated state and thereby limits the negative chronotropic effects of NO. Top right: The m2 mAchR receptor targeting to caveolae upon agonist stimulation leads to the activation of eNOS through a local increase in intracellular calcium (the sarcoplasmic reticulum is in close vicinity to caveolae) and the consecutive disruption of the caveolin-3-eNOS heterocomplex: the myocyte beating rate is slowed (see representation of heart-beating chart recording).

Bottom: Proof of principle that eNOS compartmentation in caveolae is critical for mAchR signaling in the heart [23]. Left: In cardiac myocytes expressing only recombinant myristoylation-deficient eNOS in the cytosol, the coupling between agonist-bound m2 mAchR receptor and the mutant eNOS is lost, and no effect on the myocyte beating rate can be observed. Right: When caveolin-3-derived peptides (corresponding to the CSD sequence) are introduced in myocytes, eNOS is inactivated by this excess inhibitory clamping, preventing activation of the m2mAchR signaling cascade and the associated negative chronotropism.

sequestration of mAchRs through caveolae fission through a dynamin-dependent GTP-driven process. Repeated stimulations of mAchRs led to a progressive increase in mAchR sequestration (via the detachment of caveolae from myocyte sarcolemma) and a concurrent stabilization of the inhibitory eNOS-caveolin complex. These findings suggested that caveolae fission may contribute to G-protein-coupled receptor desensitization and thereby terminate the (initially facilitated) NO signaling cascade.

11.8 Conclusions

In this chapter we have reviewed the multitude of evidence that the structural protein caveolin-1 and the caveolae themselves are essential for the control of NO production. Although caveolin-deficient mice are viable, there are today clear insights on the deregulation, in these mice, of many biological functions wherein NO is a key mediator or modulator, including angiogenesis [46,47], vasodilation [48,49], vascular permeability [61,62], lipid metabolism [36], and cardiac biology [63,64]. More specifically, we have documented that the caveolar compartmentation of eNOS plays a paradoxical role, both tonically repressing basal eNOS activity by the enzyme's interactions with caveolin, and also ensuring the efficient activation of the enzyme upon agonist stimulation.

No other caveolar protein resident has been so extensively studied as eNOS. Although many studies remain to be performed to understand the relevance of the specific locale of the dozens of molecules proposed to be associated with caveolin, it can be stated that among the responses to the many incoming signals integrated in the caveolar organelles, endothelial NO production appears as one of the major signal emanating from these signaling platforms.

Abbreviations

APT1	acyl-protein thioesterase 1
CaM	calmodulin
CBM	caveolin binding motif
CSD	caveolin scaffolding domain
eNOS	endothelial nitric oxide synthase
FC	free cholesterol
GST	glutathione S-transferase
HDL	high-density lipoprotein
HMGCoA	hexamethylglutaryl coenzyme A
iNOS	inducible NOS isoform
mAchR	muscarinic cholinergic
NO	nitric oxide
NOSIP	eNOS interacting protein
NOSTRIN	eNOS traffic inducer
oxLDL	oxidized low-density lipoprotein
VEGF	vascular endothelial growth factor

References

- T. Michel, O. Feron, J. Clin. Invest. 1997, 100, 2146–2152.
- 2 D.J. Stuehr, J. Santolini, Z. Q. Wang, C.C. Wei, S. Adak, J. Biol. Chem. 2004, 279, 36167–36170.
- **3** L.J. Ignarro, J. Physiol. Pharmacol. **2002**, 53, 503–514.
- 4 P. B. Massion, O. Feron, C. Dessy, J. L. Balligand, Circ. Res. 2003, 93, 388–398.
- 5 H.C. Champion, M.W. Skaf, J.M. Hare, Heart Fail. Rev. 2003, 8, 35–46.
- 6 P.W. Shaul, E.J. Smart, L.J. Robinson, Z. German, I.S. Yuhanna, Y. Ying, R.G. Anderson, T. Michel, J. Biol. Chem. 1996, 271, 6518–6522.
- 7 G. Garcia-Cardena, P. Oh, J. Liu, J. E. Schnitzer, W. C. Sessa, *Proc. Natl. Acad. Sci. USA* **1996**, 93, 6448–6453.
- G. Feron, L. Belhassen, L. Kobzik, T.W. Smith, R.A. Kelly, T. Michel, *J. Biol. Chem.* 1996, 271, 22810–22814.
- 9 O. Feron, J. B. Michel, K. Sase, T. Michel, Biochemistry 1998, 37, 193–200.
- 10 S. Li, J. Couet, M. P. Lisanti, J. Biol. Chem. 1996, 271, 29182–29190.
- 11 J. Couet, S. Li, T. Okamoto, T. Ikezu, M. P. Lisanti, J. Biol. Chem. 1997, 272, 6525–6533.
- 12 G. Garcia-Cardena, P. Martasek, B.S. Masters, P.M. Skidd, J. Couet, S. Li, M. P. Lisanti, W. C. Sessa, *J. Biol. Chem.* 1997, 272, 25437–25440.
- 13 J. B. Michel, O. Feron, K. Sase, P. Prabhakar, T. Michel, J. Biol. Chem. 1997, 272, 25907–25912.
- 14 O. Feron, F. Saldana, J. B. Michel, T. Michel, J. Biol. Chem. 1998, 273, 3125–3128.
- 15 J. B. Michel, O. Feron, D. Sacks, T. Michel, J. Biol. Chem. 1997, 272, 15583–15586.
- 16 H. Ju, R. Zou, V.J. Venema, R.C. Venema, J. Biol. Chem. 1997, 272, 18522–18525.
- 17 S. Ghosh, R. Gachhui, C. Crooks, C. Wu, M. P. Lisanti, D. J. Stuehr, J. Biol. Chem.
 1998, 273, 22267–22271.
- 18 P. Prabhakar, H. S. Thatte, R. M. Goetz, M. R. Cho, D. E. Golan, T. Michel, J. Biol. Chem. 1998, 273, 27383–27388.
- 19 D.C. Yeh, J.A. Duncan, S. Yamashita, T. Michel, J. Biol. Chem. 1999, 274, 33148–33154.

- 20 M. T. Weis, J. L. Crumley, L. H. Young, J. N. Stallone, *Cardiovasc. Res.* 2004, *63*, 338–346.
- 21 K. K. McDonald, S. Zharikov, E. R. Block, M. S. Kilberg, J. Biol. Chem. 1997, 272, 31213–31216.
- 22 C. Dessy, R.A. Kelly, J.L. Balligand, O. Feron, *EMBO J.* 2000, 19, 4272–4280.
- 23 O. Feron, C. Dessy, D. J. Opel, M.A. Arstall, R.A. Kelly, T. Michel, J. Biol. Chem. 1998, 273, 30249–30254.
- 24 J. Dedio, P. Konig, P. Wohlfart, C. Schroeder, W. Kummer, W. Muller-Esterl, FASEB J. 2001, 15, 79–89.
- 25 K. Zimmermann, N. Opitz, J. Dedio, C. Renne, W. Muller-Esterl, S. Oess, *Proc. Natl. Acad. Sci. USA* 2002, 99, 17167–17172.
- 26 I. Fleming, R. Busse, Am. J. Physiol Regul. Integr. Comp Physiol 2003, 284, R1-R12.
- 27 A. Brouet, P. Sonveaux, C. Dessy, S. Moniotte, J. L. Balligand, O. Feron, *Circ. Res.* 2001, 89, 866–873.
- 28 A. Brouet, P. Sonveaux, C. Dessy, J. L. Balligand, O. Feron, J. Biol. Chem. 2001, 276, 32663–32669.
- 29 C. Kupatt, C. Dessy, R. Hinkel, P. Raake, G. Daneau, C. Bouzin, P. Boekstegers, O. Feron, Arterioscler. Thromb. Vasc. Biol. 2004, 24, 1435–1441.
- 30 R. Rezzani, L. Rodella, C. Dessy, G. Daneau, R. Bianchi, O. Feron, *FEBS Lett.* 2003, 552, 125–129.
- 31 A. Bist, P. E. Fielding, C. J. Fielding, Proc. Natl. Acad. Sci. USA 1997, 94, 10693–10698.
- 32 C. J. Fielding, A. Bist, P. E. Fielding, Proc. Natl. Acad. Sci. USA 1997, 94, 3753– 3758.
- 33 O. Feron, C. Dessy, S. Moniotte, J. P. Desager, J. L. Balligand, J. Clin. Invest. 1999, 103, 897–905.
- 34 A. Blair, P. W. Shaul, I. S. Yuhanna, P. A. Conrad, E. J. Smart, J. Biol. Chem. 1999, 274, 32512–32519.
- 35 A. Uittenbogaard, P.W. Shaul, I.S. Yuhanna, A. Blair, E.J. Smart, *J. Biol. Chem.* 2000, 275, 11278–11283.
- 36 P.G. Frank, S.E. Woodman, D.S. Park, M.P. Lisanti, Arterioscler. Thromb. Vasc. Biol. 2003, 23, 1161–1168.

- 248 References
 - 37 J. F. Kincer, A. Uittenbogaard, J. Dressman, T.M. Guerin, M. Febbraio, L. Guo, E. J. Smart, *J. Biol. Chem.* 2002, 277, 23525–23533.
 - 38 O. Feron, C. Dessy, J. P. Desager, J. L. Balligand, *Circulation* 2001, 103, 113–118.
 - 39 M. Pelat, C. Dessy, P. Massion, J. P. Desager, O. Feron, J. L. Balligand, *Circulation* 2003, 107, 2480–2486.
 - 40 I.S. Yuhanna, Y. Zhu, B.E. Cox, L.D. Hahner, S. Osborne-Lawrence, P. Lu, Y.L. Marcel, R.G. Anderson, M.E. Mendelsohn, H. H. Hobbs, P.W. Shaul, *Nat. Med.* 2001, 7, 853–857.
 - 41 C. Mineo, I. S. Yuhanna, M. J. Quon, P. W. Shaul, J. Biol. Chem. 2003, 278, 9142–9149.
 - 42 J. Liu, B. Razani, S. Tang, B. I. Terman, J. A. Ware, M. P. Lisanti, J. Biol. Chem.
 1999, 274, 15781–15785.
 - 43 J. Liu, X. B. Wang, D. S. Park, M. P. Lisanti, J. Biol. Chem. 2002, 277, 10661–10668.
 - 44 C. Griffoni, E. Spisni, S. Santi, M. Riccio, T. Guarnieri, V. Tomasi, *Biochem. Biophys. Res. Commun.* 2000, 276, 756–761.
 - **45** A. Brouet, J. DeWever, P. Martinive, X. Havaux, C. Bouzin, P. Sonveaux, O. Feron, *FASEB J.* **2005**, *19*, 602–604.
 - 46 P. Sonveaux, P. Martinive, J. DeWever, Z. Batova, G. Daneau, M. Pelat, P. Ghisdal, V. Gregoire, C. Dessy, J. L. Balligand, O. Feron, *Circ. Res.* 2004, 95, 154–161.
 - 47 S. E. Woodman, A. W. Ashton, W. Schubert, H. Lee, T. M. Williams, F. A. Medina, J. B. Wyckoff, T. P. Combs, M. P. Lisanti, *Am. J. Pathol.* 2003, 162, 2059–2068.
 - 48 M. Drab, P. Verkade, M. Elger, M. Kasper, M. Lohn, B. Lauterbach, J. Menne, C. Lindschau, F. Mende, F. C. Luft, A. Schedl, H. Haller, T. V. Kurzchalia, *Science* 2001, 293, 2449–2452.
 - 49 B. Razani, J.A. Engelman, X. B. Wang, W. Schubert, X. L. Zhang, C. B. Marks, F. Macaluso, R.G. Russell, M. Li, R.G. Pestell, D. Di Vizio, H. Hou, Jr., B. Kneitz, G. Lagaud, G.J. Christ, W. Edelmann, M. P. Lisanti, J. Biol. Chem. 2001, 276, 38121–38138.
 - 50 M. Omura, S. Kobayashi, Y. Mizukami, K. Mogami, N. Todoroki-Ikeda, T. Miyake, M. Matsuzaki, *FEBS Lett.* 2001, 487, 361–366.

- 51 V. Rizzo, D. P. McIntosh, P. Oh, J. E. Schnitzer, J. Biol. Chem. 1998, 273, 34724–34729.
- 52 V. Shah, S. Cao, H. Hendrickson, J. Yao, Z. S. Katusic, Am. J. Physiol. Gastrointest. Liver Physiol. 2001, 280, G1209–G1216.
- 53 T. Murata, K. Sato, M. Hori, H. Ozaki, H. Karaki, J. Biol. Chem. 2002, 277, 44085–44092.
- 54 D.A. Pelligrino, S. Ye, F. Tan, R.A. Santizo, D. L. Feinstein, Q. Wang, *Biochem. Biophys. Res. Commun.* 2000, 269, 165–171.
- 55 P. Sonveaux, C. Dessy, A. Brouet, B. F. Jordan, V. Gregoire, B. Gallez, J. L. Balligand, O. Feron, *FASEB J.* 2002, *16*, 1979–1981.
- 56 Y. Feng, V.J. Venema, R.C. Venema, N. Tsai, M.A. Behzadian, R.B. Caldwell, *Invest. Ophthalmol. Vis. Sci.* 1999, 40, 157–167.
- 57 Y. Feng, V. J. Venema, R. C. Venema, N. Tsai, R. B. Caldwell, *Biochem. Biophys. Res. Commun.* **1999**, 256, 192–197.
- 58 M. Bucci, J. P. Gratton, R. D. Rudic, L. Acevedo, F. Roviezzo, G. Cirino, W. C. Sessa, *Nat. Med.* 2000, 6, 1362–1367.
- 59 J.P. Gratton, M.I. Lin, J. Yu, E. D. Weiss, Z. L. Jiang, T.A. Fairchild, Y. Iwakiri, R. Groszmann, K.P. Claffey, Y.C. Cheng, W.C. Sessa, *Cancer Cell* 2003, 4, 31–39.
- 60 O. Feron, T.W. Smith, T. Michel, R.A. Kelly, J. Biol. Chem. 1997, 272, 17744–17748.
- 61 W. Schubert, P. G. Frank, S. E. Woodman, H. Hyogo, D. E. Cohen, C. W. Chow, M. P. Lisanti, J. Biol. Chem. 2002, 277, 40091–40098.
- 62 W. Schubert, P. G. Frank, B. Razani, D. S. Park, C. W. Chow, M. P. Lisanti, J. Biol. Chem. 2001, 276, 48619–48622.
- 63 A. W. Cohen, D. S. Park, S. E. Woodman, T. M. Williams, M. Chandra, J. Shirani, D. S. Pereira, R. N. Kitsis, R. G. Russell, L. M. Weiss, B. Tang, L. A. Jelicks, S. M. Factor, V. Shtutin, H. B. Tanowitz, M. P. Lisanti, Am. J. Physiol. Cell Physiol. 2003, 284, C457–C474.
- 64 Y.Y. Zhao, Y. Liu, R.V. Stan, L. Fan, Y. Gu, N. Dalton, P. H. Chu, K. Peterson, J. Ross, Jr., K. R. Chien, *Proc. Natl. Acad. Sci. USA* 2002, 99, 11375–11380.

12 The Role of Caveolin-1 in Tumor Cell Survival and Cancer Progression

Dana Ravid and Mordechai Liscovitch

12.1 Introduction

Multifunctional adaptor proteins may have divergent actions in early versus late phases of cancer progression, playing both tumor-suppressing and tumor-promoting roles in a context-dependent manner. One such protein is the adaptor and caveolar coat protein caveolin-1, which is an essential structural constituent of plasma membrane caveolae (for a review, see [1,2]). The cellular localization of caveolin-1 is not limited to caveolae in all cells [3], and it exhibits diverse functions that go beyond its well-characterized role as a component of the caveolar coat (for a review, see [4]). Caveolin-1 was implicated in tumorigenesis because its expression is suppressed by oncogenic transformation, and little or no expression of caveolin-1 was initially found in various human tumors and cancer cell lines [5]. In addition, caveolin-1 has well-established growth-inhibitory properties, and it has been suggested to act as a tumor-suppressor protein [5]. However, a general tumor suppressor action cannot easily be reconciled with the fact that, in many other cancer cell lines and tumor specimens, the expression of caveolin-1 is high [6]. Furthermore, a positive correlation was often noted between the expression of caveolin-1 and the tumor cell grade and progression stage; in certain studies, the expression of caveolin-1 could be used to independently predict poor disease prognosis [6].

How can the divergent, cell type- and tumor stage-dependent changes in caveolin-1 be explained? Previous studies have indicated that in addition to its growthinhibitory action, under certain circumstances caveolin-1 acts as a survival-promoting protein [7–14]. The differential expression of caveolin-1 in various tumor cells and specimens may thus be explained by the ability of caveolin-1 to exert both antiproliferative and pro-survival effects. It was previously hypothesized that, during the early stages of cancer progression (when rapid proliferation is essential for clonal expansion), expression of caveolin-1 is down-regulated, thus suppressing its growth-inhibitory actions. Conversely, during the late, advanced stages of the disease (when survival and stress resistance are paramount), expression of caveolin-1 is up-regulated and it plays a pro-survival role [6,15]. In the following sections we

250 12 The Role of Caveolin-1 in Tumor Cell Survival and Cancer Progression

shall re-evaluate this hypothesis in view of recent evidence that relates the expression and function(s) of caveolin-1 to cancer progression and tumor cell survival.

12.2

The Caveolin-1 Gene and its Regulation During Differentiation and Transformation

Caveolin-1, which was first identified as a ~22-kDa, tyrosine-phosphorylated protein in Rous sarcoma virus-transformed cells [16], was later found to be an essential constituent of caveolae [17–19]. The caveolar membrane system mediates certain transport processes, including transcytosis, potocytosis, and clathrin-independent endocytosis [20]. Caveolin-1 binds sphingolipids and cholesterol – lipids that are characteristic constituents of lipid rafts [4]. Caveolin-1 and caveolae are also involved in mediating cellular cholesterol efflux [19,21].

Caveolin-1 is a member of a gene family that also comprises caveolin-2 and caveolin-3. Caveolin-2 is co-expressed with caveolin-1 in many cell types, including mesenchymal, endothelial, epithelial, neuronal, and glial cells [22,23]. Together, caveolin-1 and caveolin-2 form hetero-oligomeric assemblies that constitute the filamentous caveolar coat [23,24]. Caveolin-3 is selectively expressed in skeletal and heart muscle cells, where it appears to substitute functionally for caveolin-1 [25].

Caveolin-1 interacts with numerous proteins via a caveolin "scaffolding" domain (CSD) that binds short-sequence motifs that are rich in aromatic amino acids [1,26,27]. The ability of caveolin-1 to interact with many raft-localized signaling proteins has indicated that it participates in signal transduction, and that its expression may have a profound effect on cell function and fate. Indeed, the expression of caveolin-1 is altered dynamically under different physiological conditions, clearly implicating it as a regulator of cell growth and survival.

The induction of differentiation up-regulates caveolin-1 in various cell types [3,28–32]. Up-regulation of caveolin-1 is also observed upon acquisition of cell senescence [33,34]. In contrast, caveolin-1 is down-regulated upon transformation of fibroblasts by oncogenes such as Bcr-Abl, v-Abl, H-Ras, Polyoma virus middle T and Crk1 [35], or Neu-T, c-Src-Y52F and Myc [36]. These data accorded well with the many growth-inhibitory effects of caveolin-1 (for reviews, see [5,6]) and its human chromosomal location near a locus (7q31.1/D7S522) that is deleted in several forms of cancers [37,38]. In addition, genetic knockout of caveolin-1 leads to hyperplasia of pulmonary endothelial cells and mammary gland epithelial cells [39,40] and results in increased sensitivity to oncogenic and carcinogenic stimuli [41,42]. Together, these data have led to the suggestion that caveolin-1 may act as a tumor-suppressor protein [5].

However, as discussed below, this simple hypothesis is unable to explain the complex pattern of caveolin-1 expression in human tumors and the full range of its actions in cancer cells.

12.3 Divergent Expression of Caveolin-1 in Human Cancer: The Case of Lung Cancer

The expression of caveolin-1 in different cancer cell lines and tumor samples was documented in numerous studies, and the picture that emerges from these studies is that caveolin-1 expression is highly divergent [6]. Whereas in many cases caveolin-1 expression is down-regulated, in other cancer cells elevated caveolin-1 levels are maintained. High expression of caveolin-1 in cancer cells was initially demonstrated in human multidrug resistant (MDR) cancer cells [43–45] and in mouse metastatic prostate cancer cells [46]. Evidence that has accumulated since these early studies seems to indicate that the expression of caveolin-1 often depends on the tumor's stage and grade. In fact, in most cases where this relationship was examined explicitly, a positive correlation was found between a high expression of caveolin-1 and advanced tumor cell grade and/or progression stage. Such was the case in cancers from the breast [46], prostate [46,47], lung [48,49], bladder [50], kidney [51,52] and pancreas [53,54]. Furthermore, in some cases, high expression of caveolin-1 was shown to be an independent predictor of poor disease prognosis [47,49,51–53,55–58].

The divergent pattern of caveolin-1 in human cancer is well represented in studies of lung cancer tumors and cell lines. Human lung adenocarcinoma- and small cell carcinoma-derived cell lines exhibited little or no caveolin-1 expression, in contrast to high levels of caveolin-1 expression observed in CaLu-1 lung squamous carcinoma cells [59]. However, when compared with normal human lung epithelia, four other lung squamous cell carcinoma lines expressed reduced levels of caveolin-1 [60]. In another study, low-level caveolin-1 expression was similarly found in cell lines derived from two lung adenocarcinomas, a bronchioalveolar carcinoma and a large-cell lung carcinoma, whereas non-small-cell lung carcinoma cell lines (Hop-62, Hop-92) highly expressed caveolin-1 [61]. More recently, it was found that about 95% of small-cell lung cancer (SCLC) cell lines exhibit low or no caveolin-1 expression, whereas a majority (76%) of non-SCLC lines retained high caveolin-1 expression [62].

The loss of caveolin-1 expression in SCLC and in primary non-SCLC was verified immunohistochemically in tumor samples [63,64]. Caveolin-1 expression was reported to be positively correlated with metastatic potential in a series of lung adenocarcinoma cell lines established by selection for increasing invasiveness [48]. The same study found that primary lung adenocarcinoma tumors are largely caveolin-1-negative, but there was a significant trend of increased caveolin-1 expression in metastatic lung tumors and in their lymph nodes metastases [48]. Caveolin-1 was prominently down-regulated in primary lung adenocarcinomas, as revealed by DNA microarray analysis [65,66].

Nevertheless, ~25% of lung adenocarcinomas still expressed high levels of caveolin-1 protein, although this study found no correlation of caveolin-1 expression with tumor stage or lymph node status [63]. Other analyses of primary lung squamous cell carcinoma specimens revealed that 26–30% were caveolin-1-positive [49,67]. In these studies there was a significant correlation between caveolin-1 expression and advanced pathologic stage [49,67].

252 12 The Role of Caveolin-1 in Tumor Cell Survival and Cancer Progression

Furthermore, there was a statistically significant decrease in five-year survival after complete resection of patients with caveolin-1-positive tumors [49]. Together, the above data are consistent with a suppression of caveolin-1 gene expression in primary lung tumors (in particular, SCLC). However, some lung tumors and lung cancer cell lines are caveolin-1-positive (notably non-small cell lung carcinoma), and there seems to be a positive correlation between caveolin-1 expression and advanced cancer pathologic stage or metastatic potential. Thus, the status of caveolin-1 in lung cancer is as complex as it is in other forms of cancer.

12.4

Actions of Caveolin-1 in Cancer Cells: Effects of Heterologous Expression and Genetic or Functional Suppression

12.4.1

Anti-Proliferative Activity of Caveolin-1

Heterologous expression of caveolin-1 in T47-D human mammary cancer cells results in a 50% decrease in growth rate and a three- to 10-fold reduction in anchorage-independent growth [68]. Inhibition of in-vitro anchorage-independent growth, a parameter that is highly correlated with *in-vivo* tumorigenesis [69], was similarly seen in MCF-7 human breast adenocarcinoma cells [11] and SCLC cells [62]. Caveolin-1 may therefore block a matrix-independent, intrinsic growth signal (e.g., a signal that emanates from an activated oncogene). Indeed, transient transfection with caveolin-1 reduces growth rates in human mammary tumor cells [11,68] and human ovarian carcinoma cells [70]. Accordingly, a mutant caveolin-1 (P132L), identified in about 16% of primary human breast cancer specimens [71], induces morphological transformation in NIH-3T3 cells and supports anchorage-independent growth of mutant-transfected cells [71]. P132L appears to act by causing missorting of normal caveolin-1, leading to its retention at a perinuclear compartment that is probably the Golgi apparatus [72]. It should be noted that the occurrence of the P132L mutation in mammary tumors awaits confirmation in non-Japanese breast cancer patients [73,74]. Further support for its growthinhibitory action was obtained by analysis of heterozygous and homozygous caveolin-1 knockout mice. Retroviral inactivation of one of the caveolin-1 alleles results in loss of ca. 50% of caveolin-1 expression and enables anchorage-independent growth [75]. Likewise, genetic knockout of caveolin-1 results in hyperplasia of pulmonary endothelial cells and mammary gland epithelial cells [39,40]. Although spontaneous development of mammary or other tumors was not evident in caveolin-1-null mice [72], caveolin-1 gene knockout results in increased sensitivity to carcinogenic and oncogenic stimuli [41,42]. Mammary gland tumor-prone MMTV-PyMT mice exhibit accelerated formation of larger and higher-grade dysplastic foci [42]. Long-term mammary tumorigenesis in these mice was doubled and lung metastases were significantly increased [76]. Genetic disruption of caveolin-1 was shown also to cooperate with loss of the tumor suppressor INK4a. This study

showed that double knockout of caveolin-1 and INK4a stimulates proliferation of mouse embryo fibroblasts, greatly increases transformation by an activated oncogene (e.g., H-Ras-G12V; v-Src) and results in dramatic stimulation of tumor growth *in vivo* [77]. Taken together with the heterologous expression data, these results implicate caveolin-1 as an important anti-proliferative protein.

12.4.2

Pro-Apoptotic Activity of Caveolin-1

Caveolin-1 can also acts as a pro-apoptotic protein. This was originally observed in Rat-1 fibroblasts, where overexpression of caveolin 1 sensitizes the cells to both γ irradiation and ceramide-induced cell death [78]. Elevated sensitivity to apoptotic stimuli was also reported in caveolin-1-transfected OVCAR-3 ovarian carcinoma cells [65] and T24 bladder cancer cells [79], and in mouse embryo fibroblasts that transgenically overexpress caveolin-1 [80]. Activation of caspase-3 was reported to occur in three OVCA ovarian cancer cell lines upon expression of caveolin-1 [70]. Supporting these data, caveolin-1 antisense-suppressed NIH-3T3 cells were reported to be more resistant to staurosporine-induced apoptosis [79]. Similarly, retroviral disruption of the caveolin-1 gene in L929 murine fibrosarcoma cells caused resistance to tumor necrosis factor- α (TNF- α)-, hydrogen peroxide- and staurosporine-induced apoptosis, whereas forced caveolin-1 expression sensitized HepG2 human hepatocellular carcinoma cells to TNF-a [81]. Caveolin-1 expression also sensitizes HeLa human cervical carcinoma cells to sodium arsenite and hydrogen peroxide toxicity [82]. In addition, adenoviral overexpression of caveolin-1 was found to inhibit platelet-derived growth factor (PDGF)-induced proliferation of primary vascular smooth muscle cells (VSMCs), leading to apoptosis [83]. Collectively, the results of these studies indicate that the growth-inhibitory actions of caveolin-1 may sometimes be accompanied by a pro-apoptotic activity.

12.4.3

Survival-Promoting Activity of Caveolin-1

Several studies using either overexpression or antisense suppression techniques demonstrated that caveolin-1 may act as a positive regulator of cell survival in certain cancer cells. This was first shown in mouse and human prostate cancer cells, where expression of caveolin-1 protects the cells from androgen deprivation-induced apoptosis and c-Myc-induced apoptosis, respectively [7,10]. In the former case, caveolin-1 was secreted from the cells, upon testosterone treatment, acting as an autocrine or paracrine pro-survival factor [9]. Similarly, overexpression of caveolin-1 in LNCaP human prostate cancer cells significantly reduced thapsigargin-induced apoptosis [12]. The pro-survival action of caveolin-1 in prostate cancer cells was recently confirmed in the TRAMP mouse model, where caveolin-1 gene knockout attenuated tumor progression and metastasis *in vivo* and stimulated apoptosis of tumor-derived cells both *in vitro* and *in vivo* [84]. We have recently shown that caveolin-1 expression in MCF-7 cells results in inhibition of anoikis

254 12 The Role of Caveolin-1 in Tumor Cell Survival and Cancer Progression

(detachment-induced apoptosis) [11] and of detachment-induced activation of p53 [13]. Interestingly, cellular resistance to anoikis was also recently shown to be caveolin-1-dependent in pancreatic adenocarcinoma cells [85]. Antisense inhibition of caveolin-1 expression sensitizes nonmalignant intestinal epithelial cells to anoikis [14]. Taken together, these data clearly indicate that caveolin-1 can promote cell survival upon environmental challenge in both normal and cancer cells.

12.5 Molecular Mechanisms Implicated in the Pro-Survival Action of Caveolin-1

The phosphoinositide 3-kinase (PI3K)/Akt cell survival pathway has recently emerged as a major target for regulation by caveolin-1 in a variety of cancer cells [12,13,86]. A possible role of caveolin-1 in PI3K/Akt pathway activation was shown in multiple myeloma cells which, unlike most other hematopoietic-derived cells, express high levels of caveolin-1 [86]. Caveolin-1 was found to co-localize in lipid rafts fractions, and to co-exist in an immunoprecipitable complex, with receptors for insulin-like growth factor-I (IGF-I) and interleukin-6 (IL-6). These cell survival-inducing factors stimulated c-Src-mediated phosphorylation of caveolin-1 on Tyr14 and its association with PI3K subunits. The disruption of lipid raft organization by cholesterol depletion resulted in a redistribution of caveolin-1 and PI3K, an inhibition of Tyr14 phosphorylation of caveolin-1, and an abrogated IGF-I- and IL-6-induced activation of Akt and survival, thus confirming that both caveolin-1 and intact caveolae are essential for these processes [86]. Consistent with this conclusion, transient expression of caveolin-1 in LNCaP human prostate cancer cells resulted in elevated Akt phosphorylation and activation and increased resistance to thapsigargin-induced apoptosis [12]. Similarly, stable expression of caveolin-1 in MCF-7 human breast cancer cells also resulted in elevated basal phosphorylation of Akt, concomitantly increasing cellular resistance to anoikis [13].

Interestingly, a correlation between caveolin-1 expression and Akt activation was also seen in HEK-293 human embryonic kidney cells [82] and L929 mouse fibrosarcoma cells [81] although, in the latter two cases, Akt activation was reported to sensitize the cells to the apoptotic stimuli.

The mechanisms whereby caveolin-1 regulates PI3K/Akt-mediated cell survival have yet to be fully elucidated. It is clear, however, that caveolin-1 may influence this important pathway by recruiting and/or modulating its various components at different levels. First, caveolin-1 may act at the receptor level, as shown in caveolin-1-expressing MCF-7 cells, in which IGF-I receptor expression is elevated [13]. In these cells, the up-regulation of IGF-I receptors is associated with enhanced IGF-I signaling to the Erk1/2 and PI3K/Akt pathways (Fig. 12.1). In endothelial cells, caveolin-1 promotes the nongenomic action of nuclear receptors such as the estrogen receptor- α (ER α), resulting in activation of Akt and endothelial nitric oxide synthase (eNOS) [87]. Whether a similar mechanism operates in steroid-dependent human cancers, such as breast cancer and prostate cancer, has yet to be determined. In endothelial cells, caveolin-1 similarly interacts with TNF- α receptors,



Fig. 12.1 Enhanced IGF-I-mediated signaling in caveolin-1-transfected MCF-7 breast adenocarcinoma cells. Cells were serum-starved for 24 h and then stimulated with 50 ng mL⁻¹ human IGF-I for the indicated times. Cell lysates were resolved by SDS-PAGE and then

blotted with antibodies to the active phospho-Erk1/2 (pErk1/2), Erk1/2, phospho-Akt (pAkt) and actin (loading control) as indicated. (Figure reprinted, with permission, from [13].)

allowing TNF- α -induced activation of Akt [88]. An essential role of caveolin-1 in $\alpha_v\beta_3$ integrin-dependent activation of PI3K and Akt upon mechanical stress was recently noted also in VSMC [89]. In these cells, caveolin-1 is additionally required for angiotensin II receptor transactivation of the EGF receptor after Rac1 and NADPH oxidase activation, resulting in Akt activation [90]. Caveolin-1 is part of a survival signaling complex comprising urokinase-type plasminogen activator receptor, integrin $\alpha_v\beta_3/\alpha_5\beta_1$ and the SFK Yes in vitronectin-attached endothelial cells [91].

As mentioned above, caveolin-1 recruits both regulatory and catalytic subunits of PI3K upon IGF-I and IL-6 stimulation in multiple myeloma cells [86]. Phosphoinositide-dependent protein kinase-1 (PDK1), which mediates Akt Thr308 phosphorylation was also found physically to interact with caveolin-1 [92]. However, in this case a caveolin-1 scaffolding peptide suppressed the self-phosphorylation and the *in-vitro* kinase activities of PDK1. These results are inconsistent with the data showing that caveolin-1 significantly increases the activity of PDK1 and Akt [12]. Chun and co-workers suggested that these differences may result from changes in both subcellular localization and in the affinity to caveolin-1 of PDK1 compared to other factors (such as the serine/threonine protein phosphatases) [92]. Integrinlinked kinase has been proposed to serve as an Akt Ser473-kinase, and was found to interact with caveolin-1 via a CSD-binding motif [93,94]. Intriguingly, another putative Akt Ser473 kinase, namely DNA-PK, is localized in part in lipid rafts, although in this case direct interaction with caveolin-1 has yet to be demonstrated [95].

Finally, caveolin-1 may regulate the PI3K/Akt pathway by affecting lipid and protein phosphatases that turn the pathway off. Interaction of caveolin-1 via its CSD with protein phosphatases that dephosphorylate phospho-Akt, namely PP1 and PP2A, results in their inhibition and in consequent activation of Akt in LNCaP

256 12 The Role of Caveolin-1 in Tumor Cell Survival and Cancer Progression

cells [12]. Caveolin-1 is also associated with the tumor suppressor protein PTEN, a 3-phosphoinositide phosphatase that terminates the PI3K-dependent signal, although in this case the effect of caveolin-1 on PTEN activity has not been determined [96]. Interestingly, it was shown recently that caveolin-1 knockout up-regulates PTEN expression in TRAMP mouse prostate tumors, consistent with the inhibition of tumor progression caused by loss of caveolin-1 in this model [84].

The PI3K/Akt pathway is not the only survival pathway that is regulated by caveolin-1. Although caveolin-1 usually inhibits growth factor activation of the mitogenic Erk1/2 pathway, it is required for integrin-dependent activation of Erk1/2 through a complex that consists of Fyn, Shc, and Grb2 [97,98]. The Erk1/2 pathway promotes cell survival, at least in some cell types [99,100]. Caveolin-1 is similarly required for β_1 -integrin-mediated signaling to Src kinases and focal adhesion kinase (FAK) and, consequently, β_1 -integrin-dependent adhesion [101]. Caveolin-1 enables activation of c-Src upon cross-linking of the cell adhesion protein CE-CAM6 in BxPC3 human pancreatic adenocarcinoma cells, as well as c-Src-dependent tyrosine phosphorylation of FAK and the consequent inhibition of anoikis [85]. A positive regulatory role of caveolin-1 on survival signaling may also be implied by data showing that it interacts with a component of the TNF- α -NF κ B pathway [102], and that it may inhibits a caveolae-resident neutral sphingomyelinase via its CSD sequence [103].

12.6

The Role of Tyr14 Phosphorylation in Caveolin-1-Mediated Signaling

Tyr14 resides within a consensus motif for c-Src and c-Abl phosphorylation, present in caveolin-1 α (but not in caveolin-1 β), that is phosphorylated in a constitutive manner in v-Src- and v-Abl-transformed cells [16,104,105]. Phosphorylation of Tyr14 in response to cell stimulation has emerged recently as a major switch in caveolin-1 physiological function. Tyr14 phosphorylation occurs in various cell types in response to extracellular signal molecules such as insulin [106], IGF-I [107], EGF [108], fibronectin [109], IL-6 [86], PDGF [110], VEGF [111], and adrenocorticotropic hormone (ACTH) [112]. In addition, Tyr14 phosphorylation occurs in response to cell stressors such as oxidants [113], hyperosmolarity and UV irradiation [114], DNA damage-inducing drugs [115], and denial of extracellular matrix attachment [14].

Although insulin-dependent Tyr14 phosphorylation was initially believed to be mediated by a Src-family kinase (SFK) [116], more recent studies have shown that insulin-dependent Tyr14 phosphorylation is insensitive to a general SFK inhibitor, and suggested instead that it is mediated by the insulin receptor directly [117]. In contrast, EGF-, IGF-I and IL-6-induced phosphorylation of Tyr14 were inhibited by the same SFK blocker [86,118]. Similarly, a SFK is clearly involved in hyperosmotic stress-induced Tyr14 phosphorylation, since it is strongly inhibited by a dominant-negative form of Src [114]. As noted above, exposure of primary human fibroblasts to oxidative stress causes Tyr14 phosphorylation, and the SFK Fyn was shown to be

required for this response [119]. In the same cells, this phosphorylation event also depends on the presence of c-Abl [120]. The relationship between Fyn- and Ablmediated phosphorylation of caveolin-1 is unclear, as both are required for oxidantinduced Tyr14 phosphorylation, but neither kinase is required for the action of the other, suggesting that Fyn and Abl do not participate in a linear signaling pathway [119]. It was proposed that low level of oxidative stress results in the activation of Fyn and Tyr14 phosphorylation, promoting cell survival, whereas Abl is activated (and phosphorylates caveolin-1) upon high level of oxidant exposure, leading to apoptosis [119].

What is the role of the phosphorylation of caveolin on Tyr14? The formation of a phosphotyrosine residue likely constitutes a binding site for proteins with Src homology 2 (SH2) and phosphotyrosine binding (PTB) domains. Such proteins, recruited upon Tyr14 phosphorylation, include Grb7 [105] and Csk [121,122]. The recruitment of Grb7 to pTyr14 was shown to greatly enhance anchorage-independent growth in cells co-transfected with Grb7, c-Src, and caveolin-1 [105]. Similarly, caveolin-1 transfection markedly increased EGF-stimulated cell migration and this, too, was dependent on the presence of a phosphorylatable Tyr14 [105]. The recruitment of Csk, which catalyzes the C-terminal phosphorylation of SFK and inhibits their activity, is likely a part of a negative regulatory loop wherein Csk limits or abrogates SFK activation in oxidant-stressed cells [122]. In accordance with this study, crosslinking of the GPI-anchored protein CECAM6 reduced caveolin-1 Tyr14 phosphorylation, forcing its dissociation from the Csk and thus inducing Src-dependent activation of FAK [85]. Recently, Tyr14 phosphorylation was found to mediate interaction of caveolin-1 with the membrane type I matrix metalloprotease and this interaction, in turn, was shown to correlate with the proteaseinduced cell migration [123]. This interesting finding resonates with the fact that pTyr14-caveolin-1 is localized in focal adhesions, as shown in Src-transfected cells [105,124], as well as cells exposed to hyperosmotic and oxidative stress [114,120], cells plated on a fibronectin matrix [109,122,124], and cells treated with ACTH [112]. In summary, although the precise mechanisms and role of Tyr14 phosphorylation has not been fully resolved, the evidence accumulated so far clearly implicates this event as critical to at least some of the physiological functions of caveolin-1. In this context, of particular interest are the differential actions of caveolin-1 α and caveolin-1 β , which could be related to the lack of the Tyr14 residue in the latter protein.

12.7 Stress-Induced Changes in Caveolin-1 Expression

New evidence shows that not only caveolin-1 may negatively regulate survival and apoptosis but also that stressful stimuli may, in turn, positively regulate caveolin-1 expression. This phenomenon was shown to occur in NIH-3T3 mouse fibroblasts exposed to subcytotoxic levels of hydrogen peroxide [34]. Significant up-regulation of caveolin-1 was observed in mouse macrophages exposed to various unrelated

258 12 The Role of Caveolin-1 in Tumor Cell Survival and Cancer Progression

apoptotic agents, including simvastatin, camptothecin, or glucose deprivation [125]. In this case, caveolin-1 was found to co-localize with phosphatidylserine on the cell surface of the apoptotic cells, thus serving as an indicator of macrophage apoptosis. Other stress signals that were reported to increase caveolin-1 protein levels included exposure to various cytostatic drugs in lung cancer cells [45,126,127]. The acute up-regulation of caveolin-1 expression by chemotherapeutic drugs may be related to the constitutively elevated caveolin-1 levels observed in MDR human cancer cell lines [44,45,128,129] and the correlation of caveolin-1 expression with expression of the *MDR1* gene in leukemic bone marrow leukocytes [130]. Another stress condition reported to induce caveolin-1 protein levels in endothelial cells at G_0/G_1 cell-cycle phase is hypergravity stress [131]. In the latter case, the up-regulation of caveolin-1 was associated with the redistribution of caveolin-1 to an intracellular compartment [131]. Recently, caveolin-1 protein levels were found to be significantly increased, in a time-dependent manner, upon detachment of anoikis-resistant breast and MDR colon cancer cells [13]. A related phenomenon was also shown by immunofluorescence staining of caveolin-1 in kidneys of acute renal failure rats in vivo, where high-intensity caveolin-1 expression was observed in injured proximal tubules that were losing basement membrane adhesion or were apoptotic, at one to four days after ischemia-reperfusion [132].

The mechanisms involved in the regulation of caveolin-1 expression is response to stress are poorly understood. TNF- α - and IL-1-induced up-regulation of caveolin-1 in breast and ovarian carcinoma cells is mediated by the NF κ B pathway [133]. A more recent study provides strong evidence implicating FOXO transcription factors, known to be up-regulated upon oxidative stress, in the induction of caveolin-1 [134]. Finally, p53 was also shown to act as a positive transcriptional regulator of caveolin-1 expression [60,135].

Taken together, these data indicate that caveolin-1 expression is regulated by various stress and apoptosis-inducing conditions, suggesting that caveolin-1 may play an important role in the physiological stress response of both normal and cancer cells. Additional experiments are required in order to determine the mechanisms involved in caveolin-1 regulation and whether caveolin was functioning as a pro-survival protein or alternatively, causing apoptosis in these cells.

12.8 Concluding Remarks

The complex picture that emerges from the studies outlined above defies a simple explanation and indicates, instead, that caveolin-1 plays different roles in early-versus advanced-stage cancer cells. The growth-inhibitory actions of caveolin-1 are well established, and involve direct inhibition of mitogenic signaling pathways. It is likely that, during the early phase of cancer progression, caveolin-1 is down-regulated in order to suppress its growth-inhibitory actions and to allow rapid proliferation and clonal expansion.

Yet, it is clearly evident that caveolin-1 also exhibits pro-survival actions. We believe that these actions explain why the expression of caveolin-1 is often upregulated at late, advanced stages of cancer, when tumor cell survival and stress resistance are of paramount importance. Although the PI3K/Akt pathway has been revealed as a major survival pathway targeted by caveolin-1, other pathways may also participate in mediating survival of malignant cells during advanced stages of cancer progression. The molecular mechanisms utilized by caveolin-1 to modulate the activity of these pathways have still to be fully elucidated. Likewise, an intriguing question – the answer to which remains obscure – is how advanced stage cancer cells manage to circumvent the anti-proliferative actions of caveolin-1.

The ability of caveolin-1 to effect both growth-inhibitory and survival-promoting activities provides a rational explanation for the divergent changes of caveolin-1 expression in different cancer cells and tumor specimens, and revises the currently dominant view of caveolin-1 as a primarily tumor-suppressor-like protein. Furthermore, the context-dependent dual functionality of caveolin-1 highlights the potential of multifunctional adaptor proteins to affect cancer cell phenotype in an unpredictable manner, thus underscoring the complexity of cancer cell biology.

Acknowledgments

The authors are grateful to all members of their group for many helpful discussions. This project was supported in part by a grant from the DKFZ/MOST, and is currently supported by a grant from the Philip Morris Foundation and by intramural grants from the Willner Center for Vascular Biology and the Center for Scientific Excellence. M. L. is the incumbent of the Harold L. Korda Professorial Chair in Biology.

Abbreviations

ACTH CSD	adrenocorticotropic hormone caveolin scaffolding domain
eNOS	endothelial nitric oxide synthase
ER-α	estrogen receptor-α
FAK	focal adhesion kinase
IGF-I	insulin-like growth factor-I
IL	interleukin
MDR	multidrug resistance/t
PDGF	platelet-derived growth factor
PDK1	phosphoinositide-dependent protein kinase-1
PI3K	phosphoinositide 3-kinase
PTB	phosphotyrosine binding
SCLC	small-cell lung cancer

SFK	Src family kinase(s)
SH2	Src homology 2
TNF-α	tumor necrosis factor-α
VSMC	vascular smooth muscle cells

References

- Razani, B., Woodman, S. E., Lisanti, M. P. (2002) *Pharmacol. Rev.* 54, 431–467.
- 2 Parton, R.G. (2003) Nat. Rev. Mol. Cell Biol. 4, 162–167.
- 3 Li, W. P., Liu, P., Pilcher, B. K., Anderson, R. G. (2001) J. Cell Sci. 114, 1397–1408.
- 4 Liu, P., Rudick, M., Anderson, R.G. (2002) J. Biol. Chem. 277, 41295–41298.
- 5 Razani, B., Schlegel, A., Liu, J., Lisanti, M.P. (2001) *Biochem. Soc. Trans.* 29, 494–499.
- 6 Liscovitch, M., Burgermeister, E., Jain, N., Ravid, D., Shatz, M., Tencer, L. (2004) In: Mattson, M. P. (Ed.) *Membrane Microdomain Signaling. Lipid Rafts in Biology and Medicine.* Humana Press, Totowa, New Jersey, pp. 161–190.
- 7 Nasu, Y., et al. (1998) Nat. Med. 4, 1062–1064.
- 8 Li, L., et al. (2001) *Cancer Res.* 61, 4386-4392.
- **9** Tahir, S.A., et al. (2001) *Cancer Res.* 61, 3882–3885.
- 10 Timme, T. L., Goltsov, A., Tahir, S., Li, L., Wang, J., Ren, C., Johnston, R. N., Thompson, T. C. (2000) *Oncogene* 19, 3256–3265.
- Fiucci, G., Ravid, D., Reich, R., Liscovitch, M. (2002) Oncogene 21, 2365–2375.
- 12 Li, L., Ren, C. H., Tahir, S. A., Ren, C., Thompson, T.C. (2003) *Mol. Cell. Biol.* 23, 9389–9404.
- 13 Ravid, D., Maor, S., Werner, H., Liscovitch, M. (2005) Oncogene 24, 1338–1347.
- 14 Loza-Coll, M. A., Perera, S., Shi, W., Filmus, J. (2005) Oncogene 24, 1727–1737.
- 15 Shatz, M., Liscovitch, M. (2004) Leukemia Res. 28, 907–908.
- 16 Glenney, J.R.J. (1989) J. Biol. Chem. 264, 20163–20166.
- 17 Glenney, J.R.J., Soppet, D. (1992) Proc. Natl. Acad. Sci. USA 89, 10517–10521.
- 18 Rothberg, K. G., Heuser, J. E., Donzell, W. C., Ying, Y. S., Glenney, J. R., Anderson, R. G. (1992) *Cell* 68, 673–682.

- 19 Smart, E. J., Graf, G. A., McNiven, M. A., Sessa, W. C., Engelman, J. A., Scherer, P. E., Okamoto, T., Lisanti, M. P. (1999) *Mol. Cell. Biol.* 19, 7289–7304.
- 20 Anderson, R.G.W. (1998) Annu. Rev. Biochem. 67, 199–225.
- 21 Fielding, C. J., Fielding, P. E. (2001) Biochim. Biophys. Acta 1529, 210–222.
- 22 Scherer, P. E., Okamoto, T., Chun, M., Nishimoto, I., Lodish, H. F., Lisanti, M. P. (1996) *Proc. Natl. Acad. Sci. USA* 93, 131–135.
- 23 Scherer, P. E., et al. (1997) J. Biol. Chem. 272, 29337-29346.
- 24 Scheiffele, P., Verkade, P., Fra, A.M., Virta, H., Simons, K., Ikonen, E. (1998) J. Cell Biol. 140, 795–806.
- 25 Galbiati, F., Razani, B., Lisanti, M. P. (2001) Trends Mol. Med. 7, 435–441.
- 26 Li, S., Couet, J., Lisanti, M. P. (1996) J. Biol. Chem. 271, 29182–29190.
- 27 Couet, J., Li, S., Okamoto, T., Ikezu, T., Lisanti, M.P. (1997) J. Biol. Chem. 272, 6525–6533.
- 28 Scherer, P. E., Lisanti, M. P., Baldini, G., Sargiacomo, M., Mastick, C. C., Lodish, H. F. (1994) *J. Cell Biol.* 127, 1233–1243.
- 29 Galbiati, F., et al. (1998) Proc. Natl. Acad. Sci. USA 95, 10257–10262.
- 30 Campbell, L., Hollins, A. J., Al-Eid, A., Newman, G. R., von Ruhland, C., Gumbleton, M. (1999) *Biochem. Biophys. Res. Commun.* 262, 744–751.
- 31 Mikol, D. D., Hong, H. L., Cheng, H. L., Feldman, E. L. (1999) *Glia* 27, 39–52.
- 32 Fuchs, S., Hollins, A. J., Laue, M., Schaefer, U. F., Roemer, K., Gumbleton, M., Lehr, C. M. (2003) *Cell Tissue Res.* 311, 31–45.
- 33 Park, W.Y., Park, J.S., Cho, K.A., Kim, D.I., Ko, Y.G., Seo, J.S., Park, S.C. (2000) *J. Biol. Chem.* 275, 20847–20852.
- 34 Volonte, D., Zhang, K., Lisanti, M. P., Galbiati, F. (2002) *Mol. Biol. Cell* 13, 2502–2517.

- 35 Koleske, A. J., Baltimore, D., Lisanti, M. P. (1995) Proc. Natl. Acad. Sci. USA 92, 1381–1385.
- 36 Engelman, J.A., et al. (1998) J. Biol. Chem. 273, 20448–20455.
- 37 Engelman, J. A., Zhang, X. L., Lisanti, M. P. (1998) FEBS Lett. 436, 403–410.
- 38 Fra, A. M., Mastroianni, N., Mancini, M., Pasqualetto, E., Sitia, R. (1999) *Genomics* 56, 355–356.
- **39** Drab, M., et al. (2001) *Science* 293, 2449–2452.
- 40 Razani, B., et al. (2001) J. Biol. Chem. 276, 38121–38138.
- 41 Capozza, F., Williams, T. M., Schubert, W., McClain, S., Bouzahzah, B., Sotgia, F., Lisanti, M. P. (2003) *Am. J. Pathol.* 162, 2029–2039.
- 42 Williams, T.M., et al. (2003) *Mol. Biol. Cell* 14, 1027–1042.
- 43 Lavie, Y., Liscovitch, M. (1997) Mol. Biol. Cell 8, 207a.
- 44 Lavie, Y., Fiucci, G., Liscovitch, M. (1998) J. Biol. Chem. 273, 32380–32383.
- 45 Yang, C.-P. H., Galbiati, F., Volonte, D., Horwitz, S. B., Lisanti, M. P. (1998) *FEBS Lett.* 439, 368–372.
- 46 Yang, G., et al. (1998) Clin. Cancer Res. 4, 1873–1880.
- 47 Yang, G., Truong, L. D., Wheeler, T. M., Thompson, T.C. (1999) *Cancer Res.* 59, 5719–5723.
- 48 Ho, C. C., Huang, P. H., Huang, H. Y., Chen, Y. H., Yang, P. C., Hsu, S. M. (2002) *Am. J. Pathol.* 161, 1647–1656.
- **49** Yoo, S. H., et al. (2003) *Lung Cancer* 42, 195–202.
- 50 Sanchez-Carbayo, M., Socci, N. D., Charytonowicz, E., Lu, M., Prystowsky, M., Childs, G., Cordon-Cardo, C. (2002) Cancer Res. 62, 6973–6980.
- 51 Campbell, L., Gumbleton, M., Griffiths, D. F. (2003) Br. J. Cancer 89, 1909–1913.
- 52 Joo, H. J., Oh, D. K., Kim, Y. S., Lee, K. B., Kim, S. J. (2004) BJU Int. 93, 291–296.
- 53 Suzuoki, M., et al. (2002) Br. J. Cancer 87, 1140–1144.
- 54 Terris, B., Blaveri, E., Crnogorac-Jurcevic, T., Jones, M., Missiaglia, E., Ruszniewski, P., Sauvanet, A., Lemoine, N.R. (2002) *Am. J. Pathol.* 160, 1745–1754.
- 55 Kato, K., et al. (2002) Cancer 94, 929-933.
- 56 Satoh, T., et al. (2003) *Cancer* 97, 1225–1233.

- 57 Horiguchi, A., Asano, T., Asakuma, J., Sumitomo, M., Hayakawa, M. (2004) *J. Urol.* 172, 718–722.
- 58 Yang, G., Timme, T. L., Frolov, A., Wheeler, T. M., Thompson, T. C. (2005) *Cancer* 103, 1186–1194.
- 59 Racine, C., Belanger, M., Hirabayashi, H., Boucher, M., Chakir, J., Couet, J. (1999) *Biochem. Biophys. Res. Commun.* 255, 580–586.
- 60 Razani, B., Altschuler, Y., Zhu, L., Pestell, R. G., Mostov, K. E., Lisanti, M. P. (2000) *Biochemistry* 39, 13916–13924.
- **61** Ross, D. T., et al. (2000) *Nat. Genet.* 24, 227–235.
- 62 Sunaga, N., et al. (2004) Cancer Res. 64, 4277-4285.
- **63** Wikman, H., et al. (2004) *J. Pathol.* 203, 584–593.
- 64 Belanger, M. M., Roussel, E., Couet, J. (2004) Chest 125(5), 106S.
- 65 Wiechen, K., et al. (2001) Am. J. Pathol. 159, 1635–1643.
- 66 Wikman, H., Kettunen, E., Seppanen, J.K., Karjalainen, A., Hollmen, J., Anttila, S., Knuutila, S. (2002) *Oncogene* 21, 5804–5813.
- 67 Kato, T., et al. (2004) *Cancer Lett.* 214, 121–128.
- 68 Lee, S.W., Reimer, C. L., Oh, P., Campbell, D.B., Schnitzer, J. E. (1998) Oncogene 16, 1391–1397.
- 69 Shin, S.-I., Freedman, V. H., Risser, R., Pollack, R. (1975) Proc. Natl. Acad. Sci. USA 72, 4435–4439.
- 70 Syed, V., Mukherjee, K., Lyons-Weiler, J., Lau, K. M., Mashima, T., Tsuruo, T., Ho, S. M. (2005) Oncogene 24, 1774–1787.
- 71 Hayashi, K., Matsuda, S., Machida, K., Yamamoto, T., Fukuda, Y., Nimura, Y., Hayakawa, T., Hamaguchi, M. (2001) *Cancer Res.* 61, 2361–2364.
- 72 Lee, H., Park, D.S., Razani, B., Russell, R.G., Pestell, R.G., Lisanti, M.P. (2002) *Am. J. Pathol.* 161, 1357–1369.
- 73 Sloan, E. K., Stanley, K. L., Anderson, R. L. (2004) Oncogene 23, 7893–7897.
- 74 Chen, S. T., Lin, S. Y., Yeh, K. T., Kuo, S. J., Chan, W. L., Chu, Y. P., Chang, J. G. (2004) Int. J. Mol. Med. 14, 577–582.
- 75 Zou, W., McDaneld, L., Smith, L.M. (2003) Anticancer Res. 23, 4581–4586.
- 76 Williams, T. M., et al. (2004) J. Biol. Chem. 279, 51630–51646.

- 262 References
 - 77 Williams, T.M., et al. (2004) J. Biol. Chem. 279, 24745–24756.
 - 78 Zundel, W., Swiersz, L. M., Giaccia, A. (2000) Mol. Cell. Biol. 20, 1507–1514.
 - 79 Liu, J., Lee, P., Galbiati, F., Kitsis, R.N., Lisanti, M.P. (2001) Am. J. Physiol. Cell Physiol. 280, C823–C835.
 - 80 Galbiati, F., Volonte, D., Liu, J., Capozza, F., Frank, P.G., Zhu, L., Pestell, R.G., Lisanti, M.P. (2001) *Mol. Biol. Cell* 12, 2229–2244.
 - 81 Ono, K., Iwanaga, Y., Hirayama, M., Kawamura, T., Sowa, N., Hasegawa, K. (2004) Am. J. Physiol. Lung Cell. Mol. Physiol. 287, L201-L209.
 - 82 Shack, S., Wang, X.T., Kokkonen, G.C., Gorospe, M., Longo, D.L., Holbrook, N.J. (2003) Mol. Cell. Biol. 23, 2407–2414.
 - 83 Peterson, T.E., et al. (2003) Arterioscler. Thromb. Vasc. Biol. 23, 1521–1527.
 - 84 Williams, T. M., et al. (2005) J. Biol. Chem. 280, 25134–25145.
 - 85 Duxbury, M. S., Ito, H., Ashley, S. W., Whang, E. E. (2004) J. Biol. Chem. 279, 23176–23182.
 - 86 Podar, K., et al. (2003) J. Biol. Chem. 278, 5794–5801.
 - 87 Acconcia, F., Ascenzi, P., Bocedi, A., Spisni, E., Tomasi, V., Trentalance, A., Visca, P., Marino, M. (2005) *Mol. Biol. Cell* 16, 231–237.
 - 88 D'Alessio, A., Al-Lamki, R. S., Bradley, J. R., Pober, J. S. (2005) Am. J. Pathol. 166, 1273–1282.
 - **89** Sedding, D.G., et al. (2005) *Circ. Res.* 96, 635–642.
 - 90 Zuo, L., Ushio-Fukai, M., Ikeda, S., Hilenski, L., Patrushev, N., Alexander, R.W. (2005) Arterioscler. Thromb. Vasc. Biol. 25, 1824–1830.
 - **91** Cao, D. J., Guo, Y. L., Colman, R. W. (2004) *Circ. Res.* 94, 1227–1234.
 - 92 Chun, J., Kwon, T., Lee, E. J., Hyun, S., Hong, S. K., Kang, S. S. (2005) *Biochem. Biophys. Res. Commun.* 326, 136–146.
 - 93 Meyer, A., van Golen, C. M., Boyanapalli, M., Kim, B., Soules, M. E., Feldman, E. L. (2005) *Biochemistry* 44, 932–938.
 - 94 Chun, J., Hyun, S., Kwon, T., Lee, E. J., Hong, S. K., Kang, S. S. (2005) *Cell Signal*. 17, 751–760.
 - 95 Feng, J., Park, J., Cron, P., Hess, D., Hemmings, B.A. (2004) *J. Biol. Chem.* 279, 41189–41196.

- 96 Caselli, A., Mazzinghi, B., Camici, G., Manao, G., Ramponi, G. (2002) Biochem. Biophys. Res. Commun. 296, 692–697.
- 97 Wary, K. K., Mainiero, F., Isakoff, S. J., Marcantonio, E. E., Giancotti, F. G. (1996) *Cell* 87, 733–743.
- 98 Wary, K. K., Mariotti, A., Zurzolo, C., Giancotti, F. G. (1998) Cell 94, 625–634.
- 99 Martindale, J. L., Holbrook, N. J. (2002) J. Cell. Physiol. 192, 1–15.
- 100 Kohno, M., Pouyssegur, J. (2003) Prog. Cell Cycle Res. 5, 219–224.
- 101 Wei, Y., Yang, X., Liu, Q., Wilkins, J.A., Chapman, H.A. (1999) J. Cell Biol. 144, 1285–1294.
- 102 Feng, X., Gaeta, M. L., Madge, L. A., Yang, J. H., Bradley, J. R., Pober, J. S. (2001) J. Biol. Chem. 276, 8341–8349.
- 103 Veldman, R. J., Maestre, N., Aduib, O. M., Medin, J. A., Salvayre, R., Levade, T. (2001) *Biochem. J.* 355, 859–868.
- 104 Li, S., Seitz, R., Lisanti, M. P. (1996) J. Biol. Chem. 271, 3863–3868.
- 105 Lee, H., et al. (2000) Mol. Endocrinol. 14, 1750–1775.
- 106 Mastick, C. C., Brady, M. J., Saltiel, A. R. (1995) J. Cell Biol. 129, 1523–1531.
- 107 Maggi, D., Biedi, C., Segat, D., Barbero, D., Panetta, D., Cordera, R. (2002) *Biochem. Biophys. Res. Commun.* 295, 1085–1089.
- 108 Kim, Y.N., Wiepz, G.J., Guadarrama, A.G., Bertics, P.J. (2000) J. Biol. Chem. 275, 7481–7491.
- 109 Mettouchi, A., Klein, S., Guo, W., Lopez-Lago, M., Lemichez, E., Westwick, J.K., Giancotti, F.G. (2001) Mol. Cell 8, 115–127.
- 110 Fielding, P. E., Chau, P., Liu, D., Spencer, T. A., Fielding, C. J. (2004) *Biochemistry* 43, 2578–2586.
- 111 Podar, K., et al. (2004) *Cancer Res.* 64, 7500–7506.
- 112 Colonna, C., Podesta, E. J. (2005) Exp. Cell Res. 304, 432–442.
- 113 Vepa, S., Scribner, W. M., Natarajan, V. (1997) Free Radic. Biol. Med. 22, 25–35.
- 114 Volonte, D., Galbiati, F., Pestell, R. G., Lisanti, M. P. (2001) J. Biol. Chem. 276, 8094–8103.
- 115 Navakauskiene, R., Treigyte, G., Gineitis, A., Magnusson, K. E. (2004) *Proteomics* 4, 1029–1041.

- 116 Mastick, C.C., Saltiel, A.R. (1997) J. Biol. Chem. 272, 20706-20714.
- 117 Kimura, A., Mora, S., Shigematsu, S., Pessin, J.E., Saltiel, A.R. (2002) J. Biol. Chem. 277, 30153-30158.
- 118 Kim, Y.N., Dam, P., Bertics, P.J. (2002) Exp. Cell Res. 280, 134-147.
- 119 Sanguinetti, A. R., Cao, H., Corley Mastick, C. (2003) Biochem. J. 376, 159-168.
- 120 Sanguinetti, A.R., Mastick, C.C. (2003) Cell Signal. 15, 289-298.
- 121 Cao, H., Courchesne, W.E., Mastick, C.C. (2002) J. Biol. Chem. 277, 8771-8774.
- 122 Cao, H., Sanguinetti, A.R., Mastick, C.C. (2004) Exp. Cell Res. 294, 159-171.
- 123 Labrecque, L., Nyalendo, C., Langlois, S., Durocher, Y., Roghi, C., Murphy, G., Gingras, D., Beliveau, R. (2004) J. Biol. Chem. 279, 52132-52140.
- 124 Lee, H., Park, D.S., Wang, X.B., Scherer, P.E., Schwartz, P.E., Lisanti, M.P. (2002) J. Biol. Chem. 277, 34556-34567.
- 125 Gargalovic, P., Dory, L. (2003) J. Lipid Res. 135 Bist, A., Fielding, C. J., Fielding, P. E. 44, 1622-1632.

- 126 Belanger, M. M., Roussel, E., Couet, J. (2003) Anticancer Drugs 14, 281-287.
- 127 Roussel, E., Belanger, M. M., Couet, J. (2004) Anticancer Drugs 15, 961-967.
- 128 Lavie, Y., Czarny, M., Liscovitch, M. (1997) FASEB I. 11, A1346.
- 129 Bender, F.C., Reymond, M.A., Bron, C., Quest, A.F. (2000) Cancer Res. 60, 5870-5878.
- 130 Pang, A., Wing, Y., Kwong, Y.L. (2004) Leukemia Res. 28, 973-977.
- 131 Spisni, E., Bianco, M.C., Griffoni, C., Toni, M., D'Angelo, R., Santi, S., Riccio, M., Tomasi, V. (2003) J. Cell. Physiol. 197, 198-204.
- 132 Mahmoudi, M., et al. (2003) J. Pathol. 200, 396-405.
- 133 Deregowski, V., Delhalle, S., Benoit, V., Bours, V., Merville, M. P. (2002) Biochem. Pharmacol. 64, 873-881.
- 134 van den Heuvel, A.P., Schulze, A., Burgering, B. M. (2005) Biochem. J. 385, 795-802.
- (2000) Biochemistry 39, 1966-1972.

Index

а

AA see amino acid A β see amyloid β -peptide ABCA1 see ATP-binding cassette transporter-A1 Abl 129-130 - actin remodeling 130-131 - caveolin kinase 120, 129 - caveolin phosphorylation 121 acceptor, fluorescence correlation spectroscopy 59 acceptor channels, FRET 146 ACTH see adrenocorticotropic hormone actin cytoskeleton - attachment 127-129 - dynamics of caveolin 199-200 - dvnamin 83 - endocytosis 73 - stabilisation 130 actin/plasma membrane attachment sites, phosphorylation 128 actin polymerization 131 actin remodeling 127 - Abl 130-131 - insulin-induced 131 activation, Csk 125-127 activation-induced cell death (AICD), lipid rafts 157 acylation, signal transduction 92 AD see Alzheimer's disease ADAM10, metalloprotease 214 adaptor proteins, multifunctional 249 adhesion complex, focal 198-199 adhesion molecules, immunological synapse 151 adipocytes - caveolin 199 - caveolin phosphorylation 117 - signaling pathways 117

adrenocorticotropic hormone (ACTH) 256-257 affinity conversion, lipid rafts 157 AFM see atomic force microscopy aggregation, membrane proteins 32 aging, gate theory 195 aging process, caveolin 195 AICD see activation-induced cell death Alzheimer's disease (AD) 205 role of rafts 215–223 amino acid (AA), caveolins domains 25 amino-terminus, caveolin-1 123 amphiphilic dye, LAURDAN 5 amyloid β-peptide (Ab) 215-217 amyloid precursor protein (APP) - cleavage 215-223 - processing 205-223 anchored protein (AP) - GPI- 8, 54, 63, 92, 207, 214, 257 angiogenesis, caveolin-eNOS interaction 241-242 anisotropy, measurements 61 annular lipids, transmembrane signaling 143 anoikis, cellular resistance 253-254 anomalous diffusion, immunological synapse 152 antennopedia (AP) peptide 102 anti-proliferative activity, caveolin-1 252 antibodies, diffusion-based measurements 55 antigen-presenting cell, immunological synapse see APC antisense suppression techniques, caveolin-1 253 AP see also antennopedia, see anchored protein APC, immunological synapse 150 apical membrane - basolateral protein exclusion 14

- epithelial cells 9-10

266 Index

apical t-SNARE, syntaxin 3 16 APP *see* amyloid precursor protein arachidonyl ethanolamine plasmalogens 94 ARP2/3, actin polymerization 130 artificial lipid membranes, DRM 50 asialogangliosides, signal transduction 94 asymmetric membrane protein, bending forces 33-35 atomic force microscopy 149 ATP-binding cassette transporter-A1 97 autocorrelation function, FCS 147 autofluorescence, FRET 146

autophosphorylation, signal transduction 104

b

baby hamster kidney, electron micrographs 80 BACE see β -site APP cleavage enzyme

- basolateral proteins, exclusion from apical membrane 14
- bending energy, invaginated membrane domains 30
- bending forces, asymmetric membrane protein 33-35

bending rigidity, biomembranes 27–28 β-site APP cleavage enzyme (BACE) 215–218

BHK cells see baby hamster kidney bilayer see lipid bilayer

biomembranes

- bending rigidity 27-28

– *see* membranes

blocked caveolin phosphorylation 119

Bodipy-PC, fluorescence correlation

spectroscopy 57 breast cancer, caveolin-1 expression 252 budding, coat-driven/domain-induced 15 buds, formation 34–37 bulb radius (preferred), caveolae 36–37

С

- C-terminal, caveolins 25, 177 C-terminal region, endocytosis 70 Ca²⁺ signal, voltage gated channels 154 calciumbound calmodulin (CaM) 233 – eNOS enzyme activity 236 CaM *see* calciumbound calmodulin cancer progression, role of caveolin-1 249–259 carboxyl terminal fragment (CTF), APP 215 cardiac myocytes – caveolin-eNOS interaction 244–245
- eNOS expression 234
- cargo sorting, membrane traffic 15

caveola-cholesterol trafficking cycle, regulation 184-185 caveolae 1-17, 131, 175-176 - caveolin oligomer mixing 35 - cholesterol localization 182-183 constitutively internalization 81-82 - curvature 12 - disruption 40 - endocytosis 69, 75-81 endothelial nitric oxide synthase (eNOS) 233-246 - FC 102-106 - first observation 25 - formation 30-31 - function 199-200 - invaginated lipid rafts 28-31 - labeling 80 - lipid composition 12 - lipid rafts 206-207 - lipids 93-95 - membrane domains 71 membrane tension 38-42 - membrane traffic 12-17 - origin of name 10 - preferred bulb radius 36-37 proteins 95-98 scaffold motif 99 scaffolded membrane domains 10-12 shaping forces 25-42 signal transduction 91-114 signaling 115-140 - structure 96 - subpopulations 84 thermodynamic phase separation of membrane proteins 35-38 - water permeability 94 - wave 74 caveolar coat protein, caveolin-1 249 caveolar endocytosis, consensus model 85 caveolar function, dynamin 83 caveolar invaginations 84-85 caveolar-like domains (CLDs) 210 caveolar proteins, FC depletion 102 caveolar vesicles, free 73 caveolin 11, 101-102, 175 - contact site 99 - contribution to cell-cycle regulation 40 - homo-oligomers 33 insulin receptor 117–118 - internalization 127 - intracellular chaperone complexes 175 - 194

- N-/C-Terminal 25
- overexpression 176

- phosphorylation see caveolin phosphorylation - protein acylation 186 - raft immobilization 84-85 - signal transduction 95 signaling pathways 116-117 caveolin-1 - expression 251, 257-258 - regulation 250 - role in tumor cell survival and cancer progression 249-259 - tyr14 phosphorylation 256-257 caveolin-associated structures, intracellular 82 - 83caveolin binding motif (CBM) 235-236 caveolin-eNOS interaction - angiogenesis 241-242 - cardiac myocytes 244-245 - lipoproteins 239-241 - vasodilation and endothelial permeability 242-244 caveolin-eNOS regulatory cycle 236-238 caveolin family, endocytosis 70 caveolin kinase 120 caveolin localization, caveolae 182-183 caveolin oligomers, mixing on caveolae 35 caveolin phosphorylation 115-140 - adipocytes 117 - blocked 119 - insulin-induced 119 - signaling pathway 121 - stress-induced 118-119 - tyrosine see caveolin tyrosine phosphorylation caveolin protein structure 177-178 caveolin rafts, morphology 40 caveolin scaffolding domain (CSD) 235-236, 250 - pro-survival action of caveolin-1 255 - regulation of angiogenesis 242 signal transduction 98–101 caveolin systems, mutant 37-38 caveolin tyrosine kinases 123 caveolin tyrosine phosphorylation 256-257 signaling pathways 116–123 caveolins domains, amino acid 25 caveosomes 82-83 cbc see critical budding concentration CBM see caveolin binding motif CD36, scavenger receptors 187 cell compartments see compartments cell-cycle regulation, contribution of caveolin 40

cell death, activation-induced 157

cell differentiation, caveolin-1 250 cell fusion 155-156 cell membrane - basic organization 1-3 - mechanical regulation 41 phase separation 5-7 - rafts 49-51 - see also membranes cell migration, phosphorylation 130 cell shape, roles of caveolae and caveolin 195-202 cell-surface proteins, distribution patterns 144 cell survival, caveolin-1 253-256 cell types, caveolae 176 cells in culture, physiological state 180-182 cellular motility 196 cellular organelles, membrane traffic 12 cellular stresses, caveolin phosphorylation 118 central supramolecular activation cluster see cSMAC ceramidase, signal transduction 95 ceramide, signal transduction 94 chaperone complexes 188 - caveola-cholesterol trafficking cycle 184-185 - intracellular 175-194 chemical affinity, raft hypothesis 45 chemical incompatibility, influence on lipid domain shape 29-30 chemical interaction, lipid rafts 50 cholera toxin - endocytosis 75-81 - raft and caveolar endocytosis 14 cholera toxin B. diffusion-based measurements 52 cholesterol - APP processing 222 - caveolin 183-184 - cell membrane 2 - DRM 50 - fluorescence correlation spectroscopy 57 - lipid rafts 205-206 - low-density lipoprotein (LDL) 239-241 - relative concentration, caveolae 175 signal transduction 91-114 - ternary phase diagram 4 cholesterol-binding drugs, endocytosis 77 cholesterol-binding proteins, caveolae 115 cholesterol-containing domains, raft hypothesis 45 cholesterol depletion - PrP^c conversion 209

268 Index

> – strong 216 cholesterol homeostasis – endocytosis 185 - regulation 187 cholesterol localization, caveolae 182-183 cholesterol recognition amino acid consensus see CRAC cholesterol/sphingolipid-rich microdomains 91 cholesterol transport, specialized tissues 188 chronotropic effects, negative 244-245 clathrin-coated pits - endocytosis 76 internalization of PrP^c 209–210 clathrin-mediated endocytosis, caveolae 69, 176 CLDs see caveolar-like domains **CLSM 165** cluster size, lipid microdomains, ERB2 164 clustered cell-surface distribution, immunological synapse 152 clustered raft domains 8-9 cmc see critical micelle concentration co-clustering 154 co-immunoprecipitation, caveolins 117 co-localization – endocytosis 78 - immunological synapse 153 - lipid rafts 159, 162 - membrane proteins 142 co-mobility, protein 147 coat-driven budding, cargo sorting and vesicle cytoskeleton mesh, single-molecule tracking formation 15 coat proteins, caveolae 115 coats, protein 13 columnar epithelia 9 compartments, intracellular 208-210, 215 - 217confluent human umbilical vein endothelial cells 181 confocal laser scanning microscopy, domain size 144 conformational diseases 205 conformational transition, $\mbox{Pr}\mbox{P}^{c}\mbox{-}\mbox{Pr}\mbox{P}^{Sc}$ 211 - 214consensus model, caveolar endocytosis 85 constitutively internalization, caveolae 81-82 contact site, proteins 99 cooperative interactions, lipid microdomains 163 coronary arteries, endothelial cells 234 correlation spectroscopy, fluorescence 57-58 CRAC 98 CRAC-1 101

critical budding concentration (cbc) 34, 37 critical micelle concentration 34 cross-correlation coefficient, fluorescence distributions 145 CSD see caveolin scaffolding domain Csk 127, 257 - activation 125-127 - phosphocaveolin 124-125 - Src-family kinases 125 cSMAC, immunological synapse 150 CT - endocytosis 75-76, 78 - internalization by caveolae 81 CT-B, internalization 79, 84 CT-B-gold, endocytosis 76 CT-B-HRP conjugate, incubated BHK cells 80 CTF see carboxyl terminal fragment CTL, immunological synapse 150 CtxB see cholera toxin B curvature - caveolae 12 - spontaneous 27 cvclodextrin, FC depletion 104 cyclosporine, chaperone complexes 184 cysteines - caveolin 178 - protein acylation 186 cytofacial leaflet, endocytosis 183 cytokine receptor assembly, lipid rafts 156-162 53 cytotoxic T cell see CTL d Danielli-Davson model 1 deformable fluid membranes, physical mod-

eling 26-29 dehydroergosterol, signal transduction 94 depletion, cholesterol 209, 216 - FC 102-104 detergent-free isolation, caveolae 93 detergent-insoluble glycolipid (DIG) 46 detergent-resistant membranes (DRM) 2, 46, 49-51, 206, 209, 215-217 dialkylcarbocyanine see DiI 1,1'-dieicosanyl-3,3,3',3'-tetramethylindocarbocyanine perchlorate see DiI-C20 differentiated primary cells, signal transduction 91 diffusion-based measurements, raft hypothesis 52-58 diffusion constants, FRAP 56

diffusion process, membranes 58 downstream signaling cascades, tyrosine diffusion time, FCS 148 2-(4,4-difluoro-5,7-dimethyl-4-bora-3a,4a-DPPC see dipalmitoyl phosphatidylcholine diazas-indacene-3-pentanoyl)-1-hexadeca-DRM see detergent-resistant membranes noyl-sn-glycero-3-phosphocholine see Duchènne muscular dystrophy, caveolae Bodipy-PC DIG see detergent-insoluble glycolipid dvnamin diglyceride, signal transduction 94 DiI-C20, fluorescence correlation spectroscopy 57 dilauroyl phosphatidylcholine (DLPC), fluorescence correlation spectroscopy 57 dimerization - lipid microdomains 163 - signal transduction 104 1,2-dio-leoyl-sn-glycero-3-phosphoethanolamine-fluorescein see FL-DOPE dipalmitoyl phosphatidylcholine (DPPC), fluorescence correlation spectroscopy 57 1,2-dipalmitoyl-sn-glycero-3-phosphoethanolamine-fluorescein see FL-DPPE dipole moment, molecular 5 diseases - Alzheimer's see Alzheimer's disease - conformational 205 prion 205-223 distributions - clustered cell-surface 152 - fluorescence 144-145 - GM1 gangliosides 160 DLPC see dilauroyl phosphatidylcholine domain energy, in relation to domain shape and membrane tension 39 domain-induced budding, cargo sorting and vesicle formation 15 domain shape - in relation to domain energy and membrane tension 39 - lipid 29-30 domain size, determination 144-145 domains caveolin 25, 177–178 - caveolin scaffolding see caveolin scaffolding domain - macro- 143 - membrane 30, 45, 71 - MHC I molecules 152 - micro- see microdomains - raft 8-9 donor, fluorescence correlation spectroscopy 59 donor channel, FRET 146 donor photobleaching see pbFRET

- caveolar function 83 - transendothelial transport 74 dynamin-1 mutant, endocytosis 81 е Echovirus 1 75 ECM see extracellular matrix effective membrane temperature 27 EGF 117 - receptor, lipid microdomains 163 - signal transduction 96 electron microscopy, lipid rafts 51 EM see electron microscopy endocytic machinery 69 endocytic marker, constitutively internalization 81 endocytic pathways, caveolae 176 endocytic vesicles, free 83 endocytosis 69 - caveolae-mediated 85, 200-201 - cholera toxin 75-81 - independent pathways 84 endoplasmic reticulum (ER), PrPc-PrPSc transconformation 208 endoplasmic reticulum-Golgi-intermediatecompartment see ER-GIC endosomes - APP cleaving pathways 218 - caveosomes communication 82 endothelial cells, coronary arteries 234 endothelial nitric oxide synthase see eNOS endothelial permeability, caveolin-eNOS interaction 242-244 endothelial tube formation, stimulation by statins 241 endothelin receptor-A, scaffold sequence 99 endothelium-dependent vascular relaxation 242-243 energy, bending 30 energy barrier, between flat and budded caveolae states 37-39 energy transfer process, fluorescence 59 eNOS (endothelial nitric oxide synthase) 70 - and caveolae 233-246 - protein acylation 186 - signal transduction 98 eNOS-caveolin interaction

phosphorylation 123

number 40

- 270 Index
 - angiogenesis 241–242
 - cardiac myocytes 244-245
 - lipoproteins 239-241
 - vasodilation and endothelial permeability 242–244

eNOS interacting protein (NOSIP) 237–238 eNOS traffic inducer (NOSTRIN) 237–238 epithelia, columnar 9 epithelial cells, apical membrane 9–10 epitope accessibility, caveolae determination 179

ER see endoplasmic reticulum

ER-associated degradation (ERAD) 208 ER-GIC 182

ERAD *see* ER-associated degradation ErbB2, lipid microdomains 163 estrogen uptake, caveola 185–186 exclusion, due to phase separation 15 exofacial leaflet, endocytosis 183 expression

- caveolin-1, heterologous 252-254
- caveolin-1, in human cancer 251
- caveolins 178–182
- extracellular matrix (ECM) 198-199
- extracellular signals, caveolin phosphorylation 124
- extracytoplasmic proteins, raft hypothesis 48 ezetimibe, cholesterol transport 188

f

FAK see focal adhesion kinase fatty-acid linked proteins 48 FC – caveolae 102–104 - changes 104-106 - interactions 94 FC binding proteins 101–102 structural aspects 101 FC depletion, effects on caveolar proteins 102 FC recognition sites, signal transduction 99 FC-rich vacuoles, increased endocytosis 103 FCCS see fluorescence cross-correlation spectroscopy FCS *see* fluorescence correlation spectroscopy fibroblasts - phase separation 10 signaling pathways 125 fibronectin, phosphorylation 128 filamin 127 - identification 199 filipin, cholesterol localization 182

filopodia, focal adhesion complex 198-199

fission-fusion processes, endocytosis 73 FL-DOPE 52 fl-PE 52 FLIP, endocytosis 72 flotillins, signal transduction 96 flow cytometric FRET, lipid rafts 162 fluid membranes, physical modeling 26-29 fluid-mosaic model 45-46 fluorescein phosphatidylethanolamine see fl-PE fluorescence correlation spectroscopy (FCS) 57-59, 147 - diffusion-based measurements 52 - mobility measurements 161 fluorescence cross-correlation spectroscopy (FCCS) 147 - mobility measurements 161 fluorescence distributions, overlap 144-145 fluorescence lifetime, FRET 147 fluorescence loss in photobleaching see FLIP fluorescence microscopy 144-145 fluorescence recovery after photobleaching (FRAP) 10, 16, 55-57 – endocvtosis 72 fluorescence resonance energy transfer 59, 145 focal adhesion complex, caveolin 198-199 focal adhesion kinase (FAK) 199, 256-257 Foerster's resonance energy transfer see FRET formation - buds 34-37 - caveolae 30-31, 37-38 - endothelial tube 241 - PrP^{Sc} 210-211 - stress fiber 195-202 supramolecular protein complexes 141-174 - vesicle 15 Förster distance 146 Förster's resonance energy transfer see FRET FRAP see fluorescence recovery after photobleaching free-cholesterol-binding sites, peptide contributions 100 free endocytic vesicles, caveolar function 83 free energy (F), physical modeling of lipid membranes 26 FRET 4, 6-7, 16, 145 - efficiency 145 - flow cytometric 162 immunological synapse 151 lipid rafts 157 - pb see pbFRET

proximity measurements 60
frustration, bilayer order 31–32
Frye-Edidin experiment 155
functional multiprotein complexes 103
functional suppression, caveolin-1 in cancer cells 252–254
functional unit 142
Fyn
activation 132
caveolin phosphorylation 121
expression 119

inhibition 125–127overexpression 122Fyn-Abl, interaction 122–123

g

gangliosides 93 gate theory of aging 195 gel state, lipids in membranes 49-50 genetic suppression, caveolin-1 in cancer cells 252-254 GFP 6 proximity measurements 60 GFP-tagged caveolin 71 giant unilamellar vesicle (GUV) – fluorescence correlation spectroscopy 57 - membrane model system 3 - raft domains 9 glucose transport, stimulation 131 glucose transporter 4 (GLUT4) caveolin 199–200 translocation 131 glutathione S-transferase (GST)-CSD fusion proteins 235 glycoproteins, immunological synapse 151 glycosphingolipid GM1 ganglioside, lipid microdomains 164 glycosphingolipids 9 - raft hypothesis 45 glycosyl-phosphatidyl-inositol see GPI GM1 gangliosides, distribution 160 GPI, membrane traffic 4, 6-8 GPI-anchor - proximity measurements 60 - PrPc 207-208 - raft hypothesis 48 - transmembrane signaling 143 GPI-anchored proteins 8, 48, 51, 54, 63, 92, 207, 214, 257 – caveolae 175 raft immobilization 84 - signal transduction 92

green fluorescent protein *see* GFP growth conditions, caveolae structure 96 growth factor, vascular endothelial 241 growth factor-mediated signaling 104 GST *see* glutathione S-transferase GTP-binding proteins, TC10 200 GUV *see* giant unilamellar vesicle

h

H-Ras, FRAP 56 HA, single-molecule tracking 54 hairy-aspect, caveolae 25 HASM 53 HDL see high-density lipoprotein HDL uptake, caveolin-1 182 heart, role of nitric oxide 234 heart rate regulation, muscarinic cholinergic 244-245 HeLa cells, endocytosis 79 hemagglutinin see HA hetero-FRET 6 - proximity measurements 62-63 heteroclusters, cell fusion 156 heterodimerization, lipid rafts 157 heterologous expression, caveolin-1 in cancer cells 252-254 hexamethylglutaryl coenzyme A (HMGCoA) 240 high-density lipoprotein (HDL) 239 - estrogen 185 signal transduction 94 HMGCoA reductase inhibitors (statins) 240-241 homeoproteins, endocytosis 183 homo-FRET 6 - proximity measurements 60-62 homo-oligomers, caveolins 33 homoclusters - cell fusion 156 - immunological synapse 152 Hooke's law, physical modeling of lipid membranes 27 hop diffusion, single-molecule tracking 54 human cancer, role of caveolin-1 251 human coronary artery smooth muscle see HCASM human epidermal carcinoma see KB human myoepithelial cells 70 human T lymphoma cells, lipid rafts 157 hydrophobic regions, membrane proteins 31 hyperphosphorylation, caveolin phosphoryla-

tion 118

- Thy-1 8

272 Index

ICAM-1, co-localization 153 IGF-I see insulin-like growth factor-I П - co-localization 159, 162 - raft assisted signaling 159 IL-2, lipid rafts 156-162 IL-6 254-255 immunolocalization, caveolae 179, 182 immunological synapse (IS) 150 inclusions - due to sorting signals 15 membrane 31–34 incompatibility (chemical), influence on lipid domain shape 29-30 independent pathways, endocytosis 84 inducible NOS isoform (iNOS) 234 inhibition, Fyn 125-127 iNOS see inducible NOS isoform insulin absence 70 insulin in adipocytes, signaling pathways 116-117 insulin-induced actin remodeling 131 insulin-induced caveolin phosphorylation 119 insulin-like growth factor-I (IGF-I) 254-255 insulin receptor, caveolins 117-118 integrins, focal adhesion complex 198-199 interleukin see IL intermixing, cell fusion 155 internalization 75 – caveolae 81–82 - PrP^c 209-210 intracellular caveolin-associated structures, caveosomes 82-83 intracellular chaperone complexes 175-194 intracellular compartments - amyloid β-peptide (Ab) 215-217 - PrP^c conversion 208-210 intracellular polarity 195-196 invaginated caveolae, equilibrium 92 invaginated lipid rafts, caveolae 28-31 invaginated membrane domains, bending energy 30 invaginations, caveolar 84-85 IS see immunological synapse

k

K-Ras, FRAP 56 K⁺ channels, voltage-gated 154–155 KB, signal transduction 94 kinases 127 – caveolin 120, 129 – caveolin tyrosine 123 focal adhesion 199, 256–257
non-receptor tyrosine 119
phosphoinositide 254–256
receptor 106
receptor tyrosine 163–166
Src-family *see* Src-family kinase
tyrosine 116, 119
kiss-and-run, transendothelial transport 74

L

labeled molecules, local concentration 148 lactase-phlorizin hydrolase see LPH lag time, FCS 148 large unilamellar vesicles see LUV latranaculin A, FRAP 56 LAURDAN, amphiphilic dye 5 LDL see low-density lipoprotein line tension 13 linear pathway, caveolin phosphorylation 121 lipid-anchored proteins 48 lipid bilayer 1, 141 - order 31-32 - structure 28 lipid components, cellular membranes 2, 143-144, 175 lipid composition, caveolae 12 lipid domain shape, influence of chemical incompatibility 29-30 lipid membrane 141 - artificial 50 - bilayer structure 28 - physical modeling 26-29 see membranes lipid microdomains 141-174 - receptor tyrosine kinases 163-166 lipid phosphate phosphohydrolase, signal transduction 94 lipid rafts 1-17, 51-52, 141, 143, 156-162, 175 - biophysical characterization 45-68 - co-localization 153 - invaginated 28-31 - membrane traffic 12-17 - planar 92 - processing of prion protein 205-223 - signal transduction 91 - voltage-gated K⁺ channels 154-155 lipid uptake, caveolin 188 lipids, two-dimensional fluids 45 lipoproteins, caveolin-eNOS interaction 239 - 241liquid crystalline state, lipids in membranes 49 - 50

liquid-disordered phases 3-5

liquid-ordered phases 3-5 lipid rafts 92 loading, FC 102-104 local motility, endocytosis 73-74 local regulatory processes 154-155 localization - caveolae 182-183 - caveolin 178-182, 187 - cholesterol 182-183 – co- see co-localization locomotion, roles of caveolae and caveolin 195 - 202low-density lipoprotein (LDL) 103 - cholesterol 239-241 LPH, single-molecule tracking 54 lung cancer, role of caveolin-1 251 LUV 3 lysotracker, caveosomes 82

m

mAchR see muscarinic cholinergic macrodomains, lipid 143 macrophage apoptosis, mouse 258 Madin-Darby canine kidney (MDCK) - cells 16, 50 – endocytosis 72 mammary tumor cells, caveolin-1 expression 252 mbCD 58 MCF-7 cells, caveolin-1 expression 253-254 MDCK see Madin-Darby canine kidney MDR human cancer cells 258 mean barrier-free path, domain size 144 mechano-regulation/-sensitivity, caveolae 38-42 melting temperature, lipid mixtures 3 membrane-associated proteins 49 membrane cap, spherical 28 membrane components 142-143 membrane crossing, caveolin 183-184 membrane domain shape, influence of chemical incompatibility 29-30 membrane domains 45 - bending energy 30 - caveolae 71 - nonendocytic 71 - scaffolded see scaffolded membrane domains membrane extension, actin remodeling 127 membrane inclusions 31-34 membrane interactions, caveolin 177-178 membrane model, Singer-Nicolson 141 membrane molecules, single-molecule tracking 53

membrane phases, fluorescence correlation spectroscopy 57 membrane protein clusters, cell fusion 156 membrane proteins - aggregation 32 - asymmetric 33-35 - hydrophobic regions 31 - lateral organization 144-149 - reorganization 142 - thermodynamic phase separation 35-38 membrane raft hypothesis, development 48 membrane systems, phase separation 3-5 membrane temperature, effective 27 membrane tension - caveolae 38-42 - in relation to domain energy and shape 39 membrane topology 48 membrane traffic 1-17 - cargo sorting and vesicle formation 15 - caveolae and lipid rafts 12-17 - cellular organelles 12 - GPI 4, 6-8 membrane transporter proteins 97 membranes - cell see cell membrane - detergent-insoluble 50 - detergent-resistant see detergent-resistant membranes (DRM) - diffusion process 58 - lipid see lipid membrane - plasma see plasma membrane - see apical membrane metalloproteases 214 methyl-β-cyclodextrin see mbCD MHC I - co-localization 153, 162 - homoassociation 151 - transendothelial transport 75 MHC I glycoprotein, domain size 145 microdomains - cholesterol/sphingolipid (FC/SPH)-rich 91 - lipid 141-174 - plasma membrane 175 microscopy - atomic force 149 - confocal laser scanning 144 - electron 51 - fluorescence 144-145 - quantitative immunogold labeling electron 71 migratory synapses, immunological synapse

150

274 Index

mitosis, signal transduction 96 MLV 3 model membrane systems, phase separation 3 - 5modeling, lipid membranes 26-29 modulation of polarity, caveolin 195-196 molecular specificity, SNOM 149 molecules - adhesion 151 dipole moment 5 - labeled 148 - motion 52 - signaling 133 single, tracking 53–54 motility, caveolae 73-74 multi-protein complexes, signal transduction 95 multifunctional adaptor proteins 249 multilamellar vesicles see MLV multiphoton excitation, domain size 144 multiprotein complexes, functional 103 multistage caveolin lipidation cycle 105 muscarinic cholinergic (mAchR), heart rate regulation 244-245 mutant caveolin 180, 252 – caveolae formation 37–38 mYFP-GPI, proximity measurements 60 myocytes, cardiac 234, 244-245 myoepithelial cells, human 70

n

N-terminal – caveolins 25, 177 – endocytosis 70 N-WASP, actin polymerization 130 NCAM 53 negative data evaluation, caveolae determination 179 neural cell adhesion molecule see NCAM nitric oxide synthase (NOS), endothelial 233-246 NMR see nuclear magnetic resonance NO-mediated vascular relaxation 242-243 non-receptor tyrosine kinases 119 noncaveolar rafts, endocytosis 69 nonendocytic membrane domains, caveolae 71 nongenomic actions, caveola 185-186 NOS see nitric oxide synthase NOSIP see eNOS interacting protein NOSTRIN see eNOS traffic inducer NPC-1 fibroblasts, cholesterol transport 188 nuclear magnetic resonance (NMR)

- lipid microdomains 163
- lipid rafts 50

0

organelles, membrane traffic 12 ovarian cancer cells, pro-apoptotic activity of caveolin-1 253 overexpression techniques, caveolin-1 253 oxidative stress, signaling pathways 125

р

P132L, mutant caveolin-1 252 partitioned fluid, single-molecule tracking 53 pathways - APP cleaving 218 - caveolin trafficking 188 - compensatory activation, caveolae 176 - endocytic 84, 176 - PI3K/Akt 254-256 - signaling 116-123 paxillin, signaling pathways 124 pbFRET 146 pbFRET microscopy, lipid microdomains 164 PC 2-5.9 - DRM 50 PDGF 117 - signal transduction 96 PDGFR, signal transduction 105 PDK1 see phosphoinositide-dependent protein kinase-1 PE 2-3 peptide contributions, free-cholesterolbinding sites 100 percolating raft membrane 9-10 peripheral SMAC see pSMAC PHA 160 phalloidin, phosphorylation 129 phase diagram (ternary), SM/PC/cholesterol 4 phase separation - cargo sorting and vesicle formation 15 - caveolin oligomers 38 - cell membranes 5-7 - fibroblasts 10 - model membrane systems 3-5 phases - liquid-disordered 3-5 liquid-ordered 3–5 phosphatidic acid, signal transduction 94 phosphatidylcholine see PC phosphatidylethanolamine see PE phosphatidylinositol-3'-phosphate see PI(3)P phosphatidylserine see PS

phosphocaveolin 126-129 - Csk 124–125 phosphocholine see PC phosphoinositide 3-kinase (PI3K), cell survival 254-256 phosphoinositide-dependent protein kinase-1 (PDK1) 255 phospholipase D2, signal transduction 94 phosphoprotein, signaling pathways 124 phosphorylation 115 actin/plasma membrane attachment sites 128 - caveolin see caveolin phosphorylation - cell migration 130 - eNOS activity 238 - signaling pathways 116-123 - stable 129 - tyr14 256-257 - tyrosine 116-123 photobleaching - fluorescence recovery see fluorescence recovery after photobleaching - FRET 146 phytohemagglutinin see PHA PI(3)P 2 PI3K see phosphoinositide 3-kinase PI3K/Akt pathway, cell survival 254-256 picket-fence model 54, 56 placental alkaline phosphatase see PLAP planar lipid rafts, equilibrium 92 PLAP 4, 8, 50 plasma membrane 127-129 – fluid-mosaic model 46 - microdomains 175 - protein complexes 141 - see also membranes plasmalemmal vesicles see caveolae plasmalogens, caveolae 94 platelet-derived growth factor see PDGF q polarity, modulation by caveolin 195-196 polyacrylamide gel electrophoresis see PAGE polyethylene glycol, artificial frustration 32 potassium channels see K⁺ channels preadipocytes, caveolin phosphorylation 117 preassembly, IL-chains 157 prebeta-HDL, signal transduction 94 preferred bulb radius, caveolae 36-37 primary cells, differentiated 91 prion conversion, raft action 211-214 prion diseases 205-223 prion proteins internalization 81 - processing 205-223

- see also PrP^c

pro-apoptotic activity, caveolin-1 253 prostate cancer cells, survival-promoting activity of caveolin-1 253-254 proteases, metallo- 214 protein acylation, caveolae 186 protein anchor, raft hypothesis 48 protein clusters, cell fusion 155-156 protein co-mobility 147 protein coats, recyclable 13 protein complexes, supramolecular 141-174 protein organization, caveolae 187 protein patterns, plasma membrane 142 protein-protein interactions, tyrosine phosphorylation 123 protein structure, caveolin 177-178 protein transport, raft hypothesis 49 proteins - basolateral 14 - caveolae 95-98 - contact site 99 - FC binding 101-102 - GPI-anchored 51 - lipid-anchored 48 - membrane-associated 49 - prion see prion proteins proteolytic attack on PrPc, role of rafts 214 proximity measurements 59-63 PrPc - lipid rafts 205, 207-214 - see also prion protein PrP^c conversion, role of rafts 208-210 PrP^c-PrP^{Sc} transconformation 208, 210–214 PrPSc 207 - formation 210-211 PS 2-3 pSMAC, immunological synapse 150 PTEN, tumor suppressor protein 255

quantitative immunogold labeling electron microscopy, endocytosis 71

radiolabeled tracers, caveola 184 raft action, prion conversion 211-214 raft-aspect, caveolae 25 raft concept, phase separation in cell membranes 5-7 raft domains, clustered 8-9 raft hypothesis, origin 45-48 raft immobilization, caveolin 84-85 raft integrity 159 raft membrane, percolating 9-10

raft models, current 47

276 Index

raft signaling., protein acylation 186 rafts - lipid see lipid rafts noncaveolar 69, 75 role in PrP^c conversion 208–210 rapamycin, chaperone complexes 184 rat basophilic leukemia see RBL receptor, insulin binding 117 receptor autophosphorylation, signal transduction 104 receptor families, lipid microdomains 163 receptor kinase, signal transduction 106 receptor tyrosine kinases, function 163-166 recyclable protein coats 13 regulation, caveolin-1 182 regulatory cycle, caveolin-eNOS 236-238 regulatory processes, local 154-155 reorganization, membrane proteins 142 reserve rafts, diffusion-based measurements 55 resolution, SNOM 149 resonance energy transfer process, fluorescence 59 Rho GTPase activity, modulation by caveolin 195, 197-198 rigidity, bending 27-28

s

Saffman-Delbruck model, single-molecule tracking 54 saturated acyl chains, transmembrane signaling 143 scaffold hypothesis, caveolin 98-101 scaffold motifs, signal transduction 99 scaffold sequence, endothelin receptor-A 99 scaffolded membrane domains - caveolae 10-12 - caveolin see caveolin scaffolded domain scanning near-field optical microscopy 149 scavenger receptor BI see SRBI scavenger receptors, caveolae 187 SCLC see small-cell lung cancer scrapie PrP see PrPSc SDS-PAGE - caveolae 177 lipid rafts 51 SDS-resistant complexes 117 secondary structure, caveolin 177 secretase 215-223 $\alpha\mbox{-secretase},$ role of caveolae and lipid rafts 220-221 β-secretase, role of caveolae and lipid rafts

β-secretase, role of caveolae and lipid rafts 206, 214–218 γ-secretase, role of caveolae and lipid rafts 221-222 secretory synapse, immunological synapse 150 Semliki Forest virus, raft hypothesis 49 Ser1177 phosphorylation, eNOS activity 238 sex hormone action, caveola 185 SFK see Src-family kinase SH2 120 SH2-mediated binding, phosphocaveolin 126 shaping forces, caveolae 25-42 Shiga toxin, endocytosis 77 sialidase, signal transduction 95 signal transduction - caveolae 91-114 - FC changes 104-106 signaling - caveolae 115-140 - time course 105 - transmembrane 141-174 signaling cascades - activation 124 - caveolin 115 signaling complexes, caveolin 132 signaling molecules, caveolin phosphorylation 133 signaling pathways 116-133 - caveolin phosphorylation 121 - phosphorylation 256-257 signaling platforms, lipid rafts 156-162 signaling proteins - classes 96 - scaffold motif 99 Simian virus 40 (SV40) 200 - raft and caveolar endocytosis 14 - transendothelial transport 74 Singer-Nicolson fluid mosaic membrane model, validity 141 single kinase knockout mice 119 single-molecule studies, diffusion-based measurements 52-55 single-particle tracking 52 - immunological synapse 151 membrane molecules 53 SM 2-4,9 - DRM 50 small-cell lung cancer (SCLC), role of caveolin-1 251 SNARE, fluorescence correlation spectroscopy 58 SNOM 149, 165 sodium dodecyl sulfate see SDS sodium dodecyl sulfate-polyacryamide gel electrophoresis see SDS-PAGE

sorting signals, cargo sorting and vesicle formation 15 spectroscopic ruler 145 spectroscopy, fluorescence see fluorescence correlation spectroscopy spherical membrane cap 28 sphingolipid-rich microdomains 91 sphingolipids - APP processing 222 - raft hypothesis 45 sphingomyelin see SM sphingomyelinase, signal transduction 95 spontaneous curvature 27 SPT see single-particle tracking SR-BI, scavenger receptors 187 SRBI 185 Src-family kinase (SFK) 118-119, 127, 256-257 - inhibitors 119 - phosphorylation 130 - regulation 125 - signal transduction 106 Src-homology 2 see SH2 SSR, diffusion-based measurements 55 statins - reduction of caveolin abundance 240-242 - stimulation of endothelial tube formation 241 statistical confidence, FRAP 56 steady-state anisotropy, proximity measurements 61 sterol uptake, caveolin 188 stress fiber formation, roles of caveolae and caveolin 195-202 stress-induced changes, caveolin-1 expression 257-258 stress-induced pathways, opposing 130 strong cholesterol depletion 216 subcellular locations, caveolin-1 prescence 181 superstructures, ErbB2-containing 164 suppression, caveolin-1 in cancer cells 252-254 supramolecular organization, immunological synapse 150 supramolecular protein complexes - formation 141-174 - lipid rafts 162 surface scanning resistance see SSR surface tension, lipid membrane 27 survival-promoting activity, caveolin-1 253-254 SV40 see Simian virus 40 syntaxin 3, apical t-SNARE 16

synthases, endothelial nitric oxide 233–246 synthetic lipid bilayers, signal transduction 92

t

T-cell homeostasis, lipid rafts 156 T-cell receptor see TCR T cells - activation 154 raft clustering 8 T lymphocytes 154 T lymphoma cells, human 157 t-SNARE, syntaxin 3 16 TAMRA-4D5-labeled ErbB2 receptors 165 TC10, GTP-binding protein 200 TCR complex, immunological synapse 150 TCZ 52 temperature - melting, lipid mixtures 3 - membrane 27 ternary phase diagram, SM/PC/cholesterol 4 TfR 156. 160-162 TGFB 82 TGN see trans-Golgi network thermodynamic phase separation, membrane proteins 35-38 Thr495 phosphorylation, eNOS activity 238 Thy-1, GPI-anchored protein 8 TNFa 124 traffic, membrane see membrane traffic trafficking - caveolin 183-184 - prion protein 205-223 trafficking cycle, caveola 184-185 trafficking pathways, caveolin 188 TRAMP mouse model 253, 256 trans-Golgi network (TGN) 50, 215 - APP cleaving pathways 218 - endocytosis 72 transconformation, PrPc-PrPSc 210-214 transendothelial transport, endocytosis 73-74 transferrin receptor see TfR transforming growth factor beta see TGFB transient confinement zone see TCZ transmembrane proteins 48 transmembrane receptor tyrosine kinases, lipid microdomains 163 transmembrane receptors, signal transduction 96 transmembrane signaling 141-174 - complex 105 transport, transendothelial 73-74 trastuzumab, cancer therapy 165

Trojan peptide 102

278 Index

tube formation, endothelial 241 tumor cell survival, role of caveolin-1 249–259 tumor necrosis factor-alpha *see* TNFα tumors, angiogenesis 241–242 tyr14 phosphorylation 256–257 tyrosine, caveolin phosphorylation 116–123 tyrosine kinases 119 – signaling pathways 116

и

unidentified intracellular compartment (UIC) 218

V

vascular endothelial growth factor (VEGF) 241–242
– signal transduction 97
vascular functions, heart 234
vascular permeability, transendothelial transport 74 vascular relaxation, endothelium-dependent and NO-mediated 242–243 vasodilation, caveolin-eNOS interaction 242–244 VEGF *see* vascular endothelial growth factor venular endothelial cells, caveolin determination 180 vesicle formation, membrane traffic 15 vesicle trafficking, caveolin 183 vesicular vacuolar organelle *see* VVO virus, caveolae wave 74 voltage-gated K⁺ channels, lipid rafts 154–155

VVO, caveolin 180

w

water permeability, caveolae 94

γ

yellow fluorescent protein (YFP), singlemolecule tracking 54